Exhibit No.: Issues:

Capital StructureWitness:Pauline M. AhernExhibit Type:RebuttalSponsoring Party:Missouri-American Water CompanyCase No.:WR-2011-0337Date:January 19, 2012

Return on Equity

# MISSOURI PUBLIC SERVICE COMMISSION

### CASE NO. WR-2011-0337

## **REBUTTAL TESTIMONY**

### OF

# PAULINE M. AHERN, CRRA

### ON BEHALF OF

### **MISSOURI-AMERICAN WATER COMPANY**

### **BEFORE THE PUBLIC SERVICE COMMISSION**

OF THE STATE OF MISSOURI

IN THE MATTER OF MISSOURI-AMERICAN ) WATER COMPANY FOR AUTHORITY TO ) FILE TARIFFS REFLECTING INCREASED ) RATES FOR WATER AND SEWER ) SERVICE ) CASE NO. WR-2011-0338

#### **AFFIDAVIT OF PAULINE M. AHERN**

Pauline M. Ahern, being first duly sworn, deposes and says that she is the witness who sponsors the accompanying testimony entitled "Rebuttal Testimony of Pauline M. Ahern"; that said testimony and schedules were prepared by her and/or under her direction and supervision; that if inquires were made as to the facts in said testimony and schedules, she would respond as therein set forth; and that the aforesaid testimony and schedules are true and correct to the best of her knowledge.

Pauline M. Ahern

State of New Jersey County of Burlington SUBSCRIBED and sworn to Before me this  $17^{\text{th}}$  day of 3000 2012.

**Notary Public** 

My commission expires:

SHARON M. KEEFE NOTARY PUBLIC OF NEW JERSEY MY COMMISSION EXPIRES JULY 9, 2016

## TABLE OF CONTENTS

# Page No.

| INTRODUCTION                                       | 1  |
|--|----|
| PURPOSE  | 1  |
| SUMMARY  | 2  |
| CAPITAL STRUCTURE                                  | 3  |
| Double Leverage                                    | 3  |
| COMMON EQUITY COST RATE                            | 12 |
| Testimony of MoPSC Staff Witness Matthew J. Barnes | 12 |
| Discounted Cash Flow Model                         | 12 |
| Capital Asset Pricing Model                        | 21 |
| Recommended Common Equity Cost Rate                | 30 |
| MIEC WITNESS MICHAEL P. GORMAN                     | 35 |
| COMMON EQUITY COST RATE                            | 35 |
| Current Capital Market Conditions                  | 35 |
| Proxy Group Selection                              | 36 |
| Discounted Cash Flow Model (DCF)                   | 37 |
| Risk Premium Model (RPM)                           | 44 |
| Capital Asset Pricing Model                        | 47 |
| Financial Integrity                                | 50 |
| Recommended Common Equity Cost Rate                | 51 |
| BJC WITNESS BILLIE SUE LACONTE                     | 53 |
| COMMON EQUITY COST RATE                            | 53 |
| Discounted Cash Flow Model                         | 53 |
| Capital Asset Pricing Model                        | 54 |
| Recommended Common Equity Cost Rate                | 56 |
| UPDATED OVERALL COST OF CAPITAL AND                | 58 |
| RATE OF RETURN ON COMMON EQUITY                    | 58 |

#### 1 INTRODUCTION

- 2 Q. Please state your name, occupation and business address.
- A. My name is Pauline M. Ahern and I am a Principal of AUS Consultants. My
  business address is 155 Gaither Drive, Suite A, Mt. Laurel, New Jersey 08054.
- 5 Q. Are you the same Pauline M. Ahern who previously submitted prepared direct 6 testimony in this proceeding?
- 7 A. Yes, I am.
- 8 Q. Have you prepared schedules which support your rebuttal testimony?
- 9 A. Yes, I have. They have been marked for identification as Schedules PMA-18
  10 through PMA- 39.

#### 11 **PURPOSE**

12 Q. What is the purpose of this testimony?

13 Α. The purpose of this testimony is to rebut certain aspects of the Missouri Public 14 Service Commission Staff Report – Cost of Service (Staff Report, Staff Witness 15 Matthew J. Barnes), as well as the direct testimonies of Mr. Michael P. 16 Gorman, Witness for the Missouri Industrial Energy Consumers (MIEC) and 17 Ms. Billie Sue LaConte, Witness for BJC Healthcare (BJC). Specifically, I will 18 address Staff's comments relative to the concept of double leverage; its 19 application of the Discounted Cash Flow (DCF) Model and Capital Asset 20 Pricing Model (CAPM); and, its failure to reflect Missouri American Water 21 Company's (MAWC) greater unique business risks relative to its proxy group of 22 six water companies, the greater financial risk of Staff's recommended capital 23 structure ratios as well as flotation costs. Relative to the direct testimony of Mr.

Gorman, I will address his applications of the DCF, Risk Premium Model (RPM) and CAPM. Relative to the direct testimony of Ms. LaConte, I will address her applications of the DCF and CAPM. In addition, I will address Mr. Gorman's And Ms. LaConte's their failure to reflect MAWC's greater unique business risks relative to their proxy groups of water companies and flotation costs.

6 SUMMARY

7 Q. Please briefly summarize your rebuttal testimony.

A. My rebuttal testimony addresses Staff's discussion of the concept of double
 leverage and how it violates the basic financial principles of risk and return, the
 opportunity cost of capital, is discriminatory and based upon faulty
 assumptions.

12 My rebuttal testimony also describes a number of errors causing Staff's 13 recommended common equity cost rate to be well below any reasonable range 14 for MAWC because:

15 • Staff erroneously relies primarily upon the DCF model to arrive at its 16 recommended common equity cost rate despite the Commission's 17 consideration of the results of other cost of common equity models. Staff 18 uses, albeit incorrectly, the CAPM model but only as a check on its 19 flawed and understated recommendation. The Efficient Market 20 Hypothesis (EMH), upon which all the cost of common equity models are 21 premised, confirms that investors rely upon multiple cost of common 22 equity models in formulating their required rates of return.

• Staff's test of reasonableness, i.e., its CAPM analysis, is flawed.

- Staff's recommended range of common equity cost rate is not consistent
  with the expected returns on book common equity for Staff's proxy group
  of water companies.
- Staff failed to reflect MAWC's unique business risks, the greater
  financial risk inherent in Staff's recommended American Water Works
  Company's (American Water or the Parent) consolidated capital
  structure and debt cost rate, as well as flotation costs.

8 My rebuttal testimony also describes a number of errors causing bot 9 MIEC's and BJC's recommended common equity cost rate to be well below 10 any reasonable cost rate for MAWC because:

- MIEC's applications of the DCF, RPM and CAPM and BJC's application
   of the DCF and CAPM are flawed; and
- Both MIEC and BJC failed to reflect MAWC's unique business risks, the
  lower financial risk inherent in MAWC's capital structure as well as
  flotation costs.
- Finally, my rebuttal testimony provides an updated capital structure,
  senior capital cost rates and recommended common equity cost rate based
  upon current capital market conditions.
- 19 CAPITAL STRUCTURE
- 20 Double Leverage
- Q. On page 18 at lines 1-13 of the Staff Report, Staff provides the fourth reason
   for its use of American Water's consolidated capital structure, namely
   American Water's use of double leverage. Please comment.

1 Α. Company Witness William D. Rogers rebuttal testimony discusses why it is not 2 possible for American Water to use double leverage since the Parent debt was 3 incurred to finance the retirement of RWE's preferred stock and other 4 payments to RWE resulting in no cash proceeds being available to infuse equity into MAWC or any other American Water subsidiary. Consequently, the 5 6 notion that American Water employs double leverage, i.e., a mix of debt and 7 equity, to fund its equity infusions to MAWC or any of its operating subsidiaries, 8 as a rationale for using American Water's consolidated capital structure for 9 ratemaking purposes to determine MAWC's allowed overall rate of return is 10 unfounded. In addition, the very concept of double leverage and subsequent 11 use of the parent consolidated capital structure is flawed for five reasons:

- 12 1. It violates the basic financial principle of risk and return;
- 13 2. It is inconsistent with the concept of the opportunity cost of capital;
- 14 3. It discriminates against the investor, i.e., the parent, in the regulated
  15 operating utility, thus violating both the concept of fairness and the
  16 capital attraction standard;
- 17 4. It is based upon some highly problematic assumptions; and,
- 18 5. As Roger A. Morin states<sup>1</sup>: "[t]he double leverage approach is a
  19 tautology."
- Q. Please explain how double leverage violates the basic financial principle of riskand return.
- A. The basic financial principle of risk and return states that the rate of return

Morin, Roger A., New Regulatory Finance, 526 (Public Utilities Reports 2006).

1 required by investors on any investment is dependent upon the risk of that 2 investment and that investment alone. Since most investors are risk averse, this means that the higher the investor perceived risk of an investment, the 3 higher the return required by investors. As Eugene F. Brigham states<sup>2</sup>: 4 5 In a market dominated by risk-averse investors, riskier securities will 6 have higher expected returns, as estimated by the average investor, 7 than will less risky securities, for if this situation does not hold, 8 actions will occur in the market to force it to occur. (italics in original) 9 10 The risk of any investment, including investment in MAWC, is 11 independent of the ownership of the capital financing the investment. In 12 addition, it is a basic financial principle that it is the use of the funds invested 13 which gives rise to the risk of the investment, not the source of the funds. As Richard A. Brealev and Stewart C. Mvers state<sup>3</sup>: 14 The true cost of capital depends on the use to which the capital is 15 16 *put.* (italics in original) 17 18 19 20 The company cost of capital is the correct discount rate for projects 21 that have the same risk as the company's existing business. . . . In 22 principle, each project should be evaluated at its own opportunity 23 cost of capital; the true cost of capital depends on the use to which 24 the capital is put. 25 26 For example, if one were to inherit money, free of charge, and then 27 invest it in a given utility's common stock, one would require a rate of return on 28 that stock commensurate with the risks to which that common stock investment

<sup>&</sup>lt;sup>2</sup> Brigham, Eugene F., <u>Fundamentals of Financial Management</u>, 114 (The Dryden Press, 5<sup>th</sup> Ed. 1989).

<sup>&</sup>lt;sup>3</sup> Brealey, Richard A. and Myers, Stewart C., <u>Principles of Corporate Finance</u> 205,299 (McGraw-Hill Book Company, 1988).

is exposed including the financial risk inherent in that utility's capital structure.
It would be illogical to state that the required return on investment is zero just
because there was zero cost in acquiring the capital, i.e., inherited money,
which was the source of the investment. Even the Internal Revenue Service
places your cost basis, as an inheritor, on the market value of inherited
common stock on the date of death of the person who willed the stock and not
on its zero cost to you.

Just as illogical is the inevitable conclusion that, in the event that the common shares of the operating water utility subsidiary were held by both a corporate parent and by an outside investor or investors, that portion of subsidiary equity supplied by the parent would have one cost rate, i.e., the parent's weighted overall cost of capital, while the portion supplied by the outside investor or investors would have another, i.e., their investor required return based upon the risk to which their capital is put.

In view of the foregoing, using the concept of double leverage to justify
the use of American Water's consolidated capital structure and not MAWC's
ratemaking capital structure violates the basic financial principle of risk and
return, because it presumes that MAWC's investment risk is equal to that of
American Water.

Q. Please explain how double leverage is inconsistent with the concept of theopportunity cost of capital.

A. The opportunity cost of capital is that rate of return offered by investments of
 comparable risk. It is called the opportunity cost because it represents the

1 return which is given up or foregone by investing in one investment alternative 2 as opposed to an alternative investment of comparable risk. If the risk-3 adjusted cost of equity investment in an operating water utility subsidiary, such as MAWC, is 9.90% (the midpoint of Staff's recommended range common 4 5 equity cost rate) and the effective authorized return is less than 9.90% through 6 the use of a consolidated capital structure, i.e., assuming double leverage, then 7 there is no incentive for a parent company, such as American Water, to invest in that operating subsidiary. In order to do so, the parent would have to forego 8 9 the risk-adjusted return of 9.90% on alternative investments not subject to 10 double leverage, in the form of a consolidated parent capital structure.

In fact, Staff's recommended 9.90% common equity cost rate results in
 an effective authorized return on common equity ROE for MAWC of only 8.93%
 based upon an income tax rate of 35% and as derived in Schedule PMA-18

14

In fact, the use of double leverage through use of a consolidated parent capital structure presents an incentive to spin-off the subsidiary, because the utility subsidiary should then be allowed a return on equity commensurate with its own business and financial risks and not one derived from the parent company's consolidated capital structure, which presumably would be lower. If such a divestiture were to occur, any cost reducing benefits due to economies of scale and diversification would be lost to the utility's ratepayers.

Hence, double leverage in the form of the use of a consolidated parent capital structure is inconsistent with the concept of the opportunity cost of capital.

- Q. How does the use of a consolidated parent capital structure discriminate
   against the parent holding company as the investor, thus violating the concept
   of fairness and the capital attraction standard?
- 4 Α. The holding company's required return on its equity investment in the operating 5 utility subsidiary is the risk-adjusted cost of common equity of that utility which 6 is dependent upon that utility's specific business and financial risks as stated 7 previously. Double leverage, in the form of imposing the parent's consolidated 8 capital structure, requires the use of the parent holding company's overall cost 9 of capital as the operating utility subsidiary's overall cost of capital. In so doing, 10 the parent holding company investor is denied the opportunity to earn its 11 required rate of return based upon the risk to which its common equity 12 investment in that utility is exposed. In this proceeding, should Staff's 13 recommended overall rate of return be adopted, based upon an income tax 14 rate of 35%, MAWC would, in effect, be authorized an 8.93% return on equity 15 capital as discussed above. This would not be the case for a utility whose stock 16 is held not by a holding company, but by individual investors.

For example, if there are two operating utilities with identical business and financial risks, the cost of common equity for both would be identical according to the basic financial principle of risk and return as discussed previously. However, if one of the utilities is an operating subsidiary of a parent holding company and its allowed overall rate of return is based upon the parent company's consolidated capital structure, the parent holding company will not be fairly compensated for the risk it bears by investing in the subsidiary. This is

1 discriminatory. As Roger A. Morin states<sup>4</sup>:

Estimating equity costs by one procedure for publicly held utilities and by another for utilities owned by a holding company is inconsistent with financial theory and discriminates against the holding company form of ownership. Two utilities identical in all respects but their ownership format should have the same set of rates. Yet, this would not be the case under the double leverage adjustment.

8 9 10

11

12

2

3

4 5

6

7

In addition, double leverage in the form of imposing a parent consolidated capital structure containing less common equity than the regulated subsidiary will weaken the regulated utility's ability to attract capital in

- 13 violation of the capital attraction standard established in Bluefield<sup>5</sup> which states
- 14 that:

15 A public utility is entitled to such rates as will permit it to earn a return 16 on the value of the property which it employs for the convenience of 17 the public equal to that generally being made at the same time and in 18 the same general part of the country on investments in other business 19 undertakings which are attended by corresponding risks and uncertainties. . . . The return should be reasonable sufficient to 20 21 assure confidence in the financial soundness of the utility and should 22 be adequate, under efficient and economical management, to 23 maintain and support its credit and enable it to raise the money 24 necessary for the proper discharge of its public duties. 25

The regulated utility must compete in the capital markets for its debt capital and must earn a reasonable return on <u>its</u> common equity to assure potential bond holders of its creditworthiness. The use of double leverage, in the form of imposing a parent consolidated capital structure, does not permit an opportunity to earn a rate of return commensurate with publicly owned enterprises of similar risk, thereby pressuring cash flows and potentially

<sup>&</sup>lt;sup>4</sup> Morin, 525.

<sup>&</sup>lt;sup>5</sup> Bluefield Water Works Improvement Co. v. Public Serv. Comm'n, 262 U.S. 679 (1922).

impairing interest coverage and, in turn, the regulated utility's ability to attract
 debt capital at reasonable costs.

Thus, the use of double leverage, in the form of imposing a parent consolidated capital structure, is both discriminatory and patently unfair to the parent holding company. Some of the assumptions of double leverage are highly problematic and nonsensical.

Q. What are some of the problematic assumptions upon which the concept ofdouble leverage is based?

9 Α. First, double leverage, in the form of imposing a parent consolidated capital 10 structure, assumes that all of the regulated subsidiary's equity capital was 11 provided by the parent holding company. However, the retained earnings of 12 the subsidiary are not derived from the parent. Rather, retained earnings result 13 from the accumulated net income to common equity after payment of common 14 dividends and are derived from revenues collected from the regulated 15 operating subsidiary's ratepayers. Also, any debt or preferred stock issued to 16 holders other than the parent company, are not derived from the parent. In 17 addition, if the proceeds of any of the senior capital, i.e., debt and / or preferred 18 equity, at the parent level were used to specifically invest in the operations of 19 other subsidiaries or to acquire another subsidiary, the assumption that such 20 funds were available for investment in the subsidiary subject to the imposition 21 of the parent consolidated capital structure is invalid.

22 Second, double leverage assumes that the business and financial risks 23 of all the operating subsidiaries are identical and, in turn, identical to the

1 business and financial risks of the parent holding company. This is clearly non-2 sensical, given that, at the very least, other regulated operating utility 3 subsidiaries most likely operate in different states under different regulatory 4 paradigms, as is the case with AWW, which has regulated operations in twenty (20) states. In addition, the regulated operating subsidiaries of AWW are of 5 6 different sizes, and face different operating and financial risks. Clearly, the 7 risks of all American Water's regulated operating subsidiaries are not equal. 8 Once again, the risk and return principle is violated by double leverage, 9 including the imposition of a parent consolidated capital structure, because it 10 assumes the same overall cost of capital for all the subsidiaries regardless of 11 their specific risk differences.

12 Q. Please explain how "[t]he double leverage approach is a tautology."

13 A tautology is unnecessary redundancy, i.e., saying the same thing twice. The Α. 14 double leverage approach using a parent consolidated capital structure is a 15 tautology because it is not the parent's overall cost of capital that determines 16 the subsidiary's overall cost of capital because the parent's overall cost of capital is itself a weighted average of capital costs of all subsidiaries.<sup>6</sup> A 17 18 holding company is like a mutual fund, but one which holds its operating subsidiaries in its portfolio of assets instead of capital market securities, i.e., 19 20 stocks and bonds. A mutual fund's required return, based upon portfolio 21 theory, is the weighted average of the returns of the individual securities in the 22 fund. Each security in the fund has its own unique required return which is a

<sup>&</sup>lt;sup>6</sup> Morin, 526.

1 function of its individual risk profile. The concept of double leverage, including 2 the use of a parent consolidated capital structure, if applied to a mutual fund, indicates that the required return on any given individual security held by the 3 mutual fund is the weighted average required return on the mutual fund as a 4 5 whole. This defies common sense. If an investor could expect to receive the 6 same return on the individual securities as in the mutual fund as a whole why, 7 would he / she invest in the fund and pay the attendant fees which would then reduce his / her return? 8

9 Thus, the use of double leverage and use of a parent consolidated 10 capital structure transposes the direction of cause and effect on the parent's 11 overall cost of capital. Consistent with the fundamental and basic financial 12 concept of risk and return, discussed above, the overall cost of capital of a regulated operating utility subsidiary is a function of its business and financial 13 14 risks and must be found on a stand-alone basis, which requires the use of the 15 Company's own capital structure and cost rates, including the cost rate of 16 common equity capital, and not the use of the parent consolidated capital 17 structure, which assumes the weighted average overall cost of capital of the 18 consolidated parent company is that of the subsidiary.

19 COMMON EQUITY COST RATE

#### 20 Testimony of MoPSC Staff Witness Matthew J. Barnes

- 21 Discounted Cash Flow Model
- Q. Staff's range of recommended common equity cost rate, 9.40% 10.40%, with
  a midpoint of 9.90% is based exclusively upon a Discounted Cash Flow (DCF)

analysis, notwithstanding its use of the CAPM as a check. Please comment.

A. No single common equity cost rate model should be relied upon exclusively in
determining a cost rate of common equity and the results of multiple cost of
common equity models should be taken into account. Staff's exclusive reliance
upon the DCF model, notwithstanding its use of the CAPM as a check, is at
odds with the EMH, upon which the DCF is predicated.

7 The DCF model utilized by Staff is market-based since market prices are 8 employed in its application. Therefore, it is based upon the EMH which is the 9 foundation of modern investment theory, first pioneered by Eugene F. Fama<sup>7</sup> in 10 1970. As discussed in my direct testimony, pages 32 through 34, an efficient 11 market is one in which security prices reflect all relevant information all the 12 time. This implies that prices adjust instantaneously to new information, thus 13 reflecting the intrinsic fundamental economic value of a security.<sup>8</sup>

14 The semistrong form of the EMH, which asserts that all publicly available 15 information is fully reflected in securities prices, i.e., fundamental analysis 16 cannot "outperform the market", is generally held to be true because the use of 17 insider information often enables investors to "outperform the market" and earn 18 excessive returns. This means that all perceived risks are taken into account by investors in the prices they pay for securities. Investors are thus aware of all 19 20 publicly-available information, including bond ratings; discussions about 21 companies by bond rating agencies and investment analysts; as well as the

<sup>&</sup>lt;sup>7</sup> Fama, Eugene F., "Efficient Capital Markets: A Review of Theory and Empirical Work" (<u>Journal of Finance</u>, May 1970) 383-417.

<sup>&</sup>lt;sup>8</sup> Brigham (1989) 225.

- various cost of common equity methodologies (models) discussed in the
   financial literature.
- 3 Q. Do you have further academic support for the need to rely upon more than one
- 4 cost of common equity model in arriving at a recommended common equity
- 5 cost rate?

7

8 9

10

11

12

15 16

17

18

19 20

21

22

23

24

6 A. Yes. For example, Phillips<sup>9</sup> states:

Since regulation establishes a level of authorized earnings which, in turn, implicitly influences dividends per share, estimation of the growth rate from such data is an inherently circular process. For these reasons, the DCF model "suggests a degree of precision which is in fact not present" and leaves "wide room for controversy and argument about the level of k". (italics added) (p. 396)

13 14 \* \* \*

> Despite the difficulty of measuring relative risk, the comparable earnings standard is no harder to apply than is the marketdetermined standard. The DCF method, to illustrate, requires a subjective determination of the growth rate the market is contemplating. Moreover, as Leventhal has argued: 'Unless the utility is permitted to earn a return comparable to that available elsewhere on similar risk, it will not be able in the long run to attract capital.' (italics added) (p. 398)

25 Also, Morin<sup>10</sup> states:

26 Each methodology requires the exercise of considerable judgment 27 on the reasonableness of the assumptions underlying the 28 methodology and on the reasonableness of the proxies used to 29 validate a theory. The inability of the DCF model to account for 30 changes in relative market valuation, discussed below, is a vivid 31 example of the potential shortcomings of the DCF model when 32 applied to a given company. Similarly, the inability of the CAPM to 33 account for variables that affect security returns other than beta 34 tarnishes its use. (italics added)

<sup>&</sup>lt;sup>9</sup> Phillips, Jr., Charles F. <u>The Regulation of Public Utilities-Theory and Practice</u> (Public Utility Reports, Inc., 1993) 396, 398.

<sup>&</sup>lt;sup>10</sup> Morin 428-431.

1 2 No one individual method provides the necessary level of precision 3 for determining a fair return, but each method provides useful 4 evidence to facilitate the exercise of an informed judgment. 5 Reliance on any single method or preset formula is inappropriate 6 when dealing with investor expectations because of possible 7 measurement difficulties and vagaries in individual companies' 8 market data. (Morin, p. 428) 9 10 \* 11 12 The financial literature supports the use of multiple methods. 13 Professor Eugene Brigham, a widely respected scholar and finance academician, asserts: (footnote omitted) 14 15 16 Three methods typically are used: (1) the Capital Asset Pricing 17 Model (CAPM), (2) the discounted cash flow (DCF) method, and 18 (3) the bond-yield-plus-risk-premium approach. These methods are not mutually exclusive - no method dominates the others, 19 20 and all are subject to error when used in practice. Therefore, 21 when faced with the task of estimating a company's cost of 22 equity, we generally use all three methods and then choose 23 among them on the basis of our confidence in the data used for 24 each in the specific case at hand. 25 26 Another prominent finance scholar, Professor Stewart Myers, in an early pioneering article on regulatory finance, stated:<sup>2(footnote omitted)</sup> 27 28 29 Use more than one model when you can. Because estimating 30 the opportunity cost of capital is difficult, only a fool throws away 31 useful information. That means you should not use any one 32 model or measure mechanically and exclusively. Beta is helpful 33 as one tool in a kit, to be used in parallel with DCF models or 34 other techniques for interpreting capital market data. 35 36 Reliance on multiple tests recognizes that no single methodology 37 produces a precise definitive estimate of the cost of equity. As 38 stated in Bonbright, Danielsen, and Kamerschen (1988), 'no single 39 or group test or technique is conclusive.' Only a fool discards relevant evidence. (italics in original) (Morin, p. 430) 40 41 \* \* \* 42 43 44 While it is certainly appropriate to use the DCF methodology to 45 estimate the cost of equity, there is no proof that the DCF produces 46 a more accurate estimate of the cost of equity than other

1 methodologies. Sole reliance on the DCF model ignores the capital 2 market evidence and financial theory formalized in the CAPM and 3 other risk premium methods. The DCF model is one of many tools 4 to be employed in conjunction with other methods to estimate the 5 cost of equity. It is not a superior methodology that supplants other 6 financial theory and market evidence. The broad usage of the DCF 7 methodology in regulatory proceedings in contrast to its virtual 8 disappearance in academic textbooks does not make it superior to 9 other methods. The same is true of the Risk Premium and CAPM 10 *methodologies.* (italics added) (Morin, p. 431)

- 12 In view of the foregoing, it is clear that investors are aware of all of the
- 13 models available for use in determining common equity cost rate. The EMH
- 14 requires the assumption that, collectively, investors use them all.

- Q. Please comment upon Staff's estimation of the growth component for its DCFanalysis.
- On page 20, lines 10 20 of the Staff Report, Staff discusses its use of 17 Α. historical growth in dividends per share (DPS), earnings per share (EPS), book 18 value per share (BVPS) as well as projected growth in DPS, EPS, and BVPS. 19 20 More appropriately, Staff should have relied exclusively upon security analysts' 21 forecasts of EPS growth. Security analysts' forecasts take into account 22 historical information as well as all current information likely to impact the 23 future, which is critical since both cost of capital and ratemaking are 24 prospective. In addition, Myron Gordon, who first introduced the DCF model 25 adapted for utility ratemaking, came to recognize long after his book, The Cost 26 of Capital to a Public Utility was published in 1974, that the growth component 27 of his original "Gordon Model" which relied upon the sustainable growth method 28 had a serious limitation. Dr. Gordon, in a presentation on March 27, 1990 (some 16 years after the publication of his 1974 book), before the Institute for 29

- 1 Quantitative Research In Finance, in Palm Beach, Florida, entitled, The Pricing
- 2 <u>of Common Stocks</u>, stated that analysts' growth rate projections were superior
- 3 to the sustainable growth method:

The most serious limitation of the Gordon Model is the assumption that the dividend expectation can be represented with just two parameters, D and br ... We have seen that earnings and growth estimates by security analysts were found by Malkiel and Cragg to be superior to data obtained from financial statements for the explanation of variation in price among common stocks. That is, better estimates are obtained for the coefficient of the various explanatory variables. ...estimates by security analysts available from sources such as IBES are far superior to the data available to Malkiel and Cragg. Secondly, the estimates by security analysts must be superior to the estimates derived solely from financial statements. (italics added)

Also, Morin notes<sup>11</sup>:

18 Because of the dominance of institutional investors and their 19 influence on individual investors, analysts' forecasts of long-run 20 growth rates provide a sound basis for estimating required returns. 21 Financial analysts exert a strong influence on the expectations of 22 many investors who do not possess the resources to make their 23 own forecasts, that is, they are a cause of g. The accuracy of 24 these forecasts in the sense of whether they turn out to be correct 25 is not at issue here, as long as they reflect widely held 26 expectations. As long as the forecasts are typical and/or influential 27 in that they are consistent with current stock price levels, they are 28 The use of analysts' forecasts in the DCF model is relevant. 29 sometimes denounced on the grounds that it is difficult to forecast 30 earnings and dividends for only one year, let alone for longer time 31 This objection is unfounded, however, because it is periods. 32 present investor expectations that are being priced; it is the 33 consensus forecast that is embedded in price and therefore in 34 required return, and not the future as it will turn out to be.

- 35
- 36 37

4

5

6

7

8

9

10

11

12

13

14

15

16

17

38

. . . .

Published studies in the academic literature demonstrate that growth forecasts made by security analysts represent an appropriate source of DCF growth rates, are reasonable indicators

1 2

3

4 5 of investor expectations and are more accurate than forecasts based on historical growth. These studies show that investors rely on analysts' forecasts to a greater extent than on historic data only.

6 Cited on page 37 of my direct testimony, are studies performed by 7 Cragg and Malkiel<sup>12</sup> which demonstrate that analysts' forecasts are superior to 8 historical growth rate extrapolations. As noted on page 38, while some 9 question the accuracy of analysts' forecasts of EPS growth, it does not really 10 matter what the level of accuracy of those analysts' forecasts is well after the 11 fact. What is important is that they influence investors and hence the market 12 prices they pay.

13 Relative to continuing conflicts of interest and subsequent bias in 14 security analysts' forecasts of EPS growth following the 2002 financial market 15 reforms, my direct testimony at page 39, lines 5 - 16 notes that Burton A. 16 Malkiel<sup>13</sup> affirmed his belief in the superiority of analysts' earnings forecasts 17 when he testified before the Public Service Commission of South Carolina, in 18 November 2002 (see Schedule PMA-19)<sup>14</sup>:

19 There was much publicity given to tainted analysts' forecasts leading up 20 to the late 1990's. In the wake of investigations instituted by the New York 21 Attorney General, the National Association of Securities Dealers, and the

<sup>&</sup>lt;sup>12</sup> Cragg, John G. and Malkiel, Burton G. <u>Expectations and the Structure of Share Prices</u> (University of Chicago Press, 1982) Chapter 2 (Ahern Workpaper 13).

<sup>&</sup>lt;sup>13</sup> Malkiel, Burton A., the Chemical Bank Chairman's Professor of Economics at Princeton University and author of the widely-read national bestselling book on investing entitled, "A Random Walk Down Wall Street: The Time-Tested Strategy for Successful Investing (Completely Revised and Updated)" (W.W. Norton & Co. 2011).

<sup>&</sup>lt;sup>14</sup> <u>Re: South Carolina Electric & Gas Co.</u>, Docket No. 2002-223-E "Rebuttal Testimony", pp. 16-17 (S.C.P.S.C. Nov. 2002).

Securities & Exchange Commission, I believe the upward bias that existed in
 the late 1990s has diminished. In summary, I believe that current analysts'
 forecasts are more reliable than they were during the late 1990s. Therefore,
 analysts' forecasts remain the proper tool to use in performing a Gordon Model
 DCF analysis.

6 Moreover, there is no empirical evidence that investors, consistent with 7 the EMH, would discount or disregard analysts' estimates of growth in earnings "Do Analyst Conflicts Matter? 8 per share. Evidence From Stock Recommendations,"<sup>15</sup> provided in Schedule PMA-20, examined whether 9 10 conflicts of interest with investment banking ["IB"] and brokerage businesses 11 induced sell-side analysts to issue optimistic stock recommendations and 12 whether investors were misled by such biases. They conclude on page 1 of

13 Schedule PMA-20.

Overall, our findings do not support the view that conflicted
analysts are able to systematically mislead investors with
optimistic stock recommendations.

18 On page 29 of Schedule PMA-20, Agrawal and Anup state:

19 Overall, our empirical findings suggest that while analysts do 20 respond to IB and brokerage conflicts by inflating their stock 21 recommendations, the market discounts these recommendations 22 after taking analysts' conflicts into account. These findings are 23 reminiscent of the story of the nail soup told by Brealey and Myers 24 (1991), except that here analysts (rather than accountants) are the 25 ones who put the nail in the soup and investors (rather than 26 analysts) are the ones to take it out. Our finding that the market is

<sup>&</sup>lt;sup>15</sup> Agrawal, Anup and Chen, Mark A., "Do Analysts' Conflicts Matter? Evidence from Stock Recommendations", (Journal of Law and Economics, August 2008), Vol. 51.

1 not fooled by biases stemming from conflicts of interest echoes 2 similar findings in the literature on conflicts of interest in universal 3 banking (for example, Kroszner and Rajan, 1994, 1997; Gompers 4 and Lerner 1999) and on bias in the financial media (for examples, 5 Bhattacharya et al. forthcoming; Reuter and Zitzewitz 2006). 6 Finally, while we cannot rule out the possibility that some investors 7 may have been naïve, our findings do not support the notion that 8 the marginal investor was systematically misled over the last 9 decade by analysts' recommendations.

10 As discussed above and in my direct testimony, the market is 11 efficient. Therefore, investors are aware of all publicly-available information, 12 including the many available security analysts' earnings growth forecasts. 13 Investors are thus aware of the accuracy of past forecasts, whether for 14 earnings or dividends growth or for interest rates. Investors have no prior 15 knowledge of the accuracy of any available forecasts at the time of their 16 investment decision making, as that accuracy only becomes known after some 17 future period of time has elapsed.

18 Hence, consistent with the EMH upon which the cost of common equity 19 models utilized by both Staff and myself are predicated, since investors have 20 such security analysts' earnings growth rate projections available to them and 21 investors are aware of the accuracy of such projections, security analysts' 22 earnings projections should be used in a cost of common equity analysis. Staff 23 would have us ignore this reality by disregarding the largest influence on 24 individual investors who own approximately 53% on average (see Schedule 25 PMA-9) of all the common shares of the companies in my proxy group of nine 26 water proxy companies. Rate of return analysts, such as Mr. Barnes (Staff) 27 and myself, who attempt to emulate investor behavior, should not ignore this 28 reality.

- Q. What would Staff's DCF results have been if Staff had properly relied upon
   security analysts' projected growth in EPS in its DCF analysis?
- А As shown on Schedule PMA-21, had Staff relied upon security analysts' 3 projected growth in EPS, an average DCF cost rate of 10.53% results. The 4 average projected EPS growth rate ranges from 6.00% - 9.75% and when 5 added to Staff's dividend yield of 3.37%, results in a range of DCF cost rate of 6 7 9.37% - 13.12%, with a midpoint of 11.25%. DCF cost rates of 10.53% and 11.25% clearly demonstrate that Staff's DCF results, ranging from 8.97% -8 9 9.97% and Staff's recommended range of common equity cost rate of 9.40% -10 10.40% are grossly understated. Moreover, these cost rates are further 11 understated because they do not reflect either MAWC's greater unique 12 business risks relative to Staff's proxy group of six water companies, the 13 greater financial risk of Staff's recommended capital structure ratios or flotation 14 costs.
- 15

#### Capital Asset Pricing Model

16 Q. Do you have any comment regarding Staff's application of the CAPM?

A. Yes. Staff's application of the CAPM is flawed in four respects; 1) its choice of
the historical yield on 30-year U.S. Treasury bond as the risk-free rate; 2) its
use of historical market equity risk premiums which were incorrectly derived; 3)
its failure to also include a forecasted market equity risk premium; and, 4) its
failure to also apply the ECAPM to account for the fact that the Security Market
Line (SML) as described by the traditional CAPM is not as steeply sloped as
the predicted SML.

- Q. Please comment upon Staff's use of the historical yield on 30-year U.S.
   Treasury bonds as the risk-free rate.
- Α. Both the determination of cost of capital and the determination of rates for 3 utility services are prospective in nature. Therefore, it is inappropriate to use 4 5 an historical yield as the risk-free rate in a CAPM analysis. Rather, a 6 prospective yield on 30-year U.S. Treasury bonds should be used such as the 7 projections Staff provides on Schedule 5 of the Staff Report. On Schedule 5, Staff shows that the Value Line Investment Survey - Selection & Opinion 8 9 (Value Line – S&O) projects long-term U.S. Treasury bond yields of 4.90% for 2012 and 5.00% for 2013 which average 4.95%. Thus, Staff's recommended 10 11 3.04% average yield on 30-year U.S. Treasury bonds for September 2011 12 significantly understates the prospective yield.
- Q. You have stated that Staff erred in exclusively relying upon historical market
   equity risk premiums which were incorrectly derived. Please explain.
- A. Staff's derivation of historical market equity premiums is incorrect for two
   reasons. First, Staff's arithmetic historical market equity risk premium is
   incorrectly calculated. Second, Staff also relied upon the geometric historical
   market equity risk premium.
- Q. Why is Staff's arithmetic historical market equity risk premium incorrectlycalculated?
- A. Staff's arithmetic historical market equity risk premium of 6.0% is derived from
   the <u>Ibbotson<sup>®</sup> SBBI<sup>®</sup> 2011 Valuation Yearbook Market Results for Stocks</u>,
   Bonds, Bills and Inflation 1926-2010 (2011 SBBI) as the difference between

1 the arithmetic mean 1926-2010 total return on large company stocks of 11.9% 2 and the arithmetic mean 1926-2010 total return on long-term government bonds of 5.9%. (6.0% = 11.9% - 5.9%).<sup>16</sup> The correct derivation of the 3 4 historical market equity risk premium is the difference between the total return 5 on large company stocks of 11.9% and the arithmetic mean 1926-2010 income 6 return on long-term government bonds of 5.2% which results in a market equity 7 risk premium of 6.7% (6.7% = 11.9% - 5.2%). Regarding the use of the income return and not the total return for Treasury securities in deriving an equity risk 8 premium, 2011 SBBI states (see page 5 of Schedule PMA-22)<sup>17</sup>: 9

10 Another point to keep in mind when calculating the equity risk 11 premium is that the income return on the appropriate-horizon 12 Treasury security, rather than the total return, is used in the 13 calculation. The total return is comprised of three return 14 components: the income return, the capital appreciation return, 15 and the reinvestment return. The income return is defined as 16 the portion of the total return that results from a periodic cash 17 flow or, in this case, the bond coupon payment. The capital appreciation return results from the price change of a bond over 18 19 a specific period. Bond prices generally change in reaction to 20 unexpected fluctuations in yields. Reinvestment return is the 21 return on a given month's investment income when reinvested 22 into the same asset class in the subsequent months of the year. 23 The income return is thus used in the estimation of the equity risk premium because it represents the truly riskless portion of the return.<sup>2 (footnote omitted)</sup> (emphasis added) 24 25 26

27

Hence, the correct historical market equity risk premium is 6.7% and not 6.0%.

- 28 Q. Please discuss Staff's use of a geometric mean market risk premium for 1926-
- 29 2010.

<sup>&</sup>lt;sup>16</sup> <u>Ibbotson SBBI – 2011 Valuation Yearbook – Market Results for Stocks, Bonds, Bills and Inflation –</u> <u>1926-2010</u> (Morningstar, Inc., 2011) 23.

<sup>&</sup>lt;sup>17</sup> Ibbotson <u>2011 SBBI</u> 55.

A. In addition to calculating a CAPM derived common equity cost rate based upon
the historical arithmetic mean equity risk premium, albeit, incorrectly derived,
Staff also calculated a CAPM derived common equity cost rate using the longterm historical geometric mean equity risk premium. This latter calculation is
not a valid means of estimating the cost of capital based upon historical
returns.

7 The arithmetic mean return and not the geometric mean return is 8 appropriate for cost of capital purposes as noted in <u>2011</u>SBBI (see page 5 of

9 Schedule PMA-22)<sup>18</sup>:

10 The equity risk premium data presented in this book are 11 arithmetic average risk premia as opposed to geometric average risk premia. The arithmetic average equity risk premium can be 12 13 demonstrated to be most appropriate when discounting future 14 cash flows. For use as the expected equity risk premium in 15 either the CAPM or the building block approach, the arithmetic 16 mean or the simple difference of the arithmetic means of stock 17 market returns and riskless rates is the relevant number. This is 18 because both the CAPM and the building block approach are 19 additive models, in which the cost of capital is the sum of its 20 parts. The geometric average is more appropriate for reporting 21 past performance, since it represents the compound average 22 return. 23

24 The argument for using the arithmetic average is quite 25 straightforward. In looking at projected cash flows, the equity 26 risk premium that should be employed is the equity risk premium 27 that is expected to actually be incurred over the future time 28 periods. Graph 5-2 shows the realized equity risk premium for 29 each year based on the returns of the S&P 500 and the income 30 return on long-term government bonds. (The actual, observed difference between the return on the stock market and the 31 32 riskless rate is known as the realized equity risk premium.) 33 There is considerable volatility in the year-by-year statistics. At 34 times the realized equity risk premium is even negative.

35 36

Ibbotson <u>2011 SBBI</u> 56.

1 As discussed in my direct testimony at page 44, line 31 through page 2 47, line 16 and demonstrated on Schedule PMA-11, because historical total returns and equity risk premiums differ in size and direction over time, the 3 arithmetic mean provides insight into the variance and standard deviation of 4 5 returns, i.e., risk. Thus the prospect for variance, i.e., standard deviation, 6 captured in the arithmetic mean, provides the valuable insight needed by 7 investors and rate of return analysts alike to estimate the expected risk of Without such insight, investors cannot meaningfully evaluate 8 stocks. 9 prospective risk. Because the geometric mean relates the change over many 10 periods to a constant rate of change, the variance, i.e., year-to-year 11 fluctuations, and hence, risk, which is critical to rate of return analysis, is not 12 reflected in geometric mean returns / premiums.

The financial literature is quite clear on this point, that risk is measured by the variability of expected returns, i.e., the probability distribution of returns.<sup>19</sup> Pages 53 through 68 of <u>2011 SBBI</u> (see Schedule PMA-22) explain in detail why the arithmetic mean is the correct mean to use when estimating the cost of capital.

In addition, Weston and Brigham<sup>20</sup> provide the standard financial
 textbook definition of the riskiness of an asset when they state:

20 21 22 The riskiness of an asset is defined in terms of the <u>likely</u> variability of future returns from the asset. (emphasis added)

<sup>19</sup> Brigham (1989) 639.

Weston, J. Fred and Brigham, Eugene F., <u>Essentials of Managerial Finance Third Edition</u> (The Dryden Press, 1974) 272.

Morin also states<sup>21</sup>:

| 2<br>3 |    | The geometric mean answers the question of <u>what constant return</u><br>you would have to achieve in each year to have your investment |
|--------|----|--|
| 4      |    | growth match the return achieved by the stock market. The  |
| 5      |    | arithmetic mean answers the question of what growth rate is the  |
| 6      |    | best estimate of the future amount of money that will be produced  |
| 7      |    | by continually reinvesting in the stock market. It is the rate of  |
| 8      |    | return which, compounded over multiple periods, gives the mean   |
| 9      |    | of the probability distribution of ending wealth. (emphasis added)   |
| 10     |    |  |
| 11     |    | In addition, Brealey and Myers <sup>22</sup> note:   |
| 12     |    | The proper uses of arithmetic and compound rates of return from  |
| 13     |    | past investments are often misunderstood Thus the  |
| 14     |    | arithmetic average of the returns correctly measures the   |
| 15     |    | opportunity cost of capital for investments Moral: If the cost   |
| 16     |    | of capital is estimated from historical returns or risk premiums, use  |
| 17     |    | arithmetic averages, not compound annual rates of return. (italics   |
| 18     |    | in original)   |
| 19     |    |  |
| 20     |    | As previously discussed, investors gain insight into relative riskiness by   |
| 21     |    | analyzing expected future variability. This is accomplished by the use of the  |
| 22     |    | arithmetic mean of a distribution of returns / premiums. Only the arithmetic   |
| 23     |    | mean takes into account all of the returns / premiums, hence, providing  |
| 24     |    | meaningful insight into the variance and standard deviation of those returns /   |
| 25     |    | premiums.  |
| 26     | Q. | You have also stated that Staff erred in not including a forecasted market   |
| 27     |    | equity risk premium in its CAPM analysis. Please explain.  |
| 28     | A. | Staff relied exclusively upon historical market equity risk premiums which is in   |
| 29     |    | direct contrast to Staff's use of both historical and projected growth rates in its  |
| 30     |    | application of the DCF model. As stated previously, the cost of capital is   |

<sup>&</sup>lt;sup>21</sup> Morin 133. <sup>22</sup> Broalov P

 <sup>&</sup>lt;sup>2</sup> Brealey, R.A. and Myers, S.C., <u>Principles of Corporate Finance Fifth Edition</u> (McGraw-Hill Publications, Inc., 1996) 146-147.

1 prospective and while the arithmetic mean of long-term historical stock market 2 returns can provide insight into investors' expectations of stock market returns 3 because the arithmetic mean of historical returns provides investors with the 4 valuable insight needed to estimate future risk, it is also appropriate to use an 5 estimate of the forecasted or projected stock market return. One indication of 6 the forecasted stock market return can be derived using Value Line's 3-5 year 7 median total market price appreciation projections and dividend yield projections as explained in detail on pages 47 and 48 of my direct testimony 8 9 and derived in note 3 on page 3 of Schedule PMA-12. Based upon Value Line, 10 a forecasted total market return of 16.86% is indicated using the same three 11 months, July, August, and September 2011, used by Staff in developing the 12 dividend yield in its DCF analysis. When the average forecasted yield on 30-13 year U.S. Treasury bonds for 2012 and 2013, derived from Staff's Schedule 5 14 and discussed above, of 4.95% is subtracted from Value Line's forecasted total 15 market return, a forecasted market equity risk premium of 11.91% results 16 which, when averaged with the historical market equity risk premium of 6.70% 17 as reported by 2011SBBI, results in a market equity risk premium of 9.31%.

Q. You have stated that Staff also failed to apply the ECAPM to account for the
 fact that Security Market Line (SML) as described by the traditional CAPM is
 not as steeply sloped as the predicted SML. Please comment.

A. As discussed in my direct testimony at page 51, line 14 through page 52, line 4
and again at page 54, line 13 through page 56, line 8 of my direct testimony,
while numerous tests of the CAPM have confirmed its validity, these tests have

determined that "the implied intercept term exceeds the risk-free rate and the slope term is less than predicted by the CAPM."<sup>23</sup> These tests have also indicated that the expected return on a security is related to its risk by the following formula:

 $K = R_F + 0.25(R_M - R_F) + 0.75\beta(R_M - R_F)$ 

6 Some critics of the ECAPM model claim that using adjusted betas in a 7 traditional CAPM amounts to using an ECAPM but such a claim is not valid. As discussed in my direct testimony, using adjusted betas in a CAPM analysis 8 9 is not equivalent to the ECAPM. Betas are adjusted because of the regression 10 tendency of betas to converge toward 1.0 over time, i.e., over successive As discussed previously, numerous studies have 11 calculations of beta. 12 determined that the SML described by the CAPM formula at any given moment in time is not as steeply sloped as the predicted SML. As noted in my direct 13 testimony, at pages 54 and 55. Morin<sup>24</sup> states: 14

15 Some have argued that the use of the ECAPM is inconsistent with the use of adjusted betas, such as those supplied by Value 16 Line and Bloomberg. This is because the reason for using the 17 18 ECAPM is to allow for the tendency of betas to regress toward 19 the mean value of 1.00 over time, and, since Value Line betas 20 are already adjusted for such trend [sic], an ECAPM analysis 21 results in double-counting. This argument is erroneous. 22 Fundamentally, the ECAPM is not an adjustment, increase or 23 decrease, in beta. This is obvious from the fact that the expected 24 return on high beta securities is actually lower than that produced 25 by the CAPM estimate. The ECAPM is a formal recognition that 26 the observed risk-return tradeoff is flatter than predicted by the 27 CAPM based on myriad empirical evidence. The ECAPM and the 28 use of adjusted betas comprised two separate features of asset 29 pricing. Even if a company's beta is estimated accurately, the

<sup>&</sup>lt;sup>23</sup> Morin 175.

<sup>&</sup>lt;sup>24</sup> Morin 191.

| 1<br>2<br>3<br>4<br>5<br>6                                     |    | CAPM still understates the return for low-beta stocks. Even if the ECAPM is used, the return for low-beta securities is understated if the betas are understated. Referring back to Figure 6-1, the ECAPM is a return (vertical axis) adjustment and not a beta (horizontal axis) adjustment. Both adjustments are necessary.  |
|--|----|--|
| 7  |    | Moreover, the slope of the SML should not be confused with beta. As  |
| 8  |    | also noted in my direct testimony at page 55, Eugene F. Brigham, finance   |
| 9  |    | professor emeritus and the author of many financial textbooks states <sup>25</sup> :   |
| 10<br>11<br>12<br>13<br>14                                     |    | The slope of the SML reflects the degree of risk aversion in the economy – the greater the average investor's aversion to risk, then (1) the steeper is the slope of the line, (2) the greater is the risk premium for any risky asset, and (3) the higher is the required rate of return on risky assets. <sup>12</sup>   |
| 15<br>16<br>17<br>18<br>19<br>20<br>21<br>22<br>23<br>24<br>25 |    | <sup>12</sup> Students sometimes confuse beta with the slope of the SML. This is a mistake. As we saw earlier in connection with Figure 6-8, and as is developed further in Appendix 6A, beta does represent the slope of a line, but <i>not</i> the Security Market Line. This confusion arises partly because the SML equation is generally written, in this book and throughout the finance literature, as $k_i = R_F + b_i(k_M - R_F)$ , and in this form $b_i$ looks like the slope coefficient and $(k_M - R_F)$ the variable. It would perhaps be less confusing if the second term were written $(k_M - R_F)b_i$ , but this is not generally done. |
| 25<br>26   | Q. | What would Staff's CAPM results have been had Staff relied upon a correctly-   |
| 27   |    | derived historical market equity risk premium, included a forecasted market  |
| 28   |    | equity risk premium, a forecasted risk-free rate as well as the ECAPM?   |
| 29   | A. | In Column 4 on Schedule PMA-23, I have derived the traditional CAPM, the   |
| 30   |    | version applied by Staff, using a Staff provided average forecasted risk-free  |
| 31   |    | rate of 4.95% for 2012 and 2013 and an average market equity risk premium  |
| 32   |    | based upon the arithmetic mean historical market equity risk premium, correctly  |
| 33   |    | calculated as described above, coupled with a forecasted market equity risk  |

<sup>&</sup>lt;sup>25</sup> Brigham, Eugene F., <u>Financial Management – Theory and Practice</u>, 4<sup>th</sup> Ed. (The Dryden Press, 1985) 203.

premium. This results in a traditional CAPM-derived common equity cost rate
of 11.93%. In Column 5 on Schedule PMA-23, I have derived an ECAPM,
based upon the forecasted risk-free rate and correctly-derived average
historical and projected market equity risk premium. The ECAPM-derived
common equity cost rate is 12.51%.

When averaged, the traditional CAPM results of 11.93% and the 6 7 ECAPM results of 12.51% result in a CAPM of 12.23%. Such a cost rate corroborates neither Staff's range of DCF results of 8.97% - 9.97% nor its 8 9 recommended range of common equity cost rate of 9.40% - 10.40%. In 10 addition, these cost rates are further understated because they do not reflect 11 either MAWC's greater unique business risks relative to of Staff's proxy group 12 of six water companies, the greater financial risk of Staff's recommended 13 common equity ratios or flotation costs.

#### 14 **Recommended Common Equity Cost Rate**

- Q. Please discuss Staff's recommended common equity cost rate range of 9.40%
   10.40%, with a midpoint of 9.90%.
- A. Staff's recommended common equity cost rate range of 9.40% 10.40% is
  inadequate for three reasons; 1) such a cost rate range provides an insufficient
  achieved return on the book common equity of MAWC; and, 2) such a cost rate
  does not adequately reflect either MAWC's greater risk relative to Staff's proxy
  group due to its unique risks, the greater financial risk of Staff's recommended
  common equity ratios or flotation costs.

23 Q. How does Staff's recommended range of common equity cost rate of 9.40% -

10.40% with a midpoint of 9.90% compare with the expected ROEs of its prosy
 group of six water companies?

3 It is far below the level of earnings expected by Value Line for the four Α. companies in its group of six comparable water utility companies for which 4 5 Value Line publishes a projected ROE for the years 2014-2016. The latest 6 (October 21, 2011) Value Line Ratings & Reports (Standard Edition) for 7 American States Water Company, Aqua America, Inc., California Water Service Group and SJW Corporation, (there are no projections for Connecticut 8 9 Water Service, Inc. or York Water Company) indicate that Value Line expects 10 them to earn 12.0%, 12.5%, 11.0 and 8.0% on year-end book common equity 11 (see Schedule PMA-39) over the next 3-5 years averaging, 10.9%. While 12 these forecasts are for earnings on book common equity, it must be 13 remembered that the return on common equity authorized in this proceeding 14 will be applied to the book value of the common equity financed portion of 15 MAWC's and will therefore become MAWC's opportunity for earnings on book 16 value. An opportunity to earn a range of return on book common equity of 17 either Staff's recommended range of 9.40% - 10.40%, or Staff's recommended 18 midpoint of 9.90% is woefully inadequate in comparison with these expected 19 returns on book common equity of comparable water companies.

20 Such a common equity cost rate range is also inconsistent with the 21 comparability of returns standard enunciated in the <u>Hope</u> decision which 22 states:

23The return to the equity owner should be commensurate with24returns on investments in other enterprises having corresponding

1 2 risks.

Therefore, Staff's recommended common equity cost rate range should be
rejected by the MoPSC in setting rates for MAWC in this proceeding.

- Q. Previously you noted that Staff's recommended common equity cost rate range
  of 9.40% 10.40%, with a midpoint of 9.90% does not adequately reflect either
  MAWC's greater risk relative to Staff's proxy group due to its unique risks, the
  greater risk of Staff's recommended capital structure ratios or flotation costs.
  Please explain.
- 10 Α. As discussed in my direct testimony at page 18, line 1 through page 20, line 23 11 and in Company Witness Dennis R. Williams direct testimony, MAWC faces 12 unique risks due to the availability and quality of its source of supply; exposure 13 to flooding; non-contiguous service territory, concentration of investment and 14 revenues in a single metropolitan area, St. Louis; and various unique regulatory risks. Because MAWC is nearly identical in size to Staff's proxy group of six 15 16 water companies as shown on Schedule PMA-24 in my opinion, a business risk adjustment or 0.35% (slightly less than my recommended adjustment of 17 18 0.40%) is warranted.
- Q. Is there a way to quantify a financial risk adjustment due to the greater financial
  risk of Staff's recommended capital structure ratios?
- A. Although Staff arrived at its recommended common equity cost rate range of
   9.40% 10.40% by adding a credit rating differential of 0.43% to its indicated
   DCF cost range to reflect American Water's Standard & Poor's (S&P) bond
   rating of BBB+ relative to the average S&P credit rating of A for its proxy group,

1 Staff has provided no empirical support that MAWC would be assigned a bond 2 / or credit rating of BBB+ by S&P. Therefore, should the MoPSC adopt Staff's 3 recommended common equity ratio, it is necessary to adjust the common equity cost rate to reflect the greater financial risk inherent in Staff's 4 recommended capital structure ratios of 56.76% long-term debt, 0.29% 5 6 preferred stock and 42.95% common equity. Staff's recommended long-term 7 debt ratio of 56.76% is significantly higher than the average long-term debt ratio of 50.87% for Staff's proxy group of six water companies as can be 8 9 gleaned from page 1 of Schedule PMA-25. Consequently, an upward 10 adjustment to the indicated common equity cost rate based upon the six water 11 companies is necessary. An indication of the magnitude of the necessary financial risk adjustment is given by the Hamada equation<sup>26</sup>, which un-levers 12 13 and then re-levers betas based upon changes in capital structure. Using the 14 Hamada equation as described in detail on page 63, line 5 through page 65, 15 line 2 of my direct testimony, an upward adjustment for the greater financial 16 risk inherent in Staff's recommended capital structure ratios is 0.75%.

Q. You also previously noted that Staff did not reflect flotation costs in its
 recommended common equity cost rate. Please comment

A. As discussed on page 65, line 5 through page 67, line 11, of my direct
 testimony, it is necessary to include flotation costs, i.e., those costs associated
 with the sale of new issuance of common stock, in the common equity cost rate
 recommendation. There is no other mechanism in the ratemaking paradigm

<sup>&</sup>lt;sup>26</sup> Brigham, Eugene F. and Daves, Phillip R., <u>Intermediate Financial Management</u>, 9<sup>th</sup> Ed., Thomson/Southwestern, 2007, p. 533.
| 1 | with which such costs can be recovered. Using the methodology described on    |
|---|---|
| 2 | page 67, lines 5 – 11 of my direct testimony and the corrected Staff DCF cost |
| 3 | rate results in a flotation cost adjustment of 0.15%.                         |

- Q. What would Staff's recommendation be had Staff properly reflected flotation
  costs, the greater financial risk inherent in its recommended capital structure
  and MAWC's greater business risks due to its unique risks?
- A. It would be a range of 10.22% 11.22%, with a midpoint of 10.72%. (10.22% =
  8.97% + 0.15% + 0.75% + 0.35%) 11.22% = 9.97% + 0.15% + 0.75% +
  9.0.35%).
- Q. Based upon the corrected Staff DCF and CAPM discussed previously, what
   would Staff's recommendation be once flotation costs, the greater financial risk
   inherent in its recommended capital structure and MAWC's greater business
   risks due to its unique risks are reflected?
- A. As shown on Schedule PMA-26, the corrected Staff DCF is 10.53% (Line No.
  1) and the corrected Staff CAPM is 12.23% (Line No. 2). These cost rates
  average 11.38% (Line No. 3). When the flotation costs (Line No. 4), financial
  risk (Line No. 5) and business risk (Line No. 6) adjustments are added, a
  corrected indicated Staff common equity cost rate of 12.63% results as
  summarized on Schedule PMA-26.
- Q. Are you aware the MoPSC Staff has provided workpapers containing updated
   Schedules 2-1 through 21 which reflect a range of common equity cost rate of
   8.95% 9.95% with a midpoint of 9.45%?
- A. Yes. While recognizing that Mr. Barnes updated schedules as of December 8,

2011 reflect a different return on common equity than originally filed in the Staff
 Report on November 17, 2011, I have limited this rebuttal testimony to Staff's
 recommendations of the originally filed Staff Report. However, I reserve the
 right to file additional rebuttal testimony in response to any rebuttal or
 supplemental testimony or corrected Staff Report.

6

## MIEC WITNESS MICHAEL P. GORMAN

7 8

# COMMON EQUITY COST RATE

# 9 Current Capital Market Conditions

10 On page 6, line 9 through page 7, line 2, Mr. Gorman asserts that the cost of Q. 11 capital for utilities is "no higher than it was" in MAWC's last rate case when the 12 order was issued in June 2010. He bases this assertion on the decline of 13 approximately 90-100 basis points in utility bond yields since MAWC's last rate 14 case. All else equal, this would indicate an approximate 50 basis point decline in the cost of capital<sup>27</sup>. While it is true that utility bond yields have declined 15 16 since June 2010, market equity risk premiums have risen since then, providing 17 an indication that utility equity risk premiums have also risen in response to the 18 recent fragile recovery from the Great Recession. As shown on page 1 on 19 Schedule PMA-27, the projected market equity risk premium based upon a 20 forecasted total return derived from Value Line's 3-5 year average total market 21 appreciation plus average annual forecasted dividend yield at the beginning of 22 each month from June 2011 (the date of the order in MAWC's last rate case)

<sup>&</sup>lt;sup>27</sup> Morin 128-129.

through December 2011 minus the <u>Blue Chip Financial Forecast</u> consensus
estimate of about 50 economists of the expected yield on 30-year U.S.
Treasury notes for the following six quarters, also, at the beginning of each
month, has risen 131 basis points or 1.31%, from 11.09% in June 2011 to
12.40% in December 2011.

6 Likewise the actual monthly market equity risk premium for the S&P 500 7 Composite Index (S&P 500) relative to 30-day U.S. Treasury Bill yields shows increased from a negative 8.33% for May 2011 to a negative 0.84% for 8 9 November 2011. Using the actual monthly market equity risk premiums for the S&P 500 from July 1926 through May 2010 and November 2011, respectively, 10 and the Predictive Risk Premium Model<sup>TM</sup> (PRPM<sup>TM</sup>)<sup>28</sup> described in Schedule 11 12 PMA-28, predicted market equity risk premiums of 10.40% at May 2011 and 13 10.52% at November 2011 are indicated, which show a clear increase in the 14 predicted market equity risk premium.

In view of the foregoing, Mr. Gorman's conclusion that utilities' cost of
capital has declined based solely on a review of the trend in public utility bond
yields is misleading and incomplete.

18 Proxy Group Selection

- Q. Do you have any comment upon Mr. Gorman's use of a gas utility proxy groupin addition to a water utility proxy group?
- A. Yes. Mr. Gorman's use of a gas utility proxy group is inappropriate because, as

<sup>&</sup>lt;sup>28</sup> Ahern, Pauline M, Hanley, Frank J. and Michelfelder, Richard A. (2011) "A New Approach to Estimating the Cost of Common Equity Public Utilities." Journal of Regulatory Economics, 40:261-278, DOI 10.1007/s11149-011-9160-5.

1 discussed at page 7, line 13 through page 17, line 23 of my direct testimony 2 and shown on Schedules PMA-2 and PMA-3, the water utility industry faces 3 unique investment risks relative to the electric, combination electric and gas 4 and natural gas utility industries. Using a proxy group comprised of natural gas distribution companies for an ROE analysis for a water company, like MAWC, 5 6 cannot reflect specific water industry risk, and is therefore inadequate for water 7 utility cost of capital purposes. Consequently, I find it unnecessary to discuss the results pertaining to Mr. Gorman's gas utility proxy group because those 8 9 results are not reflective of the unique risks of water utilities in general, nor of 10 MAWC, specifically.

#### 11 Discounted Cash Flow Model (DCF)

Q. Please comment upon Mr. Gorman's discussion of the results of his application
of the constant growth, or single stage, DCF model.

14 Α. Mr. Gorman, as shown on page 1 of Schedule MPG-5 and on page 18, Table 4 15 of his direct testimony, derived an average constant growth DCF model cost 16 rate of 10.81% for his water proxy group and a median of 11.82%. These cost 17 rates include a negative 1.08% constant growth DCF result for Middlesex 18 Water Company (Middlesex) because the single security analysts' forecast of 19 EPS growth for Middlesex is a negative 5.00% as shown on page 2 of 20 Schedule MPG-4. Since it is illogical that investors would invest with the 21 expectation of losing money, Middlesex's negative 1.08% DCF result is not 22 Schedule PMA-29 recalculates Mr. Gorman's average and meaningful. 23 median constant growth DCF results excluding Middlesex. They are 12.51%

and 11.93%, respectively. However, these cost rates do not reflect MAWC's
 lower financial risk and greater unique business risks relative to the proxy
 group of water companies nor flotation costs which will be discussed
 subsequently.

5 Nevertheless, Mr. Gorman concludes that the constant growth DCF result 6 for his water proxy group is unreasonably high on page 18, lines 3 and 4 7 because it reflects a growth rate which he claims "is far too high to be a 8 reasonable or reliable estimate of a long-term sustainable growth rate."

9 His conclusion is based upon his contention that projected growth in
10 Gross Domestic Product (GDP) "represents a ceiling, or high-end, sustainable
11 growth rate for a utility over an indefinite period of time", because the dividend
12 growth for the market as a whole tracked the GDP growth rate during the
13 period 1926 through 2008 as noted on page 20, lines 4 – 26 of Mr. Gorman's
14 direct testimony. Those reasons, however, are not persuasive.

Hence, there is no basis for ultimately rejecting the corrected average
 constant growth DCF cost rate of 12.51% or median cost rate of 11.93% for his
 water proxy group.

Q. Why are the three-to-five year growth rate projections made by security
 analysts in earnings per share reasonable to use in a constant growth, single
 stage, DCF?

- A. Mr. Gorman's statements are contradicted by his earlier testimony at page 12,
  line 19 through page 13, line 5 where he states the following:
- [f]or purposes of determining the market-required return on
   common equity, one must attempt to estimate investors'

consensus about what the dividend or earnings growth rate will
 be, and not what an individual investor or analyst may use to
 form individual investment decisions.

4

12

5 Security analyst growth estimates have been shown to be more 6 accurate predictors of future returns than growth rates derived 7 from historical data. Assuming the market generally makes 8 rational investment decisions, forward-looking growth projections 9 are more likely to be the growth estimates considered by the 10 market that influence observable stock price than are growth 11 rates derived from only historical data.

As previously discussed in detail in this rebuttal testimony, there is a wealth of empirical and academic literature, including Cragg and Malkiel and Vander Weide and Carleton, which support the superiority of analysts' forecasts of EPS as measures of investor expectations.

Moreover, Myron Gordon, who first introduced the standard DCF model adopted for utility ratemaking, which both Mr. Gorman and I use, came to recognize that his original "Gordon Model" had a serious limitation by assuming that dividend expectations can be represented by retention growth. Dr. Gordon later came to the conclusion that security analysts' growth forecast in earnings per share were superior predictors of the variation in stock prices.

In all of the previously cited studies, the referenced analyst's growth forecasts were forecasts of growth in EPS. As the recent volatility of the stock market has shown, EPS is a prime, but not the sole, driver of market price movements Therefore, analyst's forecasts of EPS growth are extremely relevant to investors in making their investments decisions. It is the goal of rate of return analysts, such as Mr. Gorman and myself and to which he agrees, to emulate investor behavior. Therefore, consistent with the EMH, the foundation

of modern investment theory, the market prices of securities reflect all relevant
 information at all times. This implies that prices adjust instantaneously to new
 information, such as analysts' forecasts of EPS growth.

In addition, as noted above, Agrawal and Chen concluded that analysts
are not able to systematically mislead investors with optimistic stock
recommendations.

- Q. At lines 7 through 12 on page 20 of his direct testimony, Mr. Gorman quotes
  Eugene F. Brigham and Joel F. Houston, in support of his "contention that over
  the long term, a company's earnings and dividends will grow at a comparable
  rate to the growth rate of the U.S. GDP." Please comment.
- 11 A. I do not have a copy of the specific text book cited by Mr. Gorman. However,
- 12 the quotation also appears on page 164 of Intermediate Financial
- 13 <u>Management<sup>29</sup></u>. In <u>Intermediate Financial Management</u>, the quotation does not
- 14 end at the conclusion of Mr. Gorman's citation. The entire paragraph reads:
- 15 The constant growth model is often appropriate for mature 16 companies with a stable history of growth. Expected growth rates vary somewhat among companies, but dividend growth for most 17 18 mature firms is generally expected to continue to the future at 19 about the same rate as nominal grow domestic product (real GDP 20 plus inflation). On this basis, one might expect the dividends of an 21 average, or "normal," company to grow at a rate of 5 to 8 percent a 22 *vear.* (italics added for emphasis)
- 24 Continuing, on pages 165 through 167, the authors provide an example 25 of the application of the non-constant DCF, assuming a normal growth rate of 26 8% which they identify as "the assumed average for the economy." Thus,
  - 29

Brigham and Daves, 164-167.

1assuming that this same information appears in Fundamentals of Financial2Management, from which Mr. Gorman quoted, although he relied upon the3Brigham / Houston quotation to support the use of the growth in nominal GDP4for use in a non-constant DCF model, Mr. Gorman ignored the authors5recommendation of an assumed 8% normal growth rate to be used in the non -6constant DCF

- Q. At lines 13 26 on page 20 of his direct testimony, Mr. Gorman cites page 67
  of Morningstar, Inc.'s <u>2009 SBBI</u> to support using GDP growth as a maximum
  sustainable growth rate. Please comment
- 10 Α. The study reported in the 2009 SBBI relates growth in the earnings and 11 dividends of the stock market as a whole to GDP growth from 1926-2008. 12 Since the stock market as a whole, whether measured by the NYSE or the S&P 13 500, is a broad based representation of all the common stocks traded in the 14 U.S., it stands to reason that the earnings and dividends of the market as a 15 whole would track GDP growth. However, neither the 2009 SBBI nor Mr. 16 Gorman have provided any empirical support that the earnings and dividends 17 of utility companies, in general, or water companies, in particular, or indeed any 18 specific company or industry, track GDP growth.
- Q. On page 19, lines 21 23, Mr. Gorman states that "[h]ence, nominal GDP
  growth is a very conservative, albeit overstated, proxy for utility sales growth,
  rate base growth, and earnings growth." Please comment.
- A. Mr. Gorman has provided no empirical evidence that in the third stage of a
   multi-stage DCF analysis any company, especially the relatively stable and

1 mature utility companies, would grow at the average growth rate of the U.S. 2 economy. The average growth in the U.S. economy is just that, an average. 3 Some companies will grow faster and some will grow more slowly. That the 4 growth in nominal GDP is an average is demonstrated on Schedule PMA-30 which shows the nominal GDP for the years 2001-2010 as a whole and by 5 6 industry. From 2009-2010, nominal GDP grew 3.83% and 4.73% on average 7 for the nine years ending 2010. In contrast, the construction component of nominal GDP declined 5.93% from 2009 to 2010 and grew a meager 0.34% on 8 9 average for the nine years ending 2010. Likewise, the utilities component of 10 nominal GDP grew 2.83% from 2009 to 2010 and an average 6.14% for the 11 nine years ending 2010. In addition, it is a mismatch to use five- to ten-years 12 growth in GDP as a proxy either for the years eleven through perpetuity. There 13 is no evidence that a five- to ten-years growth rate in GDP accurately 14 represents the in perpetuity growth rate in GDP.

Hence, there is no valid rationale for undertaking a multi-stage DCFanalysis.

Q. Do you agree with Mr. Gorman's use of a sustainable growth constant growthDCF analysis?

A. No. As shown on pages 1 and 2 of Schedule MPG-8, he calculates sustainable
 growth for each company in his water proxy group based upon 3-5 year
 projections from <u>Value Line</u>. His allowance for growth caused by the sale of new
 common stock above book value is also based upon the five-year growth in
 shares from 2010 through 2014-2016. Hence, Mr. Gorman's sustainable growth

methodology is a short-term forecast, no longer than the security analysts' five year forecasts of EPS growth used in his first consensus analyst's growth
 constant growth DCF analysis. Moreover, he has provided no empirical support
 that sustainable growth accurately represents investors' expected growth.

5 Moreover, the sustainable growth methodology is inherently circular 6 because it relies upon an expected ROE on book common equity which is then 7 used in a DCF analysis to establish a common equity cost rate related to the 8 market value of the common stock which, if authorized as the allowed ROE in 9 this proceeding, will become the expected ROE on book common equity. Mr. 10 Gorman's 9.67% sustainable growth constant growth DCF result, which forms 11 the basis, in part, of his recommended allowed DCF derived ROE on book 12 common equity, is lower than the expected average Value Line ROE of 10.78% 13 shown on page 1 of Schedule MPG-8 for the very proxy group used to derive his recommended allowed ROE. Schedule PMA-31, an excerpt from Roger A. 14 15 Morin's book New Regulatory Finance, which corroborates the circular nature of sustainable growth. 16

17 In view of the foregoing, it is clear that Mr. Gorman's application of 18 the sustainable growth constant growth DCF is circular and ignores the basic 19 principle of rate base / rate of return regulation, namely, that the cost of equity 20 which will be authorized in this proceeding will be applied to the jurisdictional 21 <u>book value</u> rate base of MAWC and become the allowed future earned return 22 on book common equity, i.e., the expected ROE component of the sustainable 23 growth method.

In view of all of the foregoing, the use of analysts' forecasts of EPS
 growth should not be rejected when estimating today's market cost of capital.
 There is no need to reject the empirical evidence of the proven reliability of
 analysts' forecasts of EPS by turning to either a sustainable growth constant
 growth or a multi -stage DCF model.

6

#### Risk Premium Model (RPM)

- 7 Q. Do you have any comments regarding Mr. Gorman's risk premium analysis?
- A. Yes. My comments center on the time period over which he estimates the
  equity risk premium and his use of authorized returns to do so.
- Q. Do you agree with Mr. Gorman's use of the years 1986 3<sup>rd</sup> quarter 2011 to
   determine an equity risk premium?
- 12 Α. No. Mr. Gorman states on page 27, line 10 through page 28, line 6 of his direct testimony that he relied upon the period 1986 through the 3<sup>rd</sup> quarter 13 14 2011, because public utility stocks have consistently traded at a premium to 15 book value during that time. He concludes, on lines 1 and 2 on page 28, that 16 "[o]ver this time period, regulatory authorized returns were sufficient to support 17 market prices that at least exceeded book value." Use of such a short time 18 period is especially inappropriate and inconsistent in view of his use of a multi-19 stage growth DCF model and his emphasis upon long-term sustainable growth. 20 As discussed previously in this rebuttal testimony and my direct testimony, the 21 2011 SBBI makes it clear that the arbitrary selection of short historical periods 22 is highly suspect and unlikely to be representative of long-term trends in market 23 data. Page 9 of Schedule PMA-22 clearly shows that it is inappropriate to

- 1 estimate a market equity risk premium over a short period of time. For example
- 2 on page 7 the <u>2008 SBBI</u> states:

The estimate of the equity risk premium depends on the length of the data series studied. . . requires a data series long enough to give a reliable average. . . because an average of the realized equity risk premium is quite volatile when calculated using a short history, using a long series makes it less likely that the analyst can justify any number he or she wants. . .

10

15 16

17

18

19

20

- 11 As discussed in my direct testimony on page 38, lines 1 10, Bonbright,
- 12 *et al* make it very clear that the market prices of the common stocks of public
- 13 utilities are influenced by factors which are beyond the direct influence of the
- 14 regulatory process. In addition, Phillips<sup>30</sup> states:
  - Many question the assumption that market price should equal book value, believing that 'the earnings of utilities should be sufficiently high to achieve market-to-book ratios which are consistent with those prevailing for stocks of unregulated companies.'
- 22 Schedule PMA-32 demonstrates that there is no relationship between the 23 market-to-book ratios and the earned rates of return on book common equity for the S&P Industrial Index and its successor, the S&P 500 Composite Index 24 25 over a long period of time. On Schedule PMA-32, I have shown the market-to-26 book ratios, rates of return on book common equity (earnings/book ratios), 27 annual inflation rates, and the earnings/book ratios net of inflation (real rate of 28 earnings) annually for the years 1947 through 2010. In each and every year, 29 the market-to-book ratios of the S&P Industrial Index equaled or exceeded 1.00

<sup>&</sup>lt;sup>30</sup> <u>Id</u>., at p. 395.

times. In 1949, the only year in which the market-to-book ratio was 1.00 (or
100%), the real rate of earnings on book equity, adjusted for <u>deflation</u>, was
18.1% (16.3% + 1.8%). In contrast, in 1961, when the S&P Industrial Index
experienced a market-to-book ratio of 2.01 times, the real rate of earnings on
book equity for the Index was only 9.1% (9.8% - 0.7%). In 1997, the
preliminary market-to-book ratio for the Index was 5.57 times, while the
average real rate of earnings on book equity was 21.6% (23.3% - 1.7%).

8 This analysis clearly demonstrates that competitive, unregulated 9 companies have never sold below book value, on average, and have sold at 10 book value in only one year since 1947. The data show that there is no 11 relationship between earnings/book ratios and market-to-book ratios.

Because this lack of a relationship between earnings/book ratios and market-to-book ratios covers a 64-year period, 1947 through 2010, it cannot be validly argued that going forward a relationship would exist between earnings/book ratios and market-to-book ratios. The analysis shown on Schedule PMA-32, coupled with the supportive academic literature, demonstrate the following:

that while regulation is a substitute for marketplace competition, it can
 influence but not directly control market prices, and, hence, market-to book ratios; and,

21 2. that the rates of return investors expect to achieve and which
 influence their willingness to pay market prices well in excess of book
 values have no meaningful, direct relationship to rates of earnings on

1

book equity.

2 Because this lack of relationship between earnings/book ratios and 3 market-to-book ratios covers a period of nearly 65 years, it is not reasonable to 4 assume that a direct relationship will exist between rates of earnings on book 5 common equity and market-to-book ratio into the future. Schedule PMA-32 6 confirms that while regulation is a substitute for marketplace competition, it has 7 but a limited effect on, but no direct control over the market prices and hence market-to-book ratios of regulated utilities. Thus, no valid conclusion of equity 8 9 risk premia can be drawn for the 1986 to first guarter 2008 because of market-10 to-book ratios in excess of one.

Have you applied an appropriate risk premium model to Mr. Gorman'swater and gas distribution proxy groups?

13 Α. Yes. That information is shown on Schedule PMA-33. Using the same risk 14 premium methodology described in my direct testimony on page 40, line 7 15 through page 50, line 13, a risk premium indicated common equity cost rate is 16 10.61% for Mr. Gorman's proxy group of water companies based upon market 17 conditions at the time he prepared his direct testimony as summarized on page 18 1, Schedule PMA-33. However, this cost rate does not reflect MAWC's lower financial risk and greater unique business risks relative to the proxy group of 19 20 water companies nor flotation costs which will be discussed subsequently.

21

Capital Asset Pricing Model

22 Q. Please comment upon Mr. Gorman's application of the CAPM.

A. Mr. Gorman's application of the CAPM is flawed for three reasons. First, his

derivation of the market equity risk premium is incorrect. Second, his "forward looking" equity risk premium is not truly a prospective equity risk premium.
 Third, Mr. Gorman failed to utilize the ECAPM in addition to the traditional
 CAPM.

5 Q. How is Mr. Gorman's historical market equity risk premium incorrectly derived? Mr. Gorman's market equity risk premium is the difference between the 6 Α. 7 arithmetic mean 1926-2010 total return on large company stocks of 11.9% and 8 the arithmetic mean 1926-2010 total return on long-term government bonds of 9 5.9% from the 2011 SBBI which results in a 6.0% market equity risk premium. 10 As discussed previously, the correct derivation of the historical market equity 11 risk premium is the difference between the total return on large company 12 stocks of 11.9% and the arithmetic mean 1926-2010 income return on long-13 term government bonds of 5.2%, resulting in a market equity risk premium of 14 6.7%. As discussed previously, the income return on long-term government 15 bonds is the appropriate return to use in the estimation of the market equity risk 16 premium because it represents the riskless portion of the return as discussed 17 previously and noted by the 2011 SBBI in Schedule PMA-22.

Q. Why is Mr. Gorman's "forward-looking" equity risk premium not truly forward-looking?

A. Mr. Gorman derived his "forward-looking" equity risk premium by merely adding
 a current consensus analysts' inflation projection to the <u>2011 SBBI's</u> long-term
 historical arithmetic mean real market return for the years 1926-2010. It is not
 appropriate to try and match a current forecast of inflation, 2.3% from *Blue*

1 Chip Financial Forecasts, with an average real market return over a period of 2 85 years. In my opinion, investors would not attempt to do such a thing. 3 Rather, they would be influenced by a forecast such as that published by Value 4 Line which is widely subscribed to and is available in the business reference 5 section of most libraries. A more appropriate method of deriving the 6 prospective equity market return is based upon Value Line's projected 3-5 year 7 market appreciation potential, which when converted to an annual rate plus the 8 market's median expected dividend yield results in a forecasted total annual 9 market return of 18.29% for the thirteen-weeks ending October 21, 2011 and derived as explained in Note 3 on page 2 of Schedule PMA-12. 10 This 11 methodology yields a truly prospective market return which is based upon an 12 important investor-influencing publication.

Q. Why should Mr. Gorman have included an ECAPM analysis in deriving his
 CAPM-based common equity cost rate?

A. As discussed previously in this rebuttal testimony and in my direct testimony at
 page 51, line 14 through page 52, line 4 and again at page 54, line 13 through
 page 56, line 8, the empirical Security Market Line (SML) described by the
 traditional CAPM is not as steeply sloped as the predicted SML. As Morin<sup>31</sup>
 notes:

20 21 22

> 23 24

. . .low-beta securities earn returns somewhat higher than the CAPM would predict, and high-beta securities earn less than predicted.

Hence, both the traditional CAPM and ECAPM should be used in deriving

<sup>&</sup>lt;sup>31</sup> Morin,175.

1 a CAPM-based common equity cost rate. I have shown the results of applying 2 both the traditional CAPM and ECAPM to Mr. Gorman's water companies using a correctly derived historical market equity risk premium. As shown on page 1 3 of Schedule PMA-34 the traditional CAPM result is 11.71%, while the ECAPM 4 5 result is 12.39%. The average of both cost rates is 12.05%. However, this 6 cost rate does not reflect MAWC's lower financial risk and greater unique 7 business risks relative to the proxy group of water companies nor flotation costs which will be discussed subsequently. 8

9 Financial Integrity

- Q. Please comment upon Mr. Gorman's financial integrity analysis at page 37, line
  13 through page 40, lint 12 of his direct testimony.
- A. In view of S&P's revised financial matrix, Mr. Gorman's comparison to the
   former S&P financial benchmark financial ratios is misplaced and should be
   disregarded, notwithstanding his qualification on page 38, lines 6 8 of his
   direct testimony, that the "the effect of integrating the utility metrics with those
   of general corporate bonds resulted in a reduction to the transparency in S\*P's
   credit metric guideline for utilities."
- Mr. Gorman has provided no empirical evidence to assume that American Water is an appropriate "risk proxy affiliate" for MAWC. As discussed previously in this rebuttal testimony, American Water has regulated operations in twenty (20) states, thus benefiting from geographical and regulatory diversity. Also, American Water is a much large company than MAWC. Clearly, the risks of American Water on a consolidated basis are not

- 1 similar to those of MAWC.
- 2 Moreover, S&P is clear in its description of its revised ratings matrix that
- 3 they do not assign a credit, bond rating, business or financial risk profile based
- 4 upon a spot financial metrics as Mr. Gorman has done on page 33 of his direct
- 5 testimony. On pages 4 and 5 of Schedule PMA-4, S&P states:
- 6 The rating matrix indicative outcomes are what we typically 7 observe – but are not meant to be precise indications or 8 guarantees of future rating opinions. . . . Still, it is essential 9 to realize that the financial benchmarks are guidelines, neither 10 gospel nor guarantees. . . . Moreover, our assessment of 11 financial risk is not as simplistic as looking at a few ratios.
- 12

## 13 **Recommended Common Equity Cost Rate**

- Q. Do the corrected MIEC DCF, RPM and CAPM, discussed previously,
   adequately reflect flotation costs, the lower financial risk inherent in MAWC's
   capital structure and MAWC's greater business risks due to its unique risks are
   reflected?
- 18 No. As discussed in my direct testimony at page 18, line 1 through page 20, Α. 19 line 23 and in Company Witness Dennis R. Williams direct testimony, MAWC 20 faces unique risks due to the availability and quality of its source of supply; 21 exposure to flooding; non-contiguous service territory, concentration of 22 investment and revenues in a single metropolitan area, St. Louis; and various 23 unique regulatory risks. Because Mr. Gorman's proxy group is nearly identical 24 in size to my proxy group of water companies as shown on Schedule PMA-24, 25 in my opinion, my originally recommended business risk adjustment 0.40% is 26 warranted.

Q. Is there a way to quantify a financial risk adjustment due to the slightly lower
 financial risk of MAWC's capital structure ratios?

Mr. Gorman accepted the Company's capital structure ratios. Although Mr. 3 Α. Gorman concluded that they were similar to those of his proxy group, MAWC's 4 capital structure actually contains somewhat less financial risk that the proxy 5 6 group, as the proxy group's average long-term debt ratio at December 2010 7 was 50.73% as shown on Schedule PMA-25, in comparison with MAWC's requested long-term debt ratio of 49.36%. Therefore, it is necessary to adjust 8 9 the common equity cost rate to reflect the lower financial risk inherent of 10 MAWC's capital structure ratios relative to Mr. Gorman's proxy group. 11 Consequently, an upward adjustment to the indicated common equity cost rate 12 based upon the six water companies is necessary. An indication of the 13 magnitude of the necessary financial risk adjustment is given by the Hamada equation<sup>32</sup>, which un-levers and then re-levers betas based upon changes in 14 15 capital structure. Using the Hamada equation as described in detail on page 63, line 5 through page 65, line 2 of my direct testimony, a downward 16 17 adjustment for the greater financial risk inherent in MAWC's capital structure 18 ratios is 0.21%.

- Q. You also previously noted that Mr. Gorman did not reflect flotation costs in its
   recommended common equity cost rate. Please comment
- 21

32

A. As discussed on page 65, line 5 through page 67, line 11, of my direct

Brigham and Daves, 533.

testimony, it is necessary to include flotation costs, i.e., those costs associated
with the sale of new issuance of common stock, in the common equity cost rate
recommendation. There is no other mechanism in the ratemaking paradigm
with which such costs can be recovered. Using the methodology described on
page 67, lines 5 – 11 of my direct testimony and the corrected Staff DCF cost
rate results in a flotation cost adjustment of 0.16%.

- Q. Based upon the corrected MIEC DCF, RPM and CAPM discussed previously,
   what would Mr. Gorman's recommendation be once flotation costs, the lower
   financial risk inherent in MAWC's capital structure and MAWC's greater
   business risks due to its unique risks are reflected?
- A. As shown on Schedule PMA-35, the corrected MIEC DCF is 11.93% (Line No.
  1), the corrected MIEC RPM is 10.61% (Line No. 2) and the corrected MIEC
  CAPM is 12.05% (Line No. 3). These cost rates average 11.53% (Line No. 4).
  When the flotation costs (Line No. 5), financial risk (Line No. 6) and business
  risk (Line No. 7) adjustments are added / subtracted, a corrected indicated
  MIEC common equity cost rate of 12.63% results a summarized on Schedule
  PMA-35.
- 18 BJC WITNESS BILLIE SUE LACONTE

#### 19 COMMON EQUITY COST RATE

#### 20 Discounted Cash Flow Model

- 21 Q. Please comment upon Ms. LaConte's applications of the DCF model.
- A. On page 3, lines 3 7, Ms. LaConte states that she has used three applications
   of the DCF model: the constant growth version using security analyst's growth

forecasts;, the constant growth version using GDP growth; and, a two-stage
 DCF model using security analysts' growth as well as long-term GDP growth
 forecasts. As previously discussed in this rebuttal testimony, neither the use of
 GDP growth nor a multi-stage DCF, e.g., a two-stage DCF is appropriate for
 estimating the cost of common equity in for companies in general, or for
 utilities, specifically. Therefore, I will limit my comments to her constant growth
 DCF application using security analysts' growth forecasts.

As Mr. Gorman has done, Ms. LaConte included Middlesex's negative 8 9 forecasted EPS growth rate from Reuter's. This is incorrect because, as stated 10 previously, investors do not invest with the expectation of losing money. 11 Schedule PMA-36 recalculates Ms. LaConte's single stage constant growth 12 DCF analysis excluding Middlesex's negative Reuter's forecasted growth rate 13 in EPS. As shown, the average DCF result is 10.5% and the median is 9.8%. 14 As with both Staff's and MIEC's analyses, these cost rates do not adequately reflect MAWC's financial risk and greater unique business risks relative to the 15 16 proxy group of water companies nor flotation costs as will be discussed 17 subsequently.

18

23

Capital Asset Pricing Model

19 Q. Please comment upon Ms. LaConte's application of the CAPM.

A. Ms. LaConte's application of the CAPM is flawed for two reasons. First, her
 derivation of the market equity risk premium is incorrect. Second, Ms. LaConte
 failed to utilize the ECAPM in addition to the traditional CAPM.

Ms. LaConte relied exclusively upon an historical market equity risk

1 premium which is in direct contrast to her use of projected growth rates in her 2 applications of the DCF model. As stated previously, the cost of capital is 3 prospective and while the arithmetic mean of long-term historical stock market returns can provide insight into investors' expectations of stock market returns 4 5 because the arithmetic mean of historical returns provides investors with the 6 valuable insight needed to estimate future risk, it is also appropriate to use an 7 estimate of the forecasted or projected stock market return. An appropriate method of deriving the prospective equity market return is based upon Value 8 9 Line's projected 3-5 year market appreciation potential, which when converted 10 to an annual rate plus the market's median expected dividend yield results in a 11 forecasted total annual market return of 18.98% for the thirteen-weeks ending 12 November 11, 2011 and derived as explained in Note 3 on page 2 of Schedule PMA-12. This methodology yields a truly prospective market return which is 13 14 based upon an important investor-influencing publication.

Ms. LaConte also failed to utilize an ECAPM. As discussed previously in this rebuttal testimony and in my direct testimony at page 51, line 14 through page 52, line 4 and again at page 54, line 13 through page 56, line 8, the empirical Security Market Line (SML) described by the traditional CAPM is not as steeply sloped as the predicted SML. As Morin<sup>33</sup> notes:

- . . .low-beta securities earn returns somewhat higher than the CAPM would predict, and high-beta securities earn less than predicted.
  - Hence, both the traditional CAPM and ECAPM should be used in deriving

20

21

22

<sup>&</sup>lt;sup>33</sup> Morin,175.

1 a CAPM-based common equity cost rate. I have shown the results of applying 2 both the traditional CAPM and ECAPM to Ms. LaConte's water companies 3 using a correctly derived historical market equity risk premium. As shown on 4 page 1 of Schedule PMA-37, the traditional CAPM result is 12.05%, while the ECAPM result is 12.79%. The average of both cost rates is 12.42%. However, 5 6 once again, this cost rate does not reflect MAWC's lower financial risk and 7 greater unique business risks relative to the proxy group of water companies nor flotation costs which will be discussed subsequently. 8

9

#### Recommended Common Equity Cost Rate

- Q. Do the corrected BJC DCF and CAPM results discussed previously adequately
   reflect flotation costs, the lower financial risk inherent in MAWC's capital
   structure and MAWC's greater business risks due to its unique risks are
   reflected?
- 14 Α. No, they do not. As discussed in my direct testimony at page 18, line 1 through page 20, line 23 and in Company Witness Dennis R. Williams direct testimony, 15 MAWC faces unique risks due to the availability and quality of its source of 16 17 supply; exposure to flooding; non-contiguous service territory, concentration of investment and revenues in a single metropolitan area, St. Louis; and various 18 19 unique regulatory risks. Because Ms. LaConte's proxy group is identical to my 20 proxy group of water companies as shown on Schedule PMA-24, in my opinion, my originally recommended business risk adjustment 0.40% is 21 22 warranted.
- 23

Q. Is there a way to quantify a financial risk adjustment due to the slightly lower

1

#### financial risk of MAWC's capital structure ratios?

2 A. Although Ms. LaConte did not address the capital structure issue, she based 3 her recommended common equity cost rate upon the market data of my proxy group of nine water companies. As discussed on page 24, lines 7 - 17 of my 4 direct testimony, MAWC's ratemaking capital structure ratios contain less 5 6 financial risk than those of the proxy group making it is necessary to adjust the 7 common equity cost rate to reflect this lower financial risk. Consequently, an upward adjustment to the indicated common equity cost rate based upon the 8 9 six water companies is necessary. An indication of the magnitude of the necessary financial risk adjustment is given by the Hamada equation<sup>34</sup>, which 10 11 un-levers and then re-levers betas based upon changes in capital structure. 12 Using the Hamada equation as described in detail on page 63, line 5 through 13 page 65, line 2 of my direct testimony, a downward adjustment for the greater 14 financial risk inherent in MAWC's capital structure ratios is 0.21%.

# Q. You also previously noted that Ms. LaConte did not reflect flotation costs in its recommended common equity cost rate. Please comment

A. As discussed on page 65, line 5 through page 67, line 11, of my direct
testimony, it is necessary to include flotation costs, i.e., those costs associated
with the sale of new issuance of common stock, in the common equity cost rate
recommendation. There is no other mechanism in the ratemaking paradigm
with which such costs can be recovered. Using the methodology described on

Brigham and Daves, 533.

page 67, lines 5 – 11 of my direct testimony and the corrected MIEC DCF cost
 rate results in a flotation cost adjustment of 0.16%.

- Q. Based upon the corrected BJC DCF and CAPM discussed previously, what
   would Ms. LaConte's recommendation be once flotation costs, the lower
   financial risk inherent in MAWC's capital structure and MAWC's greater
   business risks due to its unique risks are reflected?
- A. As shown on Schedule PMA-38, the corrected BJC DCF is 10.49% (Line No.
  1), and the corrected BJC CAPM is 12.42% (Line No. 2). These cost rates
  average 11.46% (Line No. 3). When the flotation costs (Line No. 4), financial
  risk (Line No. 6) and business risk (Line No. 7) adjustments are added /
  subtracted, a corrected indicated BJC common equity cost rate of 11.803%
  results a summarized on Schedule PMA-38.

# 13 UPDATED OVERALL COST OF CAPITAL AND RATE OF RETURN ON COMMON 14 EQUITY

- 15
- Q. Have you updated your recommended rate of return on common equity forMAWC?
- A. Yes. Page 1 of Schedule PMA-39 shows the updated overall rate of return for
   MAWC of 9.10% using the capital structure ratios and senior capital cost rates
   at December 31, 2011 and my updated common equity cost rate
   recommendation of 11.85%. In arriving at my updated common equity cost
   rate recommendation, I have applied the same four cost of common equity
   models in an identical manner to the current market data of the same proxy
   group of water companies as in my direct testimony.

- 1 Q. Does that conclude your rebuttal testimony?
- 2 A. Yes.

Exhibit No.: Issues:

Rate of Return on Equity

Witness:Pauline M. AhernExhibit Type:Direct SchedulesSponsoring Party:Missouri-American Water CompanyCase Nos.:WR-2011-0337Date:January 19, 2012

### PUBLIC SERVICE COMMISSION OF THE STATE OF MISSOURI

### CASE NO. WR-2011-0337

#### **EXHIBIT**

### TO ACCOMPANY THE

#### **REBUTTAL TESTIMONY**

#### OF

#### PAULINE M. AHERN, CRRA

#### **ON BEHALF OF**

#### MISSOURI-AMERICAN WATER COMPANY

# JEFFERSON CITY, MISSOURI

#### Missouri-American Water Company Derivation Implied Return on Common Equity (ROE) Based upon MoPSC Staff's Recommended Overall Rate of Return

|          |                 | Mo   | oPSC Staff's Recommer  | nded Overall Rate of         | f Return                     |
|----------|-----------------|--|--|------------------------------|------------------------------|
|          |                 | Capital Structure  | Weighted Cost Rate   |                              |                              |
| Line No. | -               | Ratios (1)   | Cost Rate (1)  | After-Inc. Tax               | Before Inc. Tax              |
| 1.       | Long-Term Debt  | 56.76%   | 6.19%  | 3.51%                        | 3.51%                        |
| 2.       | Preferred Stock | 0.29%  | 9.21%  | 0.03%                        | 0.05% (2)                    |
| 3.       | Common Equity'  | 42.95%   | 9.90%  | 4.25%                        | 6.54% (2)                    |
| 4.       |                 |  |  | 7.79%                        | 10.10%                       |
|          |                 | Missouri-American<br>Water Co.'s<br>Requested Capital<br>Structure Ratios<br>(3) | Missouri-<br>American Water<br>Co.'s<br>Requested Cost<br>Rate (3) | Weighted (<br>After-Inc. Tax | Cost Rate<br>Before Inc. Tax |
| 5.       | Long-Term Debt  | 49.36%   | 6.36%  | 3.14%                        | 3.14%                        |
| 6.       | Preferred Stock | 0.27%  | 9.23%  | 0.02%                        | 0.03% (3)                    |
| 7.       | Common Equity'  | 50.37%   | <b>8.93%</b> (4)   | 4.50% (5)                    | 6.93% (6)                    |
| 8.       |                 |  |  | 7.66%                        | 10.10% (7)                   |

#### Notes:

- (1) From Schedule 21 of the MoPSC Staff Report.
- (2) Before income tax weighted cost rate of preferred stock and common equity. 0.05% = ( 0.03 / ( 1 0.35% )) and 6.54% = ( 4.25% / ( 1 0.35% )).
- (3) From Schedule 1, page 1 of the Exhibit accompanying Ms. Ahern's direct testimony.
- (4) Implied return on common equity based upon MoPSC Staff's recommended overall rate of return. 8.93% = (4.50% / 50.37%).
- (5) After income tax weighted cost of common equity based upon MoPSC Staff's recommended overall rate of return. 4.50% = ( 6.93% \* ( 1 - 0.35% )).
- (6) Before income tax weighted cost of common equity based upon MoPSC Staff's recommended overall rate of return. 6.93% = 10.10% 0.03% 3.14%.
- (7) From Line No. 4 above.

3

÷

| 1<br>2<br>3<br>4<br>5 |    | REBUTTAL TESTIMONY OF<br>BURTON G. MALKIEL<br>ON BEHALF OF<br>SOUTH CAROLINA ELECTRIC AND GAS COMPANY<br>DOCKET NO. 2002-223-E |
|-----------------------|----|--|
| 6                     |    |  |
| 7                     | Q. | PLEASE STATE YOUR NAME AND BUSINESS ADDRESS.   |
| 8                     | A. | My name is Burton G. Malkiel and my business address is Princeton  |
| 9                     |    | University, Princeton, NJ 08544-1021.  |
| 10                    |    |  |
| 11                    | Q. | HAVE YOU PREVIOUSLY FILED TESTIMONY IN THIS  |
| 12                    |    | PROCEEDING?  |
| 13                    | A. | Yes.   |
| 14                    |    |  |
| 15                    | Q. | WHAT IS THE PURPOSE OF YOUR REBUTTAL TESTIMONY?  |
| 16                    | А. | My rebuttal testimony addresses issues raised by Michael Gorman (on  |
| 17                    |    | behalf of the South Carolina Energy Users Committee ("SCEUC")) in his direct   |
| 18                    |    | testimony, including his criticism of my reliance upon the discounted cash flow  |
| 19                    |    | ("DCF") model for estimating the cost of equity capital for South Carolina Electric  |
| 20                    |    | and Gas Company ("SCE&G"), and by David C. Parcell (on behalf of the South   |
| 21                    |    | Carolina Consumer Advocate ("CA") and South Carolina Merchants Association   |
| 22                    |    | ("SCMA")) in his direct testimony, including his criticism of my reliance upon   |
| 23                    |    | securities analysts' projected growth rates in my DCF analysis and my  |

ž.

| 1  |    | These costs have been fully incurred, are real costs of assessing capital                 |
|----|----|---|
| 2  |    | markets, and should be included in any fair analysis to determine SCE&G's cost of         |
| 3  |    | equity capital. My analysis includes flotation cost, as does Commission Staff witness     |
| 4  |    | Spearman, and it is my recommendation that SCE&G be permitted to recover these            |
| 5  |    | legitimate and real costs of raising equity capital for the benefit of its operations and |
| 6  |    | customers.  |
| 7  |    |   |
| 8  | Q. | WITNESS PARCELL ON PAGE 27, LINES 12-23 EXPLAINS THAT HE                                  |
| 9  |    | USED A COMBINATION OF "FIVE INDICATORS OF GROWTH IN                                       |
| 10 |    | [HIS] DCF ANALYSIS." THEN, ON PAGE 40-43, WITNESS PARCELL                                 |
| 11 |    | CRITICIZES YOUR RELIANCE UPON SECURITIES ANALYSTS'  |
| 12 |    | PROJECTED GROWTH RATES IN YOUR DCF ANALYSIS TO THE  |
| 13 |    | EXCLUSION OF MR. PARCELL'S SELECTION OF GROWTH  |
| 14 |    | INDICATORS. PLEASE RESPOND TO WITNESS PARCELL'S   |
| 15 |    | SELECTION OF GROWTH INDICATORS AND HIS CRITICISM OF                                       |
| 16 |    | YOUR USE OF SECURITIES ANALYSTS' PROJECTED GROWTH   |
| 17 |    | RATES IN YOUR DCF ANALYSIS.   |
|    |    |   |

A. In my book with John Cragg entitled, <u>Expectations of the Structure of</u>
 <u>Share Prices<sup>5</sup></u>, Dr. Cragg and I studied analysts' forecasts over very long periods
 of time. One of the main findings of the study published in this book was that the

1 most effective predictor of future growth was securities analysts' forecasts. Constructed growth rates based either upon historical growth or retention rates and 2 historical rates of return on equity are unreliable and are not nearly as effective 3 predictors of future growth as analysts' forecasts. Consequently, Mr. Parcell's 4 DCF analysis significantly underestimates SCE&G's true cost of equity capital 5 because he utilized constructed growth rates using historical data and retention 6 rates, when the proven choice, and the most direct and most effective predictor of 7 8 future growth is analysts' forecasts.

9 Also, please note that Witness Gorman agrees with the finance 10 community's use of analysts' forecasts. On lines 10-11, page 8 of his testimony, 11 Gorman notes that "[s]ecurity analysts' growth estimates have been shown to be 12 more accurate predictors of future returns than growth rates derived from 13 historical data."

Witness Parcell also criticizes my use of securities analysts' forecasts in performing my DCF analysis, arguing that analysts' forecasts include an upward bias rendering them suspect as a reliable predictor of future growth rates. This question has been a matter of particular interest and study for me and, as stated earlier, was a focus of the Cragg-Malkiel book. Another main finding of the book was that analysts' forecasts are not always overly optimistic. In some periods they

<sup>&</sup>lt;sup>5</sup> John Cragg and Burton G. Malkiel, *Expectations and the Structure of Share Prices*, (University of Chicago Press, 1982.)

are indeed overly optimistic. In other periods they are, however, not optimistic
 enough.

In the 1990's, I agree that there was over-optimism in securities analysts' 3 forecasts. This was especially true in the late 1990's during a period that I describe 4 5 as the biggest bubble of all times in the soon to be published eighth edition of my 6 book, A Random Walk Down Wall Street. It is also true that analysts' projections 7 were tainted by their firms' investment banking connections. To support his 8 argument, Witness Parcell includes a quote from a speech delivered on March 26, 9 2002 by Federal Reserve Chairman Alan Greenspan. In relevant part Mr. 10 Greenspan states as follows:

11 12

13

14 15

16 17

18 19 "I suspect that with the underlying database publicly available, it is just a matter of time before the ex post results of analysts' recommendations are compiled and published on a regular basis. I venture to say that with such transparency, the current upward bias of analysts' earnings projections would diminish rather rapidly, because investment firms are well aware that security analysis without credibility has no market value."

I agree with Mr. Greenspan when he states that the upward bias of analysts' earnings projections would diminish rather rapidly once their work is transparent to the public. With all the publicity given to tainted analysts' forecasts in investigations instituted by the New York Attorney General, the National Association of Securities Dealers, and the Securities & Exchange Commission, I believe the upward bias, that existed in the late 1990's has indeed diminished. In

1 summary, I believe that current analysts' forecasts are more reliable than they 2 were during the late 1990's. Therefore, analysts' forecasts remain the proper tool to use in performing a Gordon Model DCF analysis. 3 4 **ON PAGES 33-37 OF HIS PREFILED DIRECT TESTIMONY, WITNESS** 5 Q. PARCELL SETS FORTH THE RESULTS OF HIS COMPARABLE 6 EARNINGS ANALYSIS. DO YOU AGREE WITH HIS ANALYSIS AND 7 8 **CONCLUSIONS?** 9 No. First I would note that all statistics based on book values are suspect. A. 10 Book values depend on depreciation policies, policies with respect to write-offs, 11 etc., and are generally not comparable among companies. Second, Mr. Parcell 12 admits that his recommended rate of return for SCE&G would lead to a fall in the price to book value ratio, i.e., the stockholders would be made worse off. If 13 companies are to be allowed rates of return that enable them to raise new capital, 14 15 then those rates cannot be ones that cause their stock prices to decline. 16 IN YOUR OPINION WOULD THE COST OF EQUITY CAPITAL 17 **Q**. ESTIMATES PROPOSED BY CA AND SCMA WITNESS PARCELL AND 18

# SCEUC WITNESS GORMAN PROVIDE SCE&G WITH FAIR AND 20 **REASONABLE RETURNS ON ITS PLANT AND FACILITIES DEVOTED**

TO PUBLIC USE? PLEASE EXPLAIN.

19

21

# Do Analyst Conflicts Matter? Evidence from Stock Recommendations

Anup Agrawal University of Alabama Mark A. Chen Georgia State University

#### Abstract

We examine whether conflicts of interest with investment banking and brokerage businesses induce sell-side analysts to issue optimistic stock recommendations and, if so, whether investors are misled by such biases. Using quantitative measures of potential conflicts constructed from a novel data set containing revenue breakdowns of analyst employers, we find that recommendation levels are indeed positively related to conflict magnitudes. The optimistic bias stemming from investment banking conflicts was especially pronounced during the late-1990s stock market bubble. However, evidence from the response of stock prices and trading volumes to upgrades and downgrades suggests that the market recognizes analysts' conflicts and properly discounts analysts' opinions. This pattern persists even during the bubble period. Moreover, the 1-year stock performance following revised recommendations is unrelated to the magnitude of conflicts. Overall, our findings do not support the view that conflicted analysts are able to systematically mislead investors with optimistic stock recommendations.

#### 1. Introduction

In April 2003, 10 of the largest Wall Street firms reached a landmark settlement with state and federal securities regulators on the issue of conflicts of interest

[Journal of Law and Economics, vol. 51 (August 2008)]

© 2008 by The University of Chicago. All rights reserved. 0022-2186/2008/5103-0019\$10.00

We thank Yacov Amihud, Chris Barry, Utpal Bhattacharya, Stan Block, Leslie Boni, Doug Cook, Ning Gao, Jeff Jaffe, Jayant Kale, Omesh Kini, Chuck Knoeber, Junsoo Lee, Jim Ligon, Steve Mann, Vassil Mihov, Anna Scherbina, Luigi Zingales, seminar participants at Georgia State University, Southern Methodist University, Texas Christian University, the University of Alabama, the University of Delaware, the 2005 American Law and Economics Association (New York University) and European Finance Association (Moscow) meetings, and the 2006 American Finance Association (Boston), Center for Research in Security Prices Forum (Chicago), and Financial Intermediation Research Society (Shanghai) meetings for valuable comments. Special thanks are due to Randy Kroszner and Sam Peltzman and to an anonymous referee for very helpful suggestions. Tommy Cooper and Yuan Zhang provided able research assistance, and Thomson Financial provided data on analyst recommendations via the Institutional Brokers Estimate System. Agrawal acknowledges financial support from the William A. Powell Jr. Chair in Finance and Banking.

#### The Journal of LAW & ECONOMICS

faced by stock analysts.<sup>1</sup> The settlement requires the firms to pay a record \$1.4 billion in compensation and penalties in response to government charges that the firms issued optimistic stock research to win favor with potential investment banking (IB) clients. Part of the settlement funds are earmarked for investor education and for provision of research from independent firms. In addition to requiring large monetary payments, the settlement mandates structural changes in the firms' research operations and requires the firms to disclose conflicts of interest in analysts' research reports.

The notion that investors are victims of biased stock research presumes that (1) analysts respond to the conflicts by inflating their stock recommendations and (2) investors take analysts' recommendations at face value. Even if analysts are biased, it is possible that investors understand the conflicts of interest inherent in stock research and rationally discount analysts' opinions. This alternative viewpoint, if accurate, would lead to very different conclusions about the consequences of analysts' research. Indeed, investors' rationality and self-interested behavior imply that stock prices should accurately reflect a consensus about the informational quality of public announcements (Grossman 1976; Grossman and Stiglitz 1980). Rational investors would recognize and adjust for analysts' potential conflicts of interest and thereby largely avoid the adverse consequences of biased stock recommendations.

In this article, we provide evidence on the extent to which analysts and investors respond to conflicts of interest in stock research. We address four questions. First, is the extent of optimism in stock recommendations related to the magnitudes of analysts' conflicts of interest? Second, to what extent do investors discount the opinions of more conflicted analysts? In particular, do stock prices and trading volumes react to recommendation revisions in a manner that rationally reflects the degree of analysts' conflicts? Third, is the medium-term (that is, 3- to 12-month) performance of recommendation revisions related to conflict severity? And, finally, did conflicts of interest affect analysts or investors differently during the late-1990s stock bubble than during the postbubble period? The answers to these questions are clearly of relevance to stock market participants, public policy makers, regulators, and the academic profession.

We use a unique, hand-collected data set that contains the annual revenue breakdown for 232 public and private analyst employers. This information allows us to construct quantitative measures of the magnitude of potential conflicts not only from IB business but also from brokerage business. We analyze a sample of over 110,000 stock recommendations issued by over 4,000 analysts during the 1994–2003 time period. Using univariate tests as well as cross-sectional regressions that control for the size of the company followed and individual analysts' experience, resources, workloads, and reputations, we attempt to shed

<sup>&</sup>lt;sup>1</sup> Two more securities firms (Deutsche Bank Securities Inc. and Thomas Weisel Partners LLC) were added to the formal settlement in August 2004.

#### Analyst Conflicts

light both on how analysts respond to pressures from IB and brokerage businesses and on how investors compensate for the existence of such conflicts of interest.

A number of studies (for example, Dugar and Nathan 1995; Lin and McNichols 1998; Michaely and Womack 1999; Dechow, Hutton, and Sloan 2000; Bradley, Jordan, and Ritter 2008) focus on conflicts faced by analysts in the context of existing underwriting relationships (see also Malmendier and Shanthikumar 2007; Cliff 2007).<sup>2</sup> Our article complements this literature in several ways. First, we take into account the pressure to generate underwriting business from both current and potential client companies. Even if an analyst's firm does not currently do IB business with a company that the analyst tracks, it might like to do so in the future. Second, we examine the conflict between research and all IB services (including advice on mergers, restructuring, and corporate control), rather than just underwriting. Third, we examine conflicts arising from brokerage business in addition to those from IB.<sup>3</sup>

Fourth, the prior empirical finding that underwriter analysts tend to be more optimistic than other analysts is consistent with two alternative interpretations: (a) an optimistic report on a company by an underwriter analyst is a reward for past IB business or an attempt to win future IB business by currying favor with the company or (b) a company chooses an underwriter whose analyst already likes the stock. The second interpretation implies that underwriter choice is endogenous and does not necessarily imply a conflict of interest. We sidestep this issue of endogeneity by not focusing on underwriting relations between an analyst's firm and the company followed. Instead, our conflict measures focus on the importance to the analyst's firm of IB and brokerage businesses, as measured by the percentage of its annual revenue derived from IB business and from brokerage commissions. Unlike underwriting relations between an analyst's firm and the company followed, the proportions of the entire firm's revenues from each of these businesses can reasonably be viewed as given, exogenous variables from the viewpoint of an individual analyst. Finally, our approach yields substantially larger sample sizes than those used in prior research, and it therefore leads to greater statistical reliability of the results.

Several articles adopt an approach that is similar in spirit to ours. For example, Barber, Lehavy, and Trueman (2007) find that recommendation upgrades (downgrades) by investment banks—which typically also have brokerage businesses—

<sup>&</sup>lt;sup>2</sup> Bolton, Freixas, and Shapiro (2007) theoretically analyze a different type of conflict of interest in financial intermediation, one faced by a financial advisor whose firm also produces financial products (such as in-house mutual funds). Mehran and Stulz (2007) provide an excellent review of the literature on conflicts of interest in financial institutions.

<sup>&</sup>lt;sup>3</sup> Hayes (1998) analyzes how pressure on analysts to generate brokerage commissions affects the availability and accuracy of earnings forecasts. Both Irvine (2004) and Jackson (2005) find that analysts' optimism increases a brokerage firm's share of the trading volume. Ljungqvist et al. (2007) find that analysts employed by larger brokerage houses issue more optimistic recommendations and more accurate earnings forecasts. However, none of these articles examines how investors' responses to analysts' recommendations and the investment performance of recommendations vary with the severity of brokerage conflicts, issues that we investigate here.
underperform (outperform) similar recommendations by non-IB brokerages and independent research firms. Cowen, Groysberg, and Healy (2006) find that fullservice securities firms—which have both IB and brokerage businesses—issue less optimistic forecasts and recommendations than do non-IB brokerage houses. Finally, Jacob, Rock, and Weber (2008) find that short-term earnings forecasts made by investment banks are more accurate and less optimistic than those made by independent research firms. We extend this line of research by quantifying the reliance of a securities firm on IB and brokerage businesses. This is an important feature of our article for at least two reasons. First, given that many securities firms operate in multiple lines of business, it is difficult to classify them by business lines. By separately measuring the magnitudes of both IB and brokerage conflicts in each firm, our approach avoids the need to rely on a classification scheme. Second, since the focus of this research is on the consequences of analysts' conflicts, the measurement of those conflicts is important. Our conclusions sometimes differ from those in classification-based studies.

We find that analysts do indeed seem to respond to pressures from IB and brokerage businesses: larger potential conflicts of interest from these businesses are associated with more positive stock recommendations. We also document that the distortive effects of IB conflicts were larger during the late-1990s stock bubble than during the postbubble period. Nonetheless, the empirical analysis yields several pieces of evidence to suggest that investors are sophisticated enough to adjust for these biases. First, the short-term reactions of both stock prices and trading volumes to recommendation upgrades are negatively and statistically significantly related to the magnitudes of potential IB or brokerage conflicts. For downgrades, the corresponding relation is negative for stock prices but positive for trading volumes. Second, the 1-year investment performance after recommendation revisions bears no systematic relation to the magnitude of conflicts. Finally, investors continued to discount conflicted analysts' opinions during the bubble period, even amid the euphoria prevailing in the market at the time. Together these results strongly support the idea that the marginal investor, taking analysts' conflicts into account, rationally discounts optimistic stock recommendations.4

The remainder of the article is organized as follows. We discuss the issues in Section 2 and describe our sample and data in Section 3. Section 4 examines the relation between recommendation levels and the degree of IB or brokerage conflict faced by analysts. Section 5 analyzes how conflicts are related to the response of stock prices or trading volumes to recommendation revisions. Section

<sup>&</sup>lt;sup>4</sup> In a companion paper (Agrawal and Chen 2005), we find that analysts appear to respond to conflicts when making long-term earnings growth projections but not short-term earnings forecasts. This finding is consistent with the idea that, with short-term forecasts, analysts worry about their deception being revealed with the next quarterly earnings release, but they have greater leeway with long-term forecasts. We also find that the frequency of forecast revisions is positively related to the magnitude of brokerage conflicts, and several tests suggest that analysts' trade generation incentives impair the quality of stock research.

6 investigates the relation between conflicts and the investment performance of recommendation revisions. Section 7 presents our results for the late-1990s stock bubble and postbubble periods, and Section 8 concludes.

# 2. Issues and Hypotheses

Investment banking activity is a potential source of analyst conflict that has received widespread attention in the financial media (for example, Gasparino 2002; Maremont and Bray 2004) as well as the academic literature (for example, Lin and McNichols 1998; Michaely and Womack 1999). When IB business is an important source of revenue for a securities firm, a stock analyst employed by the firm often faces pressure to inflate his or her recommendations. This pressure is due to the fact that the firm would like to sell IB services to a company that the analyst tracks.<sup>5</sup> The company, in turn, would like the analyst to support its stock with a favorable opinion. Thus, we expect that the more critical is IB revenue to an analyst's employer, the greater the incentives an analyst faces to issue optimistic recommendations.<sup>6</sup>

Analysts also face a potential conflict with their employers' brokerage businesses. Here, the pressure on analysts originates not from the companies that they follow but from within their employing firms. Brokerage business generates a large portion of most securities firms' revenues, and analyst compensation schemes are typically related explicitly or implicitly to trading commissions. Thus, analysts have incentives to increase trading volumes in both directions (that is, buys and sells). Given the many institutional constraints that make short sales relatively costly, many more investors participate in stock purchases than in stock sales.<sup>7</sup> Indeed, it is mostly existing shareholders of a stock who sell. This asymmetry between purchases and sales implies that the more important brokerage business is to an analyst's employer, the more pressure the analyst faces to be bullish when issuing recommendations.

Analysts who respond to the conflicts they face by issuing blatantly misleading stock recommendations can develop bad reputations that reduce their labor income and hurt their careers.<sup>8</sup> Stock recommendations, however, are not as easily evaluated as other outputs of analysts' research, such as 12-month price targets or quarterly earnings forecasts, which can be judged against public, near-

<sup>&</sup>lt;sup>5</sup> Throughout this article, we refer to an analyst's employer as a "firm" and a company followed by an analyst as a "company."

<sup>&</sup>lt;sup>6</sup> Ljungqvist, Marston, and Wilhelm (2006, forthcoming) find that, while optimistic recommendations do not help the analyst's firm win the lead underwriter or comanager positions in general, they help the firm win the comanager position in deals in which the lead underwriter is a commercial bank.

<sup>&</sup>lt;sup>7</sup> Numerous regulations in the United States increase the cost of selling shares short (see, for example, Dechow et al. 2001). Therefore, the vast majority of stock sales are regular sales rather than short sales. For example, over the 1994–2001 period, short sales comprised only about 10 percent of the annual New York Stock Exchange trading volume (New York Stock Exchange 2002).

<sup>&</sup>lt;sup>8</sup> See Jackson (2005) for a theoretical model showing that analysts' concerns about their reputations can reduce optimistic biases arising from brokerage business.

term realizations. So it is not clear whether analysts' career concerns can completely prevent them from responding to pressures to generate IB or brokerage business.

The relation between conflict severity and the short-term (2- or 3-day) stock price impact of a recommendation should depend on whether investors react to the opinion rationally or naively.<sup>9</sup> Under the rational discounting hypothesis, the relation should be asymmetric for upgrades and downgrades. For upgrades, the stock price response should be negatively related to the degree of conflict. This implication arises because analysts who face greater pressure from IB or brokerage business are likely to be more bullish in their recommendations, and rational investors should discount an analyst's optimism more heavily. For downgrades, however, the story is different. When an analyst downgrades a stock despite facing large conflicts, rational investors should find the negative opinion more convincing and should be more likely to revalue the stock accordingly. This implies that the short-term stock price response to a downgrade should be negatively related to the degree of conflict.

The rational discounting hypothesis also predicts cross-sectional relations between conflict severity and the short-term trading volume responses to recommendations. As Kim and Verrecchia (1991) demonstrate in a rational expectations model of trading, the more precise a piece of news, the more individuals will revise their prior beliefs and, hence, the more trading that will result. In the present context, investor rationality implies that an upgrade by a highly conflicted analyst represents less precise news to investors, and so such a revision should be followed by a relatively small abnormal volume. But when an analyst downgrades a stock despite a substantial conflict, the signal is regarded as being more precise, and thus the downgrade should lead to relatively large abnormal trading.

By contrast, under the naive investor hypothesis, investors are largely ignorant of the distortive pressures that analysts face and accept analysts' recommendations at face value. This implies that there should be no relation between conflict severity and the short-term response of either stock prices or trading volume to recommendation revisions. Furthermore, the absence of a systematic relation should hold true for both upgrades and downgrades.

What are the implications of the two hypotheses for the medium-term (3- to 12-month) investment performance of analysts' recommendations? Under the rational discounting hypothesis, there should be no systematic relation between the magnitude of conflicts faced by an analyst and the performance of his or her stock recommendations: the market correctly anticipates the potential distortions up front and accordingly adjusts its response. But the naive investor hypothesis predicts that performance should be negatively related to conflict

<sup>&</sup>lt;sup>9</sup> This framework follows Kroszner and Rajan (1994) and Gompers and Lerner (1999), who analyze the conflicts that a bank faces in underwriting securities of a company when the bank owns a (debt or equity) stake in it.

severity for both upgrades and downgrades. That is, investors ignore analysts' conflicts up front and pay for their ignorance later.

# 3. Sample and Data

# 3.1. Sample

Our sample of stock recommendations comes from the Institutional Brokers Estimate System (I/B/E/S) U.S. Detail Recommendations History file. This file contains data on newly issued recommendations as well as revisions and reiterations of existing recommendations made by individual analysts over the period 1993–2003. Although the exact wording of recommendations can vary considerably across brokerage houses, I/B/E/S classifies all recommendations into five categories ranging from strong buy to strong sell. We rely on the I/B/E/S classification and encode recommendations on a numerical scale from 5 (strong buy) to 1 (strong sell).

Since we are primarily interested in examining how the nature and consequences of analysts' recommendations are related to IB or brokerage business, we require measures of the importance of these business lines to analysts' employers. Under U.S. law, all registered broker-dealer firms must file audited annual financial statements with the Securities and Exchange Commission (SEC) in x-17a-5 filings.<sup>10</sup> These filings contain information on broker-dealer firms' principal sources of revenue, broken down into revenue from IB, brokerage commissions, and all other businesses (such as asset management and proprietary trading). We use these filings to obtain various financial data, including data on our key explanatory variables: the fractions of total brokerage house revenues from IB and from brokerage commissions. Beginning with the names of analyst employers contained in the I/B/E/S Broker Translation file,<sup>11</sup> we search for all available revenue information in x-17a-5 filings from 1994 to 2003.<sup>12</sup> For publicly traded broker-dealer firms, we also use 10-K annual report filings over the sample period to gather information on revenue breakdowns, if necessary. We thus obtain annual data from 1994 to 2003 on IB revenue, brokerage revenue, and other revenue for 188 privately held and 44 publicly traded brokerage houses.<sup>13</sup> For each brokerage house, we match recommendations to the latest broker-year revenue data preceding the recommendation date. Over the sample period, we

<sup>&</sup>lt;sup>10</sup> The Securities Exchange Act, sections 17(a)–17(e), requires these filings. We accessed them from Thomson Financial's Global Access database and the Securities and Exchange Commission's (SEC's) public reading room in Washington, D.C.

<sup>&</sup>lt;sup>11</sup> We use the file supplied directly by the Institutional Brokers Estimate System (I/B/E/S) on CD-ROM. This file does not recode the name of an acquired brokerage firm to that of its acquirer for years before the merger.

<sup>&</sup>lt;sup>12</sup> The electronic availability of x-17a-5 filings is very limited prior to 1994, the year the SEC first mandated electronic form filing. Hence, we do not search for revenue information prior to 1994.

<sup>&</sup>lt;sup>13</sup> We exclude a small number of firm-years in which the total revenue is negative (for example, because of losses from proprietary trading).

are able to match in this fashion 110,493 I/B/E/S recommendations issued by 4,089 analysts.

All broker-dealer firms are required to publicly disclose their balance sheets as part of their x-17a-5 filings. But a private broker-dealer firm can withhold the public disclosure of its income statement, which contains the revenue breakdown information needed for this study, if the SEC deems that such disclosure would harm the firm's competitive position. Thus, our sample of private securities firms is limited to broker-dealers that disclose their revenue breakdowns in x-17a-5 filings. We examine whether this selection bias affects our main results by separately analyzing the subsample of publicly traded securities firms, for which public disclosure of annual revenue information is mandatory. Our findings do not appear to be affected by this selection bias. All of our results for the subsample of publicly traded securities firms are qualitatively similar to the results for the full sample reported in the article. In the Appendix, we describe the characteristics of disclosing and nondisclosing private securities firms, shed some light on the firms' income statement disclosure decisions, and use a selectivity-corrected probit model to examine whether the resulting selection bias can explain analysts' response to conflicts in these private firms. We find no evidence that selection bias affects our results for these firms.

# 3.2. Characteristics of Analysts, Their Employers, and Companies Followed

We next measure characteristics of analysts, their employers, and the companies they cover. Prior research (for example, Clement 1999; Jacob, Lys, and Neale 1999) finds that analysts' experience and workloads affect the accuracy and credibility of their research. Using the I/B/E/S Detail History files, we measure an analyst's experience and workloads in terms of all research activity reported in I/B/E/S, including stock recommendations, quarterly and annual earningsper-share forecasts, and long-term earnings growth forecasts. We measure general research experience as the number of days since an analyst first issued research on any company in the I/B/E/S database and company-specific research experience as the number of days since an analyst first issued research on a particular company. We measure an analyst's workload as the number of different companies or the number of different four-digit I/B/E/S sector industry groups (S/I/Gs)<sup>14</sup> for which the analyst issued research in a given calendar year.

The amount of resources devoted to investment research within brokerage houses also affects the quality of analysts' research (Clement 1999). Larger houses have access to better technology, information, and support staff. Accordingly, we use three measures of brokerage house size: the number of analysts issuing stock recommendations for a brokerage house over the course of a calendar year, book value of total assets, and net sales. All of our subsequent results are qual-

<sup>&</sup>lt;sup>14</sup> The I/B/E/S sector industry group numbers are six-digit codes that provide information on the industry sectors and subsectors for companies in the I/B/E/S database. We use the first four digits, which correspond to broad industry groupings.

Table 1Revenue Sources (%) of Analysts' Employers

|                                    | Inves<br>Ban | tment<br>king | Brok<br>Comn | erage<br>nission | Sample |
|------------------------------------|--------------|---------------|--------------|------------------|--------|
| Recommendation Level               | Mean         | Median        | Mean         | Median           | Size   |
| 5 (Strong buy)                     | 13.94        | 11.81         | 29.87        | 24.09            | 28,901 |
| 4 (Buy)                            | 13.81        | 11.21         | 26.68        | 17.22            | 37,478 |
| 3 (Hold)                           | 12.68        | 11.13         | 28.44        | 24.07            | 37,883 |
| 2 (Sell)                           | 11.61        | 10.55         | 23.13        | 16.12            | 4,875  |
| 1 (Strong sell)                    | 16.27        | 14.90         | 33.44        | 24.95            | 1,356  |
| p-Value (4 and 5) versus (1 and 2) | .0000        | .0000         | .0000        | .0023            |        |

Note. Shown are the percentages of analyst employer revenues from investment banking and brokerage commissions, by recommendation level. Data are for 110,493 stock recommendations and are drawn from the Institutional Brokers Estimate System U.S. Detail Recommendations History file for 1994–2003.

itatively similar under each of the three size measures. To save space, we report results only of tests based on the first size measure.

To capture the degree to which investors believe that individual analysts have skill in providing timely and accurate research, we use two measures of analysts' reputation. The first is based on *Institutional Investor* (*II*) magazine's All-America Research Team designation. Each year around October 15, *II* mails an issue to subscribers that lists the names of analysts who receive the most votes in a poll of institutional money managers. About 300–400 analysts are identified. We construct a variable that indicates, for each recommendation revision, whether the recommending analyst was named to the first, second, third, or honorable mention team in the latest annual survey. As a complementary, objective measure of analysts' reputation, we use a variable based on the *Wall Street Journal*'s (*WSJ*'s) annual All-Star Analysts Survey. The *WSJ* All-Star Analysts are determined by an explicit set of criteria relating to past stock-picking performance and forecasting accuracy.<sup>15</sup> The survey covers about 50 industries annually and names the top five stock pickers and top five earnings forecasters in each industry.<sup>16</sup>

Tables 1 and 2 report summary data on the characteristics of our sample. In Table 1, both the mean and the median percentages of analyst employer revenues derived from IB decline monotonically over the first four recommendation levels, but these values are the highest for strong sell recommendations. Similarly, it is the brokerage firms issuing strong sell recommendations that generally derive

<sup>&</sup>lt;sup>15</sup> We recognize that the performance metrics used in the *Wall Street Journal* (*WSJ*) All-Star Analysts Survey are public information and can, in principle, be replicated by investors. However, to the extent that computing and evaluating analysts' performance is a costly activity, being named an All-Star Analyst can still affect an analyst's reputation and credibility.

<sup>&</sup>lt;sup>16</sup> Since the I/B/E/S Broker Translation File provides only analysts' last names and first initials, in some instances it is not possible to ascertain from the I/B/E/S data alone whether an analyst in our sample was named to the *Institutional Investor (II)* or *WSJ* team. For these cases, we determine team membership of analysts from NASD BrokerCheck, an online database (http://www.nasd.com, accessed October 2004) that provides the full names of registered securities professionals as well as their employment and registration histories for the past 10 years. The database also keeps track of analysts' name changes (such as those resulting from marriage).

# Table 2

Characteristics of Analysts, Firms, and Companies Followed

| Characteristic                                  | Mean     | Median   | SD        | Sample<br>Size |
|---|----------|----------|-----------|----------------|
| Investment banking revenue (%)                  | 13.60    | 11.25    | 11.93     | 94,892         |
| Brokerage commission revenue (%)                | 28.74    | 24.07    | 24.75     | 94,892         |
| Analyst's company-specific experience (years)   | 2.42     | 1.20     | 3.29      | 85,531         |
| Analyst's general experience (years)            | 6.41     | 4.90     | 5.32      | 85,531         |
| Analysts employed by a firm                     | 86.34    | 60       | 79.73     | 94,618         |
| Companies followed by an analyst                | 17.24    | 15       | 12.93     | 84,016         |
| Four-digit I/B/E/S S/I/Gs followed by an        |          |          |           |                |
| analyst   | 3.05     | 3        | 1.90      | 84,014         |
| Institutional Investor All-America stock picker | .005     | 0        | .07       | 85,531         |
| Institutional Investor All-America Research     |          |          |           |                |
| Team member                                     | .035     | 0        | .18       | 85,531         |
| Wall Street Journal All-Star stock picker       | .018     | 0        | .13       | 85,531         |
| Wall Street Journal All-Star Analyst            | .136     | 0        | .34       | 85,531         |
| Market capitalization (\$ millions)             | 8,804.46 | 1,367.22 | 27,758.81 | 81,333         |
| Analyst following                               | 9.14     | 7        | 6.88      | 92,869         |

Note. Data are for 94,892 recommendation revisions and are drawn from the Institutional Brokers Estimate System (I/B/E/S) U.S. Detail Recommendations History file for 1994–2003. Recommendation revisions include recommendation changes as well as initiations, resumptions, and discontinuations of coverage. Analysts' experience is measured from all analyst research activity reported in I/B/E/S, including earnings-per-share forecasts, long-term earnings growth forecasts, and stock recommendations. An analyst is considered to be a top stock picker or team member if he or she appeared in the relevant portion of the most recent analyst survey by *Institutional Investor* or the *Wall Street Journal* at the time of a recommendation revision. Market capitalization is measured 12 months before the end of the current month, and analyst following is measured on the basis of stock recommendation coverage. Market capitalization values are inflation adjusted (with Consumer Price Index numbers and with 2003 as the base year). S/I/G = sector industry group.

the highest percentage of their total revenues from brokerage commissions. Notably, in each of the five categories, the mean percentage of revenue from commissions is about twice as large as the mean percentage of revenue from IB. This fact underscores the importance of trading commissions as a source of revenue for many securities firms. The last column shows that about 95 percent of the recommendations in the sample are at levels 5 (strong buy), 4 (buy), or 3 (hold). Levels 1 (strong sell) and 2 (sell) represent only about 1 percent and 4 percent of all recommendations, respectively.

The data in Table 2 provide a flavor of our sample of analysts and their employers. As noted by Hong, Kubik, and Solomon (2000), careers as analysts tend to be relatively short. The median recommendation is made by an analyst with under 5 years of experience, of which just over a year was spent following a given stock. Stock analysts tend to be highly specialized, following a handful of companies in a few industries. The median recommendation is made by an analyst following 15 companies in three industries who works for a securities firm employing 60 analysts. Being named as an All-America Research Team member by *II* is a rare honor, received by under 5 percent of all analysts in our sample. Finally, the typical company followed is large, with mean (median) market capitalization of about \$8.8 billion (\$1.4 billion) in inflation-adjusted

2003 dollars. Over the time span of a year, a company is tracked by a mean (median) of 9.1 (7) analysts.

# 4. Conflicts and the Levels of Analyst Recommendations Net of the Consensus

In this section, we examine whether the level of an analyst's stock recommendation net of the consensus (that is, median) recommendation level is related to the conflicts that he or she faces. We start by ascertaining the level of the outstanding recommendation on each stock by each analyst following it at the end of each quarter (March, June, September, December) from 1995 through 2003. An analyst's recommendation on a stock is included only if it is newly issued, reiterated, or revised in the preceding 12 months.

We estimate a regression explaining individual analysts' net stock recommendation levels at the end of a quarter (which is the recommendation level minus the median recommendation level across all analysts following a stock during the quarter).<sup>17</sup> The regression pools observations across analysts, stocks, and quarters and includes our two main explanatory variables: the percentage of an analyst employer's total revenues from IB and the percentage from brokerage commissions. Following Jegadeesh et al. (2004) and Kadan et al. (forthcoming), who find that momentum is an important determinant of analysts' recommendations, we control for the prior 6-month stock return.

The regression also controls for other factors that can affect the degree of analysts' optimism, such as the size of the company followed and the resources, reputation, experience, and workload of an analyst. As a measure of the resources available to an analyst, a dummy variable is used for a large brokerage house, and it equals one if the firm ranks in the top quartile of all houses in terms of the number of analysts employed during the year. The size of the company followed is measured by the natural logarithm of its market capitalization, measured 12 months before the end of the month. We measure an analyst's reputation by dummy variables that equal one if the recommending analyst was named in the most recent year as an All-America Research Team member by II or as an All-Star Analyst by the WSJ. An analyst's company-specific research experience is measured by the natural logarithm of one plus the number of days an analyst has been producing research (including earnings-per-share forecasts, long-term growth forecasts, or stock recommendations) on the company. We measure an analyst's workload by the natural logarithm of one plus the number of companies for which he or she produces forecasts or recommendations in the current year.

Finally, we control for industry and time period effects by adding dummy variables for I/B/E/S two-digit S/I/G industries and for each calendar quarter (March 1995, June 1995, and so forth). Since net recommendation levels can

<sup>&</sup>lt;sup>17</sup> To ensure meaningful variation in the dependent variable, we omit stocks followed by only one analyst in a quarter.

### Table 3

#### Ordered Probit Analysis of Recommendation Levels Net of the Consensus

| Explanatory Variable                                   | Coefficient | z-Statistic |
|--|-------------|-------------|
| Investment banking revenue (%)                         | .4167       | 17.35       |
| Brokerage commission revenue (%)                       | .0363       | 3.00        |
| Prior 6-month stock return                             | 0068        | -2.89       |
| Large brokerage house dummy                            | 0639        | -8.60       |
| Company size   | .0038       | 2.89        |
| Institutional Investor All-America Research Team dummy | .0032       | .15         |
| Wall Street Journal All-Star Analyst dummy             | 0196        | -2.23       |
| Company-specific research experience                   | .0012       | 1.42        |
| Number of companies followed                           | .0070       | 4.64        |

Note. The results are from ordered probit regressions explaining individual analysts' stock recommendation levels net of the consensus (that is, median) recommendation level at the end of each quarter (March, June, September, December) for 1995-2003. Observations are excluded if the analyst issued no new or revised recommendation in the preceding 12 months. The regression includes observations pooled across analysts, stocks, and quarters. Data on recommendations are drawn from the Institutional Brokers Estimate System (I/B/E/S) U.S. Detail Recommendations History file for 1994–2003. Investment banking or brokerage commission revenue refer to the percentage of the brokerage firm's total revenues derived from investment banking or brokerage commissions. The large brokerage house dummy is an indicator variable that equals one if a brokerage house is in the top quartile of all houses, based on the number of analysts issuing stock recommendations listed in I/B/E/S in a given calendar year. Company size is the natural logarithm of the market capitalization of the company followed, measured 12 months prior to the end of the current month. The Institutional Investor All-America Research Team and Wall Street Journal All-Star Analyst dummies are indicator variables that equal one if the recommending analyst was listed as an All-America Research Team member or All-Star Analyst in the most recent analyst ranking. Company-specific research experience is the natural log of one plus the number of days that an analyst has been issuing I/B/E/S research on a company. Number of companies followed equals the natural log of one plus the number of companies followed by an analyst in the current calendar year. The regression includes dummy variables for two-digit I/B/E/S sector industry group industries and for calendar quarters. Test statistics are based on a robust variance estimator. The number of observations is 213,011; the *p*-value of the  $\chi^2$  test is <.0001.

take ordered values from -4 (strongly pessimistic) to 4 (strongly optimistic) in increments of .5, we estimate the regression as an ordered probit model.<sup>18</sup> The *Z*-statistics are based on a robust (Huber-White sandwich) variance estimator.

Table 3 shows the regression estimate. The coefficients of IB revenue percentage and commission revenue percentage are both positive. This finding implies that greater conflicts with IB and brokerage businesses lead an analyst to issue a higher recommendation on a stock relative to the consensus. Stocks followed by busier analysts and stocks of larger companies receive higher recommendations relative to the consensus. Stocks that experience a price run-up over the prior 6 months, stocks followed by analysts at large brokerage houses, and stocks followed by *WSJ* All-Star Analysts all receive lower recommendations relative to the consensus. All of these relations are highly statistically significant.

To provide a sense of the magnitude of the main effects of interest, we show in Table 4 the derivatives of the probability of each net recommendation level

<sup>&</sup>lt;sup>18</sup> Notice that recommendation levels can take integer values from 1 to 5, and the median recommendation can take values from 1 to 5 in increments of .5. See Greene (2003) for a detailed exposition of the ordered probit model.

|       | (<br>- |
|-------|--------|
|       |        |
| 4     |        |
| Table | ,      |
|       | •      |

| • | ۲  | 2  |
|---|--|--|
|   | a  | د  |
| - | -  | 1  |
| - | 5  | 2  |
| Ľ | G  | 2  |
| 5 |  | 1  |
|   | 2  | 1  |
| • | ÷  | ÷  |
|   | c  | 5  |
|   | 2  | 5  |
| • | 2  | 1  |
|   | ä  | 8  |
|   | ã  | ĩ  |
|   | È  | 1  |
|   | ă  | 5  |
| , | ç  | 2  |
| 1 |  | 1  |
|   | t  | 2  |
| _ | Ċ  | S  |
|   | C  | 5  |
|   | ł  | •  |
| F | 2  | ł  |
| - |  | ÷  |
|   | à  | 5  |
|   | ř  | 4  |
| _ | ٩  | 2  |
|   | ç  | 2  |
| , | -  | 2  |
| 5 | -  | ,  |
|   | a  | د  |
| _ | ē  | 5  |
| 5 | ÷  | 5  |
|   | •  |  |
|   | ē  | 5  |
| ٩ | f  | ł  |
|   |  |  |
|   | -  | •  |
|   | Ę  | 5  |
| • |  | 5  |
| • |  |  |
| • |  | n LIVII  |
| : |  | TOTIOT   |
|   | 11110n                                   | LITU ULU                                       |
|   | ctribition                               | TIOTIN OTTO                                    |
|   | 115T T10111010                           | VISITI D MILLOIT                               |
|   | 11111111111111111111111111111111111111   | LINULUUUUUUUUUUUUUUUUUUUUUUUUUUUUUUUUUU        |
|   |  | L DISTINGUINALIN                               |
| - |  | TO DISTINGTING                                 |
| - |  | TOTIN PIPETRATIONI                             |
| - | mple Distribution                        | IIDIN PIPIPIPIPIPIPIPIPIPIPIPIPIPIPIPIPIPIP    |
| - | ampe listribution                        | aupto Distruction                              |
| - | Sample Distribution                      | number Plantourion                             |
|   | 1 Sample Distribution                    | TOTINOTTION AND A DISTORTION AND A DISTORT     |
|   | ad Sample Distribution                   | TOTINOTHER ATAINING AT                         |
|   | and Sample Distribution                  | TIM DUILDIN DISTINCTION                        |
|   | and Sample Distribution                  | Intra output Automation and                    |
|   | te and Sample Dietribution               | is and campic Pistingation                     |
|   | offe and Sample Distribution             | Coante Campie Pistination                      |
|   | terts and Sample Distribution            | icces and campic Pistination                   |
|   | there and sample littlini                | Autors and Campic Pistinguine                  |
|   | HTPCTs and Sample Distribution           | THESE AND CAMPTER PROVIDENCE                   |
|   | I HTPOTE and Sample Distribution         | T THESE and Cample Plantouron                  |
|   | al Htterts and Sample Distribution       | in the second and cample planto and            |
|   | The share and sample lister but to the   | man Encode and Cample Planto and               |
|   | ornal Effects and Sample Distribution    | Emilia Lincols and Campic Pistinguin           |
|   | roinal Htterts and Sample Distribution   | inguint threes and campic Pistinguin           |
|   | arginal Htterts and Sample Distribution  | angum ruccio ana campic Pistiloanon            |
|   | Marginal Htterts and Sample Distribution | Mail Full Full of a state of the second action |

Recommendation Level Net of the Consensus

|                                  | -4    | -3.5  | -3    | -2.5  | -2    | -1.5  | -1    | 5     | 0     | ŝ     | -     | 1.5   | 7     | 2.5    | ю      |
|----------------------------------|-------|-------|-------|-------|-------|-------|-------|-------|-------|-------|-------|-------|-------|--------|--------|
| Investment banking revenue (%)   | 00031 | 0002  | 0026  | 0010  | 0199  | 0086  | 0744  | 0321  | .0123 | .0325 | .0671 | .0077 | .0188 | 0002   | .0003  |
| Brokerage commission revenue (%) | 00003 | 00001 | 0002  | 00000 | 0017  | 0008  | 0065  | 0028  | .0011 | .0028 | .0059 | 7000. | .0016 | .00002 | .00003 |
| Observed frequency               | .0001 | .000  | .0016 | .0007 | .0176 | .0094 | .1241 | .0948 | .4940 | .0937 | .1289 | .0111 | .0233 | 0002   | .0003  |
|                                  |       | •     |       | •     | •     |       |       |       |       |       |       |       |       |        |        |

Note. Shown is the derivative of the probability of each net recommendation level with respect to investment banking or brokerage revenue percentage, estimated from the ordered probit regression in Table 3. Investment banking and brokerage commission revenue refer to the percentage of the brokerage firm's total revenues derived from investment banking and brokerage commissions. The last row shows observed frequency of each net recommendation level as a proportion of the sample of 213,011 observations.

516

with respect to IB revenue and commission revenue percentages.<sup>19</sup> Thus, for example, a 1-standard-deviation increase in IB revenue percentage increases the probability of an optimistic recommendation (that is, a net recommendation level greater than zero) by .1193 × (.0325 + .0671 + ... + .0003) = .0151. Compared to the unconditional probability of an optimistic recommendation by an analyst, this represents an increase of about 5.9 percent (.0151/.2575). The effect of a change in commission revenue percentage is much smaller. A 1-standard-deviation increase in commission revenue percentage increases the probability of an optimistic recommendation by .2475 × .01105 = .0027, or about 1 percent (.0027/.2575) of the unconditional probability. Thus, despite possible concerns about a loss of reputation, analysts seem to respond to conflicts of interest, particularly those stemming from IB.

# 5. Conflicts and Investor Response to Recommendation Revisions

# 5.1 Stock Price Response

This section examines whether an analyst's credibility with investors is related to the degree of conflict faced. We interpret the reaction of stock prices to a recommendation revision as an indication of an analyst's credibility. Our analysis focuses on revisions in recommendation levels, rather than on recommendation levels per se, because revisions are discrete events that are likely to be salient for investors, and previous research finds that revisions have significant information content (see, for example, Womack 1996; Jegadeesh et al. 2004). To capture the effects of the most commonly observed and economically important types of revisions, we structure our tests around four basic categories: added to strong buy, added to buy or strong buy, dropped from strong buy, and dropped from buy or strong buy.<sup>20</sup> These four categories are defined to include initiations, resumptions, and discontinuations of coverage because such events also reflect analysts' positive or negative views about a company.<sup>21</sup> Thus, for example, we consider a stock to be added to strong buy from a lower level or (*b*) coverage is

<sup>&</sup>lt;sup>19</sup> Notice that, for each explanatory variable, these derivatives sum to zero across all the net recommendation levels.

<sup>&</sup>lt;sup>20</sup> Our analysis focuses on these four types of revisions instead of the other four (added to strong sell, and so forth) because, as shown in Table 1, sell and strong sell recommendations are quite rare. But note that dropped-from-buy and dropped-from-buy-or-strong-buy revisions can entail movement to the sell or strong sell category.

<sup>&</sup>lt;sup>21</sup> We use the I/B/E/S Stopped Recommendations file to determine instances in which a brokerage firm discontinued coverage of a company. This file contains numerous cases in which an analyst stops coverage of a stock only to issue a new recommendation a month or two later. Conversations with I/B/E/S representatives indicate that such events likely represent pauses in coverage due to company quiet periods or analysts' reassignments within a brokerage house. We define a stopped coverage event to be a true stoppage only if the analyst does not issue a recommendation on the stock over the subsequent 6 months.

initiated or resumed at the level of strong buy.<sup>22</sup> Defining revisions in this fashion yields a sample of 94,892 recommendation revisions made over the 1994–2003 period.

# 5.1.1. Average Response

We compute the abnormal return on an upgraded or downgraded stock over day *t* as the return (including dividends) on the stock minus the return on the Center for Research in Security Prices equal-weighted market portfolio of New York Stock Exchange (NYSE), American Stock Exchange, and NASDAQ stocks. The cumulative abnormal return (CAR) on the stock over days  $t_1$  to  $t_2$  relative to the revision date (day 0) is measured as the sum of the abnormal returns over those days. Table 5 shows mean and median CARs for three windows: days -1 to 0, -1 to 1, and -5 to 5. The *t*-statistics for the difference of the mean abnormal returns from zero are computed as in Brown and Warner (1985) and are shown in parentheses. The *p*-values for the Wilcoxon test are reported in parentheses with the medians.

It is clear from Table 5 that recommendation revisions have large effects on stock prices. For example, when a stock is added to the strong-buy list, it experiences a mean abnormal return of about 2 percent over the 2-day revision period. Downgrades have even larger effects on stock prices than do upgrades. Strikingly, the 2-day mean abnormal return around the dropped-from-strong-buy list is -4 percent. Median values are consistently smaller in magnitude than are means, and this finding indicates that some revisions lead to price reactions of a very large magnitude. Mean and median 2-day abnormal returns are statistically different from zero for all four groups of forecast revisions. The magnitudes of abnormal returns are somewhat larger over the 3-day and 11-day windows than over the 2-day window. Overall, these returns are consistent with those found by prior research that examines the average stock price impact of recommendation revisions (for example, Womack 1996; Jegadeesh et al. 2004).

# 5.1.2. Cross-Sectional Analysis

Table 6 contains cross-sectional regressions of stock price reactions to recommendation revisions over days -1 to 1. The main explanatory variables of interest in these regressions are our revenue-based measures of the magnitudes of IB and brokerage conflicts. We include controls for the size of an analyst's employer, the size of the company followed, and measures of an analyst's reputation, experience, and workload.<sup>23</sup> We estimate a separate regression for each

<sup>&</sup>lt;sup>22</sup> Note that the definitions of our four recommendation revision groups imply that stocks can be added to a group more than once on a given day. Nonetheless, excluding days on which a stock experiences multiple revisions does not change any of our qualitative results.

<sup>&</sup>lt;sup>23</sup> Prior research finds that analysts who have more experience, carry lower workloads, or are employed by larger firms tend to generate more precise research (see, for example, Clement 1999; Jacob, Lys, and Neale 1999; Mikhail, Walther, and Willis 1997). In addition, more reputed analysts tend to generate timelier and more accurate research (see, for example, Stickel 1992; Hong and Kubik 2003). We expect such analysts to be more influential with investors.

| Mean                                  | Days – | -1 to 0           |        | D                              | tys $-1$ to $1$              |        | Ď                              | 1ys -5 to 5                  |        |
|---------------------------------------|--------|-------------------|--------|--------------------------------|------------------------------|--------|--------------------------------|------------------------------|--------|
| Recommendation Revision (t-Statistic) |        | fedian<br>-Value) | N      | Mean<br>( <i>t</i> -Statistic) | Median<br>( <i>p</i> -Value) | Ν      | Mean<br>( <i>t</i> -Statistic) | Median<br>( <i>p</i> -Value) | Ν      |
| Upgrades:                             | r      | 0100              | 073 FC | 0100                           | 0120                         |        | 6360                           | 1010                         | 11.00  |
| Ациец to strong buy (49.53)*          | *      | 6010.             | 74,000 | .0240<br>(46.89)*              | 0000.)                       | 000,47 | . (26.84)*                     | (000 <sup>.</sup> )          | 24,499 |
| Added to buy or strong buy .0149      | 6      | .0071             | 36,879 | .0165                          | .0085                        | 36,875 | .0207                          | .0128                        | 36,780 |
| (46.47)*                              | *      | (000)             |        | $(42.01)^{*}$                  | (000)                        |        | $(27.53)^{*}$                  | (000)                        |        |
| Downgrades:                           |        |                   |        |                                |                              |        |                                |                              |        |
| Dropped from buy or strong buy –.0337 |        | 0126              | 33,322 | 0358                           | 0155                         | 33,262 | 0491                           | 0287                         | 33,197 |
| (-56.21)*                             | *      | (000)             |        | $(-48.75)^{*}$                 | (000.)                       |        | $(-34.92)^{*}$                 | (000)                        |        |
| Dropped from strong buy –.0399        | - 6    | 0153              | 22,825 | 0427                           | 0183                         | 22,795 | 0570                           | 0326                         | 22,767 |
| (-49.88)*                             | *      | (000)             |        | $(-43.58)^{\star}$             | (000)                        |        | $(-30.38)^{*}$                 | (000)                        |        |

Cumulative Abnormal Returns surrounding Revisions in Analyst Stock Recommendations Table 5

Recommendation revisions include recommendation changes and initiations, resumptions, and discontinuations in coverage. Day 0 is the revision date. Recommendation revisions are classified according to the level of any existing recommendation and whether coverage is being initiated or dropped. For example, a revision by an analyst is classified as added to strong buy if the new recommendation is strong buy and (*a*) the previous recommendation was lower than strong buy or (*b*) analyst coverage by the brokerage house is resummed or initiated. A recommendation is classified from strong buy of the previous recommendation is lower than strong buy or (*b*) research coverage by the brokerage house is resummed or initiated. A recommendation is strong buy if the previous recommendation was lower than strong buy and (*a*) the new recommendation is lower than strong buy or (*b*) research coverage of the company is stopped. The *t*-statistics for the difference from zero are computed as in Brown and Warner (1985). The *p*-values for the difference from zero are from a Wilcoxon test. \*Statistically significant at the 1% level in two-tailed tests.

| ns         |
|------------|
| 10         |
| . <u></u>  |
| R          |
| nc         |
| äti        |
| ndi        |
| nei        |
| I          |
| 3          |
| R          |
| ng         |
| iđ         |
| I          |
| Ĕ          |
| su         |
| Ŧ          |
| ò          |
| 1          |
| Ι          |
| iys        |
| Ď          |
| er         |
| 0          |
| ns         |
| 'n         |
| Set        |
| al J       |
| Ë          |
| 101        |
| PI         |
| e          |
| ÷          |
| ula        |
| E          |
| õ          |
| of         |
| JS         |
| <u>ē</u> . |
| ess        |
| žo         |
| ž          |
| Jal        |
| ior        |
| ect        |
| s-S        |
| š0.        |
| .5         |

Table 6

| Explanatory Variable                                   | Added to<br>Strong Buy | Added to Buy or<br>Strong Buy | Dropped from Buy or<br>Strong Buy | Dropped from<br>Strong Buy |
|--|------------------------|-------------------------------|-----------------------------------|----------------------------|
| Intercept  | .0369                  | .0412                         | 2294                              | 2224                       |
| 4  | (7.66)**               | $(11.21)^{**}$                | $(-31.31)^{**}$                   | $(-29.25)^{**}$            |
| Investment banking revenue (%)                         | 0262                   | 0139                          | 0200                              | 0354                       |
| 2  | $(-5.65)^{**}$         | $(-3.57)^{**}$                | $(-2.74)^{**}$                    | $(-3.92)^{**}$             |
| Brokerage commission revenue (%)                       | 0187                   | 0148                          | 0089                              | 0013                       |
| 2  | $(-6.51)^{**}$         | $(-6.43)^{**}$                | $(-2.39)^{*}$                     | (29)                       |
| Large brokerage house dummy                            | .0116                  | .0088                         | 0242                              | 0220                       |
|  | $(7.46)^{**}$          | $(6.88)^{**}$                 | $(-12.79)^{**}$                   | $(-10.25)^{**}$            |
| Company size   | 0056                   | 0041                          | 0004                              | .0018                      |
|  | $(-16.13)^{**}$        | $(-15.40)^{**}$               | (97)                              | $(3.77)^{**}$              |
| Institutional Investor All-America Research Team dummy | .0159                  | .0122                         | 0148                              | 0207                       |
|  | $(4.11)^{**}$          | $(3.82)^{**}$                 | $(-2.93)^{**}$                    | $(-3.28)^{**}$             |
| Wall Street Journal All-Star Analyst dummy             | .0015                  | .0013                         | 0011                              | .0045                      |
| •  | (.81)                  | (.84)                         | (48)                              | (1.78)                     |
| Company-specific research experience                   | .0017                  | .0019                         | .0039                             | .0018                      |
| 4  | $(8.42)^{**}$          | $(12.49)^{**}$                | $(7.37)^{**}$                     | $(3.21)^{**}$              |
| Number of companies followed                           | 0012                   | 0016                          | .0007                             | .0008                      |
| 4  | $(-2.97)^{**}$         | $(-5.37)^{**}$                | (1.49)                            | (1.31)                     |
| Observations   | 19,440                 | 28,665                        | 28,618                            | 19,632                     |
| Adjusted $R^2$   | .038                   | .0240                         | .028                              | .035                       |
| <i>P</i> -Value of <i>F</i> -test                      | <.0001                 | <.0001                        | <.0001                            | <.0001                     |
|  |                        |                               |                                   |                            |

519

Note. Shown are coefficient estimates and (in parentheses) *r*-statistics from ordinary least squares regressions. Day 0 is the recommendation revision date. Data on recommendations revenue teler to the percentages of a brokerage fine is total revenues derived from investment banking and brokerage commissions. The angle or not durm is *s* total revenues derived from investment banking and brokerage commissions. The large brokerage house dummy is an indicator variable that equals one if a brokerage house is in the top quartile of all houses, based on the number of analysts issuing stock recommendations listed in *JBE/S* in a given calendar year. Company size is the natural logarithm of the market capitalization of the company followed, measured 12 months prior to the end of the current month. The *Institutional Investor* All-America Research Team and *Wall Street Journal* All-Star Analyst dummies are indicator variables that equals one fue number of any analyst in the market capitalization of the company followed, measured 12 months prior to the end of the current month. The *Institutional Investor* All-America Research Team and *Wall Street Journal* All-Star Analyst dummies are indicator variables that equal one if the recommending analyst was listed as an All-America Research Team and *Wall Street Journal* All-Star Analyst tanking. Company-seclific research experience is the natural log of one plus the number of days that an analyst in the current analyst transing. Companies followed equals the attranal log of one plus the number of companies followed by an analyst in the current calendar year. All sector industry group industries (not reported). The *t*-statistics are based on a robust variance estimator. \*Statistically significant at the 5% level in two-tailed tests.

\*\* Statistically significant at the 1% level in two-tailed tests.

520

of the four groups of recommendation revisions. The *t*-statistics based on a robust variance estimator are reported in parentheses.

The coefficient on IB revenue percentage is statistically significantly negative for both upgrades and downgrades. The coefficient on brokerage commission revenue percentage is also negative in all four regressions; it is statistically significant in all cases, except for the dropped-from-strong-buy revisions.<sup>24</sup> Collectively, these results favor the rational discounting hypothesis over the naive investor hypothesis. The magnitudes of these effects are nontrivial. For instance, a 1-standard-deviation increase in IB revenue percentage leads to a change of about -.31 (-.42) percentage points in the 3-day abnormal return around the move to (from) a strong buy recommendation. Similarly, a 1-standard-deviation increase in brokerage commission revenue percentage leads to a change of about -.37 (-.22) percentage points in the corresponding abnormal return around the move to (from) a buy or strong buy recommendation.<sup>25</sup>

The results for control variables are also noteworthy. The dummy variable for a large analyst employer is positively (negatively) related to the market reaction to upgrades (downgrades). This finding is consistent with the idea that revisions by analysts employed at larger brokerage houses (which tend to be more reputable) have more credibility with investors. The size of the company followed is negatively (positively) related to the market reaction to upgrades (downgrades), which is consistent with the notion that, for larger companies, an analyst's recommendation competes with more alternative sources of information and advice.

Revisions by *II* All-America Research Team analysts are positively (negatively) related to the stock price reaction to upgrades (downgrades), which suggests that they wield more influence with investors. This is a notable finding; we are unaware of previous work documenting a relation between an analyst's reputation and the stock price reaction to both upgrades and downgrades. As the coefficient on the *WSJ* All-Star Analyst dummy indicates, however, being designated as a *WSJ* All-Star Analyst does not seem to enhance the credibility of an analyst's recommendations.<sup>26</sup> The absence of an effect here is somewhat

<sup>24</sup> These and all subsequent regression results in this article are qualitatively similar when we winsorize the dependent variable at the first and ninety-ninth percentiles of its distribution.

<sup>26</sup> Although *II* All-America Research Team and *WSJ* All-Star Analyst dummies both measure aspects of an analyst's reputation, they are not highly correlated. The correlation coefficient is .14 across all upgrades and .13 across all downgrades.

<sup>&</sup>lt;sup>25</sup> For each group of revisions (such as added to strong buy), we also estimate the regression after excluding similar revision events that a stock experiences within 3 days of a given revision event. These results are qualitatively similar to those reported in Tables 6 and 8. We also examine the possibility that investors perceived the conflicts to be more severe, and hence discounted them more, in securities firms that were charged by regulators (that is, the 10 firms that were part of the global analyst settlement) than in other firms. We do this by interacting both investment banking (IB) revenue percentage and brokerage commission revenue percentage variables in the regression with binary (0, 1) dummy variables for securities firms that are part of the global analyst settlement and firms that are not. We find no significant differences between the two groups of firms in their coefficients on IB revenue percentage and commission revenue percentage.

surprising given that the WSJ has a much broader readership base than that of *II*. One explanation is that *II* analyst rankings are based on an opinion poll of money managers, who control substantial assets and therefore directly affect stock prices, while WSJ rankings are based on strictly quantitative measures of analysts' past stock-picking or forecasting performance.

The market reaction to upgrades is positively related to an analyst's companyspecific research experience. This finding suggests that more experienced analysts tend to be more influential with investors. But the reaction to downgrades is also positively related to analysts' experience. Finally, the stock price reaction to upgrades is negatively related to analysts' workload. This finding suggests that busier analysts' opinions tend to get discounted by the market. All of these relations are statistically significant.

# 5.2. Response of Trading Volume

In this section, we measure analysts' credibility via changes in the volume of trade around recommendation revisions.<sup>27</sup> Revisions of analysts' recommendations can affect trading volumes by inducing investors to rebalance their portfolios to reflect updated beliefs.

# 5.2.1. Average Response

We compute the abnormal volume for a trading day t as the mean-adjusted share turnover for stock i:<sup>28</sup>

$$e_{it} = v_{it} - v_i, \tag{1}$$

where  $v_{ii}$  is the trading volume of stock *i* over day *t* divided by common shares outstanding on day *t* and  $v_i$  is the mean of  $v_{ii}$  over days -35 to -6.

The cumulative abnormal volume (CAV) for stock *i* over days  $t_1$  to  $t_2$  is measured in the following way:

$$CAV^{i}t_{1},t_{2} = \sum_{t=t_{1}}^{t_{2}} e_{it}.$$
 (2)

Table 7 shows mean and median CAV values over three windows surrounding revisions in analyst stock recommendations. Over the 2-day revision period, the mean abnormal volume is positive for both upgrades and downgrades, but its magnitude is substantially larger for downgrades. The move to (from) the strongbuy list increases a stock's trading volume by a mean of about .9 percent (2.6 percent) of the outstanding shares, compared to a normal day's volume. For longer windows, the mean abnormal volumes are substantially higher for down-

<sup>&</sup>lt;sup>27</sup> Many prior studies have used trading volume to examine investors' response to informational events (see, for example, Shleifer 1986; Jain 1988; Jarrell and Poulsen 1989; Meulbroek 1992; Sanders and Zdanowicz 1992).

<sup>&</sup>lt;sup>28</sup> This approach has been used in a number of prior studies (for example, Shleifer 1986; Vijh 1994; Michaely and Vila 1996).

| Cumulative Adnormal  | Irading volum   | es surrounan  | ng Announ                    | cements of Ke                    | VISIONS II 200                       | ock kecomi                  | nendations by                              | Analysts                      |                            |
|--|---|---|------------------------------|----------------------------------|--------------------------------------|-----------------------------|--|-------------------------------|----------------------------|
|  | D   | tys $-1$ to $0$   |                              | Ď                                | ays -1 to 1                          |                             | Ď  | ays –5 to 5                   |                            |
| Recommendation revision  | Mean<br>( <i>t</i> -Statistic)  | Median<br>( <i>p</i> -Value)                            | N                            | Mean<br>( <i>t</i> -Statistic)   | Median<br>( <i>p</i> -Value)         | Ν                           | Mean<br>( <i>t</i> -Statistic)             | Median<br>( <i>p</i> -Value)  | Ν                          |
| Upgrades:  |   |   |                              |                                  |                                      |                             |  |                               |                            |
| Added to strong buy  | .0086   | .0011   |                              | .0097                            | .0015                                |                             | .0071                                      | .0030                         |                            |
|  | $(8.89)^{*}$  | (000)   | 24,506                       | $(8.18)^{*}$                     | (000)                                | 24,502                      | $(3.13)^{*}$                               | (000)                         | 24,488                     |
| Added to buy or strong buy   | .0053   | .0002   |                              | .0058                            | .0004                                |                             | .0020                                      | .0008                         |                            |
|  | $(5.08)^{*}$  | (000)   | 36,800                       | $(4.54)^{*}$                     | (000)                                | 36,796                      | (.818)                                     | (000)                         | 36,766                     |
| Downgrades:  |   |   |                              |                                  |                                      |                             |  |                               |                            |
| Dropped from buy or strong buy   | .0217   | .0010   |                              | .0265                            | .0014                                |                             | .0381                                      | .0039                         |                            |
|  | $(114.47)^{*}$  | (000)   | 33, 291                      | $(114.14)^{*}$                   | (000)                                | 33,232                      | $(85.70)^{*}$                              | (000)                         | 33,175                     |
| Dropped from strong buy  | .0259   | .0017   |                              | .0315                            | .0025                                |                             | .0453                                      | .0057                         |                            |
| -<br>)   | $(128.76)^{*}$  | (000)   | 22,808                       | $(127.86)^{*}$                   | (000)                                | 22,779                      | $(96.03)^{*}$                              | (000)                         | 22,756                     |
| Note. The abnormal volume for stock <i>i v</i> , is the average volume over days $-35$ outstanding on the same day. The <i>p</i> -valu *Statistically significant at the 1% leve | t on day $t$ is complete to $-6$ relative to use are from a Wi el in two-tailed tee | uted from dail<br>o the recomme<br>lcoxon test.<br>sts. | y Center for<br>ndation revi | Research in Secusion date (day 0 | ırity Prices dat.<br>). All share vo | a as $e_{ii} = v_{ii}^{-1}$ | - $v_i$ , where $v_i$ is ormalized by divi | the volume on<br>ding by comm | day <i>t</i> and on shares |

| ysts            |
|-----------------|
| Anal            |
| bу              |
| Recommendations |
| Stock ]         |
| Е.              |
| f Revisions     |
| s of            |
| Announcement    |
| surrounding     |
| Volumes         |
| Trading         |
| Abnormal        |
| Cumulative      |
|                 |

Table 7

grades. The median values are lower than the mean values. Each mean and median abnormal volume is statistically greater than zero, with a *p*-value below .01. Clearly, revisions of stock recommendations by analysts generate trading.

# 5.2.2. Cross-Sectional Analysis

Table 8 presents cross-sectional regressions explaining CAVs over days -1 to 1 surrounding the recommendation revisions. The explanatory variables in the regressions are the same as in regressions of CARs in Section 5.1.2. The results provide strong support for the rational discounting hypothesis. The coefficients on both the IB revenue percentage and commission revenue percentage variables are generally statistically significant and negative (positive) for both groups of upgrades (downgrades). The magnitudes of these effects are nontrivial. For example, a 1-standard-deviation increase in IB revenue percentage leads to a change in the 3-day abnormal volume around the addition (omission) of a stock to (from) the strong-buy list of about -.12 percent (.36 percent) of the outstanding shares; a corresponding change in the commission revenue percentage results in a change in the abnormal volume of about -.15 percent (.22 percent).

Recommendation revisions by larger brokerage houses generate more trading. The abnormal volume is also larger for revisions involving smaller companies. Revisions by *II* All-America Research Team members generate statistically significantly more abnormal volume for the dropped from buy or strong-buy group. Upgrades (downgrades) by more experienced analysts result in larger (smaller) abnormal volumes, and upgrades by busier analysts are less credible.

# 6. Conflicts and the Performance of Recommendation Revisions

We next consider the investment performance of analysts' recommendation revisions over periods of up to 12 months. Here, the choice of the benchmark used to compute abnormal returns is somewhat more important than it is in Section 5.1, where we measure abnormal returns over a few days around the revision. But the results here are likely to be less sensitive to the benchmark employed than are those in studies of long-run stock performance, where the time period of interest can be as long as 5–10 years (see, for example, Agrawal, Jaffe, and Mandelker 1992; Agrawal and Jaffe 2003).

# 6.1. Average Performance

We use an approach similar to Barber, Lehavy, and Trueman (2007). To evaluate the performance of stocks over a given window, say, months 1–12 following the month of their inclusion (month 0) in a given group of revisions such as the added-to-strong-buy list, we form a portfolio p that initially invests \$1 in each recommendation. Each recommended stock remains in the portfolio until month 12 or the month that the stock is either downgraded or dropped from coverage by the securities firm, whichever is earlier. If multiple securities firms recommend a stock in a given month, the stock appears multiple times in the

| Revisions      |
|----------------|
| Recommendation |
| surrounding    |
| +1             |
| 1 tc           |
| s<br>S         |
| Day            |
| over           |
| Volumes        |
| Trading        |
| Abnormal       |
| Cumulative     |
| of (           |
| Regressions    |
| nal            |
| s-Section      |
| Cros           |

Table 8

| Explanatory Variable                                   | Added to<br>Strong Buy | Added to Buy or<br>Strong Buy | Dropped from Buy or<br>Strong Buy | Dropped from<br>Strong Buy |
|--|------------------------|-------------------------------|-----------------------------------|----------------------------|
| Intercept  | .0083                  | .0042                         | .0946                             | .0828                      |
| -  | $(2.65)^{**}$          | (1.90)                        | $(13.72)^{**}$                    | $(15.01)^{**}$             |
| Investment banking revenue (%)                         | 0100                   | 0085                          | .0140                             | .0304                      |
| 5  | $(-3.31)^{**}$         | $(-2.26)^{*}$                 | $(2.18)^{*}$                      | $(3.63)^{**}$              |
| Brokerage commission revenue (%)                       | 0057                   | 0059                          | .0087                             | .0055                      |
|  | (-1.76)                | $(-4.13)^{**}$                | $(2.76)^{**}$                     | (1.45)                     |
| Large brokerage house dummy                            | .0058                  | .0038                         | .0168                             | .0171                      |
| -  | $(3.72)^{**}$          | $(4.50)^{**}$                 | $(11.12)^{**}$                    | $(9.48)^{**}$              |
| Company size   | 0031                   | 0018                          | 0023                              | 0041                       |
| -  | $(-9.54)^{**}$         | $(-12.30)^{**}$               | $(-7.60)^{**}$                    | $(-11.40)^{**}$            |
| Institutional Investor All-America Research Team dummy | .0035                  | .0033                         | .0084                             | .0046                      |
| -  | (1.74)                 | (1.88)                        | $(2.32)^{*}$                      | (1.21)                     |
| Wall Street Journal All-Star Analyst dummy             | .000                   | .0013                         | .0023                             | - 0006                     |
| •  | (.74)                  | (1.42)                        | (1.36)                            | (29)                       |
| Company-specific research experience                   | .0010                  | .0010                         | 0041                              | 0019                       |
| 4  | $(8.39)^{**}$          | $(11.19)^{**}$                | $(-6.18)^{**}$                    | $(-4.11)^{**}$             |
| Number of companies followed                           | - 0009                 | 0013                          | 0001                              | 0005                       |
| a.   | $(-3.49)^{**}$         | $(-6.23)^{**}$                | (38)                              | (66-)                      |
| Observations   | 19,431                 | 28,653                        | 28,594                            | 19,619                     |
| Adjusted R <sup>2</sup>                                | .025                   | .019                          | .030                              | .042                       |
| <i>p</i> -Value of <i>F</i> -test                      | <.0001                 | <.0001                        | <.0001                            | <.0001                     |
|  | ;                      |                               |                                   |                            |

Note. Shown are coefficient estimates and (in parentheses) *t*-statistics from ordinary least squares regressions. Day 0 is the recommendation revision date. Data on recommendations are drawn from the Institutional Brokersg Entimate System (*I/B/E/S*) U.S. Detail Recommendations History file for 1994–2003. Investment banking and brokerage commissions. The lange brokerage commissions are drawn from the institutional Brokers Estimate System (*I/B/E/S*) U.S. Detail Recommendations History file for 1994–2003. Investment banking and brokerage commissions. The large brokerage commissions are drawn from the institutional Brokers of this total revenues derived from investment banking and brokerage commissions issuing stock recommendations listed in *I/B/E/S* in a given calendar year. Company size is the natural logarithm of the market capitalization of the company followed, measured 12 months prior to the end of the current month. The *Institutional Investor* All-America Research Team and *Wall Street Journal* All-Star Analyst dummies are indicator variables that equals on the *Institutional Investor* All-America Research Team member or All-Star Analyst in the most are indicator variables that equals one if the neutral log of the number of days that an analyst has been issuing *I/B/E/S* research are manks. Number of companys is the number of companies followed by an analyst in the current calendar year and two-digit *I/B/E/S* sector industry group industries (not reported). The *t*-statistics are based on a combany. Number of companies followed by an analyst in the current calendar year and two-digit *I/B/E/S* sector industry group industries (not reported). The *t*-statistics are based on a robust variance estimator.

\* Statistically significant at the 5% level in two-tailed tests.

\*\* Statistically significant at the 1% level in two-tailed tests.

portfolio that month, once for each securities firm with a strong buy recommendation. The portfolio return for calendar month t is given by

$$R_{pt} = \sum_{i=1}^{n_t} x_{it} \times R_{it} / \sum_{i=1}^{n_t} x_{io}$$
(3)

where  $R_{it}$  is the month *t* return on recommendation *i*,  $x_{it}$  is one plus the compound return on the recommendation from month 1 to month t - 1 (that is,  $x_{it}$  equals one for a stock that was recommended in month *t*), and  $n_t$  is the number of recommendations in the portfolio. This calculation yields a time series of monthly returns for portfolio *p*.

We compute the abnormal performance of portfolio p as the estimate of the intercept term  $\alpha_p$  from the Fama and French (1993) three-factor model. Accordingly, we estimate the following time-series regression for portfolio p:

$$R_{pt} - R_{ft} = \alpha_p + \beta_{1p}(R_{mt} - R_{ft}) + \beta_{2p}SMB_t + \beta_{3p}HML_t + \varepsilon_{pp},$$
  
$$t = \text{January 1994 to December 2003,}$$
(4)

where  $R_f$  is the risk-free rate,  $R_m$  is the return on the value-weighted market index, SMB equals the monthly return on a portfolio of small firms minus the return on a portfolio of big firms, and HML is the monthly return on a portfolio of firms with high book-to-market ratio minus the return on a portfolio of firms with low book-to-market ratio. The error term in the regression is denoted  $\varepsilon$ . The time series of monthly returns on  $R_m - R_f$ , SMB, and HML are obtained from Kenneth French's Web site.<sup>29</sup> We repeat this procedure for each time window of interest, such as months 1–3, and for each group of revisions, such as the dropped-from-strong-buy list.

Table 9 shows the performance of analysts' recommendation revisions. Over the period of 3 months following the month of recommendation revision, the average abnormal returns for upgrades are positive, and the returns for downgrades are negative. The magnitudes of these returns are nontrivial. For example, the addition of a stock to the strong-buy list has an abnormal monthly return of about .875 percent, or about 2.62 percent over the 3-month period. The pattern is generally similar over longer windows. For example, over months 1–12, the abnormal monthly return for the added-to-strong-buy list is .679 percent, or about 8.15 percent over the 12-month period. The abnormal returns are significantly different from zero for upgrades in all cases; they are statistically insignificant for downgrades in all cases except one.

<sup>&</sup>lt;sup>29</sup> Kenneth R. French, Fama/French Factors (file F-F\_Research\_Data\_Factors.zip at http://mba.tuck.dartmouth.edu/pages/faculty/ken.french/data\_library.html).

|             |            | Table 9     |                   |           |
|-------------|------------|-------------|-------------------|-----------|
| ledium-Term | Investment | Performance | of Recommendation | Revisions |

|  | Month                                | ns 1–3                             | Month                                | ns 1–6                              | Month                                | s 1–12                          |
|--|--------------------------------------|------------------------------------|--------------------------------------|-------------------------------------|--------------------------------------|---------------------------------|
| Portfolio  | Abnormal<br>Monthly<br>Return<br>(%) | t-Statistic                        | Abnormal<br>Monthly<br>Return<br>(%) | <i>t</i> -Statistic                 | Abnormal<br>Monthly<br>Return<br>(%) | t-Statistic                     |
| Added to strong buy<br>Added to buy or strong buy<br>Dropped from buy or strong buy<br>Dropped from strong buy | .875<br>.586<br>361<br>367           | 6.12**<br>4.49**<br>-1.60<br>-1.58 | .758<br>.511<br>260<br>395           | 6.12**<br>4.82**<br>-1.28<br>-2.00* | .679<br>.503<br>072<br>231           | 5.70**<br>5.38**<br>44<br>-1.49 |

Note. Abnormal returns are reported for three event windows relative to the month of revision (month 0) and are computed using an approach similar to that in Barber, Lehavy, and Trueman (2007). The abnormal return is the estimated intercept from a time-series regression of 114 monthly portfolio returns using the Fama and French (1993) three-factor model.

Statistically significant at the 5% level in two-tailed tests.

\*\* Statistically significant at the 1% level in two-tailed tests.

#### 6.2. Cross-Sectional Analysis

Table 10 shows the results of a regression similar to that in Section 5.1.2, except that the dependent variable here is the average monthly abnormal return for a firm over months 1–12 following the month of a recommendation revision. We compute this abnormal return by estimating a time-series regression similar to that in equation (4) over months 1–12 for each stock in a sample of recommendation revisions. The intercept from this regression is our estimate of the performance of the recommendation revision. Observations involving recommendation revisions on a stock that occur within 12 months of an earlier revision are omitted from each regression.<sup>30</sup>

In each regression result reported in Table 10, the coefficients of IB revenue percentage and commission revenue percentage are not statistically significantly different from zero. These results favor the rational discounting hypothesis, at least for the marginal investor. The performance of both groups of recommendation upgrades is negatively related to company size; the performance of one group of downgrades is positively related to the dummy variable for *WSJ* All-Star Analysts. None of the other variables is statistically significant.

# 7. Bubble versus Postbubble Periods

We next exploit the fact that our sample spans both the late-1990s U.S. stock bubble and a postbubble period. During the bubble period, initial public offerings, merger activities, and stock prices were near record highs, and media attention was focused on analysts' pronouncements. We therefore examine whether analysts' behavior and investors' responses to analysts' recommendations differed during the bubble and postbubble periods. Given the euphoria on Wall

Ν

<sup>&</sup>lt;sup>30</sup> The results are qualitatively similar when we include these observations.

| 1              |
|----------------|
| Months         |
| over           |
| Revisions      |
| Recommendation |
| following      |
| Returns        |
| Abnormal       |
| Monthly        |
| Average        |
| of             |
| Regressions    |
| -Sectional     |
| Cross          |

Table 10

12

|  | Added to                  | Added to Buy or           | Dropped from Buy or   | Dropped from          |
|--|---------------------------|---------------------------|-----------------------|-----------------------|
| Explanatory Variable                             | Strong Buy                | Strong Buy                | Strong Buy            | Strong Buy            |
| Intercept  | .0523                     | .0089                     | 0646<br>(6 01)**      | 0821<br>(6 55)**      |
| Investment banking revenue (%)                   | (1.01)<br>0089            | (.47)<br>0018             | (0.01)                | 0068                  |
| Brokerage commission revenue (%)                 | (22.1 - 1)                | (29)                      | (1.04)                | (6/)                  |
| Large brokerage house dummy                      | (1000)                    | (1.54)<br>0027            | (1.21)<br>.0016       | ()                    |
| Company size                                     | 0013                      | (-1.52)<br>(0017)         |                       |                       |
| Institutional Investor All-America analyst dummy | (-2.74)                   | (-4.18)                   | (-1.71)<br>0016       | $(+0.000)^{-0.000}$   |
| Wall Street Journal All-Star Analyst dummy       | (                         | (.01)                     | (44)<br>0029          | ()                    |
| Company-specific research experience             | (1.24) $(0004)$           | (.12)<br>.0004            | (-1.42).0004          | .0004                 |
| Number of companies followed                     | (1.08)<br>0011<br>(-1.61) | (1.50)<br>0008<br>(-1.79) | (, 6)<br>0002<br>(45) | (.52)<br>0002<br>(47) |
| Observations                                     | (1.01) (6,411             | 8,851                     | 10,644                | 8,368                 |
| Adjusted R <sup>2</sup>                          | .026                      | .023                      | .019                  | .020                  |
| <i>p</i> -Value of <i>F</i> -test                | <:0001                    | <.0001                    | <.0001                | <:0001                |
|  |                           |                           |                       | •                     |

Investment banking and brokerage commission revenue data refer to the percentage of the brokerage firm's total revenues derived from investment banking and brokerage commissions. The large brokerage house dummy is an indicator variable that equals one if a brokerage house is in the top quartile of all houses, based on the number of analysts issuing stock recommendations on *ID*/*IFS* in a given calendar year. Company size is the natural logarithm of the market capitalization of the company followed, measured 12 months prior to the end of the current month. The *Institutional Investo* All-America Research Team and *Wall Street Journal* All-Star Analyst dummies are indicator variables that equal one if the recommending analyst was listed as an All-America Research Team member or All-Star Analyst in the most recent analyst ranking. Company-specific research experience is the natural log of one plus the number of days that an analyst has been issuing *IVBFIS* research on a company. Number of company followed by an analyst in the current calendar year. Must respressions include dummy variables for calendar-year and two-digit *I/B/ES* sector industry group industries (not reported). The *t*-statistics are based on a robust regressions include dummy variables for calendar-year and two-digit *I/B/ES* sector industry group industries (not reported). The *t*-statistics are based on a robust **Note.** Shown are the coefficient estimates and (in parentheses) *t*-statistics from ordinary least squares regressions. Month 0 is the month of recommendation revision. The abnormal return is the estimated intercept from a time-series regression of monthly portfolio returns in accordance with the Fama and French (1993) three-factor model. Data on recommendations are drawn from the Institutional Brokers Estimate System (I/B/E/S) U.S. Detail Recommendations History file for 1994–2003. variance estimator. \* Statistically significant at the 5% level in two-tailed tests. \*\* Statistically significant at the 1% level in two-tailed tests.

Schedule PMA-20 Page 25 of 35

# Table 11

Ordered Probit Regression of Recommendation Levels Net of the Consensus for Bubble versus Postbubble Periods

|                                | Bubble | Postbubble | <i>p</i> -Value |
|--------------------------------|--------|------------|-----------------|
| Investment banking revenue (%) | .5103* | .3089*     | <.001           |
| Brokerage revenue (%)          | 1868*  | .2286*     | <.001           |

Note. The explanatory variables are as in Table 3, except that (a) the investment banking revenue and brokerage commission revenue percentage variables are interacted with dummy variables for the bubble or postbubble period and (b) calendar-quarter dummies are replaced with a postregulation indicator (which is equal to one for quarters after May 2002). Shown are the coefficient estimates of investment banking and brokerage revenue percentage variables for the bubble and postbubble periods and the p-value for the difference in the coefficient estimate between the two periods. All test statistics use robust variance estimators. \* Statistically significant at the 1% level in two-tailed tests.

Street and among investors during the bubble, analysts appear to have been under acute pressure to generate IB fees and brokerage commissions. As for the response of investors, the rational discounting hypothesis predicts greater discounting of analysts' opinions during this period in response to heightened conflicts, while the naive investor hypothesis predicts less discounting.

We estimate regressions similar to those for relative recommendation levels (Table 3), those for announcement abnormal returns (Table 6), those for announcement abnormal volumes (Table 8), and those for 12-month investment performance of recommendation revisions (Table 10), except that we now interact IB revenue percentage and commission revenue percentage with dummy variables for the bubble (January 1996-March 2000) and postbubble (April 2000–December 2003) periods. Accordingly, we restrict the sample period for these regressions to January 1996–December 2003. For regressions corresponding to those with results shown in Table 3, we also replace the calendar-quarter dummies with a postregulation indicator (equal to one for quarters ending after May 2002). In May 2002, both the NYSE and the National Association of Securities Dealers considerably tightened the regulations on the production and dissemination of sell-side analyst research.<sup>31</sup> The findings of Barber et al. (2006) and Kadan et al. (forthcoming) suggest that these regulations exerted a downward pressure on recommendation levels. The regression results are presented in Tables 11 and 12. To save space, we report only the coefficient estimates for IB revenue percentage and commission revenue percentage.

The results in Table 11 show that analysts appear to have inflated their recommendations in response to IB conflicts during both the bubble and postbubble periods. But the magnitude of this effect is substantially greater during the bubble period than during the postbubble period. This difference is statistically significant. The magnitude of the effect is smaller for brokerage conflicts than for IB conflicts during both periods. In fact, the effect for brokerage conflicts is negative

<sup>&</sup>lt;sup>31</sup> See NYSE Amended Rule 472, "Communications with the Public," and National Association of Securities Dealers Rule 2711, "Research Analysts and Research Reports."

| Table 12           Ordinary Least Squares Regressions of Abnormal Returns, Abnormal Volumes, and           Abnormal Stock Performance for Bubble and Postbubble Periods |
|---|
|---|

|   | Addeo       | to Strong ]                   | 3uy             | Added to                  | Buy or Stro                | ng Buy                  | Dropp                    | ed from Buy<br>strong Buy | / or               | Dropped                 | from Stron                | g Buy           |
|---|-------------|-------------------------------|-----------------|---------------------------|----------------------------|-------------------------|--------------------------|---------------------------|--------------------|-------------------------|---------------------------|-----------------|
|   | Bubble      | Postbubble                    | <i>p</i> -Value | Bubble                    | Postbubble                 | <i>p</i> -Value         | Bubble                   | Postbubble                | <i>p</i> -Value    | Bubble                  | Postbubble                | <i>p</i> -Value |
| CARs, days -1 to 1:   |             |                               |                 |                           |                            |                         |                          |                           |                    |                         |                           |                 |
| Investment banking revenue (%)  | $0248^{**}$ | 0120                          | .083            | $0121^{**}$               | 0080                       | .517                    | 0125                     | 0379**                    | .027               | $0361^{**}$             | $0345^{**}$               | 908.            |
| Brokerage revenue (%)   | $0114^{**}$ | $0105^{**}$                   | .827            | 0099**                    | $0110^{**}$                | .720                    | 0063                     | 0208**                    | .003               | .0017                   | $0114^{*}$                | .024            |
| CAVs, days $-1$ to 1:   |             |                               |                 |                           |                            |                         |                          |                           |                    |                         |                           |                 |
| Investment banking revenue (%)  | 0076        | 0052                          | .655            | 0065                      | $0082^{*}$                 | 669.                    | .0257**                  | .0130                     | .214               | .0555**                 | .0153                     | .002            |
| Brokerage revenue (%)   | 0042        | 0008                          | .376            | $0054^{**}$               | 0031                       | .179                    | $.0106^{*}$              | .0139**                   | .521               | .0046                   | .0141**                   | .056            |
| Average monthly CARs, months 1-12:  |             |                               |                 |                           |                            |                         |                          |                           |                    |                         |                           |                 |
| Investment banking revenue (%)  | 0016        | 0151                          | .273            | .0000                     | .0083                      | .420                    | 0085                     | .0223**                   | .003               | 0123                    | 0051                      | .564            |
| Brokerage revenue (%)   | .0069       | .0108                         | .511            | .0086                     | .0096                      | .842                    | .0035                    | .0136                     | .101               | 0036                    | 1600.                     | .019            |
| <b>Note.</b> The explanatory variables are as are interacted with dummy variables f | in Tables ( | 5, 8, and 10,<br>ble or postb | except 1        | hat the inv<br>riod. Show | estment bar<br>n are the c | aking rev<br>oefficient | renue and l<br>estimates | of the inves              | mmissic<br>tment h | n revenue<br>anking and | percentage v<br>brokerage | ariables        |

are increased with duminy variables for the bubble and postbubble period. Shown are the coefficient estimates of the investment banking and brokerage revenue percentage variables for the bubble and postbubble periods and the p-value for the difference in the coefficient estimate between the two periods. Day (month) 0 is the recommendation revision date. All test statistics use robust variance estimators. CAR = cumulative abnormal return; CAV = cumulative abnormal volume. \*Statistically significant at the 9% level in two-tailed tests. \*Statistically significant at the 1% level in two-tailed tests.

530

during the bubble; it is positive and statistically significantly higher during the postbubble period.

Table 12 shows that, in regressions of 3-day abnormal returns, the coefficients of both IB revenue percentage and commission revenue percentage are negative and statistically significant during the bubble period for both groups of upgrades. For the added-to-strong-buy group, the coefficient of IB revenue percentage is significantly lower during the bubble period than during the postbubble period. For downgrades, the coefficients of both variables are generally negative in both periods, and they are statistically significantly lower during the postbubble period.

In regressions of 3-day abnormal volumes, the coefficients of IB revenue percentage and commission revenue percentage are negative for upgrades and positive for downgrades in all cases, both during and after the bubble. These coefficients are not statistically significantly different between the bubble and postbubble periods for both groups of upgrades and one group of downgrades. For the dropped-from-strong-buy group, the coefficient of IB revenue percentage is statistically significantly larger during the bubble period than during the postbubble period, but the coefficient of the commission revenue percentage is statistically significantly smaller. In regressions of 12-month postrecommendation stock performance, the coefficients of both variables are statistically insignificant both during and after the bubble period in nearly all cases, and this finding is consistent with the results shown in Table 10 for the full sample period.

Overall, analysts appear to respond to IB conflicts both during and after the bubble, but the magnitude of their response declines during the postbubble period. Perversely, while analysts do not seem to respond to brokerage conflicts during the bubble, they appear to do so after the bubble. Perhaps the intense regulatory and media focus on IB conflicts has led analysts to look for alternative avenues. Did investors discount conflicted analysts' opinions more during the bubble than in the postbubble period? The answer to this question is unclear. However, our evidence does not support the notion that investors threw caution to the wind during the bubble.

# 8. Summary and Conclusions

Following the collapse of the late-1990s U.S. stock market bubble, there has been a widespread hue and cry from investors and regulators over the conflicts of interest faced by Wall Street stock analysts. The discovery of e-mail messages, in which analysts were privately disparaging stocks that they were touting publicly, led to the landmark \$1.4 billion settlement between a number of leading Wall Street firms and securities regulators in April 2003. The settlement requires the firms to disclose IB conflicts in analyst reports and imposes a variety of restrictions designed to strengthen the firewalls that separate research from IB. Part of the settlement funds are set aside for investor education and for research produced by independent firms. The settlement basically presumes that analysts

respond to the conflicts by inflating their stock recommendations and that investors take analysts' recommendations at face value.

Consistent with the view of the media and regulators, we find that optimism in stock recommendations is positively related to the importance of both IB and brokerage businesses to an analyst's employer. This pattern is more pronounced during the late-1990s stock market bubble with respect to IB conflicts. However, we provide several pieces of empirical evidence that suggest that investors are sophisticated enough to adjust for this bias. First, the short-term reactions of both stock prices and trading volumes to recommendation upgrades vary negatively with the magnitude of potential IB or brokerage conflicts faced by analysts. For instance, over the 3 days surrounding an upgrade to strong buy, a 1-standarddeviation increase in the proportion of revenue from IB is associated with a .31 percentage point decrease in abnormal returns and a .12 percentage point decrease in abnormal volume. These results suggest that investors ascribe lower credibility to an analyst's upgrade when the analyst is subject to greater pressures to issue an optimistic view. For downgrades, conflict severity varies negatively with the short-term stock price reaction and positively with the short-term trading volume impact. This pattern is consistent with the idea that investors perceive an analyst to be more credible if he or she is willing to voice an unfavorable opinion on a stock despite greater pressures to be optimistic.

Second, we find no evidence that the 1-year investment performance of recommendation revisions is related to the magnitude of analysts' conflicts, either for upgrades or for downgrades. This finding suggests that, on average, investors properly discount an analyst's opinions for potential conflicts at the time the opinion is issued. Finally, investors discounted conflicted analysts' opinions during the late-1990s stock bubble, even in the face of the prevailing market euphoria. This evidence does not support the popular view that recommendations of sell-side analysts led investors to throw caution to the wind during the bubble period.

Overall, our empirical findings suggest that while analysts do respond to IB and brokerage conflicts by inflating their stock recommendations, the market discounts these recommendations after taking analysts' conflicts into account. These findings are reminiscent of the story of the nail soup told by Brealey and Myers (1991), except that here analysts (rather than accountants) are the ones who put the nail in the soup and investors (rather than analysts) are the ones to take it out. Our finding that the market is not fooled by biases stemming from conflicts of interest echoes similar findings in the literature on conflicts of interest in universal banking (for example, Kroszner and Rajan 1994, 1997; Gompers and Lerner 1999) and on bias in the financial media (for example, Bhattacharya et al., forthcoming; Reuter and Zitzewitz 2006). Finally, while we cannot rule out the possibility that some investors may have been naive, our findings do not support the notion that the marginal investor was systematically misled over the last decade by analysts' recommendations.

# Appendix

This Appendix describes the characteristics of disclosing and nondisclosing private securities firms, sheds some light on their decisions to publicly disclose their income statements, and examines whether the resulting selection bias affects our main results in Table 3. Table A1 provides summary statistics of recommendation levels and characteristics of disclosing and nondisclosing private securities firms. Compared with nondisclosing firms, disclosing firms tend to be smaller and more liquid and issue somewhat more optimistic stock recommendations. The mean recommendation level is slightly higher for disclosing firms than for nondisclosing firms. The median disclosing firm is smaller and holds more liquid assets than the median nondisclosing firm. All these differences are statistically significant. The two groups of firms have similar financial leverage ratios and 2-year growth rates in total assets.

We next examine cross-sectional determinants of a private securities firm's decision to disclose its income statement. In an excellent review of the corporate disclosure literature, Healy and Palepu (2001) point out that a firm is more willing to voluntarily disclose financial information when it needs to raise external financing and when it is less concerned that the disclosure would damage its competitive position in product markets. Ceteris paribus, firms with greater growth opportunities, higher financial leverage, and less liquid resources are more likely to need external financing. They are more likely to be open with potential investors by disclosing financial information, including their income statements. Similarly, smaller firms are likely to have greater need for external financing as they try to grow. In addition, given the intense competition in the securities business, smaller private firms are also likely to be more willing to disclose their profits and profitability because they have less business at stake. For both reasons, smaller firms are likely to be more willing to disclose financial information. We control for firm size by the natural logarithm of one plus total assets in millions of dollars, for growth opportunities by the 2-year growth rate of total assets, for financial leverage by the ratio of long-term debt to total assets, and for liquidity by the ratio of cash and equivalents to total assets. We estimate a probit regression of DISCLOSER, which equals one for a disclosing firm and is zero otherwise.

In accordance with the predictions of corporate disclosure theory, the coefficients on firm size and liquidity are negative, and the coefficient on growth is positive. Contrary to the prediction, however, the coefficient on leverage is negative. All of these coefficients are highly statistically significant. The pseudo- $R^2$ -value of this model is .08. To save space, these results are not shown in a table.

Finally, we examine whether the selection bias caused by a private securities firm's disclosure choice (and, consequently, the availability of data on IB revenue percentage and commission revenue percentage) affects our main results in Table 3. While there is no Heckman selectivity correction for the ordered probit model, there is one for the regular probit model. So we define a binary variable to

|                                |            | Mean          |                   |            | Median        |                         |            |               |
|--------------------------------|------------|---------------|-------------------|------------|---------------|-------------------------|------------|---------------|
|                                |            |               | <i>P</i> -Value   |            |               | <i>p</i> -Value of Rank | Sam        | ple Size      |
| Variable                       | Disclosers | Nondisclosers | of <i>t</i> -Test | Disclosers | Nondisclosers | Sum Test                | Disclosers | Nondisclosers |
| Recommendation level:          |            |               |                   |            |               |                         |            |               |
| Level                          | 3.902      | 3.810         | <.001             | 4          | 4             | <.001                   | 62,417     | 181,068       |
| Level minus median level       | .036       | .010          | <.001             | 0          | 0             | <.001                   | 62,417     | 181,068       |
| Firm size:                     |            |               |                   |            |               |                         |            |               |
| Total assets (\$ millions)     | 383.37     | 1,863.52      | <.001             | 4.05       | 28.43         | <.001                   | 365        | 615           |
| Book equity (\$ millions)      | 26.40      | 68.98         | <.001             | 1.97       | 10.56         | <.001                   | 365        | 615           |
| Financial leverage:            |            |               |                   |            |               |                         |            |               |
| Long-term debt to total assets | .0539      | .0653         | .253              | 0          | .002          | .004                    | 365        | 615           |

 Table A1

 Summary Statistics for Disclosing and Nondisclosing Private Securities Firms

**Note.** Disclosers are brokers that publicly disclose their income statements, while nondisclosers are brokers that do not disclose them. The statistics for recommendation level are computed from individual analysts' recommendation levels at the end of each quarter in the sample. The median recommendation level is computed at the end of each quarter and is based on all analysts recommending a stock. The statistics for broker characteristics are computed across broker years. The firm size statistics are inflation adjusted (with Consumer Price Index numbers and with 2003 as the base year). The 2-year growth rate is (Total assets, / Total assets)<sup>1/2</sup> – 1.

615 615 615 541

365 365 365 246

.004 <.001 .0001 .099

.002 .018 .052 .020

0 0 .101 .052

.253 .295 .001 .440

.0653 .1823 .1816 .0697

.0539 .0685 .2392 .0849

Liquidity: cash and equivalents to total assets

2-Year growth rate

Total debt to total assets

measure an optimistic recommendation that equals one if an analyst's recommendation level on a stock exceeds the consensus level and equals zero otherwise. We then replace the dependent variable in the regression in Section 4 with this optimistic recommendation dummy. Using the subsample of private securities firms, we estimate the resulting equation in two ways: (a) with a regular probit model and (b) with a Heckman selectivity-corrected probit model, where we use the equation described in the second paragraph of this Appendix as the selection equation. When we use approach b, the coefficient of the selection term (that is, the inverse Mills ratio) is statistically significant in the second-stage probit regression. What is more important for our purposes is that the sign, magnitude, and statistical significance of our main explanatory variables, the IB revenue percentage and the commission revenue percentage, are similar in the regular probit and the Heckman-corrected probit regressions. These results do not support the idea that our main findings are driven by the selection bias caused by a private securities firm's decision to disclose its revenue breakdown. To save space, these results are not shown in a table.

# References

- Agrawal, Anup, and Mark Chen. 2005. Analyst Conflicts and Research Quality. Working paper. University of Alabama, Culverhouse College of Business, Tuscaloosa.
- Agrawal, Anup, and Jeffrey F. Jaffe. 2003. Do Takeover Targets Under-Perform? Evidence from Operating and Stock Returns. *Journal of Financial and Quantitative Analysis* 38: 721–46.
- Agrawal, Anup, Jeffrey F. Jaffe, and Gershon N. Mandelker. 1992. The Post-merger Performance of Acquiring Firms: A Re-Examination of an Anomaly. *Journal of Finance* 47:1605–21.
- Barber, Brad, Reuven Lehavy, Maureen McNichols, and Brett Trueman. 2006. Buys, Holds, and Sells: The Distribution of Investment Banks' Stock Ratings and the Implications for the Profitability of Analysts' Recommendations. *Journal of Accounting and Economics* 41:87–117.
- Barber, Brad, Reuven Lehavy, and Brett Trueman. 2007. Comparing the Stock Recommendation Performance of Investment Banks and Independent Research Firms. *Journal of Financial Economics* 85:490–517.
- Bhattacharya, Utpal, Neal Galpin, Rina Ray, and Xiaoyun Yu. Forthcoming. The Role of the Media in the Internet IPO Bubble. *Journal of Financial and Quantitative Analysis*.
- Bolton, Patrick, Xavier Freixas, and Joel Shapiro. 2007. Conflicts of Interest, Information Provision, and Competition in the Financial Services Industry. *Journal of Financial Economics* 85:297–330.
- Bradley, Daniel J., Bradford D. Jordan, and Jay R. Ritter. 2008. Analyst Behavior following IPOs: The "Bubble Period" Evidence. *Review of Financial Studies* 21:103–33.
- Brealey, Richard A., and Stewart C. Myers. 1991. *Principles of Corporate Finance*. 3rd ed. New York: McGraw-Hill.
- Brown, Stephen J., and Jerold B. Warner. 1985. Using Daily Stock Returns: The Case of Event Studies. *Journal of Financial Economics* 14:3–31.

- Clement, Michael B. 1999. Analyst Forecast Accuracy: Do Ability, Resources, and Portfolio Complexity Matter? *Journal of Accounting and Economics* 27:285–303.
- Cliff, Michael T. 2007. Do Affiliated Analysts Mean What They Say? *Financial Management* 36(4):5–29.
- Cowen, Amanda, Boris Groysberg, and Paul Healy. 2006. Which Type of Analyst Firms Are More Optimistic? *Journal of Accounting and Economics* 41:119–46.
- Dechow, Patricia M., Amy P. Hutton, Lisa Meulbroek, and Richard G. Sloan. 2001. Short Interest, Fundamental Analysis, and Market Efficiency. *Journal of Financial Economics* 61:77–106.
- Dechow, Patricia, Amy Hutton, and Richard Sloan. 2000. The Relation between Analysts' Forecasts of Long-Term Earnings Growth and Stock Price Performance following Equity Offerings. *Contemporary Accounting Research* 17:1–32.
- Dugar, Amitabh, and Siva Nathan. 1995. The Effect of Investment Banking Relationships on Financial Analysts' Earnings Forecasts and Investment Recommendations. *Contemporary Accounting Research* 12:131–60.
- Fama, Eugene F., and Kenneth R. French. 1993. Common Risk Factors in the Returns on Stocks and Bonds. *Journal of Financial Economics* 33:3–56.
- Gasparino, Charles. 2002. Ghosts of E-mails Continue to Haunt Wall Street: In Grubman Inquiry, Preschool Is Pressed on Twins' Admission. *Wall Street Journal*, November 18.
- Gompers, Paul, and Josh Lerner. 1999. Conflict of Interest in the Issuance of Public Securities: Evidence from Venture Capital. *Journal of Law and Economics* 42:1–28.
- Greene, William H. 2003. *Econometric Analysis*. 5th ed. Upper Saddle River, N.J.: Prentice Hall.
- Grossman, Sanford J. 1976. On the Efficiency of Competitive Stock Markets Where Traders Have Diverse Information. *Journal of Finance* 31:573–85.
- Grossman, Sanford J., and Joseph E. Stiglitz. 1980. On the Impossibility of Informationally Efficient Markets. *American Economic Review* 70:393–408.
- Hayes, Rachel M. 1998. The Impact of Trading Commission Incentives on Analysts' Stock Coverage Decisions and Earnings Forecasts. *Journal of Accounting Research* 36:299–320.
- Healy, Paul M., and Krishna G. Palepu. 2001. Information Asymmetry, Corporate Disclosure, and the Capital Markets: A Review of the Empirical Disclosure Literature. *Journal of Accounting and Economics* 31:405–40.
- Hong, Harrison, and Jeffrey D. Kubik. 2003. Analyzing the Analysts: Career Concerns and Biased Earnings Forecasts. *Journal of Finance* 58:313–51.
- Hong, Harrison, Jeffrey D. Kubik, and Amit Solomon. 2000. Security Analysts' Career Concerns and Herding of Earnings Forecasts. *Rand Journal of Economics* 31:121–44.
- Irvine, Paul. 2004. Analysts' Forecasts and Brokerage-Firm Trading. *Accounting Review* 79:125–49.
- Jackson, Andrew R. 2005. Trade Generation, Reputation, and Sell-Side Analysts. *Journal* of Finance 60:673–717.
- Jacob, John, Thomas Z. Lys, and Margaret A. Neale. 1999. Expertise in Forecasting Performance of Security Analysts. *Journal of Accounting and Economics* 28:51–82.
- Jacob, John, Steve Rock, and David P. Weber. 2008. Do Non-investment Bank Analysts Make Better Earnings Forecasts? Journal of Accounting, Auditing, and Finance 23:23–61.
- Jain, Prem C. 1988. Response of Hourly Stock Prices and Trading Volume to Economic News. *Journal of Business* 61:219–31.
- Jarrell, Gregg A., and Annette B. Poulsen. 1989. Stock Trading before the Announcement

of Tender Offers: Insider Trading or Market Anticipation? *Journal of Law, Economics, and Organization* 5:225–48.

- Jegadeesh, Narasimhan, Joonghyuk Kim, Susan D. Krische, and Charles M. C. Lee. 2004. Analyzing the Analysts: When Do Recommendations Add Value? *Journal of Finance* 59:1083–124.
- Kadan, Ohad, Leonardo Madureira, Rong Wang, and Tzachi Zach. Forthcoming. Conflicts of Interest and Stock Recommendations: The Effects of the Global Settlement and Related Regulations. *Review of Financial Studies*.
- Kim, Oliver, and Robert E. Verrecchia. 1991. Trading Volume and Price Reactions to Public Announcements. *Journal of Accounting Research* 29:302–21.
- Kroszner, Randall S., and Raghuram G. Rajan. 1994. Is the Glass-Steagall Act Justified? A Study of the U.S. Experience with Universal Banking before 1933. American Economic Review 84:810–32.

. 1997. Organization Structure and Credibility: Evidence from Commercial Bank Securities Activities before the Glass-Steagall Act. *Journal of Monetary Economics* 39: 475–516.

- Lin, Hsiou-Wei, and Maureen McNichols. 1998. Underwriting Relationships, Analysts' Earnings Forecasts, and Investment Recommendations. *Journal of Accounting and Economics* 25:101–27.
- Ljungqvist, Alexander, Felicia Marston, Laura T. Starks, Kelsey D. Wei, and Hong Yan. 2007. Conflicts of Interest in Sell-Side Research and the Moderating Role of Institutional Investors. *Journal of Financial Economics* 85:420–56.
- Ljungqvist, Alexander, Felicia Marston, and William J. Wilhelm, Jr. 2006. Competing for Securities Underwriting Mandates: Banking Relationships and Analyst Recommendations. *Journal of Finance* 61:301–40.
- Malmendier, Ulrike, and Devin Shanthikumar. 2007. Are Small Investors Naïve about Incentives? *Journal of Financial Economics* 85:457–89.
- Maremont, Mark, and Chad Bray. 2004. In Latest Tyco Twist, Favored Analyst Got Private Eye, Gratis. *Wall Street Journal*, January 21.
- Mehran, Hamid, and René Stulz. 2007. The Economics of Conflicts of Interest in Financial Institutions. *Journal of Financial Economics* 85:267–96.
- Meulbroek, Lisa K. 1992. An Empirical Analysis of Illegal Insider Trading. *Journal of Finance* 47:1661–99.
- Michaely, Roni, and Jean-Luc Vila. 1996. Trading Volume with Private Valuation: Evidence from the Ex-Dividend Day. *Review of Financial Studies* 9:471–509.
- Michaely, Roni, and Kent Womack. 1999. Conflict of Interest and the Credibility of Underwriter Analyst Recommendations. *Review of Financial Studies* 12:653–86.
- Mikhail, Michael B., Beverly R. Walther, and Richard H. Willis. 1997. Do Security Analysts Improve Their Performance with Experience? *Journal of Accounting Research* 35 (Suppl.):131–57.
- New York Stock Exchange. 2002. *Fact Book for the Year 2001*. New York: New York Stock Exchange.
- Reuter, Jonathan, and Eric Zitzewitz. 2006. Do Ads Influence Editors? Advertising and Bias in the Financial Media. *Quarterly Journal of Economics* 121:197–227.
- Sanders, Ralph W., and John S. Zdanowicz. 1992. Target Firm Abnormal Volume around

the Initiation of Change in Control Transactions. *Journal of Financial and Quantitative Analysis* 27:109–29.

- Shleifer, Andrei. 1986. Do Demand Curves for Stocks Slope Down? *Journal of Finance* 41:579–90.
- Stickel, Scott E. 1992. Reputation and Performance among Security Analysts. *Journal of Finance* 47:1811–36.

Vijh, Anand. 1994. S&P 500 Trading Strategies and Stock Betas. *Review of Financial Studies* 7:215–51.

Womack, Kent. 1996. Do Brokerage Analysts' Recommendations Have Investment Value? Journal of Finance 51:137–67.

|   | Co<br><u>Using or</u>  | <u>Missouri</u><br>rrection o<br>Ily Securi  | -America<br>f MoPSC<br>ty Analys | an Water Co<br>Witness B<br>sts' Projecte                                  | ompany<br>arnes' DCF<br>ed Growth in EPS                       |  |   |
|---|------------------------|--|----------------------------------|--|--|--|---|
|   |                        | [1]  |                                  | [2]  | [3]  | [4]  | [2]   |
| MoPSC Witness Barnes Proxy Group<br>of Six Water Companies  | Exp<br>An<br>Divid     | ected<br>nual<br>end (1)                     | Av<br>High/L<br>Pri              | erage<br>.ow Stock<br>ce (1)   | Projected<br>Dividend Yield<br>(1)                             | Average<br>Projected EPS<br>Growth (2)             | Estimated Cost<br>of Common<br>Equity (3)                     |
| American States Water Co.<br>Aqua American, Inc.<br>California Water Service Group<br>Connecticut Water Service, Inc.<br>SJW Corp.<br>York Water Co.<br>Average | <del>လ လ လ လ လ လ</del> | 1.18<br>0.69<br>0.65<br>0.93<br>0.75<br>0.75 | აააააა                           | 33.83<br>21.36<br>18.15<br>26.09<br>22.87<br>17.07<br>17.07<br>sed Dividen | 3.49%<br>3.23%<br>3.56%<br>3.56%<br>3.05%<br>3.37%<br>d Yield: | 6.33%<br>8.88%<br>6.00%<br>9.75%<br>6.00%<br>7.16% | 9.82%<br>12.11%<br>9.58%<br>9.56%<br>9.05%<br>10.53%<br>3.37% |
|   |                        |  | Propos<br>Indicat                | sed Kange (<br>ted Range c   | of Growtn:<br>of Common Equity (                               | Cost Rate  | 6.00% - 9.75%<br>9.37% - 13.12%                               |

# Notes:

- From Schedule 17 of the Staff Report.
   From Schedule 15, column 4 of the Staff Report, supplemented with the 4.0% 5-year EPS growth rate for Connecticut Water Service, Inc. and the 6.0% 5-year EPS grwoth rate for York Water Co. from <u>The Value Line Investment Survey</u>, July 22, 2011. Column 3 + Column 4. (3)

# **Ibbotson® SBBI®** 2011 Valuation Yearbook

Market Results for Stocks, Bonds, Bills, and Inflation 1926–2010



# M RNINGSTAR®

#### 2011 Ibbotson® Stocks, Bonds, Bills, and Inflation® Valuation Yearbook

Stocks, Bonds, Bills, and Inflation® and SBBI® are registered trademarks of Morningstar, Inc. Ibbotson® and Ibbotson Associates® are registered trademarks of Ibbotson Associates, a wholly owned subsidiary of Morningstar, Inc., and are used with permission.

The information presented in this publication has been obtained with the greatest of care from sources believed to be reliable, but is not guaranteed to be complete, accurate or timely. Morningstar and its affiliated companies expressly disclaim any liability, including incidental or consequential damages, arising from the use of this publication or any errors or omissions that may be contained in it.

© 2011 Morningstar. All rights reserved. No part of this publication may be reproduced or used in any other form or by any other means—graphic, electronic, or mechanical, including photocopying, recording, taping, or information storage and retrieval systems—without Morningstar's prior, written permission. To obtain permission, please call Product Sales or write to the address below. Your request should specify the data or other information you wish to use and the manner in which you wish to use it. In addition, you will need to include copies of any charts, tables, and/or figures that you have created based on that information. There is a minimum \$1500 processing fee per request. There may be additional fees depending on your proposed usage.

Published by: Morningstar, Inc. 22 W. Washington Chicago, Illinois 60602

Main (312) 696-6000 Product Sales (888) 298-3647 Fax (312) 696-6010 global.morningstar.com/SBBIYearbooks

ISBN 978-0-9792402-9-4 ISSN 1523-343x

Ibbotson Associates<sup>®</sup> is a leading authority on asset allocation with expertise in capital market expectations and portfolio implementation. Approaching portfolio construction from the top-down through a research-based investment process, its experienced consultants and portfolio managers serve mutual fund firms, banks, brokerdealers, and insurance companies worldwide. Ibbotson Associates' methodologies and services address all investment phases, from accumulation to retirement and the transition between the two. Visit Ibbotson.com for contact information, published research, product fact sheets and other information.

For more information about Morningstar's software and data products for individuals, advisors, and institutions, see "Investment Tools and Resources" at the back of this book, or call (800) 735-0700.

Additional copies of the 2011 Ibbotson<sup>®</sup> SBBI<sup>®</sup> Valuation Yearbook may be obtained for \$175 per book, plus shipping and handling. Archived editions (2010 and prior) are available in limited quantities for \$200 per book, plus shipping and handling. For purchasing or other information related to volume discounts or companion publications, please call (888) 298-3647, or write to the address above.

# Chapter 5 The Equity Risk Premium

The expected equity risk premium can be defined as the additional return an investor expects to receive to compensate for the additional risk associated with investing in equities as opposed to investing in riskless assets. It is an essential component in several cost of equity estimation models, including the buildup method, the capital asset pricing model (CAPM), and the Fama-French three factor model. It is important to note that the expected equity risk premium, as it is used in discount rates and cost of capital analysis, is a forward-looking concept. That is, the equity risk premium that is used in the discount rate should be reflective of what investors think the risk premium will be going forward.

Unfortunately, the expected equity risk premium is unobservable in the market and therefore must be estimated. Typically, this estimation is arrived at through the use of historical data. The historical equity risk premium can be calculated by subtracting the long-term average of the income return on the riskless asset (Treasuries) from the long-term average stock market return (measured over the same period as that of the riskless asset). In using a historical measure of the equity risk premium, one assumes that what has happened in the past is representative of what might be expected in the future. In other words, the assumption one makes when using historical data to measure the expected equity risk premium is that the relationship between the returns of the risky asset (equities) and the riskless asset (Treasuries) is stable. The stability of this relationship will be examined later in this chapter.

Since the expected equity risk premium must be estimated, there is much controversy regarding how the estimation should be conducted. A variety of different approaches to calculating the equity risk premium have been utilized over the years. Such studies can be categorized into four groups based on the approaches they have taken. The first group of studies tries to derive the equity risk premium from historical returns between stocks and bonds as was mentioned above. The second group, embracing a supply side model, uses fundamental information such as earnings, dividends, or overall economic productivity to measure the expected equity risk premium. A third group adopts demand side models that derive the expected returns of equities through the payoff demanded by investors for bearing the risk of equity investments.<sup>1</sup> The opinions of financial professionals through broad surveys are relied upon by the fourth and final group.

The range of equity risk premium estimates used in practice is surprisingly large. Using a low equity risk premium estimate as opposed to a high estimate can have a significant impact on the estimated value of a stream of cash flows. This chapter addresses many of the controversies surrounding estimation of the equity risk premium and focuses primarily on the historical calculation but also discusses the supply side model.

# **Calculating the Historical Equity Risk Premium**

In measuring the historical equity risk premium one must make a number of decisions that can impact the resulting figure; some decisions have a greater impact than others. These decisions include selecting the stock market benchmark, the risk-free asset, either an arithmetic or a geometric average, and the time period for measurement. Each of these factors has an impact on the resulting equity risk premium estimate.

# The Stock Market Benchmark

The stock market benchmark chosen should be a broad index that reflects the behavior of the market as a whole. Two examples of commonly used indexes are the S&P 500° and the New York Stock Exchange Composite Index. Although the Dow Jones Industrial Average is a popular index, it would be inappropriate for calculating the equity risk premium because it is too narrow.

We use the total return of our large company stock index (currently represented by the S&P 500) as our market benchmark when calculating the equity risk premium. The S&P 500 was selected as the appropriate market benchmark because it is representative of a large sample of companies across a large number of industries. As of December 31, 1993, 88 separate industry groups were included in the index, and the industry composition of the index has not changed since. The S&P 500 is also one of
the most widely accepted market benchmarks. In short, the S&P 500 is a good measure of the equity market as a whole. Table 5-1 illustrates the equity risk premium calculation using several different market indices and the income return on three government bonds of different horizons.

| Table 5-1: Equity Risk Premium | with Different | Market Indices |             |
|--------------------------------|----------------|----------------|-------------|
| . <u></u>                      | Equity Risk Pr | emia           |             |
|                                | Long-          | Intermediate-  | Short-      |
|                                | Horizon (%)    | Horizon (%)    | Horizon (%) |
| S&P 500                        | 6.72           | 7.22           | 8.22        |
| Total Value-Weighted NYSE      | 6.52           | 7.03           | 8.02        |
| NYSE Deciles 1-2               | 5.99           | 6.50           | 7.49        |

Data from 1926-2010.

The equity risk premium is calculated by subtracting the arithmetic mean of the government bond income return from the arithmetic mean of the stock market total return. Table 5-2 demonstrates this calculation for the long-horizon equity risk premium.

| Table 5-2: Long-Horizon Equity | Risk Premium ( | Cal | culation  |    |            |
|--------------------------------|----------------|-----|-----------|----|------------|
|                                | Arithmetic Me  | an  |           |    |            |
|                                | Market Total   |     | Risk-Free | E  | quity Risk |
| Long-Horizon                   | Return (%)     |     | Rate (%)  | Pr | remium (%) |
| S&P 500                        | 11.88          |     | 5.17      | =  | 6.72*      |
| Total Value-Weighted NYSE      | 11.69          |     | 5.17      | =  | 6.52       |
| NYSE Deciles 1–2               | 11.15          | —   | 5.17      | =  | 5.99*      |

Data from 1926-2010. \*difference due to rounding.

Data for the New York Stock Exchange is obtained from Morningstar and the Center for Research in Security Prices (CRSP) at the University of Chicago's Graduate School of Business. The "Total" series is a capitalization-weighted index and includes all stocks traded on the New York Stock Exchange except closed-end mutual funds, real estate investment trusts, foreign stocks, and Americus Trusts. Capitalization-weighted means that the weight of each stock in the index, for a given month, is proportionate to its market capitalization (price times number of shares outstanding) at the beginning of that month. The "Decile 1-2" series includes all stocks with capitalizations that rank within the upper 20 percent of companies traded on the New York Stock Exchange, and it is therefore a largecapitalization index. For more information on the Center for Research in Security Pricing data methodology, see Chapter 7.

The resulting equity risk premia vary somewhat depending on the market index chosen. It is expected that using the 'Total" series will result in a higher equity risk premium than using the "Decile 1–2" series, since the "Decile 1–2" series is a large-capitalization series. As of September 30, 2010, deciles 1–2 of the New York Stock Exchange contained the largest 274 companies traded on the exchange. The "Total" series includes smaller companies that have had historically higher returns, resulting in a higher equity risk premium.

The higher equity risk premium arrived at by using the S&P 500 as a market benchmark is more difficult to explain. One possible explanation is that the S&P 500 is not restricted to the largest 500 companies; other considerations such as industry composition are taken into account when determining if a company should be included in the index. Some smaller stocks are thus included, which may result in the higher equity risk premium of the index. Another possible explanation would be what is termed the "S&P inclusion effect." It is thought that simply being included among the stocks listed on the S&P 500 augments a company's returns. This is due to the large quantity of institutional funds that flow into companies that are listed in the index.

Comparing the S&P 500 total returns to those of another large-capitalization stock index may help evaluate the potential impact of the "S&P inclusion effect." Prior to March 1957, the S&P index that is used throughout this publication consisted of 90 of the largest stocks. The index composition was then changed to include 500 large-capitalization stocks that, as stated earlier, are not necessarily the 500 largest. Deciles 1-2 of the NYSE contained just over 200 of the largest companies, ranked by market capitalization, in March of 1957. The number of companies included in the deciles of the NYSE fluctuates from quarter to quarter, and by September of 2010, deciles 1-2 contained 274 companies. Though one cannot draw a causal relationship between the change in construction and the correlation of these two indices, this analysis does indicate that the "S&P inclusion effect" does not appear to be very significant in recent periods.

Another possible explanation could be differences in how survivorship is treated when calculating returns. The Center for Research in Security Prices includes the return for a company in the average decile return for the period following the company's removal from the decile, whether caused by a shift to a different decile portfolio, bankruptcy, or other such reason. On the other hand, the S&P 500 does not make this adjustment. Once a company is no longer included among the S&P 500, its return is dropped from the index. However, this effect may be lessened by the advance announcement of companies being dropped from or added to the S&P 500. In many instances throughout this publication we will present equity risk premia using both the S&P 500 and the NYSE "Deciles 1–2" portfolio to provide a comparison between these largecapitalization benchmarks.

## The Market Benchmark and Firm Size

Although not restricted to include only the 500 largest companies, the S&P 500 is considered a large company index. The returns of the S&P 500 are capitalization weighted, which means that the weight of each stock in the index, for a given month, is proportionate to its market capitalization (price times number of shares outstanding) at the beginning of that month. The larger companies in the index therefore receive the majority of the weight. The use of the NYSE "Deciles 1-2" series results in an even purer large company index. Yet many valuation professionals are faced with valuing small companies, which historically have had different risk and return characteristics than large companies. If using a large stock index to calculate the equity risk premium, an adjustment is usually needed to account for the different risk and return characteristics of small stocks. This will be discussed further in Chapter 7 on the size premium.

## The Risk-Free Asset

The equity risk premium can be calculated for a variety of time horizons when given the choice of risk-free asset to be used in the calculation. The *2011 lbbotson® Stocks, Bonds, Bills, and Inflation® Classic Yearbook* provides equity risk premia calculations for short-, intermediate-, and long-term horizons. The short-, intermediate-, and long-horizon equity risk premia are calculated using the income return from a 30-day Treasury bill, a 5-year Treasury bond, and a 20-year Treasury bond, respectively.

Although the equity risk premia of several horizons are available, the long-horizon equity risk premium is preferable for use in most business-valuation settings, even if an investor has a shorter time horizon. Companies are entities that generally have no defined life span; when determining a company's value, it is important to use a long-term discount rate because the life of the company is assumed to be infinite. For this reason, it is appropriate in most cases to use the long-horizon equity risk premium for business valuation.

#### 20-Year versus 30-Year Treasuries

Our methodology for estimating the long-horizon equity risk premium makes use of the income return on a 20-year Treasury bond; however, the Treasury currently does not issue a 20-year bond. The 30-year bond that the Treasury recently began issuing again is theoretically more correct due to the long-term nature of business valuation, yet lbbotson Associates instead creates a series of returns using bonds on the market with approximately 20 years to maturity. The reason for the use of a 20-year maturity bond is that 30-year Treasury securities have only been issued over the relatively recent past, starting in February of 1977, and were not issued at all through the early 2000s.

The same reason exists for why we do not use the 10-year Treasury bond—a long history of market data is not available for 10-year bonds. We have persisted in using a 20-year bond to keep the basis of the time series consistent.

## Income Return

Another point to keep in mind when calculating the equity risk premium is that the income return on the appropriatehorizon Treasury security, rather than the total return, is used in the calculation. The total return is comprised of three return components: the income return, the capital appreciation return, and the reinvestment return. The income return is defined as the portion of the total return that results from a periodic cash flow or, in this case, the bond coupon payment. The capital appreciation return results from the price change of a bond over a specific period. Bond prices generally change in reaction to unexpected fluctuations in yields. Reinvestment return is the return on a given month's investment income when reinvested into the same asset class in the subsequent months of the year. The income return is thus used in the estimation of the equity risk premium because it represents the truly riskless portion of the return.<sup>2</sup>

Yields have generally risen on the long-term bond over the 1926–2010 period, so it has experienced negative capital appreciation over much of this time. This trend has turned around since the 1980s, however. Graph 5-1 illustrates the yields on the long-term government bond series

compared to an index of the long-term government bond capital appreciation. In general, as yields rose, the capital appreciation index fell, and vice versa. Had an investor held the long-term bond to maturity, he would have realized the yield on the bond as the total return. However, in a constant maturity portfolio, such as those used to measure bond returns in this publication, bonds are sold before maturity (at a capital loss if the market yield has risen since the time of purchase). This negative return is associated with the risk of unanticipated yield changes.



For example, if bond yields rise unexpectedly, investors can receive a higher coupon payment from a newly issued bond than from the purchase of an outstanding bond with the former lower-coupon payment. The outstanding lower-coupon bond will thus fail to attract buyers, and its price will decrease, causing its yield to increase correspondingly, as its coupon payment remains the same. The newly priced outstanding bond will subsequently attract purchasers who will benefit from the shift in price and yield; however, those investors who already held the bond will suffer a capital loss due to the fall in price. Anticipated changes in yields are assessed by the market and figured into the price of a bond. Future changes in yields that are not anticipated will cause the price of the bond to adjust accordingly. Price changes in bonds due to unanticipated changes in yields introduce price risk into the total return. Therefore, the total return on the bond series does not represent the riskless rate of return. The income return better represents the unbiased estimate of the purely riskless rate of return, since an investor can hold a bond to maturity and be entitled to the income return with no capital loss.

## **Arithmetic versus Geometric Means**

The equity risk premium data presented in this book are arithmetic average risk premia as opposed to geometric average risk premia. The arithmetic average equity risk premium can be demonstrated to be most appropriate when discounting future cash flows. For use as the expected equity risk premium in either the CAPM or the building block approach, the arithmetic mean or the simple difference of the arithmetic means of stock market returns and riskless rates is the relevant number. This is because both the CAPM and the building block approach are additive models, in which the cost of capital is the sum of its parts. The geometric average is more appropriate for reporting past performance, since it represents the compound average return.

The argument for using the arithmetic average is quite straightforward. In looking at projected cash flows, the equity risk premium that should be employed is the equity risk premium that is expected to actually be incurred over the future time periods. Graph 5-2 shows the realized equity risk premium for each year based on the returns of the S&P 500 and the income return on long-term government bonds. (The actual, observed difference between the return on the stock market and the riskless rate is known as the realized equity risk premium.) There is considerable volatility in the year-by-year statistics. At times the realized equity risk premium is even negative.





Data from 1926-2010.

To illustrate how the arithmetic mean is more appropriate than the geometric mean in discounting cash flows, suppose the expected return on a stock is 10 percent per year with a standard deviation of 20 percent. Also assume that only two outcomes are possible each year: +30 percent and -10 percent (i.e., the mean plus or minus one standard deviation). The probability of occurrence for each outcome is equal. The growth of wealth over a two-year period is illustrated in Graph 5-3.





The most common outcome of \$1.17 is given by the geometric mean of 8.2 percent. Compounding the possible outcomes as follows derives the geometric mean:

$$((1+0.30)\times(1-0.10))^{1/2}-1=0.082$$

However, the expected value is predicted by compounding the arithmetic, not the geometric, mean. To illustrate this, we need to look at the probability-weighted average of all possible outcomes:

| (0.25 × \$1.69) =   | \$0.4225 |
|---------------------|----------|
| + (0.50 × \$1.17) = | \$0.5850 |
| + (0.25 × \$0.81) = | \$0.2025 |
| Fotal               | \$1.2100 |
|                     |          |

Therefore, \$1.21 is the probability-weighted expected value. The rate that must be compounded to achieve the terminal value of \$1.21 after 2 years is 10 percent, the arithmetic mean:

$$1 \times (1+0.10)^2 = 1.21$$

The geometric mean, when compounded, results in the median of the distribution:

$$(1+0.082)^2 = (1.17)^2$$

The arithmetic mean equates the expected future value with the present value; it is therefore the appropriate discount rate.

## **Appropriate Historical Time Period**

The equity risk premium can be estimated using any historical time period. For the U.S., market data exists at least as far back as the late 1800s. Therefore, it is possible to estimate the equity risk premium using data that covers roughly the past 100 years.

Our equity risk premium covers the time period from 1926 to the present. The original data source for the time series comprising the equity risk premium is the Center for Research in Security Prices. CRSP chose to begin their analysis of market returns with 1926 for two main reasons. CRSP determined that the time period around 1926 was approximately when quality financial data became available. They also made a conscious effort to include the period of extreme market volatility from the late twenties and early thirties; 1926 was chosen because it includes one full business cycle of data before the market crash of 1929. These are the most basic reasons why our equity risk premium calculation window starts in 1926.

Implicit in using history to forecast the future is the assumption that investors' expectations for future outcomes conform to past results. This method assumes that the price of taking on risk changes only slowly, if at all, over time. This "future equals the past" assumption is most applicable to a random time-series variable. A time-series variable is random if its value in one period is independent of its value in other periods.

## Does the Equity Risk Premium Revert to Its Mean Over Time?

Some have argued that the estimate of the equity risk premium is upwardly biased since the stock market is currently priced high. In other words, since there have been several years with extraordinarily high market returns and realized equity risk premia, the expectation is that returns and realized equity risk premia will be lower in the future, bringing the average back to a normalized level. This argument relies on several studies that have tried to determine whether reversion to the mean exists in stock market prices and the equity risk premium.<sup>3</sup> Several academics contradict each other on this topic; moreover, the evidence supporting this argument is neither conclusive nor compelling enough to make such a strong assumption.

Our own empirical evidence suggests that the yearly difference between the stock market total return and the U.S. Treasury bond income return in any particular year is random. Graph 5-2, presented earlier, illustrates the randomness of the realized equity risk premium.

A statistical measure of the randomness of a return series is its serial correlation. Serial correlation (or autocorrelation) is defined as the degree to which the return of a given series is related from period to period. A serial correlation near positive one indicates that returns are predictable from one period to the next period and are positively related. That is, the returns of one period are a good predictor of the returns in the next period. Conversely, a serial correlation near negative one indicates that the returns in one period are inversely related to those of the next period. A serial correlation near zero indicates that the returns are random or unpredictable from one period to the next. Table 5-3 contains the serial correlation of the market total returns, the realized long-horizon equity risk premium, and inflation.

#### Table 5-3: Interpretation of Annual Serial Correlations

|                                   | Serial      | Inter-    |
|-----------------------------------|-------------|-----------|
| Series                            | Correlation | pretation |
| Large Company Stock Total Returns | 0.02        | Random    |
| Equity Risk Premium               | 0.02        | Random    |
| Inflation Rates                   | 0.64        | Trend     |

Data from 1926-2010.

The significance of this evidence is that the realized equity risk premium next year will not be dependent on the realized equity risk premium from this year. That is, there is no discernable pattern in the realized equity risk premium—it is virtually impossible to forecast next year's realized risk premium based on the premium of the previous year. For example, if this year's difference between the riskless rate and the return on the stock market is higher than last year's, that does not imply that next year's will be higher than this year's. It is as likely to be higher as it is lower. The best estimate of the expected value of a variable that has behaved randomly in the past is the average (or arithmetic mean) of its past values.

Table 5-4 also indicates that the equity risk premium varies considerably by decade. The complete decades ranged from a high of 17.9 percent in the 1950s to a low of -3.7 percent in the 2000s. This look at historical equity risk premium reveals no observable pattern.

| _      |       |       |       |       |       |       |       |       | 2001- |
|--------|-------|-------|-------|-------|-------|-------|-------|-------|-------|
| 1920s* | 1930s | 1940s | 1950s | 1960s | 1970s | 1980s | 1990s | 2000s | 2010  |
| 17.6   | 2.3   | 8.0   | 17.9  | 4.2   | 0.3   | 7.9   | 12.1  | -3.7  | -1.1  |

Schedule PMA-22 Page 9 of 18

Finnerty and Leistikow perform more econometrically sophisticated tests of mean reversion in the equity risk premium. Their tests demonstrate that—as we suspected from our simpler tests—the equity risk premium that was realized over 1926 to the present was almost perfectly free of mean reversion and had no statistically identifiable time trends.<sup>4</sup> Lo and MacKinlay conclude, "the rejection of the random walk for weekly returns does not support a mean-reverting model of asset prices."

#### Choosing an Appropriate Historical Period

The estimate of the equity risk premium depends on the length of the data series studied. A proper estimate of the equity risk premium requires a data series long enough to give a reliable average without being unduly influenced by very good and very poor short-term returns. When calculated using a long data series, the historical equity risk premium is relatively stable.<sup>5</sup> Furthermore, because an average of the realized equity risk premium is quite volatile when calculated using a short history, using a long series makes it less likely that the analyst can justify any number he or she wants. The magnitude of how shorter periods can affect the result will be explored later in this chapter.

Some analysts estimate the expected equity risk premium using a shorter, more recent time period on the basis that recent events are more likely to be repeated in the near future; furthermore, they believe that the 1920s, 1930s, and 1940s contain too many unusual events. This view is suspect because all periods contain "unusual" events. Some of the most unusual events of the last hundred years took place quite recently, including the inflation of the late 1970s and early 1980s, the October 1987 stock market crash, the collapse of the high-yield bond market, the major contraction and consolidation of the thrift industry, the collapse of the Soviet Union, the development of the European Economic Community, the attacks of September 11, 2001 and the more recent liquidity crisis of 2008 and 2009.

It is even difficult for economists to predict the economic environment of the future. For example, if one were analyzing the stock market in 1987 before the crash, it would be statistically improbable to predict the impending shortterm volatility without considering the stock market crash and market volatility of the 1929–1931 period. Without an appreciation of the 1920s and 1930s, no one would believe that such events could happen. The 85-year period starting with 1926 is representative of what can happen: it includes high and low returns, volatile and quiet markets, war and peace, inflation and deflation, and prosperity and depression. Restricting attention to a shorter historical period underestimates the amount of change that could occur in a long future period. Finally, because historical event-types (not specific events) tend to repeat themselves, long-run capital market return studies can reveal a great deal about the future. Investors probably expect "unusual" events to occur from time to time, and their return expectations reflect this.

#### A Look at the Historical Results

It is interesting to take a look at the realized returns and realized equity risk premium in the context of the above discussion. Table 5-5 shows the average stock market return and the average (arithmetic mean) realized long-horizon equity risk premium over various historical time periods. Similarly, Graph 5-5 shows the average (arithmetic mean) realized equity risk premium calculated through 2010 for different ending dates. The table and the graph both show that using a longer historical period provides a more stable estimate of the equity risk premium. The reason is that any unique period will not be weighted heavily in an average covering a longer historical period. It better represents the probability of these unique events occurring over a long period of time.

| Table 5-5: | Stock Market Return an | d Equity Risk Premium ( | Over Time    |
|------------|------------------------|-------------------------|--------------|
|            |                        | Large Company           |              |
|            |                        | Stock Arithmetic        | Long-Horizon |
| Length     | Period                 | Mean Total              | Equity Risk  |
| (Yrs.)     | Dates                  | Return (%)              | Premium (%)  |
| 85         | 1926-2010              | 11.8                    | 6.7          |
| 70         | 1941-2010              | 12.6                    | 7.0          |
| 60         | 1951-2010              | 12.3                    | 6.1          |
| 50         | 1961-2010              | 11.2                    | 4.4          |
| 40         | 1971-2010              | 11.8                    | 4.5          |
| 30         | 1981-2010              | 12.2                    | 5.0          |
| 20         | 1991-2010              | 11.0                    | 5.3          |
| 15         | 1996-2010              | 8.9                     | 3.7          |
| 10         | 2001-2010              | 3.6                     | -1.1         |
| 5          | 2006-2010              | 5.2                     | 0.8          |

Data from 1926-2010.

#### Schedule PMA-22 Page 10 of 18



Data from 1926-2010.

Looking carefully at Graph 5-4 will clarify this point. The graph shows the realized equity risk premium for a series of time periods through 2010, starting with 1926. In other words, the first value on the graph represents the average realized equity risk premium over the period 1926-2010. The next value on the graph represents the average realized equity risk premium over the period 1927-2010, and so on, with the last value representing the average over the most recent five years, 2006-2010. Concentrating on the left side of Graph 5-5, one notices that the realized equity risk premium, when measured over long periods of time, is relatively stable. In viewing the graph from left to right, moving from longer to shorter historical periods, one sees that the value of the realized equity risk premium begins to decline significantly. Why does this occur? The reason is that the severe bear market of 1973-1974 is receiving proportionately more weight in the shorter, more recent average. If you continue to follow the line to the right, however, you will also notice that when 1973 and 1974 fall out of the recent average, the realized equity risk premium jumps up by nearly 1.2 percent.

Additionally, use of recent historical periods for estimation purposes can lead to illogical conclusions. As seen in Table 5-5, the bear market in the early 2000's and in 2008 has caused the realized equity risk premium in the shorter historical periods to be lower than the long-term average.

The impact of adding one additional year of data to a historical average is lessened the greater the initial time period of measurement. Short-term averages can be affected considerably by one or more unique observations. On the other hand, long-term averages produce more stable results. A series of graphs looking at the realized equity risk premium will illustrate this effect. Graph 5-5 shows the average (arithmetic mean) realized long-horizon equity risk premium starting in 1926. Each additional point on the graph represents the addition of another year to the average. Although the graph is extremely volatile in the beginning periods, the stability of the long-term average is quite remarkable. Again, the "unique" periods of time will not be weighted heavily in a long-term average, resulting in a more stable estimate.



Data from 1926-2010.





#### Data from 1926-2010.

Some practitioners argue for a shorter historical time period, such as 30 years, as a basis for the equity risk premium estimation. The logic for the use of a shorter period is that historical events and economic scenarios present before this time are unlikely to be repeated. Graph 5-6 shows the equity risk premium measured over 30-year periods, and it appears from the graph that the premium has been trending downwards. The 30-year equity risk premium remained close to 4 percent for several years in the 1980s and 1990s. However, it has fallen and then risen in the most recent 30-year periods.

The key to understanding this result lies again in the years 1973 and 1974. The oil embargo during this period had a tremendous effect on the market. The equity risk premium for these years alone was -21 and -34 percent, respectively. Periods that include the years 1973 and 1974 result in an average equity risk premium as low as 3.1 percent. In the most recent 30-year periods that excludes 1973 and 1974, the average rises to over 6 percent. The 2000s have also had an enormous effect on the equity risk premium.

It is difficult to justify such a large divergence in estimates of return over such a short period of time. This does not suggest, however, that the years 1973 and 1974 should be excluded from any estimate of the equity risk premium; rather, it emphasizes the importance of using a long historical period when measuring the equity risk premium in order to obtain a reliable average that is not overly influenced by short-term returns. The same holds true when analyzing the poor performance of the early 2000s and 2008.

# Does the Equity Risk Premium Represent Minority or Controlling Interest?

There is quite a bit of confusion among valuation practitioners regarding the use of publicly traded company data to derive the equity risk premium. Is a minority discount implicit in this data? Recall that the equity risk premium is typically derived from the returns of a market index: the S&P 500, the New York Stock Exchange (NYSE), or the NYSE Deciles 1-2. (The size premia that are covered in Chapter 7 are derived from the returns of companies traded on the NYSE, in addition to those on the NYSE AMEX and NASDAQ). Both the S&P 500 and the NYSE include a preponderance of companies that are minority held. Does this imply that an equity risk premium (or size premium) derived from these data represents a minority interest premium? This is a critical issue that must be addressed by the valuation professional, since applying a minority discount or a control premium can have a material impact on the ultimate value derived in an appraisal.

Since most companies in the S&P 500 and the NYSE are minority held, some assume that the risk premia derived from these return data represent minority returns and therefore have a minority discount implicit within them. However, this assumption is not correct. The returns that are generated by the S&P 500 and the NYSE represent returns to equity holders. While most of these companies are minority held, there is no evidence that higher rates of return could be earned if these companies were suddenly acquired by majority shareholders. The equity risk premium represents expected premiums that holders of securities of a similar nature can expect to achieve on average into the future. There is no distinction between minority owners and controlling owners.

The discount rate is meant to represent the underlying risk of being in a particular industry or line of business. There are instances when a majority shareholder can acquire a company and improve the cash flows generated by that company. However, this does not necessarily have an impact on the general risk level of the cash flows generated by the company. When performing discounted cash flow analysis, adjustments for minority or controlling interest value may be more suitably made to the projected cash flows than to the discount rate. Adjusting the expected future cash flows better measures the potential impact a controlling party may have while not overstating or understating the actual risk associated with a particular line of business.

Appraisers need to note the distinction between a publicly traded value and a minority interest value. Most public companies have no majority or controlling owner. There is thus no distinction between owners in this setting. One cannot assume that publicly held companies with no controlling owner have the same characteristics as privately held companies with both a controlling interest owner and a minority interest owner.

## **Other Equity Risk Premium Issues**

There are a number of other issues that are commonly brought up regarding the equity risk premium that, if correct, would reduce its size. These issues include:

- 1. Survivorship bias in the measurement of the equity risk premium
- 2. Utility theory models of estimating the equity risk premium
- Reconciling the discounted cash flow approach to the equity risk premium
- 4. Over-valuation effects of the market
- 5. Changes in investor attitudes toward market conditions
- 6. Supply side models of estimating the equity risk premium

In this section, we will examine each of these issues.

## Survivorship

One common problem in working with financial data is properly accounting for survivorship. In working with company-specific historical data, it is important for researchers to include data from companies that failed as well as companies that succeeded before drawing conclusions from elements of that data.

The same argument can be made regarding markets as a whole. The equity risk premium data outlined in this book represent data on the United States stock market. The United States has arguably been the most successful stock market of the twentieth century. That being the case, might equity risk premium statistics based only on U.S. data overstate the returns of equities as a whole because they only focus on one successful market?

In a recent paper, Goetzmann and Jorion study this question by looking at returns from a number of world equity markets over the past century.<sup>6</sup> The Goetzmann-Jorion paper looks at the survivorship bias from several different perspectives. They conclude that once survivorship is taken into consideration the U.S. equity risk premium is overstated by approximately 60 basis points.<sup>7</sup> The non-U.S. equity risk premium was found to contain significantly more survivorship bias.

While the survivorship bias evidence may be compelling on a worldwide basis, one can question its relevance to a purely U.S. analysis. If the entity being valued is a U.S. company, then the relevant data set should be the performance of equities in the U.S. market.

## **Equity Risk Premium Puzzle**

In 1985, Mehra and Prescott published a paper that discussed the equity risk premium from a utility theory perspective. The point that Mehra and Prescott make is that under existing economic theory, economists cannot justify the magnitude of the equity risk premium. The utility theory model employed was incapable of obtaining values consistent with those observed in the market.

This is an interesting point and may be worthy of further study, but it does not do anything to prove that the equity risk premium is too high. It may, on the other hand, indicate that theoretical economic models require further refinement to adequately explain market behavior.

# Discounted Cash Flow versus Capital Asset Pricing Model

Two of the most commonly used cost of equity models are the discounted cash flow model and the capital asset pricing model. We should be able to reconcile the two models. In its basic form, the discounted cash flow model states that the expected return on equities is the dividend yield plus the expected long-term growth rate. The capital asset pricing model states that the expected return on equities is the risk-free rate plus the equity risk premium.<sup>8</sup> For the discounted cash flow model we can obtain an estimate of the long-term growth rate for the entire economy by looking at its component parts. Real Gross Domestic Product growth has averaged approximately three percent over long periods of time. Long-term expected inflation is currently in the range of one percent. Combining these two numbers produces an expected long-term growth rate of about four percent. Dividend yields have been between two percent and three percent historically. The discounted cash flow expected equity return is thus between six percent and seven percent using these assumptions.

If we try to reconcile this expected equity return with that found using the capital asset pricing model, we find a significant discrepancy. The yield on government bonds has been about three percent. If the two models are to reconcile, the equity risk premium must be in the three to four percent range instead of the seven to eight percent range we have observed historically.

It is not easy to explain why these two models are so difficult to reconcile. While it is possible to modify the assumptions slightly, doing so still does not produce the desired results. One explanation might be that one or both of the models are too simplistic and therefore lack the ability to resolve this inconsistency.

#### **Market Bubbles**

Another criticism of using the historical equity risk premium is that the market is overvalued. This aroument is often offered after stock prices have seen a sustained increase. The logic of the argument is that abnormally high market returns drive the historical equity risk premium higher while at the same time driving the expected equity risk premium lower. As evidence of the market being overvalued, one can look at the price/earnings multiple of the market. Graph 5-7 attempts to demonstrate the relationship between the price/earnings multiple and the subsequent period's equity risk premium. If the above argument held, one would expect to find a low equity risk premium associated with a high price/earnings multiple from the prior period. One would also expect a high equity risk premium to be associated with a low price/earnings multiple in the prior period. From the graph there does not seem to be a clear indication of the market being overvalued or undervalued with respect to the next period's realized equity risk premium.

**Graph 5-7:** Price-Earnings Multiple versus Subsequent Year's Realized Equity Risk Premium



Data from 1926–2010. Source: Historical price/earnings ratios from Standard & Poor's Security Price Index Record and Compustat database.

There are yet other problems with this theory. First, the equity risk premium is measured over a long historical time period. Several years of strong market returns have a relatively small impact on the ultimate equity risk premium estimate. Second, we are attempting to forecast a long-term equity risk premium. Even if the market were to underperform over several consecutive time periods, this should not have a significant impact on expected long-term returns. Finally, one ratio does not necessarily tell the whole story. The price/earnings ratio shows the current stock price divided by the historical earnings per share. Stock prices should, on the other hand, incorporate expectations of future earnings growth. A high market price/earnings ratio may indicate that investors expect significant future earnings growth.

## **Change in Investor Attitudes**

There is no law that states that investor attitudes must remain constant over time. With the advent of 401(k) investing and the increase in education of the investing public, the market may have changed. In fact, stock returns have become less volatile over time. Graph 5-8 demonstrates a relative decline in the rolling 60-month standard deviation of both large and small stocks. (Standard deviation is a measure of the returns' volatility or risk.) This may suggest that we have moved into a new market regime in which stocks are less volatile and therefore require a lower risk premium than in the past.<sup>9</sup>

í

**Graph 5-8:** Roiling 60-Month Standard Deviation for Large and Small Stocks



Data from January 1926-December 2010.

There are two arguments against this rationale. First, it could easily be argued that we have moved through a series of market regimes during the 85-year history of the equity risk premium calculation window used in this book. Given that markets and investor attitudes have changed over time and the equity risk premium has remained relatively constant, there is no reason to believe that a new market regime will have any greater or lesser impact than any other time period.

A second argument relates to the demand for investments. If investors are more comfortable with the market and with stock investing, they will probably place more money into the market. This influx of funds will increase the demand for stocks, which will ultimately increase, not decrease, the equity risk premium.

## **Supply Model**

Long-term expected equity returns can be forecasted by the use of supply side models. The supply of stock market returns is generated by the productivity of the corporations in the real economy. Investors should not expect a much higher or lower return than that produced by the companies in the real economy. Thus, over the long run, equity returns should be close to the long-run supply estimate.

Roger G. Ibbotson and Peng Chen forecast the equity risk premium through a supply side model using historical data.10 They utilized an earnings model as the basis for their supply side estimate; historically, the growth in corporate earnings has been in line with the growth of overall economic productivity. The earnings model breaks historical returns into four pieces, with only three historically being supplied by companies: inflation, income return, and growth in real earnings per share. The growth in the P/E ratio, the fourth piece, is a reflection of investors' changing prediction of future earnings growth. The past supply of corporate growth is forecasted to continue; however, a change in investors' predictions is not. P/E rose dramatically from 1980 through 2001 because people believed that corporate earnings were going to grow faster in the future. This growth of P/E drove a small portion of the rise in equity returns over the same period.

Graph 5-9 illustrates the price to earnings ratio calculated using one-year and three-year average earnings from 1926 to 2010. The P/E ratio, using one-year average earnings, was 10.22 at the beginning of 1926 and ended the year 2010 at 16.79—an average increase of 0.59 percent per year. The highest P/E was 136.55 recorded in 1932, while the lowest was 25.06 recorded in 1948.

Ibbotson Associates revised the calculation of the P/E ratio from a one-year to a three-year average earnings for use in equity forecasting. This is because reported earnings are affected not only by the long-term productivity, but also by "one-time" items that do not necessarily have the same consistent impact year after year. The three-year average is more reflective of the long-term trend than the year-by-year numbers. The P/E ratio calculated using the three-year average of earnings had an increase of 1.66 percent per year.





The historical P/E growth factor using three-year earnings of 1.66 percent per year is subtracted from the forecast because it is not believed that P/E will continue to increase in the future. The market serves as the cue. The current P/E ratio is the market's best guess for the future of corporate earnings and there is no reason to believe, at this time, that the market will change its mind.

Thus, the supply of equity returns only includes inflation, the growth in real earnings per share, and income return:

$$\begin{split} SR = & \left[ (1 + CPI) \times (1 + g_{REPS}) - 1 \right] + Inc + Rinv \\ & 9.24\%^* = \left[ (1 + 2.99\%) \times (1 + 1.88\%) - 1 \right] + 4.11\% + 0.21\% \\ & * difference due to rounding \end{split}$$

#### where:

The forward-looking earnings model calculates the longterm supply of U.S. equity returns to be 9.24 percent.





📓 Inflation 🗉 Growth in Earnings Per Share 📓 P/E Growth Rate 📓 Income Return

Data from 1926–2010. Results add up geometrically, not arithmetically. The darkest shade in the graph represents reinvested returns and an interaction factor between the return components.

Graph 5-10 illustrates the decomposition of historical equity returns from 1926–2010. It also illustrates the historical components that are supplied by companies: inflation, income return, and growth in real earnings per share. Once again the main difference between the historical and forecast equity returns is the exclusion of growth in P/E ratio in the forecasted earnings model. Graph 5-11: Historical and Forecast Equity Risk Premium



Data from 1926-2010. Results add up geometrically, not arithmetically. The darkest shade in the graph represents reinvested returns and an interaction factor between the return components.

Table 5-6: Supply Side and Historical Equity Risk Premium Over Time

| Period |           |        | Arithmetic Average |                   |
|--------|-----------|--------|--------------------|-------------------|
| Length | Period    |        | Supply Side Equity | Historical Equity |
| (Yrs.) | Dates     | g(P/E) | Risk Premium (%)   | Risk Premium (%)  |
| 85     | 1926-2010 | 0.60   | 5.96               | 6.72              |
| 84     | 1926-2009 | 0.96   | 5.55               | 6.67              |
| 83     | 1926-2008 | 0.79   | 5.53               | 6.47              |
| 82     | 1926-2007 | 1.15   | 5.74               | 7.06              |
| 81     | 1926-2006 | 0.75   | 6.22               | 7.13              |
| 80     | 1926-2005 | 0.65   | 6.29               | 7.08              |
| 79     | 1926-2004 | 0.83   | 6.18               | 7.17              |
| 78     | 1926–2003 | 1.09   | 5.94               | 7.19              |
| 77     | 1926-2002 | 1.17   | 5.65               | 6.97              |
| 76     | 1926-2001 | 1.53   | 5.71               | 7.43              |
| 75     | 1926-2000 | 1.49   | 6.06               | 7.76              |
| 74     | 19261999  | 1.52   | 6.32               | 8.07              |
| 73     | 1926      | 1.40   | 6.35               | 7.97              |
| 72     | 1926-1997 | 1.20   | 6.37               | 7.77              |
| 71     | 19261996  | 0.87   | 6.46               | 7.50              |
| 70     | 1926-1995 | 0.74   | 6.47               | 7.37              |
| 69     | 19261994  | 0.59   | 6.32               | 7.04              |
| 68     | 19261993  | 0.90   | 6.17               | 7.22              |
| 67     | 1926–1992 | 1.15   | 5.98               | 7.29              |
| 66     | 1926-1991 | 1.12   | 6.12               | 7.39              |
| 65     | 19261990  | 0.67   | 6.36               | 7.16              |
| 64     | 1926-1989 | 0.60   | 6.72               | 7.45              |
| 63     | 1926-1988 | 0.32   | 6.78               | 7.21              |
| 62     | 1926-1987 | 0.36   | 6.74               | 7.20              |

Data from 1926-2010.

The Supply Side equity risk premium is calculated to be 3.91 percent on a geometric basis.

SERP = 
$$\frac{(1+SR)}{(1+CPI)\times(1+RRf)} - 1$$
  
3.91%\* =  $\frac{(1+9.24\%)}{(1+2.99\%)\times(1+2.08\%)} - 1$   
\*difference due to rounding.

where: SERP the supply side equity risk premium; SR the supply of the equity return; = CPI Consumer Price Index (inflation); and, == RRf

the real risk-free rate. ==

Graph 5-11 compares the historical equity risk premium, which includes the P/E ratio, to the supply side equity risk premium calculated from 1926 to 2010 on a geometric basis. Contrary to several recent studies on equity risk premium that declare the forward-looking equity risk premium to be close to zero, or even negative, lbbotson and Chen have found the long-term supply of equity risk premium to be only slightly lower than the straight historical estimate.

The supply side equity risk premium calculated earlier is a geometric calculation. An arithmetic calculation, as mentioned earlier in the chapter, is most appropriate when discounting future cash flows. For use as the expected equity risk premium in either the CAPM or the buildup approach, the arithmetic calculation is the relevant number. There are several ways to convert the geometric average into an arithmetic average. One method is to assume the returns are independently lognormally distributed over time, where the arithmetic and geometric averages roughly follow the following relationship:

$$R_{A} = R_{G} + \frac{\sigma^{2}}{2}$$
  
5.99% = 3.88% +  $\frac{20.51\%}{2}$ 

where:

 $R_A$  = the arithmetic average;

 $R_6$  = the geometric average;

 $\sigma$  = the standard deviation of equity returns.

2

As stated in IRS Ruling 59-60, although valuation is a forward-looking process, it must be based on facts available as of the required date of appraisal. Therefore, Ibbotson provides data critical to the valuation process as far back as 1926, such as the historical equity risk premium and size premium presented in Appendix A of this book. Similarly, Table 5-6 presents the supply side equity risk premium, on an arithmetic basis, beginning in 1926 and ending in each of the last 25 years.

As mentioned earlier, one of the key findings of the lbbotson and Chen study is that P/E increases account for only a small portion of the total return of equity. The reason we present supply side equity risk premium going back only 25 years is because the P/E ratio rose dramatically over this time period, which caused the growth rate in the P/E ratio calculated from 1926 to be relatively high. The subtraction of the P/E growth factor from equity returns has been responsible for the downward adjustment in the supply side equity risk premium compared to the historical estimate. Beyond the last 25 years, the growth factor in the P/E ratio has not been dramatic enough to require an adjustment.

This section has briefly reviewed some of the more common arguments that seek to reduce the equity risk premium. While some of these theories are compelling in an academic framework, most do little to prove that the equity risk premium is too high. When examining these theories, it is important to remember that the equity risk premium data outlined in this book (both the historical and supply side estimates) are from actual market statistics over a long historical time period.

#### **Taxes and Equity Risk Premium Calculations**

All of the risk premium statistics included in this publication are derived from market returns earned by an investor. The investor receives dividends and realizes price appreciation after the corporation has paid its taxes. Therefore, it is implicit that the market return data represents returns after corporate taxes but before personal taxes. When performing a discounted cash flow analysis, both the discount rate and the cash flows should be on the same tax basis. Most valuation settings rely on after-tax cash flows; the use of an after-tax discount rate would thus be appropriate in most cases. However, there are some instances (usually because of regulatory or legal statute reasons) in which it is necessary to calculate a pre-tax value. In these cases, a pre-tax cost of capital or discount rate should be employed. There is no easy way, however, to accurately modify the return on a market index to a pre-tax basis. This modification would require estimating pre-tax returns for all of the publicly traded companies that comprise the market benchmark.

This presents a problem when a pre-tax discounted cash flow analysis is required. Although not completely correct, the easiest way to convert an after-tax discount rate to a pre-tax discount rate is to divide the after-tax rate by (1 minus the tax rate). This adjustment should be made to the entire discount rate and not to its component parts (i.e., the equity risk premium). Take note that this is a "quick and dirty" way to approximate pre-tax discount rates.

The tax rate to use in this "quick and dirty" method presents yet another problem. As seen in the discussion of the weighted average cost of capital in Chapter 1, companies do not always pay the top marginal tax rate. New research has shown some progress in quantifying the expected future tax rates. See Chapter 1 for more detail.

## Endnotes

<sup>1</sup> Page 53 Ibbotson, Roger G., Jeffrey J. Diermeier, and Laurence B. Siegel. "The Demand for Capital Market Returns: A New Equilibrium Theory." *Financial Analysts Journal*, January/February, vol. 40, no. 1, 1984, pp. 22-33. Mehra, Rajnish and Edward Prescott. "The Equity Premium: A Puzzle," *Journal of Monetary Economics*, vol. 15, no. 2, 1985, pp. 145–161.

<sup>2</sup> Page 55 Please note that the appropriate forward-looking measure of the riskless rate is the yield to maturity on the appropriate-horizon government bond. This differs from the riskless rate used to measure the realized equity risk premium historically. Chapter 4 includes a thorough discussion of riskless rate selection in this context.

<sup>3</sup> Page 58 Fama, Eugene F., and Kenneth R. French. "Permanent and Temporary Components of Stock Prices," *Journal of Political Economy*, April 1988, pp. 246–273.

Poterba, James M., and Lawrence H. Summers. "Mean Reversion in Stock Prices," *Journal of Financial Economics*, October 1988, pp. 27–59.

Lo, Andrew W., and A. Craig MacKinlay. "Stock Market Prices Do Not Follow Random Walks: Evidence from a Simple Specification Test," *The Review of Financial Studies*, Spring 1988, pp. 41–66.

Finnerty, John D., and Dean Leistikow. "The Behavior of Equity and Debt Risk Premiums: Are They Mean Reverting and Downward-Trending?" *The Journal of Portfolio Management*, Summer 1993, pp. 73–84.

Ibbotson, Roger G., and Scott L. Lummer. "The Behavior of Equity and Debt Risk Premiums: Comment," *The Journal of Portfolio Management*, Summer 1994, pp. 98–100.

Finnerty, John D., and Dean Leistikow. "The Behavior of Equity and Debt Risk Premiums: Reply to Comment," *The Journal of Portfolio Management*, Summer 1994, pp. 101–102.

Page 59 Though the study performed by Finnerty and Leistikow demonstrates that the traditional equity risk premium exhibits no mean reversion or drift, they conclude that, "the processes generating these risk premiums are generally mean-reverting."This conclusion is completely unrelated to their statistical findings and has received some criticism. In addition to examining the traditional equity risk premia, Finnerty and Leistikow include analyses on "real" risk premia as well as separate risk premia for income and capital gains. In their comments on the study, lbbotson and Lummer show that these "real" risk premia adjust for inflation twice, "creating variables with no economic content." In addition, separating income and capital gains does not shed light on the behavior of the risk premia as a whole. <sup>5</sup> Page 59 This assertion is further corroborated by data presented in *Global Investing: The Professional's Guide to the World of Capital Markets* (by Roger G. Ibbotson and Gary P. Brinson and published by McGraw-Hill, New York). Ibbotson and Brinson constructed a stock market total return series back to 1790. Even with some uncertainty about the accuracy of the data before the mid-nineteenth century, the results are remarkable. The real (adjusted for inflation) returns that investors received during the three 50-year periods and one 51-year period between 1790 and 1990 did not differ greatly from one another (that is, in a statistically significant amount). Nor did the real returns differ greatly from the overall 201-year average. This finding implies that because real stock market returns have been reasonably consistent over time, investors can use these past returns as reasonable bases for forming their expectations of future returns.

- <sup>6</sup> Page 62 Goetzmann, William, and Philippe Jorion."A Century of Global Stock Markets," Working Paper 5901, National Bureau of Economic Research, 1997. <sup>7</sup> Page 62 Note that the equity risk premium referred to in the Goetzmann and Jorion paper is not the same as the equity risk premium covered in this publication. Among other differences, their equity risk premium is based on a longer history of data and does not take dividend income or reinvestment into account.
- <sup>a</sup> Page 62 The discounted cash flow model is a modification of the Gordon Growth model, which states that: where P<sub>0</sub> is the price of the security today, D<sub>1</sub> is the dividend from next period, k is the cost of equity, and g is the expected growth rate in dividends. The capital asset pricing model is stated as  $k_i = \beta_i$  (ERP)+ $r_f$  where  $k_i$  is the cost of equity for company i,  $\beta_i$  is the beta for company i, ERP is the equity risk premium, and  $r_f$  is the risk-free rate. For the market as a whole, the capital asset pricing model can be written as  $k = ERP + r_f$  because the market beta, by definition, is 1. For more information on these models, see Chapter 4.

 <sup>9</sup> Page 63 Note that the recent increase in market volatility, particularly in 1998, may also place into question the validity of this argument.
 <sup>10</sup> Page 64 lbbotson, Roger G., and Peng Chen,"Long-Run Stock Returns: Participating in the Real Economy." *Financial Analysts Journal*, January/ February, vol. 59, no.1, 2003, pp. 88–98.

## <u>Missouri-American Water Company</u> Capital Asset Pricing Model (CAPM) Cost-of-Common-Equity Estimates for MoPSC Staff's Proxy Group of Four Water Companies Corrected to Reflect a Projected Risk-Free Rate, a Market Equity Risk Premium which Accounts for a Properly Derived Historical and projected Market Equity Risk Premium <u>as well as the Empirical Capital Asset Pricing Model (ECAPM)</u>

|   | <u>1</u>                           | <u>2</u>                   | <u>3</u>              | <u>4</u>                             | <u>5</u>                  | <u>6</u>                                       |
|---|------------------------------------|----------------------------|-----------------------|--------------------------------------|---------------------------|--|
| MoPSC Staff's Proxy Group of<br>Six Water Companies | Value Line<br>Adjusted<br>Beta (1) | Market Risk<br>Premium (2) | Risk-Free<br>Rate (3) | Traditional<br>CAPM Cost<br>Rate (4) | ECAPM<br>Cost Rate<br>(5) | Indicated<br>Common<br>Equity Cost<br>Rate (6) |
| American States Water Co.                           | 0.75                               | 9.31 %                     | 4.95 %                | 11.93 %                              | 12.51 %                   | 12.22 %  |
| Aqua American, Inc.                                 | 0.65                               | 9.31                       | 4.95                  | 11.00                                | 11.82                     | 11.41  |
| California Water Service                            | 0.70                               | 9.31                       | 4.95                  | 11.47                                | 12.17                     | 11.82  |
| Connecticut Water Service                           | 0.80                               | 9.31                       | 4.95                  | 12.40                                | 12.86                     | 12.63  |
| SJW Corp.   | 0.90                               | 9.31                       | 4.95                  | 13.33                                | 13.56                     | 13.45  |
| York Water Co.                                      | 0.70                               | 9.31                       | 4.95                  | 11.47                                | 12.17                     | 11.82  |
| Average   |                                    |                            |                       | <u>11.93</u> %                       | <u>    12.51 </u> %       | <u>    12.23 </u> %                            |

#### Notes

- (1) From Column 2 of Schedule 18 of Mr. Barnes' Direct Exhibit.
- (2) Average of the Ibbotson long-term arithmetic mean risk premium of 6.70% and the projected 3-5 year return of the market as calculated by the 13 week average market appreciation potential published by Value Line ended September 30, 2011 minus Mr. Barnes' projected risk-free rate. The average risk premium is 9.31%. ((6.70% + 11.91%) / 2 = 9.31%)
- (3) Average of the projected risk-free rate for the years 2012 and 2013 as shown on Schedule 5 of Mr. Barnes' Direct Exhibit. ((4.90% + 5.00%) / 2 = 4.90%)
- (4) Calculated as shown on page 2 of Schedule PMA-12, note 3.
- (5) Calculated as shown on page 2 of Schedule PMA-12, note 4.
- (6) Average of Columns 4 and 5.

|          | Derivati<br><u>Ibbotson Associates' Siz</u>                                  | <u>Missouri-</u><br>on of Inves<br><u>Premia f</u> | -American Wa<br>stment Risk Ac<br>or the Decile F | ter Company<br>djustment Based upon<br>Portfolios of the NYSE/  | AMEX/NASDAQ  |  |  |
|----------|--|--|---|---|--|--|--|
|          |  |  |   |   | N  | က၊   | 41   |
| Line No. |  | N<br>M<br>M  | <i>A</i> arket Capitali<br>Ilions)                | zation (1)<br>(times larger)  | Applicable Decile of<br>the NYSE/AMEX/<br>NASDAQ (2)   | Applicable Size<br>Premium (3)   | Spread from<br>Applicable Size<br>Premium for (4)            |
| ť.       | Missouri-American Water Company  |  |   |   |  |  |  |
|          | Based Upon the MoPSC Staff's Proxy Group of Six Water<br>a. Companies        | ф  | 827.693   |   | 7 - 8  | 2.27%  |  |
|          | Based Upon the MIEC Witness Gorman's Proxy Group of Eight b. Water Companies | ŝ  | 764.919   |   | 7 - 8  | 2.27%  |  |
|          | Based Upon the BJC Witness LaConte's Proxy Group of Nine c. Water Companies  | \$   | 746.628   |   | ω  | 2.65%  |  |
| 5        | MoPSC Staff's Proxy Group of Six Water Companies                             | ÷  | 868.516   | 1.0 ×   | 7 - 8  | 2.27%  | 0.00%  |
| Э        | MIEC Witness Gorman's Proxy Group of Eight Water<br>Companies                | \$<br>   | ,319.694  | 1.7 ×   | 6 - 7  | 1.85%  | 0.42%  |
| 4.       | BJC Witness LaConte's Proxy Group of Nine Water Companies                    | \$<br>-  | ,205.786  | 1.6 ×   | 6 - 7  | 1.85%  | 0.80%  |
|          |  | Ŭ  | (A)   | (B)   | (C )   | ( <u>C</u> )   | (E)  |
|          |  | ă  | ecile   | Number of<br>Companies<br>( millions )  | Recent Total Market<br>Capitalization<br>( millions )  | Recent Average<br>Market<br>Capitalization<br>(millions)   | Size Premium<br>(Return in<br>Excess of<br>CAPM) (2)         |
|          | Largest  |  | - 0 m 4 m m v m o Ó                               | 168<br>187<br>187<br>185<br>230<br>230<br>230<br>287<br>361<br>1320   | <ul> <li>\$ 8,586,385,656</li> <li>1,873,378,709</li> <li>1,022,604,243</li> <li>594,702,185</li> <li>482,327,242</li> <li>302,40,48,414</li> <li>239,018,595</li> <li>181,744,805</li> <li>119,075</li> </ul> | <ul> <li>\$ 51,109,438</li> <li>\$ 10,350,159</li> <li>\$ 5,468,472</li> <li>\$ 3,214,606</li> <li>\$ 2,264,447</li> <li>\$ 1,565,828</li> <li>\$ 1,565,828</li> <li>\$ 1,062,538</li> <li>\$ 370,152</li> <li>\$ 370,152</li> <li>\$ 370,152</li> </ul> | -0.38%<br>0.81%<br>1.01%<br>1.82%<br>1.88%<br>2.94%<br>6.36% |
|          | Notes:<br>(1<br>(2   | From p<br>Gleane                                   | ages 2 - 4 of t<br>ed from Colum                  | his Schedule.<br>In (D) on the bottom of the portom of the portor of the bottom of the bottom of the bottom of the portor of the po | ,<br>of this page. The appropris   | From Ibbotson 2011 Ye<br>ite decile (Column (A))   | arbook<br>corresponds to the                                 |
|          | (3)  | Corresp<br>Line No<br>example                      | or of the 0.42% let the 0.42%                     | remium to the decile is<br>3 – Line No. 2 Colum<br>in Column 4, Line No.  | provided on Column (E) on<br>n 3 and Line No. 1b, Colu<br>2 is derived as follows 0.42°  | the bottom of this page.<br>mn 3 – Line No. 3 of (<br>6 = 2.27% - 1.85%.   | Column 3 etc For   |

Schedule PMA-24 Page 1 of 4

|  |   | <del>.</del> –I   |  | ~  |   | ကျ   |  | 41   | ای   |                          | Q  |
|--|---|---|--|--|---|--|--|--|--|--------------------------|--|
| Company  | Exchange  | Common Stock Shares<br>Outstanding at Fiscal<br>Year End 2010<br>(millions)   | Book \<br>Share<br>Year En   | /alue per<br>at Fiscal<br>d 2010 (1)   | Total Com<br>Fiscal Ye<br>(mi   | mon Equity at<br>ar End 2010<br>Ilions)  | Avera<br>Low St<br>(7/11 -                   | je High /<br>ock Price<br>9/11) (2)                      | Market-to-Book<br>Ratio (3)                          | Capit<br>(1              | Market<br>alization (4)<br>nillions)                             |
| Missouri-American Water Company  |   | NA  |  | NA   | ¢   | 415.717 (5)  |  | NA   |  |                          |  |
| Based Upon the MoPSC Staff's Proxy Group of<br>Six Water Companies   |   |   |  |  |   |  |  | ·  | 199.1 % (  | 9) \$                    | 827.693 (7)  |
| MoPSC Staff's Proxy Group of Six Water<br>Companies  |   |   |  |  |   |  |  |  |  |                          |  |
| Companies<br>American States Water Co.<br>Aqua America, Inc.<br>California Water Service Group<br>Connecticut Water Service, Inc.<br>SJW Corporation<br>York Water Company | NYSE<br>NYSE<br>NYSE<br>NASDAQ<br>NYSE<br>NASDAQ                          | 18.631<br>138.449<br>41.666<br>8.677<br>18.552<br>12.692  | Ф  | 20.264<br>8.481<br>10.453<br>13.134<br>13.747<br>7.190   | θ   | 377.541<br>1,174.254<br>1,35.526<br>113.963<br>255.032<br>91.257                     | \$   | 33.830<br>21.360<br>18.150<br>26.090<br>22.870<br>17.070 | 166.9 %<br>251.9<br>173.6<br>198.6<br>166.4<br>237.4 | \$                       | 630.282<br>2,957.271<br>756.238<br>226.379<br>424.274<br>216.653 |
| Average  | VA= Not Available   | 39.778  | Ф  | 12.212   | ¢   | 407.929  | Ф  | 23.228   | 199.1 %  | θ                        | 868.516  |
|  | Notes: (1)<br>(2)<br>(3)<br>(4)<br>(4)<br>(5)<br>(6)<br>(6)<br>(6)<br>(7) | Column 3 / Column 1.<br>From Schedule 16.<br>Column 4 / Column 2.<br>Column 5 * Column 3.<br>From Financial Statements of<br>The market-to-book ratio of 1<br>Staff's proxy group of six wa<br>Missouri-American Water Co<br>21, 2011 of the MoPSC Staf | of Missouri-A<br>Missouri-Am<br>Missouri-Am<br>ter company's co<br>ompany's co<br>fs proxy gro | merican Wate<br>erican Wate C<br>erican Water C<br>ss at October 2<br>mmon stock, if<br>up of six water<br>33 million. | r Company for<br>Sompany on Oc<br>11, 2011.<br>traded, would<br>companies, 19 | Fiscal Year End 20<br>tober 21, 2011 is a<br>trade at a market-t<br>9.1%, and Missou | 010.<br>assumed t <sub>i</sub><br>o-book rat | o be equal to t<br>o equal to the                        | he market-to-book ratii<br>average market-to-bo      | o of the M<br>ok ratio a | IoPSC<br>It October<br>October 21,                               |
| Source of Information: 2   | 2010 Annual Forms 10k   |   |  |  |   |  |  |  |  |                          |  |

Market Capitalization of Missouri-American Water Company Market Capitalization of Missouri-American Water Company and the MoPSC Staff's Proxy Group of Six Water Companies

urce of Information: 2010 Annual Forms 10K yahoo.finance.com

Schedule PMA-24 Page 2 of 4

|  |  | τI   |   | N   |   | ကျ  |                                 | 41                                | ام   | JD   |       |
|--|--|--|---|---|---|---|---------------------------------|-----------------------------------|--|--|-------|
| Company  | Exchange   | Common Stock Shares<br>Outstanding at Fiscal<br>Year End 2010<br>(millions )   | Book /<br>Share<br>Year Er  | /alue per<br>at Fiscal<br>d 2010 (1)  | Total Com<br>Fiscal Ye  | nmon Equity at<br>sar End 2010<br>iillions)   | 13-Wee<br>Stock                 | k Average<br>Price (2)            | Market-to-Book<br>Ratio (3)                    | Market<br>Capitalization (4)<br>( millions )               |       |
| Missouri-American Water Company  |  | NA   |   | NA  | ÷   | 415.717 (5)   |                                 | AN                                |  |  |       |
| Based Upon the MIEC Witness Gorman's<br>Proxy Group of Eight Water Companies |  |  |   |   |   |   |                                 |                                   | 184.0 % (                                      | () \$ 764.919 (  | <br>۲ |
| MIEC Witness Gorman's Proxy Group of Eight<br>Water Companies                |  |  |   |   |   |   |                                 |                                   |  |  |       |
| American States Water Co.<br>American Water Works Co., Inc.                  | NYSE<br>NYSE   | 18.631<br>174.996  | Ŷ   | 20.264<br>23.614  | ÷   | 377.541<br>4,132.272  | Ф                               | 33.950<br>29.040                  | 167.5 %<br>123.0                               | \$ 632.517<br>5,081.884                                    |       |
| Aqua America, Inc.<br>California Water Service Group                         | NYSE   | 138.449<br>41.666  |   | 8.481<br>10.453   |   | 1,174.254<br>435.526  |                                 | 21.380<br>17.930                  | 252.1<br>171.5                                 | 2,960.040<br>747.071                                       |       |
| Connecticut Water Service, Inc.<br>Middlesex Water Company                   | NASDAQ<br>NASDAQ   | 8.677<br>15.566  |   | 13.134<br>11.132  |   | 113.963<br>173.279  |                                 | 26.170<br>17.740                  | 199.3<br>159.4                                 | 227.073<br>276.141   |       |
| SJW Corporation<br>York Water Company  | NYSE<br>NASDAQ   | 18.552<br>12.692   |   | 13.747<br>7.190   |   | 255.032<br>91.257   |                                 | 22.570<br>16.870                  | 164.2<br>234.6                                 | 418.708<br>214.115   |       |
| Average  | IA= Not Available  | 53.654   | φ   | 13.502  | θ   | 844.141   | θ                               | 23.206                            | 184.0 %  | \$ 1,319.694   |       |
|  | Notes: (1)<br>(2)<br>(3)<br>(3)<br>(4)<br>(5)<br>(5)<br>(6)<br>(6) | Column 3 / Column 1.<br>From Schedule MPG-9.<br>Column 4 / Column 2.<br>Column 5 * Column 3.<br>From Financial Statements of Nitness Gorman's Proxy Gr<br>Mitness Gorman's Proxy Gr<br>Missouri-American Water Co<br>Missouri-American Water Co<br>2040ber 21, 2011 wild the | f Missouri-A<br>Missouri-Am<br>oup of Eight<br>mpany's co<br>s Gorman's | vmerican Water<br>erican Water C<br>Water Compar<br>mmon stock, if<br>Proxy Group o<br>seen \$764,919 | - Company for<br>ompany on O<br>intes at Octobe<br>itraded, would<br>if Eight Water<br>million. | Fiscal Year End 20<br>ctober 21, 2011 is a<br>r 21, 2011.<br>trade at a market-tr<br>Companies, 184%, | 10.<br>issumed to<br>-book rati | ) be equal to t<br>o equal to the | he market-to-book rati<br>average market-to-bo | of the MIEC<br>k ratio at October<br>cet capitalization on |       |
| Source of Information: 2   | 010 Annual Forms 10K   |  |   |   |   |   |                                 |                                   |  |  |       |

Missouri-American Water Company Market Capitalization of Missouri-American Water Company and the MIEC Witness Gorman's Proxy Group of Eight Water Companies

> rce of Information: 2010 Annual Forms 10 yahoo.finance.com

|  |  | ۲-۱  |  |  |   | ကျ   |  |                               | ای   | Q  |            |
|--|--|--|--|--|---|--|--|-------------------------------|--|--|------------|
| Company  | Exchange   | Common Stock Shares<br>Outstanding at Fiscal<br>Year End 2010<br>(millions )   | Book V<br>Share<br>Year En   | ʻalue per<br>at Fiscal<br>d 2010 (1)   | Total Com<br>Fiscal Ye<br>(mi   | mon Equity at<br>ar End 2010<br>Ilions)  | 8/2011 -<br>Averagi<br>Price                 | 11/2011<br>e Stock<br>e (2)   | Market-to-Book<br>Ratio (3)                      | Market<br>Capitalization<br>( millions ) | (4)        |
| Missouri-American Water Company  |  | NA   |  | NA   | Ş   | 415.717 (5)  |  | NA                            |  |  |            |
| Based Upon the BJC Witness LaConte's Proxy Group of Nine Water Companies |  |  |  |  |   |  |  |                               | 179.6 % (6                                       | ) \$ 746.6                               | 528 (7)    |
| BJC Witness LaConte's Proxy Group of Nine<br>Water Companies             |  |  |  |  |   |  |  |                               |  |  |            |
| American States Water Co.<br>American Water Works Co. Inc.               | NYSE<br>NYSE   | 18.631<br>174 996  | Ф  | 20.264<br>23.614   | ÷   | 377.541<br>4 132 272   | θ  | 34.040<br>29.630              | 168.0 %<br>125.5                                 | \$ 634.1<br>5.185.1                      | 194<br>131 |
| Aqua America, Inc.   | NYSE   | 138.449  |  | 8.481  |   | 1,174.254  |  | 21.590                        | 254.6  | 2,989.1                                  | 115        |
| Artesian Kesources Corp.<br>Califomia Water Service Group                | NASDAQ<br>NYSE   | 41.666   |  | 12.459<br>10.453   |   | 95.146<br>435.526  |  | 17.680                        | 143.8<br>169.1                                   | 136.7<br>736.6                           | 779<br>355 |
| Connecticut Water Service, Inc.  | NASDAQ   | 8.677  |  | 13.134   |   | 113.963  |  | 22.620                        | 172.2  | 196.2                                    | 270        |
| Middlesex vvater Company<br>SJW Corporation                              | NYSE   | 13.300   |  | 13.747   |   | 255.032  |  | 17.320<br>26.270              | 191.4  | 487.3                                    | 349        |
| York Water Company   | NASDAQ   | 12.692   |  | 7.190  |   | 91.257   |  | 16.850                        | 234.4  | 213.8                                    | 361        |
| Average  | IA= Not Available  | 48.541   | ф  | 13.386   | φ   | 760.919  | θ  | 22.679                        | 179.6 %  | \$ 1,205.7                               | 786        |
|  | Notes: (1)<br>(2)<br>(3)<br>(4)<br>(4)<br>(5)<br>(6)<br>(6)<br>(7) | Column 3 / Column 1.<br>From Schedule BSL-2.<br>Column 4 / Column 2.<br>Column 5 * Column 3.<br>From Financial Statements of<br>Witness LaConte's Proxy Gr<br>Missouri-American Water Co<br>21, 2011 of the BJC Witness<br>October 21, 2011 would ther | of Missouri-A<br>Missouri-Am<br>oup of Nine<br>mpany's cine<br>t-LaConte's F | merican Wate<br>merican Water C<br>Water Compar<br>mon stock, if<br>mors Stock, 628<br>een \$746.628 | r Company for l<br>company on Oc<br>iosmpany on Oc<br>iosat October<br>traded, would i<br>rine Water Co<br>million. | Fiscal Year End 20<br>tober 21, 2011 is a<br>21, 2011.<br>z1e at a market-to<br>mpanies, 179.6%, | 10.<br>ssumed to<br>+book ratio<br>and Misso | be equal to t<br>equal to the | he market-to-book ratio<br>average market-to-boo | of the BJC<br>k ratio at Octobe          | er<br>on   |
| Source of Information: 2   | 010 Annual Forms 10K   |  |  |  |   |  |  |                               |  |  |            |

Market Capitalization of Missouri-American Water Company Market Capitalization of Missouri-American Water Company and the BJC Witness LaConte's Proxy Group of Nine Water Companies

> Jurce of Information: 2010 Annual Forms 10K yahoo.finance.com

#### Capital Structure Based upon Total Permanent Capital for the MIEC Witness Gorman's Proxy Group of Eight Water Companies 2006 - 2010, Inclusive

|   |             |             |             |             |                   | <u>5 YEAR</u>  |
|---|-------------|-------------|-------------|-------------|-------------------|----------------|
|   | <u>2010</u> | <u>2009</u> | <u>2008</u> | <u>2007</u> | 2006              | <u>AVERAGE</u> |
| American States Water Co.                           |             |             |             |             |                   |                |
| Long-Term Debt                                      | 44.30 %     | 46 95 %     | 46 25 %     | 46.99 %     | 48 61 %           | 46 62 %        |
| Preferred Stock                                     | 0.00        | 0.00        | 0.00        | 0.00        | 0.00              | 0.00           |
| Common Equity                                       | 55 70       | 53.05       | 53 75       | 53.01       | 51 39             | 53 38          |
| Total Capital                                       | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %          | 100.00 %       |
| Amorican Water Works Co                             |             |             |             |             |                   |                |
| Inc.  |             |             |             |             |                   |                |
| Long-Term Debt                                      | 56.73 %     | 56.98 %     | 53.75 %     | 51.05 %     | 46.93 %           | 53.08 %        |
| Preferred Stock                                     | 0.29        | 0.30        | 0.32        | 0.31        | 0.06              | 0.26           |
| Common Equity                                       | 42.98       | 42 72       | 45.93       | 48 64       | 53.01             | 46.66          |
| Total Capital                                       | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %          | 100.00 %       |
| Aqua America Inc                                    |             |             |             |             |                   |                |
| Long Torm Dobt                                      | E7 OE 9/    | E6 E0 %     | E4 01 0/    | EE 00 0/    | E1 EE 0/          | EE 06 %        |
| Proferred Stock                                     | 0.02        | 0.02        | 0.00        | 0.00 //     | 0.10              | 0.06           |
|   | 12.02       | 43.30       | 45 70       | 0.09        | 48.35             | 44.88          |
| Total Capital                                       | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %          | 100.00 %       |
| California Water Service<br>Group                   |             |             |             |             |                   |                |
| Long-Term Debt                                      | 52.51 %     | 47.93 %     | 41.88 %     | 42.86 %     | 43.47 %           | 45.73 %        |
| Preferred Stock                                     | 0.00        | 0.00        | 0.00        | 0.51        | 0.51              | 0.20           |
| Common Equity                                       | 47.49       | 52.07       | 58.12       | 56.63       | 56.02             | 54.07          |
| Total Capital                                       | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %          | 100.00 %       |
| Connecticut Water Service.<br>Inc.                  |             |             |             |             |                   |                |
| Long-Term Debt                                      | 49.32 %     | 50.59 %     | 46.94 %     | 47.76 %     | 44.42 %           | 47.81 %        |
| Preferred Stock                                     | 0.34        | 0.35        | 0.39        | 0.44        | 0.49              | 0.40           |
| Common Equity                                       | 50.34       | 49.06       | 52.67       | 51.80       | 55.09             | 51.79          |
| Total Capital                                       | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %          | 100.00 %       |
| Middlesex Water Company                             |             |             |             |             |                   |                |
| Long-Term Debt                                      | 43.91 %     | 47.35 %     | 49.10 %     | 49.48 %     | 48.78 %           | 47.72 %        |
| Preferred Stock                                     | 1.07        | 1.24        | 1.22        | 1.46        | 2.95              | 1.59           |
| Common Equity                                       | 55.02       | 51.41       | 49.68       | 49.06       | 48.27             | 50.69          |
| Total Capital                                       | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %          | 100.00 %       |
| SJW Corporation                                     |             |             |             |             |                   |                |
| Long-Term Debt                                      | 53.79 %     | 49.52 %     | 46.08 %     | 47.79 %     | 41.83 %           | 47.80 %        |
| Preferred Stock                                     | 0.00        | 0.00        | 0.00        | 0.01        | 0.01              | 0.00           |
| Common Equity                                       | 46.21       | 50.48       | 53.92       | 52.20       | 58.16             | 52.20          |
| Total Capital                                       | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %          | 100.00 %       |
| York Water Company                                  |             |             |             |             |                   |                |
| Long-Term Debt                                      | 48.28 %     | 47.16 %     | 55.31 %     | 51.17 %     | 48.82 %           | 50.15 %        |
| Preferred Stock                                     | 0.00        | 0.00        | 0.00        | 0.00        | 0.00              | 0.00           |
| Common Equity                                       | 51.72       | 52.84       | 44.69       | 48.83       | 51.18             | 49.85          |
| Total Capital                                       | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %    | 100.00 %          | 100.00 %       |
| MIEC Witness Gorman's<br>Proxy Group of Eight Water |             |             |             |             |                   |                |
| Long Torm Dobt                                      | E0 70 0/    | E0 39 9/    | 40.40.9/    | 40.40.9/    | 46.00.0/          | 40.04.04       |
| Long-Term Debt                                      | 50.73 %     | 50.38 %     | 49.19 %     | 49.12 %     | 46.80 %           | 49.24 %        |
| Common Equity                                       | 0.22        | 0.24        | 0.25        | 0.35        | 0.52              | 0.32           |
| Total Capital                                       | 49.05       | 49.38       |             | 100.00 %    | 5∠.68<br>100.00 % | 100.00 %       |
| i Jiai Dapitai                                      | 100.00 /6   | 100.00 /0   | 100.00 /0   | 100.00 /0   | 100.00 /6         | 100.00 %       |

Source of Information

EDGAR Online's I-Metrix Database

Annual Forms 10-K

# Missouri-American Water Company Brief Summary of MoPSC Staff's Corrected Common Equity Cost Rate

|     |  | MoPSC Staff's<br>Proxy Group of Six |
|-----|--|-------------------------------------|
| No. | Principal Methods  | Water Companies                     |
| 1.  | Discounted Cash Flow Model (DCF) (1)                                   | 10.53 %                             |
| 2.  | Capital Asset Pricing Model (CAPM) (2)                                 | 12.23                               |
| 3.  | Indicated Common Equity Cost Rate before Adjustment for Business Risks | 11.38 %                             |
| 4.  | Flotation Cost Adjustment (3)  | 0.15                                |
| 5.  | Financial Risk Adjustment (4)  | 0.75                                |
| 6.  | Business Risk Adjustment (5)   | 0.35                                |
| 7.  | Indicated Common Equity Cost Rate                                      | 12.63 %                             |

- Notes: (1) From Schedule PMA-19.
  - (2) From Schedule PMA-21.
  - (3) From Ms. Ahern's electronic rebuttal workpapers.
  - (4) Financial risk adjustment to reflect the greater financial risk inherent the MoPSC Staff's recommended capital structure relative to Staff's proxy group of six water companies.
  - (5) Business risk adjustment to reflect Missouri-American Water Company's greater unique business risks relative to the proxy group as detailed in Ms. Ahern's direct testimony.





Schedule PMA-27 Page 1 of 2 Equity Risk Premium May 2010 - November 2011



ORIGINAL ARTICLE

# New approach to estimating the cost of common equity capital for public utilities

Pauline M. Ahern · Frank J. Hanley · Richard A. Michelfelder

Published online: 26 August 2011 © Springer Science+Business Media, LLC 2011

**Abstract** The regulatory process for setting public utilities' allowed rate of return on common equity has generally used the Gordon DCF, CAPM and Risk Premium specifications to estimate the cost of common equity. Despite the widely known problems with these models, there has been little movement to adopt more recently developed asset pricing models to provide additional evidence for estimating the cost of capital. This paper presents, validates empirically and applies a general yet simple consumption-based asset pricing specification to model the risk-return relationship for stocks and estimate the cost of common equity for public utilities. The model is not necessarily superior to other models in its practical results, yet these results do indicate that it should be used to provide additional estimates of the cost of common equity. Additionally, the model raises doubts as to whether assets such as utility stocks are a consumption (business cycle) hedge.

**Keywords** Public utilities · Cost of capital · GARCH · Consumption asset pricing model

JEL Classification G12 · L94 · L95

P. M. Ahern · F. J. Hanley AUS Consultants, Mt. Laurel, NJ 08054, USA e-mail: pahern@ausinc.com

F. J. Hanley e-mail: Fhanley@ausinc.com

R. A. Michelfelder (⊠) School of Business, Rutgers University, Camden, NJ 08102-1656, USA e-mail: richmich@rutgers.edu

# **1** Introduction

Following electricity deregulation with the National Energy Policy Act of 1992, the estimation of the cost of common equity capital remains a critical component of the utility rate-of-return regulatory process. Since the cost of common equity is not observable in capital markets, it must be inferred from asset pricing models. The models that are commonly applied in regulatory proceedings are the Gordon (1974) Discounted Cash Flow (DCF), the Capital Asset Pricing (CAPM) and Risk Premium Models. There are other tools used to estimate the cost of common equity such as comparable earnings or earnings-to-price ratios, but they are not asset pricing models. The empirical literature on the CAPM is vast {Fama and French (2004)} and the CAPM is used by a number of US regulatory jurisdictions. The DCF model has not been empirically tested to the same extent as the CAPM, yet it is considered by many US regulatory jurisdictions.

The purpose of this paper is to present, test empirically and apply a recently developed general consumption-based asset pricing model that estimates the risk-return relationship directly from asset pricing data and, when estimated with recently developed time series methods, produces a prediction of the equity risk premium that is driven by its predicted volatility. The predicted risk premium is then added to a riskfree rate of return to provide an estimate of the cost of common equity. We predict two forms of the equity risk premium with the model, the risk premium net of the risk-free rate and the equity-to-debt risk premium (equity risk premium net of the relevant bond yield for the company's stock). Either can be applied to predict the common equity cost of capital for a public utility. Although the model is tested and applied to public utilities for rate of return regulation, it can be used to estimate the cost of capital for any stock. Section 2 reviews the asset pricing models typically used in public utility rate cases and the generalized consumption asset pricing model we propose to estimate the cost of common equity. Section 3 discusses the data and the empirical testing of the consumption asset pricing model. Section 4 reviews the application of the model and compares it with the DCF and CAPM results. Section 5 is the conclusion.

# 2 DCF, CAPM and consumption asset pricing model

# 2.1 DCF and CAPM approaches

The standard DCF model frequently used in estimative the cost rate of common equity in regulatory proceedings is defined by the following equation:

$$k = D_0 \left( 1 + g \right) / P_0 + g,$$

where k is the expected return on common equity;  $D_0$  is the current dividend per share; g is the expected dividend per share growth rate; and  $P_0$  is the current market price.

The DCF was developed by Gordon (1974) specifically for regulatory purposes. Underlying the DCF model is the theory that the present value of an expected future stream of net cash flows during the investment holding period can be determined by discounting those cash flows at the cost of capital, or the investors' capitalization rate. DCF theory indicates that an investor buys a stock for an expected total return rate which is derived from cash flows received in the form of dividends plus appreciation in market price (the expected growth rate) over the investment holding period. Mathematically, the expected dividend yield  $(D_0(1 + g)/P_0)$  on market price plus an expected growth rate equals the capitalization rate, i.e., the expected return on common equity.

The standard DCF contains several restrictive assumptions, the most contentious of which during utility cost of capital proceedings is typically that dividends per share (DPS), book value per share (BVPS), earnings per share (EPS) as well as market price grow at the same rate in perpetuity. There is also considerable contention over the proper proxy for g, prospective or historical growth in DPS, BVPS, EPS and market price and over what time period. In addition, although the standard DCF described above is a single stage annual growth model, there is considerable discussion over the use of multiple stage growth models during regulatory proceedings. Some analysts use the discrete version and others use the continuous version of the DCF model. Solving these models for k, the cost of common equity, results in differing equations to solve for k. The equation above is from the discrete version. The continuous version uses the current dividend yield and is not adjusted by g, which results in a lower estimate for k. Because of these and other restrictive assumptions that require numerous subjective judgments in application, it is often difficult for regulatory commissions to reconcile the frequently large disparities in rates of return on common equity recommended by various parties in a public utility rate case.

The CAPM model is defined by the following equation:

$$k = R_f + \beta \left( R_m - R_f \right),$$

where k is the expected return on common equity;  $R_f$  is the expected risk-free rate of return;  $\beta$  is the expected beta; and  $R_m$  is the expected market return.

CAPM theory defines risk as the co-variability of a security's returns with the market's returns or  $\beta$ , also known as systematic or market risk, with the market beta being defined as 1.0. Because CAPM theory assumes that all investors hold perfectly diversified portfolios, they are presumed to be exposed only to systematic risk and the market (according to the model) will not reward them a risk premium for unsystematic or non-market risk. In other words, the CAPM presumes that investors require compensation only for systematic or market risks which are due to macroeconomic and other events that affect the returns on all assets. Mathematically, the CAPM is applied by adding a forward-looking risk-free rate of return to an expected market equity risk premium adjusted proportionately by the expected beta to reflect the systematic risk.

As with the DCF, there is considerable contention during regulatory cost of capital proceedings as to the proper proxies for all components of the CAPM: the  $R_f$ , the  $R_m$ , as well as  $\beta$ . In addition, the CAPM assumption that the market will only reward investors for systematic or market risk is extremely restrictive when estimating the expected return on common equity for a single asset such as a single jurisdictional regulated operating utility. Additionally, this assumption requires that the investor have a perfectly diversified portfolio, that is, one with no unsystematic risk. Since

this assumption is not applicable, estimating the cost of common equity capital for a single utility's common equity undoubtedly will not reflect the risk actually faced by the imperfectly diversified investor.

As will be discussed in the next section, our application of the risk premium approach, the consumption asset pricing model and GARCH<sup>1</sup> rest on minimal assumptions and restrictions and therefore requires considerably less judgment in its application.

## 2.2 Risk premium approach, consumption asset pricing models, and GARCH

A widely used model to estimate the cost of common equity capital for public utilities is the risk premium approach. This approach often estimates the expected rate of return as the long-term historic mean of the realized risk premium above an historic yield plus the current yield of the relevant bond applicable to a specific utility or peer group of utilities. Litigants in public utility rate proceedings debate the choice of inputs to estimate the risk premium as well as how far back to reach into history to collect data for calculating an average that is representative of a forward-looking premium.

It is surprising that, as popular as the risk premium method is in public utility rate cases, the intuitively appealing general consumption-based asset pricing model, with its minimal assumptions and strong theoretical foundation, has not been applied to estimate the cost of common equity capital for public utilities. The model provides projections of the conditional expected risk premium on an asset based on its relation to its predicted conditional volatility. This model generalizes the well known special case asset pricing models such as the Merton (1973) intertemporal capital asset pricing model, Campbell (1993) intertemporal asset pricing model, and the habit-persistence model of Campbell and Cochrane (1999), which are special cases of the general model. The relation of the model to their specialized cases can be found in Cochrane (2006) and Cochrane (2007). The approach of consumption asset pricing models is to make investment decisions that maximize investors' utility from the consumption that they ultimately desire, not returns.

Even if the model is not used to project directly the expected risk premium, it can, at a minimum, be used to verify that the risk premia data chosen for estimating the cost of capital is empirically validated by fitting the model well. The model can be used to predict the equity risk premia net of the risk-free rate (equity risk premium) or to predict the equity-to-debt risk premium for a firm. We perform both of these empirical tests in this paper. The general consumption-based asset pricing model developed in Michelfelder and Pilotte (2011) and based on Cochrane (2004) provides the relationship of the ex ante risk premium to an asset's own volatility in return:

$$E_t[R_{i,t+1}] - R_{f,t} = -\frac{vol_t[M_{t+1}]}{E_t[M_{t+1}]}vol_t[R_{i,t+1}]corr_t[M_{t+1}, R_{i,t+1}].$$
 (1)

<sup>&</sup>lt;sup>1</sup> GARCH refers to the generalized autoregressive conditional heteroskedasticity regression model which is discussed below.

where  $vol_t$  is the conditional volatility,  $corr_t$  is the conditional correlation, and  $M_{t+1}$  is the stochastic discount factor (SDF).

The SDF is the intertemporal marginal rate of substitution in consumption, or,  $M_{t+1} = \beta \frac{U_{c,t+1}}{U_{c,t}}$ , where the  $U_c$ 's are the marginal utilities of consumption in the next period, t + 1, and the current period, t, and  $\beta$  is the discount factor for period t to t + 1. Equation 1 shows that the algebraic sign of the relation between the expected risk premium and the conditional volatility of an asset's risk premium is determined by the correlation between the asset's return and the SDF. That is, the direction of the relation between the asset return and the ratio of intertemporal marginal utilities in consumption inversely determines the relation between the expected risk premium and conditional volatility. When the correlation is equal to negative one, the asset's conditional expected risk premium is perfectly positively correlated with its conditional volatility. A positive relation between the conditionally expected risk premium and volatility obtains when  $-1 < corr_t < 0$ . A negative relation obtains when  $0 < corr_t < 1$ . For an asset that represents a perfect hedge against shocks to the marginal utility of consumption, with  $corr_t = 1$ , there will be a perfect negative correlation between the conditionally expected risk premium and its volatility.<sup>2</sup> Therefore, estimates of the relation between the first two conditional moments of a public utility stock's returns provide a direct test of the effectiveness of a public utility stock, or any asset, as a consumption hedging asset. In Eq. 1,  $vol_t[M_{t+1}]/E_t[M_{t+1}]$  is the slope of the meanvariance frontier. If this slope changes over time, the estimated relation between the stock's risk and return will vary over time. This model can also be viewed simplistically as the projected expected risk premium as a function of its own projected risk, given information available at time t.

Note that the model allows for the expected risk premium to be negative if the asset hedges shocks to the marginal utility of consumption. Investors are willing to accept an expected rate of return lower than the risk-free rate of return if the pattern of volatility is such that returns are expected to rise with expected reductions in consumption. Simply, investors are willing to *pay* a premium for a higher level of returns volatility that has the desired pattern of returns. These desired returns patterns have a tendency to offset drops in consumption. Therefore, this model shows that investors may not be averse to volatility, but rather to the timing of expected changes in returns.

Summarizing, several conclusions can be drawn from the general model of asset pricing. First, the sign of the relation between a stock's risk premium and conditional volatility depends on the extent to which the stock serves as an intertemporal hedge against shocks to the marginal utility of consumption. Second, the relation between stock risk and return may be time-varying depending on changes in the slope of the mean-variance frontier. Third, hedging assets have desired patterns of volatility that result in expected rates of return that are less than the risk-free rate. We do not expect

 $<sup>^2</sup>$  A hedging asset is one that has a positive increase in returns that is coincident with a positive shock in the ratio of intertemporal marginal utilities of consumption. Note that if we assume a concave utility function in consumption, as consumption declines, the marginal utility of consumption rises relative to last period marginal utility. If we think of a decline in consumption as a contraction in the business cycle, the hedging asset delivers positive changes in returns when the business cycle is moving into a contraction, and therefore the asset is a business cycle hedge.

that public utility stocks serve as a hedging asset as they are not viewed as defensive stocks (they do not rise in value during downturns in the stock market) due to asymmetric regulation and returns as discussed in detail in Kolbe and Tye (1990). Under asymmetric regulation, utility regulators have a tendency to allow the return on equity to fall below the allowed return during downturns in the business cycle and to reduce the return should it rise above the allowed return during expansions. Therefore we expect that the parameter estimates of the return-risk relationship to be positive as utility stocks are hypothesized to not be hedges.

We use the GARCH model to estimate the general asset pricing model since the GARCH model accommodates ARCH effects that improve the efficiency of the parameter estimates. It also provides a volatility forecasting model for the conditional volatility of the asset's risk premium. The conditional volatility projection is used, in turn to predict the expected risk premium. We also use the GARCH-in-Mean model (GARCH-M) since it specifies that the conditional expected risk premium is a linear function of its conditional volatility. There is a vast body of literature that estimates asset pricing models with the GARCH and GARCH-M methods and therefore we will not attempt to summarize them here.

The GARCH-M model was initially developed and tested by Engle et al. (1987) to estimate the relationship between US Treasury and corporate bond risk premia and their expected volatilities. The GARCH-M model is specified as:

$$R_{t+1} - R_{f,t+1} = \alpha \sigma_{t+1}^2 + \varepsilon_{t+1} \tag{2}$$

$$\sigma_{t+1}^2 = \beta_0 + \beta_1 \sigma_t^2 + \beta_2 \varepsilon_t^2 + \eta_{t+1}$$
(3)

$$\varepsilon_t | \psi_{t-1} \sim T(0, \sigma_t^2) \tag{4}$$

where  $R_{t+1}$  is the expected total return on the public utility stock index or individual utility stock;  $R_{f,t+1}$  is the risk-free rate of return or the yield on an index of public utility bonds of a specified bond rating for the equity-to-debt premium;  $\sigma_{t+1}^2$  is the conditional or predicted variance of the risk premium that is conditioned on past information ( $\psi_{t-1}$ ); and  $\varepsilon_t$  is the error term that is conditional on  $\psi_{t-1}$ .

The conditional distribution of the error term is specified as the non-unitary variance T-distribution due to the thick-tailed distribution of the risk premia data. If the error distribution is thick-tailed, using an approximating distribution that accommodates thick tails improves the efficiency of the estimates. The parameter,  $\alpha$ , is the return-to-risk coefficient as specified in Eq. 1 as:

$$\alpha = -\frac{vol_t[M_{t+1}]}{E_t[M_{t+1}]}corr_t[M_{t+1}, R_{i,t+1}]$$
(5)

Note that the coefficient will be positive if the conditional correlation between the SDF and the asset return is negative, indicating that the stock is not a hedging asset. Recall that the SDF is the ratio of intertemporal marginal utilities. Assuming a concave utility function, an upward shock in the ratio implies falling consumption, therefore an associated rise (positive correlation) in the return ( $R_i$ ) would offset the reduction

in consumption, thereby causing the sign of  $\alpha$  to be negative. The parameter,  $\alpha$ , is also the ratio of risk premium to variance, or, the Sharpe ratio.

The intercept in Eq. 2 is restricted to zero as specified by the general asset pricing model specification. The restriction on the intercept equal to zero has been found to be robust in producing consistently positive and significant relationships between equity risk premia and risk in GARCH-M models. This is discussed in Lanne and Saikkonen (2006) and Lanne and Luoto (2007). We have found the same results in our modeling in this paper, although we have excluded these results for brevity (available upon request). Therefore we specify the prior assumption that the intercept or the "excess" return, i.e., the return not associated with risk to be equal to zero and drop the intercept from the model.

The consumption asset pricing model is estimated in the empirical section of the paper and applied in the applications section of the paper. The model is tested to (1) determine if equity-to-debt risk premium indices for utilities of differing risk specified by differing bond ratings are validated by the asset pricing model and therefore have some empirical support for risk premium prediction and application to utility cost of capital estimation, (2) determine whether equity risk premia can be predicted and fit the model and therefore be used to estimate the cost of common equity, (3) empirically test the consumption asset pricing model, and (4) ascertain whether utility stocks are assets that hedge shocks to the marginal utility of consumption.

If utility stocks are hedging assets then the cost of common equity should reflect a downward adjustment to a specified risk-free rate to reflect investors' preferences for a hedge and the compensation that they are willing to pay for it.

# 3 Data and empirical results

We use portfolios as represented by public utility stock and bond indices to estimate the conditional return-risk relationship for the equity-to-debt premium. The equityto-debt risk premium data employed for estimating Eq. 1 with the GARCH-M conditional return-risk regressions are monthly total returns on the Standard and Poor's Public Utilities Stock Index (utility portfolio), and the monthly Moody's Public Utility Aa, A, and Baa yields for the debt cost. We also obtained equity risk premia for the utility portfolio using the Fama-French specified risk-free rate of return, which is the holding period return on a 1-month US Treasury Bill. The data range from January 1928 to December 2007 with 960 observations. The return-risk relationships for the equity-to-debt premia are risk-differentiated by their own bond rating.

As a check, we also estimate Eq. 1 with the GARCH-M for large common stock returns using the monthly Ibbotson Large Company Common Stocks Portfolio total returns and the Ibbotson US Long-Term Government income returns as the risk-free rate. Additionally, as another check, we do the same for the University of Chicago's Center for Research in Security Prices value-weighted stock index (CRSP) using the Fama-French risk-free rate. This is the Fama-French specification of the market equity risk premium. The data range from January 1926 to December 2007 with 984 observations for the Large Company Common Stock estimation and the data ranges

| ive statistics: public utility and large company common stocks equity-to-debt and equity |
|--|

| Utility bond rating             | Mean   | Std. Dev. | Skewness | Kurtosis | JB         |
|---------------------------------|--------|-----------|----------|----------|------------|
| Aa                              | 0.0037 | 0.0568    | 0.0744   | 10.07    | 2,001.2*** |
| А                               | 0.0035 | 0.0568    | 0.0632   | 10.06    | 1,991.8*** |
| Baa                             | 0.0031 | 0.0568    | 0.0375   | 10.02    | 1,973.6*** |
| Ibbotson                        |        |           |          |          |            |
| Large common stocks             | 0.0054 | 0.0554    | 0.4300   | 12.84    | 3,954.7*** |
| CRSP value-weighted stock index | 0.0062 | 0.0544    | 0.2309   | 10.92    | 2,519.1*** |

The public utility equity-to-debt risk premia monthly time series is from January 1928 to December 2007 with 960 observations. The equity risk premium monthly time series for the Large Common Stocks and the CRSP index are January 1926 to December 2007 with 984 observations, and January 1926 to December 2007 with 984 observations, respectively. The public utility stocks equity-to-debt risk premia are calculated as the total return on the S&P Public Utilities Index of stocks minus the Moody's Public Utility Aa, A, and Baa Indices yields to maturity. The Large Company Common Stocks Portfolio minus the Ibbotson Large Company Common Stocks Portfolio minus the Ibbotson Long-Term US Government Bonds Portfolio income yield. The CRSP equity risk premia, or the Fama-French market risk premia are the CRSP total returns on the value-weighted equity index minus the 1-month holding period return on a 1 month Treasury Bill. The Jarque-Bera (JB) statistic is a goodness-of-fit measure of the departure of the distribution of a data series from normality, based on the levels of skewness and excess kurtosis. The JB statistic is  $\chi^2$  distributed with 2° of freedom. \*\*\* Significant at 0.01 level, one-tailed test

from January 1928 to January 2007 with 960 observations (same as the utilities) for the CRSP estimation.

Table 1 displays the descriptive statistics for these data. We have estimated the mean, standard deviation, skewness and kurtosis parameters, as well as the Jarque-Bera (JB) statistic to test the distribution of the data. The means of the utility equity-to-debt risk premia fall as the risk (bond rating) declines. This is consistent with the notion that larger yields are subtracted from stock returns the lower the bond rating. Intertemporally, there is an inverse relationship between risk premia and interest rates (See Brigham et al. (1985) and Harris et al. (2003)). The mean for risk premia will have a tendency to be larger during low interest rate periods.

Not surprisingly, large company common stocks have the highest mean risk premia as the majority of these firms are not rate-of-return regulated firms with a ceiling on their ROE's close to their cost of capital. Interestingly, the standard deviations of the utility stock returns are similar and slightly higher than large company common stocks. Skewness coefficients are small and positive except for Ibbotson large company common stock returns and CRSP returns that have large positive skewness. This suggests that large unregulated stocks have a tendency to have more and larger positive shocks in returns than do utilities that are rate of return regulated. The kurtosis values show that all of the risk premia are thick-tail distributed. This is also found in the significant JB statistics that test the null hypothesis that the data are normally distributed. The null hypothesis is rejected for all assets. The high kurtosis, low skewness, and significant JB statistics show that the risk premia data are substantially thick-tailed, except for non-utility stocks that are both skewed and thick-tailed. Therefore, robust estimation methods are required to produce efficient regression estimates with non-normal data. Additionally, although not shown but available upon request, the serial correlation and

Table 1 Descript

risk premia

ARCH Lagrange Multiplier tests show that residuals from OLS regressions of risk premia on volatilities follow an ARCH process. Therefore, the GARCH-M method will improve the efficiency of the estimates. We specify the regression error distribution as a non-unitary variance T-distribution so that thick-tails could be accommodated in the estimation and therefore produce increasingly efficient parameter estimates.

We used maximum likelihood estimation with the likelihood function specified with the non-unitary-variance T-distribution as the approximating distribution of the residuals to accommodate the thick-tailed nature of the error distribution. The equations are estimated as a system using the Marquardt iterative optimization algorithm. The chosen software for estimating the model was EViews<sup>©</sup> version 6.0 (2007).

Table 2 shows the GARCH-M estimations for the consumption asset pricing Eq. 1. We have estimated Eq. 1 for the utility equity risk premia using the Fama-French risk-free rate in addition to the equity-to-debt risk premia risk-differentiated by bond ratings and the two measures of the market equity risk premium. The chosen measure of volatility is the variance of risk premium (in contrast to other such measures such as the standard deviation or the log of variance. Although these results are not shown for brevity, they are robust to these other measures of volatility). The slope, which is the predicted return-to-predicted risk coefficient and Sharpe ratio, is positive and significant at the 99% level for all assets except the utility stock returns with Baa bonds, which is significant at the 95% level. Given that all slopes are positive, public utility stocks are not found to hedge shocks to the marginal utility of consumption. Note that the reward-to-risk slope rises as bond rating rises. This suggests that lower risk utility stocks provide a higher incremental risk-premium for an increase in conditional volatility. This is consistent with other studies that find that lower risk assets, such as shorter maturity bonds, have higher Sharpe Ratios than longterm bonds and stocks. See Pilotte and Sterbenz (2006) and Michelfelder and Pilotte (2011).

The variance equation shows that all GARCH coefficients ( $\beta$ 's) are significant at the 1% level and the sums of  $\beta_1$  and  $\beta_2$  are close to, but less than 1.0, indicating that the residuals of the risk premium equation follow a GARCH process and that the persistence of a volatility shock on returns and stock prices for utility stocks is temporary. The estimates of the non-unitary variance T-distribution degrees of freedom parameter are low and statistically significant, indicating that the residuals are well approximated by the T. Similar values for the log-likelihood functions (Log-L) show that each of the regressions has a similar goodness-of-fit. Chi-squared distributed likelihood ratio tests (not shown but available upon request) that compare the goodness of fit among the T and normal specifications of the likelihood function of the GARCH-M regressions show that the T has a significantly better fit than the normal distribution.

The GARCH-M results for the large company common stocks portfolio are similar to those of the utility stocks. Not surprisingly, large company common stocks do not hedge shocks to the marginal utility of consumption and volatility shocks temporarily affect their valuations. The exception is that the return-risk slope is substantially higher than utility stock slopes. This is partially due to the risk-free nature of the risk-free rates used with the non-utility equity risk premia compared to the

| Utility bond rating                   | α                     | $\beta_0$             | $\beta_1$             | $\beta_2$             | Log-L   | T dist. D.F.          |
|---------------------------------------|-----------------------|-----------------------|-----------------------|-----------------------|---------|-----------------------|
| Aa                                    | 1.5183***<br>(0.5308) | 0.0000**<br>(0.0000)  | 0.8791***<br>(0.0230) | 0.1031***<br>(0.0219) | 1,604.4 | 9.9254***<br>(3.0272) |
| А                                     | 1.4536***<br>(0.5308) | 0.0000**<br>(0.0000)  | 0.8790***<br>(0.0230) | 0.1033***<br>(0.0220) | 1,605.0 | 9.9381***<br>(3.0408) |
| Baa                                   | 1.3318**<br>(0.5303)  | 0.0000**<br>(0.0000)  | 0.8789***<br>(0.0229) | 0.1040***<br>(0.0220) | 1,605.2 | 10.0***<br>(3.0540)   |
| Fama-French $R_f$                     | 2.1428***<br>(0.5318) | 0.0000**<br>(0.0000)  | 0.8811***<br>(0.0232) | 0.0979***<br>(0.0212) | 1,601.0 | 9.8773***<br>(2.9700) |
| Ibbotson                              |                       |                       |                       |                       |         |                       |
| Large company<br>common<br>stocks     | 2.7753***<br>(0.5513) | 0.0001***<br>(0.0000) | 0.8381***<br>(0.0269) | 0.1186***<br>(0.0332) | 1,620.8 | 8.8457***<br>(2.1613) |
| CRSP<br>value-weighted<br>stock index | 3.3873***<br>(0.5673) | 0.0001***<br>(0.0000) | 0.8330***<br>(0.0270) | 0.1149***<br>(0.0358) | 1,598.9 | 8.8571***<br>(1.9505) |

Table 2 Estimation of return-risk relation: public utility and large company common stocks

The results below are the GARCH-in-Mean regressions for the risk premium  $(R_{t+1} - R_{f,t+1})$  on the conditional variance of the risk premium  $(\sigma_{t+1}^2)$  in the mean equation. The intercept in the mean equation is restricted to be equal to zero. The public utility equity-to-debt risk premia monthly time series is from January 1928 to December 2007 with 960 observations. The equity risk premium monthly time series for the Large Company Common Stocks and the CRSP index are January 1926 to December 2007 with 984 observations, and January 1926 to December 2007 with 984 observations, respectively. The public utility stocks equity-to-debt risk premia are calculated as the total return on the S&P Public Utilities Index of stocks minus the Moody's Public Utility Aa, A, and Baa Indices yields to maturity. The Large Company Common Stocks Portfolio minus the Ibbotson Long-Term US Government Bonds Portfolio income yield. The CRSP equity risk premia, or the Fama-French market risk premia are the CRSP total returns on the value-weighted equity index minus the 1-month holding period return on a 1 month Treasury Bill. The estimated model is:

$$R_{t+1} - R_{f,t+1} = \alpha \sigma_{t+1}^2 + \varepsilon_{t+1} \text{ where } \alpha = -\frac{vol_t[M_{t+1}]}{E_t[M_{t+1}]} corr_t[M_{t+1}, R_{i,t+1}]$$

 $\sigma_{t+1}^2 = \beta_0 + \beta_1 \sigma_t^2 + \beta_2 \varepsilon_t^2 + \eta_{t+1}$ 

The conditional distribution of the error term is the non-unitary variance T-distribution to accommodate the kurtosis of the risk premia and error term. Standard errors are in parentheses. \*\*\*, \*\*, \* denote significance at the 0.01, 0.05, and 0.10 levels, respectively for two-tail tests

utility bond yields that reflect risk. The utility stocks slope value of 2.1428 using the Fama-French risk-free rate is closer to the higher CRSP value of 3.3873 that is also based on the Fama-French risk-free rate. This is inconsistent with previous results herein and in other papers that find that Sharpe Ratios are lower for higher risk assets unless this finding can be interpreted as utility stocks having more risk than non-regulated stocks. The standard deviations on Table 1 suggest that utility stock return volatilities are as high as the stock returns of non-regulated firms. However, similar model estimates of portfolios of common stocks yield unstable results, such as negative as well as positive return-risk slopes when the intercept is not restricted to zero. See Campbell (1987), Glosten et al. (1993), Harvey (2001), and Whitelaw (1994). Stock market results are highly sensitive to empirical model specification. Many studies do not consider the impact of a zero-intercept prior restriction on the stability of their results. This simple innovation has led to more consistent results in modeling stock market risk-return relationships, and therefore we have included it in this paper.

The estimation of the consumption asset pricing model for utility stock equitydebt risk premia shows that the use of bond-rating risk-differentiated risk premia are validated as their risk-return relationships are well-fitted by theoretical and empirical models of risk and return. Therefore, these data impound good representations of the risk and reward relationship.

One concern is the intertemporal stability of the alphas. Figure 1 plots the utility stock portfolio alpha (using the Fama-French  $R_f$  to calculate the premium) and its standard error for 240 month rolling regressions of the model estimated with GARCH-M in the same manner as described above to review the intertemporal stability of the alpha. A 20-year period was used for each estimation to trade off timeliness with sufficient observation of up and down stock market regimes and business cycles. This resulted in 720 estimated alphas from 1947 to 2007. The results show that the utility alpha is stable to the extent that the algebraic sign is always positive and generally significant, therefore the nature of utility stocks are assets that are not and have never been hedges during the second half of the twentieth century up to the present. The value of the alpha does change substantially. The mean of the alpha is 4.40 with a range from -0.11 (insignificantly different from 0) to 11.66. As a comparison, the alpha for the CRSP value-weighted stock index was also estimated with rolling regressions in the same manner and for the same time period. Figure 2 is a plot of the CRSP alpha and standard error. Note that the general stock market alpha is similar to that of utility stocks. They are all positive and almost all statistically significant and follow a strikingly similar cycle. Figure 3 plots both the utility and stock market alphas and demonstrates the similarity. The correlation coefficient between the utility and stock market alphas is 0.88. Recalling that the alpha is a Sharpe Ratio, we see that return to risk ratio does change substantially. This is consistent with the results in Pilotte and Sterbenz (2006).

One other interesting observation is that the standard errors of the alphas are highly stable over the study period and are very similar in magnitude regardless of the size of the corresponding alpha. Whereas the alpha follows a cyclical pattern, the volatility in alpha is highly stationary around a constant, long-run mean.

The GARCH-M model estimations of the consumption asset pricing model were specified with variance as the measure of volatility. We also performed the same model estimations with alternative specifications of volatility such as the standard deviation and the log of variance and the results were not sensitive to this specification.

# **4** Application

We apply the model in this section to compare the cost of common equity capital estimates with the DCF and CAPM models. Using EViews<sup>©</sup> Version 6.0, we estimated the model coefficients ( $\alpha$ ,  $\beta's$ ) over rolling 24 month periods ending December 2008.



Fig. 1 Rolling 240 month utility stock alphas 1947–2007



Fig. 2 Rolling 240 month CRSP value-weighted alphas 1947–2007

We repeated the estimation over 5, 10, 15, 20 and 79 year periods.<sup>3</sup> Predicted monthly variances ( $\sigma_{t+1}^2$ ) were generated from these estimations to produce predicted risk premiums that were calculated by multiplying the predicted variance by the " $\alpha$ " slope

 $<sup>\</sup>frac{3}{3}$  We did not include the results of the 10 and 15 year estimations to abbreviate the amount of empirical results presented since they added no material insights beyond those already presented.


Fig. 3 Rolling 240 month CRSP and utility alphas 1947–2007

Table 3 Estimates of expected risk premia

|                          | Mean (% | )      | Range (%)    |               | Standard | deviation (%) |
|--------------------------|---------|--------|--------------|---------------|----------|---------------|
|                          | Average | Spot   | Average      | Spot          | Average  | Spot          |
| Ibbotson Associates data |         |        |              |               |          |               |
| 79-years                 | 9.59    | 5.76   | 8.74–9.96    | 2.62-22.60    | 0.32     | 5.24          |
| 20-years                 | 6.77    | 6.94   | 4.99-8.50    | 2.24-28.95    | 0.95     | 6.88          |
| 5-years                  | 4.20    | 10.25  | -98.49-11.62 | -100.00-39.65 | 22.00    | 26.61         |
| S&P Utility Index        |         |        |              |               |          |               |
| 79-years                 | 5.28    | 2.90   | 4.30-5.28    | 1.65-8.15     | 0.32     | 1.60          |
| 20-years                 | 3.93    | 3.51   | 2.78-5.03    | 2.18-6.88     | 0.57     | 1.11          |
| 5-years                  | 31.82   | 326.63 | 7.77–156.97  | 6.12-6465.74  | 31.47    | 1283.51       |

coefficient. To test the stability of the predicted risk premia over time, the predicted risk premia were calculated using either the predicted variance over each entire time period or the last monthly (spot) predicted variance. Table 3 presents the mean predicted risk premia, the range of predicted premia and the standard deviations for each time period. It is clear from the results that the risk premia are more stable over the rolling 24 month period when calculated using the average predicted variance compared with using the spot variance. Secondly, the 20 and 79 year means are substantially more stable and reasonable in magnitude than the 5 year means.

Next, given the lessons from the analyses above, we apply the model to mechanically<sup>4</sup> estimate the cost of common equity for 8 utility companies using the model and

<sup>&</sup>lt;sup>4</sup> The term "mechanically" in this context means that the resulting values have been developed in a consistent manner with the same inputs across all utility stocks but no subjective judgment was used to develop final values for each specific utility stock application.

the DCF and CAPM as comparisons. We also calculated the realized market return for comparison. Two publicly-traded electric, electric and gas combination, gas, and water utilities respectively were chosen for the application. The Gordon (1974) DCF and CAPM models are used in many utility regulatory jurisdictions in the US.

The DCF was applied using a dividend yield,  $D_0/P_0$ , derived by dividing the yearend indicated dividend per share  $(D_0)$  by the year-end spot market price  $(P_0)$ . The dividend yield is grown by the year-end I/B/E/S five year projected earnings per share growth rate (g) to derive  $D_0(1+g)/P_0$ . The one-year predicted dividend yield is then added to the I/B/E/S five-year projected EPS growth rate to obtain the DCF estimate of the cost of common equity capital, k. This study was conducted for the 5 years ending 2008.

The CAPM was applied by multiplying the Value Line beta ( $\beta$ ) available at yearend for each company by the long-term historic arithmetic mean market risk premium  $(R_m - R_f)$ .  $R_m - R_f$  is derived as the spread of the total return of large company common stocks over the income return on long-term government bonds from the Ibbotson SBBI 2009 Valuation Yearbook. The resulting company-specific market equity risk premium is then added to a projected consensus estimate of the yield on 30-year U.S. Treasury rate provided by Blue Chip Financial Forecasts as the risk-free rate ( $R_f$ ) to obtain the CAPM result. This study was also conducted over the 5 years ending 2008.

Figures 4–11 show the histograms of the cost of common equity capital estimations for each of the eight public utility stocks and the realized market returns in the forthcoming year. The consumption asset pricing model appears to track more consistently with the CAPM than with the DCF which seems to produce generally lower values than the other methods. The consumption asset pricing model results are similar to the CAPM. The model and the CAPM compete as the best predictor of the rate of return on the book value of common equity (not shown but available upon request), but none of the expected returns were good predictors of market returns. That does not infer that they were not good predictors of expected market returns. These results are an initial indicator that the consumption asset pricing model provides reasonable and stable results. This paper does not suggest at this early juncture that the consumption asset pricing model is superior to the CAPM or DCF, although it is based on far less restrictive assumptions than these other models. For example, both the DCF and CAPM assume that markets are efficient. Many assume that the DCF requires that the market-to-book ratio to always equal one, whereas the long-term value for the Standard and Poor's 500 is equal to 2.34. The CAPM assumes that investors demand higher returns for higher volatility and that the minimum required return is the risk-free rate, whereas the consumption asset pricing model allows for investors to require returns less than the risk-free rate for stocks that may have relatively higher volatility but are hedging assets that have desirable return fluctuation patterns that offset downturns in the business cycle. Unlike the CAPM, the model prices the risk to which investors are actually exposed, whether it's systematic risk or not. Some investors are diversified and some are not; the model prices whatever risk to which the aggregate of investors of the specific stock is exposed.

We find that the consumption asset pricing model should be used in combination with other cost of common equity pricing models as additional information in the devel-



\* Market returns calculated for the following years: 2005 -2009

Cost of Common Equity Results for Southern Company Compared to Market Return\*



PRPM CAPM DCF Actual

\* Market returnscalculated for the following years: 2005 -2009

Cost of Common Equity Results for Consolidated Edison Compared to Market Return\*

PRPM CAPM DCF Actual







Figs. 4-11 Comparison of the cost of common equity estimates and market

opment of a cost of common equity capital recommendation. Practitioners may find the modeling methods and the use of relatively advanced econometric methods rather cumbersome. The software for performing these estimations is readily available from  $EViews^{\textcircled{O}}$  and  $SAS^{\textcircled{O}}$ ; two commonly available software packages at utilities, consult-



Cost of Common Equity Results for National Fuel Gas Co. Compared to

eturniscaiculated for the following years: 2005 -2009





Missing DCF Cost of Capital Estimates Due to Unavailable Growth Rate



PRPM CAPM DCF Actual







Missing DCF Cost of Capital Estimate Due to Unavailable Growth Rate

Figs. 4-11 continued

ing firms and financial firms. Recent Ph.D. and M.S. holding members of research departments of investment and consulting firms have ready access to the model and methods discussed in this paper, although it will require years for these tools, like any "new" technology, to diffuse into standard use. Another problem is that the model requires a substantial time series history on stock returns data to develop stable estimates of risk premia This is problematic especially for the electric and gas utility industries that have consolidated with many mergers in the recent past. This problem can be addressed by developing and predicting the value-weighted risk premium of a portfolio of similar stocks such as electric utilities that have nuclear generating assets. The specific stock in question would be included in the returns index with a weight based on market capitalization that would go to 0 when the stock price history is no longer existent reaching back into the past.

### **5** Conclusion

The purpose of this paper is to introduce, test empirically and apply a general consumption based asset pricing model that is based on a minimum of assumptions and restrictions that can be used to predict the risk premium to be applied in estimating the cost of common equity for public utilities in regulatory proceedings. The results support the simple consumption-based asset pricing model that predicts the ex ante risk premium with a conditionally predicted volatility in risk premium. The estimates of the cost of common equity from the consumption asset pricing model compare well with rates of return on the book value of common equity and with the CAPM, although both the model and the CAPM results are substantially higher than the DCF. This is quite common in the practice of the cost of common equity in the utility industry. The results of the model are stable and consistent over time. Therefore the model should be considered as it provides additional evidence on the cost of common equity in general and specifically in public utility regulatory proceedings. Secondly, the use of bondrated yields to predict risk differentiated equity-to-debt risk premia is supported by the empirical evidence and therefore should be applied in estimating the cost of common equity. Finally, the robust empirical evidence on the positive risk-return relationship also shows that utility stocks are not a consumption hedge and are not good hedging securities against contractions in the economy. The model and estimation methodology presented in this paper provide a relatively simple tool to determine whether any asset is a hedge to adverse changes in the business cycle through the level of consumption in the economy.

**Acknowledgments** We would like to thank Dylan D'Ascendis, Sal Giunta, Selby Jones, III and Alison McVicker for highly capable research assistance, participants at the Center for Research in Regulated Industries Eastern Conferences and the Society of Utility Regulatory and Financial Analysts Annual Financial Forum, two anonymous reviewers and the editor for helpful comments.

### References

Brigham, E. F., Shome, D. K., & Vinson, S. R. (1985). The risk premium approach to measuring a utility's cost of capital. *Financial Management*, 14, 33–45.

Campbell, J. Y. (1987). Stock returns and the term structure. Journal of Financial Economics, 18, 373–399.

- Campbell, J. Y. (1993). Intertemporal asset pricing without consumption data. American Economic Review, 83, 487–512.
- Campbell, J. Y., & Cochrane, J. H. (1999). By force of habit: A consumption-based explanation of aggregate stock market behavior. *Journal of Political Economy*, 107, 205–251.
- Cochrane, J. H. (2004). Asset pricing. Revised Edition. Princeton, NJ: Princeton University Press.
- Cochrane, J. H. (2006). Financial markets and the real economy. NBER Working Paper.
- Cochrane, J. H. (2007). Portfolio theory. Manuscript. University of Chicago.
- Engle, R. F., Lilein, D., & Robins, R. (1987). Estimation of time varying risk premia in the term structure: The ARCH-M model. *Econometrica*, 55, 391–407.
- EViews<sup>©</sup>. (2007). Version 6.0. Quantitative Micro Software, LLC.
- Fama, E., & French, K. (2004). The capital asset pricing model: Theory and evidence. Journal of Economic Perspectives, 18, 25–46.
- Glosten, L. R., Jaganathan, R., & Runkle, D. E. (1993). Relationship between the expected value and the volatility of the nominal excess returns on stocks. *Journal of Finance*, 48, 1779–1801.
- Gordon, M. (1974). The cost of capital to a public utility. East Lansing, MI: MSU Public Utility Studies.
- Harris, R. S., Marston, F. C., Mishra, D. R., & O'Brien, T. J. (2003). Ex ante cost of equity estimate of S&P 500 firms: The choice between global and domestic CAPM. *Financial Management*, 32, 51–66.
- Harvey, C. R. (2001). The specification of conditional expectations. *Journal of Empirical Finance*, 8, 573– 637.
- Kolbe, A. L., & Tye, W. B. (1990). The *Duquense* opinion: How much "Hope" is there for investors in regulated firms. *Yale Journal on Regulation*, 8, 113–157.
- Lanne, M., & Luoto, J. (2007). Robustness of risk-return relationship in the U.S. stock market. Helsinki Center of Economic Research, Discussion Paper No. 168.
- Lanne, M., & Saikkonen, P. (2006). Why is it so difficult to uncover the risk-return tradeoff in stock returns? *Economic Letters*, 92, 118–125.
- Merton, R. C. (1973). An intertemporal capital asset pricing model. Econometrica, 41, 867-887.
- Michelfelder, R. A., & Pilotte, E. A. (2011). Treasury bond risk and return, the implications for the hedging of consumption and lessons for asset pricing. *Journal of Economics and Business* (forthcoming).
- Pilotte, E., & Sterbenz, F. (2006).Sharpe and treynor ratios on treasury bonds. *Journal of Business*, 79, 149–180.
- Whitelaw, R. W. (1994). Time-variation and covariations in the expectation and volatility of stock market returns. *Journal of Finance*, 49, 515–541.

### <u>Missouri-American Water Company</u> Correction of MIEC Witness Gorman's Constant Growth DCF Model to Reflect the Exclusion of Middlesex Water Company's DCF Results <u>Due to its Negative EPS Growth Rate Forecast</u>

|  |                 | <u>1</u>   | <u>2</u>  |              | <u>3</u>   | <u>4</u>   | <u>5</u>   |
|--|-----------------|--|---|--------------|--|--|--|
| MIEC Witness Gorman's Proxy Group<br>of Eight Water Companies  | 13<br>Aver<br>P | - Week<br>age Stock<br>rice (1)                                      | Analysts'<br>Growth (1)   | Ann<br>Divic | ualized<br>dend (1)  | Adjusted<br>Dividend<br>Yield (1)                              | Indicated<br>Common<br>Equity Cost<br>Rate (2)                         |
| American States Water Co.<br>American Water Works Co., Inc.<br>Aqua America, Inc.<br>California Water Service Group<br>Connecticut Water Service, Inc.<br>Middlesex Water Company<br>SJW Corporation<br>York Water Company | \$              | 33.95<br>29.04<br>21.38<br>17.93<br>26.17<br>17.74<br>22.57<br>16.87 | 9.58 %<br>9.55<br>7.78<br>8.00<br>8.00<br>(5.00)<br>14.00<br>6.00 | \$           | 1.12<br>0.92<br>0.62<br>0.62<br>0.95<br>0.73<br>0.69<br>0.52 | 3.61 %<br>3.47<br>3.13<br>3.71<br>3.93<br>3.92<br>3.49<br>3.29 | 13.19 %<br>13.02<br>10.91<br>11.71<br>11.93<br>NA (3)<br>17.49<br>9.29 |
| Average  |                 |  |   |              |  |  | <u>12.51</u> %   |
| Median   |                 |  |   |              |  |  | <u>11.93</u> %   |

NA= Not Available NMF = Not Meaningful Figure

Notes:

(1) From Schedule MPG-5.

(2) Column 2 + column 4.

(3) Middlesex's DCF results are not applicable due to its negative EPS growth forecast.

Source of Information: Schedule MPG-5

| Industry Title  | 2001   | 2002   | 2003  | 2004   | 2005  | 2006   | 2007  | 2008  | 2009  | 2010                                       | Change<br>2009 - 2010              | Change<br>2001 - 2010            |
|---|--|--|---|--|---|--|---|---|---|--|------------------------------------|----------------------------------|
| varue adventury (minitor) or upplication<br>Private industries<br>Agriculture, forestry, fishing, and hunting<br>Ferms  | 10,286,167<br>9,010,772<br>98,554<br>76,214  | 10,642,316<br>9,289,346<br>94,405<br>72.304  | 11,142,143<br>9,706,821<br>115,556<br>92,412  | 11,867,800<br>10,360,100<br>142,700<br>118,300   | 12,638,400<br>11,052,500<br>127,100<br>102,000  | 13,398,900<br>11,731,100<br>122,500<br>93,100  | 14,061,800<br>12,301,900<br>144,700<br>114,900  | 14,369,100<br>12,514,000<br>160,100<br>131,100  | 14,119,000<br>12,196,500<br>133,100<br>104,000  | 14,660,400<br>12,696,500<br>154,100        | 3.83%<br>4.10%<br>15.78%           | 4.73%<br>4.54%<br>6.26%          |
| Forestry, fishing, and related activities<br>Mining<br>Oil and gas extraction<br>Mining, except oil and gas   | 22,340<br>119,296<br>72,914<br>27,567  | 22,101<br>109,462<br>66,598<br>27,552  | 23,144<br>134,761<br>89,248<br>27,988   | 24,500<br>159,300<br>106,600<br>31,000   | 25,100<br>192,000<br>128,600<br>36,300  | 29,400<br>229,000<br>147,800<br>40,700   | 29,800<br>254,200<br>162,900<br>41,900  | 29,000<br>317,100<br>210,800<br>47,900  | 29,200<br>240,800<br>141,700<br>48,900  | 281,400                                    | 16.86%                             | 15.10%                           |
| Support activities for mining<br>Utilities<br>Construction<br>Manufacturing<br>Durable goods<br>Durable goods<br>Nood products<br>Noometallic mineral products  | 18,815<br>177,579<br>490,525<br>1,343,930<br>758,754<br>27,541<br>40,714   | 15,312<br>180,973<br>494,328<br>1,355,537<br>767,751<br>27,277<br>41,446   | 17,525<br>191,852<br>515,929<br>1,374,012<br>766,154<br>28,277<br>40,479  | 21,700<br>208,000<br>554,400<br>1,482,700<br>822,000<br>31,700<br>42,700                           | 27,200<br>205,700<br>611,700<br>1,568,000<br>877,600<br>33,000<br>45,300                  | 40,500<br>236,200<br>651,100<br>1,651,500<br>923,100<br>30,200<br>45,400                     | 49,400<br>248,800<br>657,200<br>1,698,900<br>942,800<br>28,200<br>44,400                    | 58,400<br>262,600<br>623,400<br>1,647,600<br>927,300<br>25,100<br>39,400                      | 50,200<br>268,100<br>537,500<br>1,584,800<br>867,200<br>20,900<br>38,200                      | 275,700<br>505,600<br>1,717,500<br>961,200 | 2.83%<br>-5.93%<br>8.37%<br>10.84% | 6.14%<br>0.34%<br>3.09%<br>2.96% |
| Finnary metals<br>Fabricated metal products<br>Machinery<br>Computer and leatronic products<br>Electrical equipment, appliances, and components<br>Motor vehicles, bodies and trailers, and parts<br>Other transportation equipment   | 39,551<br>39,551<br>103,119<br>128,239<br>42,970<br>107,684<br>69,575<br>31745   | 41,352<br>105,210<br>95,897<br>132,573<br>42,568<br>121,309<br>67,349<br>67,349  | 37,811<br>104,332<br>92,849<br>136,948<br>42,465<br>125,550<br>65,092<br>30,334   | 52,200<br>112,700<br>159,600<br>38,900<br>617,600<br>67,200  | 53,700<br>120,400<br>183,300<br>39,900<br>76,000<br>76,000                                | 99,700<br>125,600<br>200,000<br>45,600<br>81,500<br>81,500                                   | 59,000<br>134,300<br>126,300<br>45,800<br>92,400<br>34700                                   | 61,500<br>135,100<br>204,100<br>81,100<br>95,700<br>95,700                                    | 43,400<br>121,900<br>112,700<br>505,400<br>51,700<br>78,200<br>90,700                         |  |                                    |                                  |
| Miscellaneous manufacturing<br>Nondurable goods<br>Food and beverage and tobacco products<br>Tevidie mills and lexille product mills<br>Apparel and leather and allied products<br>Paper products<br>Printing and related support activities  | 55,583<br>585,176<br>173,463<br>25,282<br>18,953<br>53,722<br>38,890   | 61,489<br>587,489<br>176,942<br>18,094<br>18,269<br>53,531<br>37,437   | 607,858<br>607,858<br>174,005<br>19,410<br>16,414<br>49,844<br>36,753   | 67,900<br>660,600<br>168,900<br>26,800<br>16,400<br>54,200<br>37,000                               | 690,400<br>690,400<br>172,100<br>23,500<br>16,000<br>53,800<br>37,500                     | 728,400<br>181,400<br>151,200<br>15,300<br>59,300  | 78,800<br>756,100<br>179,900<br>21,700<br>14,900<br>58,600<br>38,500                        | 22,500<br>720,600<br>181,200<br>13,500<br>53,800<br>37,000                                    | 79,000<br>717,600<br>206,100<br>17,600<br>11,700<br>56,100<br>32,800                          | 756,300                                    | 5.39%                              | 3.25%                            |
| Petroleum and coal products<br>Chemical products<br>Plastics and rubber products<br>Wholesale trade<br>Retail trade<br>Ari transportation<br>Ari transportation<br>Real transportation<br>Truck transportation  | 59,698<br>153,163<br>62,005<br>613,005<br>613,004<br>613,004<br>703,893<br>302,591<br>25,195<br>25,576<br>8,2576<br>8,2577<br>8,2778<br>8,2778<br>8,2778<br>8,2778<br>8,2778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27778<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27788<br>8,27778<br>8,27788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,27778778<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,277788<br>8,2777787788<br>8,2777877787787777877 | 44,927<br>169,567<br>63,018<br>63,018<br>63,018<br>63,018<br>731,156<br>731,156<br>302,405<br>24,796<br>21,740<br>7,400<br>7,400<br>97,196 | 77,499<br>171,402<br>62,532<br>638,065<br>638,065<br>768,924<br>768,924<br>319,754<br>53,231<br>53,231<br>8,336<br>8,336<br>101,997 | 106,300<br>186,800<br>64,500<br>64,500<br>794,700<br>347,000<br>56,100<br>56,100<br>8,700<br>8,700 | 139,300<br>182,700<br>65,600<br>838,800<br>369,700<br>369,700<br>55,700<br>9,300<br>9,300 | 140,000<br>207,900<br>65,100<br>769,600<br>875,000<br>395,500<br>30,600<br>12,400<br>125,300 | 149,700<br>223,200<br>69,500<br>813,300<br>886,100<br>405,400<br>60,200<br>31,700<br>13,500 | 151,900<br>59,400<br>822,900<br>840,200<br>61,8,700<br>61,800<br>34,800<br>124,700            | 120,000<br>56,700<br>56,700<br>780,800<br>819,600<br>619,600<br>619,600<br>613,000<br>113,100 | 807,700<br>862,800<br>406,500              | 3.45%<br>5.27%<br>4.36%            | 3.52%<br>2.51%<br>3.82%          |
| Transit and ground passenger transportation<br>Pipeline transportation<br>Other transportation and support activities<br>Warehousing and storage<br>Information<br>Publishing industries (includes software)<br>Motion picture and sound recording industries   | 18,447<br>14,700<br>70,335<br>25,303<br>451,078<br>100,072<br>47,504   | 19,198<br>10,340<br>73,383<br>26,343<br>499,693<br>51,613<br>51,613  | 19,641<br>10,227<br>75,986<br>27,214<br>508,606<br>127,979<br>48,164  | 20,900<br>11,500<br>83,900<br>31,400<br>564,100<br>140,800<br>56,300                               | 21,200<br>10,400<br>35,300<br>592,600<br>151,200<br>56,300                                | 22,600<br>96,400<br>37,200<br>133,900<br>133,900   | 24,000<br>12,400<br>96,500<br>40,000<br>633,300<br>151,200<br>151,200                       | 23,400<br>16,200<br>41,700<br>652,500<br>61,000<br>61,000                                     | 23,400<br>12,000<br>94,700<br>39,400<br>639,300<br>147,700                                    | 670,300                                    | 4.85%                              | 5.40%                            |
| broadsung and relecommunications<br>Information and data processing services<br>Finance, insurance, real estate, rental, and leasing<br>Finance and insurance<br>Federal Reserve banks, credit intermediation, and related activities<br>Restricts, commodity contracts, and investments<br>Securities, commodity contracts, and investments<br>Insurance carriers and related activities | 261,285<br>36,215<br>36,215<br>376,573<br>376,573<br>178,174<br>260,391  | 267,916<br>58,393<br>2,222,273<br>863,537<br>436,009<br>153,512<br>248,565   | 208,031<br>63,831<br>2,316,138<br>903,149<br>461,626<br>139,840<br>272,985  | 289,000<br>78,000<br>2,409,700<br>929,200<br>433,300<br>144,000<br>319,600                         | 311,400<br>73,600<br>2,606,500<br>1,028,500<br>470,700<br>183,000<br>337,500              | 317,500<br>82,100<br>2,777,600<br>1,105,500<br>483,500<br>214,500<br>367,400                 | 347,700<br>71,700<br>2,891,300<br>1,110,400<br>476,900<br>199,700<br>392,400                | 359,100<br>77,000<br>2,974,900<br>514,300<br>188,900<br>350,900                               | 355,800<br>76,000<br>3,040,300<br>1,171,600<br>514,000<br>175,200                             | 3,093,700<br>1,235,200                     | 1.76%<br>5.43%                     | 4.84%<br>5.25%                   |
| Funds, trusts, and other financial vehicles<br>Real estate and rental and leasing<br>Real estate<br>Real and leasing services and lessons of intangible assets<br>Professional and business services<br>Professional, scientific, and lechnical services<br>Legal services<br>Computer vesterms design and related services   | 23,540<br>1,316,113<br>1,163,925<br>1,152,188<br>1,170,671<br>701,126<br>147,957<br>116,245  | 25,451<br>1,358,736<br>1,210,895<br>147,841<br>1,198,309<br>718,942<br>153,874<br>109,835  | 28,697<br>1,412,989<br>1,267,747<br>145,242<br>1,259,374<br>745,479<br>163,531<br>109,653   | 32,400<br>1,480,400<br>1,337,600<br>1,246,800<br>810,500<br>180,700<br>117,700                     | 37,300<br>1,577,900<br>1,424,900<br>153,100<br>1,461,800<br>875,600<br>194,500<br>129,300 | 40,200<br>1,672,100<br>1,488,600<br>183,400<br>1,571,400<br>952,200<br>201,900<br>144,300    | 41,500<br>1,780,800<br>1,595,100<br>185,700<br>1,700,500<br>1,028,700<br>215,300<br>160,900 | 46,300<br>1,874,500<br>1,688,900<br>1,768,800<br>1,768,800<br>1,093,600<br>225,600<br>170,600 | 57,800<br>1,868,700<br>1,686,500<br>182,100<br>1,701,300<br>1,068,500<br>219,200<br>169,600   | 1,858,500<br>1,771,900<br>1,103,900        | -0.55%<br>4.15%<br>3.31%           | 4.58%<br>5.71%<br>6.38%          |
| Miscellaneous professional: scientific, and technical services<br>Management of companies and enterprises<br>Administrative and waste management services<br>Administrative and support services<br>Waste management and termediation services  | 436,924<br>174,551<br>294,995<br>266,163<br>28,831   | 455,234<br>178,324<br>301,044<br>270,403<br>30,641   | 472,295<br>192,219<br>321,676<br>287,870<br>33,806  | 512,100<br>203,100<br>333,200<br>299,000<br>34,300   | 551,800<br>217,700<br>368,500<br>331,300<br>37,200  | 606,000<br>234,300<br>385,000<br>350,400<br>34,600   | 652,500<br>257,300<br>414,500<br>375,700<br>38,800  | 697,400<br>260,100<br>374,100<br>40,900   | 679,700<br>246,500<br>386,300<br>347,600<br>38,700  | 256,300<br>411,800                         | 3.98%<br>6.60%                     | 5.20%<br>4.40%                   |

| industry Title  | 2001      | 2002      | 2003      | 2004      | 2005      | 2006      | 2007      | 2008      | 2009      | 2010       | Change<br>2009 - 2010 | Change<br>2001 - 2010 |
|---|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|------------|-----------------------|-----------------------|
| value added (millions of dollars)<br>Educational services, health care, and social assistance | 729,193   | 789,798   | 847,263   | 906,400   | 953,400   | 1,015,200 | 1,078,300 | 1,148,900 | 1,212,900 | 1,274,400  | 5.07%                 | 8.31%                 |
| Educational services  | 88,692    | 98,855    | 106,063   | 116,000   | 120,100   | 128,700   | 137,300   | 147,000   | 154,900   | 162,600    | 4.97%                 | 9.26%                 |
| Health care and social assistance   | 640,500   | 690,943   | 741,200   | 790,400   | 833,300   | 886,500   | 941,000   | 1,001,900 | 1,057,900 | 1,111,700  | 5.09%                 | 8.17%                 |
| Ambulatory health care services   | 313,504   | 335,271   | 357,840   | 381,400   | 406,100   | 432,300   | 458,200   | 486,100   | 514,200   |            |                       |                       |
| Hospitals and nursing and residential care facilities   | 272,008   | 296,068   | 318,996   | 341,200   | 354,400   | 377,700   | 400,000   | 428,800   | 452,700   |            |                       |                       |
| Social assistance   | 54,989    | 59,605    | 64,363    | 67,800    | 72,800    | 76,400    | 82,900    | 87,000    | 91,000    |            |                       |                       |
| Arts, entertainment, recreation, accommodation, and food services                             | 391,174   | 411,149   | 426,918   | 456,700   | 481,600   | 511,300   | 545,200   | 535,400   | 513,100   | 531,100    | 3.51%                 | 3.97%                 |
| Arts, entertainment, and recreation   | 94,816    | 102,140   | 105,752   | 113,300   | 117,300   | 126,600   | 134,400   | 135,200   | 127,300   | 131,200    | 3.06%                 | 4.26%                 |
| Performing arts, spectator sports, museums, and related activities                            | 49,826    | 55,338    | 59,005    | 62,200    | 63,800    | 68,100    | 72,400    | 73,200    | 70,900    |            |                       |                       |
| Amusements, gambling, and recreation industries   | 44,990    | 46,802    | 46,747    | 51,100    | 53,500    | 58,400    | 62,000    | 62,000    | 56,400    |            |                       |                       |
| Accommodation and food services   | 296,358   | 309,010   | 321,166   | 343,400   | 364,300   | 384,700   | 410,800   | 400,200   | 385,800   | 399,900    | 3.65%                 | 3.88%                 |
| Accommodation   | 91,314    | 92,764    | 93,791    | 100,600   | 108,700   | 113,600   | 123,200   | 118,100   | 108,700   |            |                       |                       |
| Food services and drinking places   | 205,044   | 216,246   | 227,376   | 242,800   | 255,600   | 271,100   | 287,600   | 282,100   | 277,100   |            |                       |                       |
| Other services, except government   | 264,193   | 284,966   | 289,672   | 303,000   | 318,500   | 332,000   | 344,600   | 340,900   | 335,400   | 343,800    | 2.50%                 | 3.35%                 |
| Government  | 1,275,395 | 1,352,970 | 1,435,322 | 1,507,700 | 1,585,900 | 1,667,800 | 1,759,900 | 1,855,100 | 1,922,500 | 1,963,900  | 2.15%                 | 6.00%                 |
| Federal   | 385,084   | 416,585   | 447,295   | 478,400   | 501,800   | 526,500   | 552,300   | 580,200   | 611,500   | 637,700    | 4.28%                 | 7.29%                 |
| General government  | 324,936   | 351,840   | 382,921   | 412,000   | 438,700   | 460,600   | 486,000   | 517,100   | 551,700   |            |                       |                       |
| Government enterprises  | 60,148    | 64,745    | 64,374    | 66,400    | 63,100    | 65,900    | 66,200    | 63,100    | 59,900    |            |                       |                       |
| State and local   | 890,311   | 936,385   | 988,027   | 1,029,300 | 1,084,100 | 1,141,300 | 1,207,600 | 1,274,900 | 1,311,000 | 1,326,100  | 1.15%                 | 5.44%                 |
| General government  | 814,699   | 859,593   | 909,258   | 947,300   | 997,700   | 1,051,300 | 1,118,600 | 1,180,500 | 1,208,500 |            |                       |                       |
| Government enterprises  | 75,612    | 76,792    | 78,769    | 81,900    | 86,400    | 000'06    | 89,100    | 94,400    | 102,500   |            |                       |                       |
|   |           |           |           |           |           |           |           |           |           |            |                       |                       |
| Private goods-producing industries  | 2,052,305 | 2,053,732 | 2,140,258 | 2,339,200 | 2,498,800 | 2,654,100 | 2,755,000 | 2,748,200 | 2,496,300 | 2,658,600  | 6.50%                 | 3.28%                 |
| Private services-producing industries <sup>3</sup>  | 6,958,467 | 7,235,615 | 7,566,564 | 8,020,900 | 8,553,700 | 9,077,000 | 9,546,900 | 9,765,800 | 9,700,300 | 10,038,000 | 3.48%                 | 4.92%                 |
| Information-communications-technology-producing industries <sup>4</sup>                       | 380,771   | 422,572   | 438,412   | 496,200   | 537,400   | 560,300   | 580,200   | 607,100   | 599,800   | 684,100    | 14.05%                | 8.85%                 |

# NEW REGULATORY

1

# FINANCE

Roger A. Morin, PhD

2006 PUBLIC UTILITIES REPORTS, INC. Vienna, Virginia expectations relative to history, historical growth rates become suspect as a measure of investor expectations.

Yet another issue associated with historical growth is that reliance on history to measure investor expectations renders the replication of that growth a self-fulfilling prophecy. Reliance on forecast growth rates avoids this inherent circularity.

The major point of all this is that it is perilous to apply historical growth when a utility is in a transition between growth paths. When payout ratios, equity return, and market-to-book ratios are changing, reliance on historical growth is hazardous. Such transitions can occur under variable inflation environments, and under fundamental structural shifts, such as deregulation.

Given the choice of variables, length of historical period, and the choice of statistical methodologies, the number of permutations and combinations of historical growth rates is such that other methods and proxies for expected growth must be explored. Historical growth rates constitute a useful starting point and provide useful information as long as the necessary conditions and assumptions outlined in this section are not dramatically violated. Although historical information provides a primary foundation for expectations, investors use additional information to supplement past growth rates. Extrapolating past history alone without consideration of historical trends and anticipated economic events would assume either that past rates will persist over time or that investors' expectations are based entirely on history.

# 9.4 Growth Estimates: Analysts' Forecasts

Since investor growth expectations are the quantities desired in the DCF model, the use of forecast growth published by investment services merits serious consideration. The growth rates assumed by investors can be determined by a study of the analyses of future earnings and projected long-run growth rates made by the investment community. The anticipated long-run growth rates actually used by institutional investors to determine the desirability of investing in different securities influence investors' growth anticipations.

Typically, growth forecasts are in the form of earnings per share over periods ranging from one to 5 years, and are supported by extensive financial analysis.<sup>10</sup>

297

<sup>&</sup>lt;sup>10</sup> Analysts do not generally disseminate their methods of forecasting and do not generally recommend the purchase or sale of a security based on any single growth variable or growth estimating technique. A professional financial analyst is reluctant to reveal the premises and methods of his professional judgment and recommendations. Moreover, analysts' buy/sell recommendations result from complex judgments that cannot be reduced to a single variable or to simple mechanistic equations or models. Several methods and algorithms, involving both quantitative and qualitative factors, are likely to be used in arriving at a final growth forecast, including historical indicators.

### New Regulatory Finance

The average growth rate estimate from all the analysts that follow the company measures the consensus expectation of the investment community for that company. In most cases, it is necessary to use earnings forecasts rather than dividend forecasts due to the extreme scarcity of dividend forecasts compared to the widespread availability of earnings forecasts. Given the paucity and variability of dividend forecasts, using the latter would produce unreliable DCF results. In any event, the use of the DCF model prospectively assumes constant growth in both earnings and dividends. Moreover, as discussed below, there is an abundance of empirical research that shows the validity and superiority of earnings forecasts relative to historical estimates when estimating the cost of capital.

The uniformity of growth projections is a test of whether they are typical of the market as a whole. If, for example, 10 out of 15 analysts forecast growth in the 7%-9% range, the probability is high that their analysis reflects a degree of consensus in the market as a whole. As a side note, the lack of uniformity in growth projections is a reasonable indicator of higher risk. Chapter 3 alluded to divergence of opinion amongst analysts as a valid risk indicator.

Because of the dominance of institutional investors and their influence on individual investors, analysts' forecasts of long-run growth rates provide a sound basis for estimating required returns. Financial analysts exert a strong influence on the expectations of many investors who do not possess the resources to make their own forecasts, that is, they are a cause of g. The accuracy of these forecasts in the sense of whether they turn out to be correct is not at issue here, as long as they reflect widely held expectations. As long as the forecasts are typical and/or influential in that they are consistent with current stock price levels, they are relevant. The use of analysts' forecasts in the DCF model is sometimes denounced on the grounds that it is difficult to forecast earnings and dividends for only one year, let alone for longer time periods. This objection is unfounded, however, because it is present investor expectations that are being priced; it is the consensus forecast that is embedded in price and therefore in required return, and not the future as it will turn out to be.

### **Empirical Literature on Earnings Forecasts**

Published studies in the academic literature demonstrate that growth forecasts made by security analysts represent an appropriate source of DCF growth rates, are reasonable indicators of investor expectations and are more accurate than forecasts based on historical growth. These studies show that investors rely on analysts' forecasts to a greater extent than on historic data only.

Academic research confirms the superiority of analysts' earnings forecasts over univariate time-series forecasts that rely on history. This latter category

### Chapter 9: Discounted Cash Flow Application

includes many *ad hoc* forecasts from statistical models, ranging from the naive methods of simple averages, moving averages, etc. to the sophisticated time-series techniques such as the Box-Jenkins modeling techniques. The literature suggests that analysts' earnings forecasts incorporate all the public information available to the analysts and the public at the time the forecasts are released. This finding implies that analysts have already factored historical growth trends into their forecast growth rates, making reliance on historical growth rates somewhat redundant and, at worst, potentially double counting growth rates which are irrelevant to future expectations. Furthermore, these forecasts are statistically more accurate than forecasts based solely on historical earnings, dividends, book value equity, and the like.

### Summary of Empirical Research

Important papers include Brown and Rozeff (1978), Cragg and Malkiel (1968, 1982), Harris (1986), Vander Weide and Carleton (1988), Lys and Sohn (1990), and Easterwood and Nutt (1999).

The study by Brown and Rozeff (1978) shows that analysts, as proxied by Value Line analysts, make better forecasts than could be obtained using only historical data, because analysts have available not only past data but also a knowledge of such crucial factors as rate case decisions, construction programs, new products, cost data, and so on. Brown and Rozeff test the accuracy of analysts' forecasts versus forecasts based on past data only, and conclude that their evidence of superior analyses means that analysts' forecasts should be used in studies of cost of capital. Their evidence supports the hypothesis that Value Line analysts consistently make better predictions than historical time-series models.

Using the IBES consensus earnings forecasts as proxies for investor expectation, Harris (1986) estimates the cost of equity using expected rather than historical earnings growth rates. In his review of the literature on financial analysts' forecasts, Harris concludes that a growing body of knowledge shows that analysts' earnings forecasts are indeed reflected in stock prices. Elton, Gruber, and Gultekin (1981) show that stock prices react more to changes in analysts' forecasts of earnings than they do to changes in earnings themselves, suggesting the usefulness of analysts' forecasts as surrogates for market expectations. In an extensive National Bureau of Economic Research study using analysts' earnings forecasts, Cragg and Malkiel (1982) present detailed empirical evidence that the average analyst's expectation is more similar to expectations being reflected in the marketplace than historical growth rates, and that it is the best possible source of DCF growth rates. The authors show that historical growth rates do not contain any information that is not already impounded in analysts' growth forecasts. They conclude that the expectations formed by Wall Street professionals get quickly and thoroughly impounded

### New Regulatory Finance

into the prices of securities and that the company valuations made by analysts are reflected in security prices.

Vander Weide and Carleton (1988) update the Cragg and Malkiel study and find overwhelming evidence that the consensus analysts' forecasts of future growth is superior to historically oriented growth measures in predicting the firm's stock price. Their results also are consistent with the hypothesis that investors use analysts' forecasts, rather than historically oriented growth calculations, in making stock buy-and-sell decisions. A study by Timme and Eiseman (1989) produced similar results.

Using virtually all publicly available analyst earnings forecasts for a large sample of companies (over 23,000 individual forecasts by 100 analyst firms), Lys and Sohn (1990) show that stock returns respond to individual analyst earnings forecasts, even when they are closely preceded by earnings forecasts made by other analysts or by corporate accounting disclosures. Using actual and IBES data from 1982–1995, Easterwood and Nutt (1999) regress the analysts' forecast errors against either historical earnings changes or analysts' forecasting errors in the prior years. Results show that analysts tend to underreact to negative earnings information, but overreact to positive earnings information.

The more recent studies provide evidence that analysts make biased forecasts and misinterpret the impact of new information.<sup>11</sup> For example, several studies in the early 1990s suggest that analysts either systematically underreact or overreact to new information. Easterwood and Nutt (1999) discriminate between these different reactions and reported that analysts underreact to negative information, but overreact to positive information. The recent studies do not necessarily contradict the earlier literature. The earlier research focused on whether analysts' earnings forecasts are better at forecasting future earnings than historical averages, whereas the recent literature investigates whether the analysts' earnings forecasts are unbiased estimates of future earnings. It is possible that even if the analysts' forecasts are biased, they are still closer to future earnings than the historical averages, although this hypothesis has not been tested in the recent studies. One way to assess the concern that analysts' forecasts may be biased upward is to incorporate into the analysis the growth forecasts of independent research firms, such as Value Line, in addition to the analyst consensus forecast. Unlike investment banking firms and stock brokerage firms, independent research firms such as Value Line have no incentive to distort earnings growth estimates in order to bolster interest in common stocks.

<sup>&</sup>lt;sup>11</sup> Other relevant papers corroborating the superiority of analysts' forecasts as predictors of future returns versus historical growth rates include: Fried and Givoly (1982), Moyer, Chatfield and Kelley (1985), and Gordon, Gordon and Gould (1989).

### Chapter 9: Discounted Cash Flow Application

Some argue that analysts tend to forecast earnings growth rates that exceed those actually achieved and that this optimism biases the DCF results upward. The magnitude of the optimism bias for large rate-regulated companies in stable segments of an industry is likely to be very small. Empirically, the severity of the optimism problem is unclear for regulated utilities, if a problem exists at all. It is interesting to note that Value Line forecasts for utility companies made by independent analysts with no incentive for over- or understating growth forecasts are not materially different from those published by analysts in security firms with incentives not based on forecast accuracy, and may in fact be more robust. If the optimism problem exists at all, it can be circumvented by relying on multiple-stage DCF models that substitute long-term economic growth for analysts' growth forecasts in the second and/ or third stages of the model.

Empirical studies have also been conducted showing that investors who rely primarily on data obtained from several large reputable investment research houses and security dealers obtain better results than those who do not.<sup>12</sup> Thus, both empirical research and common sense indicate that investors rely primarily on analysts' growth rate forecasts rather than on historical growth rates alone.

Ideally, one could decide which analysts make the most reliable forecasts and then confine the analysis to those forecasts. This would be impractical since reliable data on past forecasts are generally not available. Moreover, analysts with poor track records are replaced by more competent analysts, so that a poor forecasting record by a particular firm is not necessarily indicative of poor future forecasts. In any event, analysts working for large brokerage firms typically have a following, and investors who heed a particular analyst's recommendations do exert an influence on the market. So, an average of all the available forecasts from large reputable investment houses is likely to produce the best DCF growth rate.

Growth rate forecasts are available online from several sources. For example, Value Line Investment Analyzer, IBES (Institutional Brokers' Estimate System), Zacks Investment Research, Reuters, First Call, Yahoo Finance, and Multex Web sites provide analysts' earnings forecasts on a regular basis by reporting on the results of periodic (usually monthly) surveys of the earnings growth forecasts of a large number of investment advisors, brokerage houses, and other firms that engage in fundamental research on U.S. corporations. These firms include most large institutional investors, such as pension funds, banks, and insurance companies. Representative of industry practices, the Zacks Investment Research Web site is a central location whereby investors

<sup>&</sup>lt;sup>12</sup> Examples of these studies include Stanley, Lewellen and Schlarbaum (1981) and Touche Ross Co. (1982).

### New Regulatory Finance

are able to research the different analyst estimates for any given stock without necessarily searching for each individual analyst. Zacks gathers and compiles the different estimates made by stock analysts on the future earnings for the majority of U.S. publicly traded companies. Estimates of earnings per share for the upcoming 2 fiscal years, and a projected 5-year growth rate in such earnings per share are available at monthly intervals. The forecast 5-year growth rates are normalized in order to remove short-term distortions. Forecasts are updated when analysts formally change their stated predictions.

Exclusive reliance on a single analyst's growth forecast runs the risk of being unrepresentative of investors' consensus forecast. One would expect that averages of analysts' growth forecasts, such as those contained in IBES or Zacks, are more reliable estimates of investors' consensus expectations likely to be impounded in stock prices.<sup>13</sup> Averages of analysts' growth forecasts rather than a single analyst's growth forecasts are more reliable estimates of investors' consensus expectations.

One problem with the use of published analysts' forecasts is that some forecasts cover only the next one or two years. If these are abnormal years, they may not be indicative of longer-run average growth expectations. Another problem is that forecasts may not be available in sufficient quantities or may not be available at all for certain utilities, for example water utilities, in which case alternate methods of growth estimation must be employed.

Some financial economists are uncomfortable with the assumption that the DCF growth rates are perpetual growth rates, and argue that above average growth can be expected to prevail for a fixed number of years and then the growth rate will settle down to a steady-state, long-run level, consistent with that of the economy. The converse also can be true whereby below-average growth can be expected to prevail for a fixed number of years and then the growth rate will resume a higher steady-state, long-run level. Extended DCF models are available to accommodate such assumptions, and were discussed in Chapter 8.

## Earnings versus Dividend Forecasts

Casual inspection of the Zacks Investment Research, First Call Thompson, and Multex Web sites reveals that earnings per share forecasts dominate the information provided. There are few, if any, dividend growth forecasts. Only Value Line provides comprehensive long-term dividend growth forecasts. The wide availability of earnings forecasts is not surprising. There is an abundance of evidence attesting to the importance of earnings in assessing investors'

<sup>&</sup>lt;sup>13</sup> The earnings growth rates published by Zacks, First Call, Reuters, Value Line, and IBES contain significant overlap since all rely on virtually the same population of institutional analysts who provide such forecasts.

### Chapter 9: Discounted Cash Flow Application

expectations. The sheer volume of earnings forecasts available from the investment community relative to the scarcity of dividend forecasts attests to their importance. The fact that these investment information providers focus on growth in earnings rather than growth in dividends indicates that the investment community regards earnings growth as a superior indicator of future longterm growth. Surveys of analytical techniques actually used by analysts reveal the dominance of earnings and conclude that earnings are considered far more important than dividends. Finally, Value Line's principal investment rating assigned to individual stocks, Timeliness Rank, is based primarily on earnings, accounting for 65% of the ranking.

### Historical Growth Rates Versus Analysts' Forecasts

Obviously, historical growth rates as well as analysts' forecasts provide relevant information to the investor with regard to growth expectations. Each proxy for expected growth brings information to the judgment process from a different light. Neither proxy is without blemish; each has advantages and shortcomings. Historical growth rates are available and easily verifiable, but may no longer be applicable if structural shifts have occurred. Analysts' growth forecasts may be more relevant since they encompass both history and current changes, but are nevertheless imperfect proxies.

### 9.5 Growth Estimates: Sustainable Growth Method

The third method of estimating the growth component in the DCF model, alternately referred to as the "sustainable growth" or "retention ratio" method, can be used by investment analysts to predict future growth in earnings and dividends. In this method, the fraction of earnings expected to be retained by the company, b, is multiplied by the expected return on book equity, r, to produce the growth forecast. That is,

### $g = b \times r$

The conceptual premise of the method, enunciated in Chapter 8, Section 8.4, is that future growth in dividends for existing equity can only occur if a portion of the overall return to investors is reinvested into the firm instead of being distributed as dividends.

For example, if a company earns 12% on equity, and pays all the earnings out in dividends, the retention factor, b, is zero and earnings per share will not grow for the simple reason that there are no increments to the asset base (rate base). Conversely, if the company retains all its earnings and pays no dividends, it would grow at an annual rate of 12%. Or again, if the company earns 12% on equity and pays out 60% of the earnings in dividends, the

### New Regulatory Finance

retention factor is 40%, and earnings growth will be  $40\% \times 12\% = 4.8\%$  per year.

In implementing the method, both 'b' and 'r' should be the rate that the market expects to prevail in the future. If no explicit forecast of 'b' is available, it is reasonable to assume that the utility's future retention ratio will, on average, remain unchanged from its present level. Or, it can be estimated by taking a weighted average of past retention ratios as a proxy for the future on the grounds that utilities' target retention ratios are usually, although not always, stable.<sup>14</sup>

Both historical and forecast values of 'r' can be used to estimate g, although forecast values are superior. The use of historical realized book returns on equity rather than the expected return on equity is questionable since reliance on achieved results involves circular reasoning. Realized returns are the results of the regulatory process itself, and are also subject to tests of fairness and reasonableness. As a gauge of the expected return on book equity, either direct published analysts' forecasts of the long-run expected return on equity, or authorized rates of return in recent regulatory cases can be used as a guide. As a floor estimate, it seems reasonable for investors to expect allowed equity returns by state regulatory commissions to be in excess of the current cost of debt to the utility in question.

Another way of obtaining the expected 'r' is to examine its fundamental determinants. Since earnings per share, E, can be stated as dividends per share, D, divided by the payout ratio (1 - b), the earnings per share capitalized by investors can be inferred by dividing the current dividend by an expected payout ratio. Provided that a utility company follows a fairly stable dividend policy, the possibility of error is less when estimating the payout than when estimating the expected return on equity or the expected growth rate. Using this approach, and denoting book value per share by B, the expected return on equity is:

$$r = E/B = (D/(1 - b)) / B$$
 (9-9)

Estimates of the expected payout ratio can be inferred from historical 10-year average payout ratio data for utilities, assuming a stable dividend policy has been pursued. Since individual averages frequently tend to regress toward the grand mean, the historical payout ratio needs to be adjusted for this tendency, using statistical techniques for predicting future values based on this tendency of individual values to regress toward the grand mean over time.

An application of the sustainable growth method is shown in example 9-1.

<sup>&</sup>lt;sup>14</sup> Statistically superior predictions of future averages are made by weighting individual past averages with the grand mean, with the variance within the individual averages and the variance across individual averages serving as weights.

### Chapter 9: Discounted Cash Flow Application

|   | EXAMPLE 9-1   |
|---|---|
| Southeastern Electric's su<br>rate case testimony. As<br>authorized rates of return<br>utilities as reported by V<br>with a standard deviation<br>were authorized to earn 1<br>from 10% to 12%. As<br>average 2006 payout rat<br>by Value Line was 60%,<br>40%, with a standard dev<br>fun target retention ratio<br>Electric. It is therefore r<br>retention ratio ranging fr<br>value of 40%. In Table 9<br>45% and assumed return<br>to produce sustainable g<br>a likely value of 4.6%. | istainable growth rate is required for upcoming<br>a gauge of the expected return on equity,<br>in in recent decisions for eastern U.S. electric<br>(alue Line for 2005 and 2006 averaged 11%,<br>of 1%. In other words, the majority of utilities<br>1%, with the allowed return on equity ranging<br>a gauge of the expected retention ratio, the<br>io of 34 eastern electric utilities as compiled<br>which indicates an average retention ratio of<br>iation of 5%. This was consistent with the long-<br>indicated by the management of Southeastern<br>easonable to postulate that investors expect a<br>om 35% to 45% for the company with a likely<br>0.4 below, expected retention ratios of 35% to<br>is on equity from 10% to 12% are multiplied<br>prowth rates ranging from 3.8% to 5.4% with |
| SUSTAINABLE   | TABLE 9-4<br>GROWTH METHOD ILLUSTRATION   |
| Expected  | Expected Return on Book Equily (r)  |
| Retention Ratio (b)   | 10%   |
| 35%<br>40%  | 3.5%<br>4.0%<br>4.4%  |
| 45%   | 4.5% 5.0%   |

•• • •

。 2. 我们的问题,我们的我们的我们的我们的我们的,你不是你的人们都是我们的是我们的是我的人们的。"

It should be pointed out that published forecasts of the expected return on equity by analysts such as Value Line are sometimes based on end-of-period book equity rather than on average book equity. The following formula<sup>15</sup>

$$\frac{r}{r_a} = \frac{E/B_l}{E/B_a} = \frac{B_a}{B_l} + \frac{B_l + B_{l-1}}{2B_l}$$

Solving for  $r_a$ , a formula for translating the return on year-end equity into the return on average equity is obtained, using reported beginning-of-the year and end-ofyear common equity figures:

$$r_a = r \frac{2B_t}{B_t + B_{t-1}}$$

305

<sup>&</sup>lt;sup>15</sup> The return on year-end common equity, r, is defined as  $r = E/B_t$ , where E is earnings per share, and  $B_t$  is the year-end book value per share. The return on average common equity,  $r_a$ , is defined as:  $r_a = E/B_a$  where  $B_a =$  average book value per share. The latter is by definition:  $B_a = (B_t + B_{t-1})/2$  where  $B_t$  is the year-end book equity per share and  $B_{t-1}$  is the beginning-of-year book equity per share. Dividing r by  $r_a$  and substituting:

### New Regulatory Finance

adjusts the reported end-of-year values so that they are based on average common equity, which is the common regulatory practice:

$$r_a = r_t \frac{2B_t}{B_t + B_{t-1}} \tag{9-10}$$

The sustainable growth method can also be extended to include external financing. From Chapter 8, the expanded growth estimate is given by:

$$g = br + sv$$

where b and r are defined as previously, s is the expected percent growth in number of shares to finance investment, and v is the profitability of the equity investment. The variable s measures the long-run expected stock financing that the utility will undertake. If the utility's investments are growing at a stable rate and if the earnings retention rate is also stable, then s will grow at a stable rate. The variable s can be estimated by taking a weighted average of past percentage increases in the number of shares. This measurement is difficult, however, owing to the sporadic and episodic nature of stock financing, and smoothing techniques must be employed. The variable v is the profitability of the equity investment and can be measured as the difference of market price and book value per share divided by the latter, as discussed in Chapter 8.

There are three problems in the practical application of the sustainable growth method. The first is that it may be even more difficult to estimate what b, r, s, and v investors have in mind than it is to estimate what g they envisage. It would appear far more economical and expeditious to use available growth forecasts and obtain g directly instead of relying on four individual forecasts of the determinants of such growth. It seems only logical that the measurement and forecasting errors inherent in using four different variables to predict growth far exceed the forecasting error inherent in a direct forecast of growth itself.

Second, there is a potential element of circularity in estimating g by a forecast of b and ROE for the utility being regulated, since ROE is determined in large part by regulation. To estimate what ROE resides in the minds of investors is equivalent to estimating the market's assessment of the outcome of regulatory hearings. Expected ROE is exactly what regulatory commissions set in determining an allowed rate of return. In other words, the method requires an estimate of return on equity before it can even be implemented. Common sense would dictate the inconsistency of a return on equity recom-

### Chapter 9: Discounted Cash Flow Application

mendation that is different than the expected ROE that the method assumes the utility will earn forever. For example, using an expected return on equity of 11% to determine the growth rate and using the growth rate to recommend a return on equity of 9% is inconsistent. It is not reasonable to assume that this regulated utility company is expected to earn 11% forever, but recommend a 9% return on equity. The only way this utility can earn 11% is that rates be set by the regulator so that the utility will in fact earn 11%. One is assuming, in effect, that the company will earn a return rate exceeding the recommended cost of equity forever, but then one is recommending that a different rate be granted by the regulator. In essence, using an ROE in the sustainable growth formula that differs from the final estimated cost of equity is asking the regulator to adopt two different returns.

The circularity problem is somewhat dampened by the self-correcting nature of the DCF model. If a high equity return is granted, the stock price will increase in response to the unanticipated favorable return allowance, lowering the dividend yield component of market return in compensation for the high g induced by the high allowed return. At the next regulatory hearing, more conservative forecasts of r would prevail. The impact on the dual components of the DCF formula, yield and growth, are at least partially offsetting.

Third, the empirical finance literature discussed earlier demonstrates that the sustainable growth method of determining growth is not as significantly correlated to measures of value, such as stock price and price/earnings ratios, as other historical growth measures or analysts' growth forecasts. Other proxies for growth, such as historical growth rates and analysts' growth forecasts, outperform retention growth estimates. See for example Timme and Eiseman (1989).

In summary, there are three proxies for the expected growth component of the DCF model: historical growth rates, analysts' forecasts, and the sustainable growth method. Criteria in choosing among the three proxies should include ease of use, ease of understanding, theoretical and mathematical correctness, and empirical validation. The latter two are crucial. The method should be logically valid and consistent, and should possess an adequate track record in predicting and explaining security value. The retention growth method is the weakest of the three proxies on both conceptual and empirical grounds. The research in this area has shown that the first two growth proxies do a better job of explaining variations in market valuation (M/B and P/E ratios) and are more highly correlated to measures of value than is the retention growth proxy.

### Missouri-American Water Company Market-to-Book Ratios, Earnings / Book Ratios and Inflation for Standard & Poor's Industrial Index and the Standard & Poor's 500 Composite Index

from 1947 through 2010

|         |   |           | <u>1101111947 tillougi</u>                | 12010              |                        |                     |                       |
|---------|---|-----------|---|--------------------|------------------------|---------------------|-----------------------|
|         | Market                                    | -         |   |                    |                        |                     |                       |
|         | to-Boo                                    | k         | Earning                                   | s/                 |                        |                     |                       |
| Veer    | Botio (1                                  | 1)        | Book Boti                                 | 2 (2)              |                        |                     |                       |
| Tear    | Italio (                                  | 08D 500   | Book Rain                                 | 0 (2)<br>0 8 D 500 |                        |                     |                       |
|         |   | 5&P 500   |   | S&P 500            |                        |                     |                       |
|         | S&P Industrial                            | Composite | S&P Industrial                            | Composite          |                        |                     |                       |
|         | Index (3)                                 | Index (3) | Index (3)                                 | Index (3)          | Inflation (4)          | Earnings / Book Rat | io - Net of Inflation |
|         | <u>, , , , , , , , , , , , , , , , , </u> |           | <u>, , , , , , , , , , , , , , , , , </u> |                    |                        |                     |                       |
| 1947    | 1.23                                      | NA        | 13.0 %                                    | NA                 | 9.0 %                  | 4.0 %               | NA                    |
| 1948    | 1.13                                      | NA        | 17.3                                      | NA                 | 2.7                    | 14.6                | NA                    |
| 1949    | 1.00                                      | NA        | 16.3                                      | NA                 | (1.8)                  | 18 1                | NA                    |
| 1050    | 1.10                                      | NIA       | 10.0                                      | NIA                | (1.6)                  | 10.5                | NIA                   |
| 1950    | 1.10                                      | NA        | 10.3                                      | INA                | 5.6                    | 12.5                | NA                    |
| 1951    | 1.27                                      | NA        | 14.4                                      | NA                 | 5.9                    | 8.5                 | NA                    |
| 1952    | 1.29                                      | NA        | 12.7                                      | NA                 | 0.9                    | 11.8                | NA                    |
| 1953    | 1.21                                      | NA        | 12.7                                      | NA                 | 0.6                    | 12.1                | NA                    |
| 1054    | 1.45                                      | NA        | 13.5                                      | NA                 | (0.5)                  | 14.0                | NIA                   |
| 1954    | 1.45                                      |           | 13.5                                      |                    | (0.3)                  | 14.0                | N/A                   |
| 1955    | 1.61                                      | NA        | 16.0                                      | NA                 | 0.4                    | 15.6                | NA                    |
| 1956    | 1.92                                      | NA        | 13.7                                      | NA                 | 2.9                    | 10.8                | NA                    |
| 1957    | 1.71                                      | NA        | 12.5                                      | NA                 | 3.0                    | 9.5                 | NA                    |
| 1958    | 1 70                                      | ΝΔ        | 9.8                                       | ΝΔ                 | 1.8                    | 8.0                 | NΔ                    |
| 1050    | 1.04                                      | NA        | 11.2                                      | NIA                | 1.0                    | 0.0                 | NIA                   |
| 1959    | 1.94                                      | INA       | 11.2                                      | INA                | 1.5                    | 9.7                 | INA                   |
| 1960    | 1.82                                      | NA        | 10.3                                      | NA                 | 1.5                    | 8.8                 | NA                    |
| 1961    | 2.01                                      | NA        | 9.8                                       | NA                 | 0.7                    | 9.1                 | NA                    |
| 1962    | 1.83                                      | NA        | 10.9                                      | NA                 | 1.2                    | 9.7                 | NA                    |
| 1963    | 1 94                                      | NA        | 11.4                                      | NA                 | 17                     | 97                  | NA                    |
| 1000    | 0.40                                      | NIA NIA   | 10.0                                      |                    | 1.7                    |                     | 101                   |
| 1964    | 2.18                                      | NA        | 12.3                                      | NA                 | 1.2                    | 11.1                | NA                    |
| 1965    | 2.21                                      | NA        | 13.2                                      | NA                 | 1.9                    | 11.3                | NA                    |
| 1966    | 2.00                                      | NA        | 13.2                                      | NA                 | 3.4                    | 9.8                 | NA                    |
| 1967    | 2.05                                      | NA        | 12.1                                      | NA                 | 3.0                    | 9.1                 | NA                    |
| 1069    | 2.00                                      | NA        | 12.1                                      | NIA                | 4.7                    | 7.0                 | NIA                   |
| 1900    | 2.17                                      | INA       | 12.0                                      | INA                | 4.7                    | 7.9                 | INA                   |
| 1969    | 2.10                                      | NA        | 12.1                                      | NA                 | 6.1                    | 6.0                 | NA                    |
| 1970    | 1.71                                      | NA        | 10.4                                      | NA                 | 5.5                    | 4.9                 | NA                    |
| 1971    | 1.99                                      | NA        | 11.2                                      | NA                 | 3.4                    | 7.8                 | NA                    |
| 1072    | 2 16                                      | NΔ        | 12.0                                      | ΝΔ                 | 3.4                    | 8.6                 | ΝΔ                    |
| 1072    | 2.10                                      | N/A       | 14.0                                      | NA                 | 5.4                    | 5.0                 | NA                    |
| 1973    | 1.96                                      | NA        | 14.0                                      | INA                | 0.0                    | 5.8                 | NA                    |
| 1974    | 1.39                                      | NA        | 14.8                                      | NA                 | 12.2                   | 2.6                 | NA                    |
| 1975    | 1.34                                      | NA        | 12.3                                      | NA                 | 7.0                    | 5.3                 | NA                    |
| 1976    | 1.51                                      | NA        | 14.5                                      | NA                 | 48                     | 97                  | NA                    |
| 1077    | 1 29                                      | NA        | 14.6                                      | NA                 | 6.9                    | 7.9                 | NIA                   |
| 1977    | 1.56                                      | ING.      | 14.0                                      | ING.               | 0.0                    | 7.0                 | INA<br>NA             |
| 1978    | 1.25                                      | NA        | 15.3                                      | NA                 | 9.0                    | 6.3                 | NA                    |
| 1979    | 1.23                                      | NA        | 17.2                                      | NA                 | 13.3                   | 3.9                 | NA                    |
| 1980    | 1.31                                      | NA        | 15.6                                      | NA                 | 12.4                   | 3.2                 | NA                    |
| 1081    | 1 24                                      | NΔ        | 14.9                                      | ΝΔ                 | 8.9                    | 6.0                 | ΝΔ                    |
| 1000    | 1.27                                      | N/A       | 14.5                                      | NA                 | 0.5                    | 0.0                 | NA                    |
| 1982    | 1.17                                      | NA        | 11.3                                      | INA                | 3.9                    | 7.4                 | NA                    |
| 1983    | 1.45                                      | NA        | 12.2                                      | NA                 | 3.8                    | 8.4                 | NA                    |
| 1984    | 1.46                                      | NA        | 14.6                                      | NA                 | 4.0                    | 10.6                | NA                    |
| 1985    | 1.67                                      | NA        | 12.2                                      | NA                 | 3.8                    | 8.4                 | NA                    |
| 1986    | 2 02                                      | NΔ        | 11.5                                      | ΝΔ                 | 1 1                    | 10.4                | ΝΔ                    |
| 1007    | 2.02                                      | NIA       | 11.0                                      | NIA                | 1.1                    | 14.0                | NIA                   |
| 1987    | 2.50                                      | NA        | 15.7                                      | INA                | 4.4                    | 11.3                | NA                    |
| 1988    | 2.13                                      | NA        | 19.0                                      | NA                 | 4.4                    | 14.6                | NA                    |
| 1989    | 2.56                                      | NA        | 18.5                                      | NA                 | 4.7                    | 13.8                | NA                    |
| 1990    | 2.63                                      | NA        | 16.3                                      | NA                 | 6.1                    | 10.2                | NA                    |
| 1991    | 2 77                                      | NA        | 10.8                                      | NA                 | 3.1                    | 77                  | NΔ                    |
| 1002    | 2.20                                      | NA        | 12.0                                      | NIA                | 2.0                    | 10.1                | NIA                   |
| 1992    | 3.29                                      | INA       | 13.0                                      | INA                | 2.9                    | 10.1                | INA                   |
| 1993    | 3.72                                      | NA        | 15.7                                      | NA                 | 2.8                    | 12.9                | NA                    |
| 1994    | 3.73                                      | NA        | 23.0                                      | NA                 | 2.7                    | 20.3                | NA                    |
| 1995    | 4.06                                      | 2.64      | 22.9                                      | 16.0 %             | 2.5                    | 20.4                | 13.5 %                |
| 1996    | 4 79                                      | 3 00      | 24.8                                      | 16.8               | 33                     | 21.5                | 13.5                  |
| 1007    | 5.00                                      | 0.00      | 24.0                                      | 10.0               | 0.0                    | 21.0                | 14.6                  |
| 1997    | 5.88                                      | 3.53      | 24.0                                      | 10.3               | 1.7                    | 22.9                | 14.0                  |
| 1998    | 7.13                                      | 4.16      | 21.3                                      | 14.5               | 1.6                    | 19.7                | 12.9                  |
| 1999    | 8.27                                      | 4.76      | 25.2                                      | 17.1               | 2.7                    | 22.5                | 14.4                  |
| 2000    | 7,51                                      | 4.51      | 23.9                                      | 16.2               | 3.4                    | 20.5                | 12.8                  |
| 2001    | NA  | 3 50      | NA  | 7.4                | 1.6                    | NA                  | 5.8                   |
| 2001    |   | 0.00      |   | 1.4                | 1.0                    | NA NA               | 5.0                   |
| 2002    | NA  | 2.93      | NA  | 8.3                | 2.4                    | NA                  | 5.9                   |
| 2003    | NA  | 2.78      | NA  | 14.1               | 1.9                    | NA                  | 12.2                  |
| 2004    | NA  | 2.91      | NA  | 15.3               | 3.3                    | NA                  | 12.0                  |
| 2005    | NA  | 2.78      | NA  | 16.4               | 3.4                    | NA                  | 13.0                  |
| 2006    | NIA                                       | 2.75 (5)  | NIA                                       | 17.0               | 0. <del>1</del><br>2 E | N/A                 | 147                   |
| 2000    | INA                                       | 2.10 (0)  | INA                                       | 11.2               | 2.5                    | INA<br>NA           | 14.7                  |
| 2007    | NA  | 2.77 (5)  | NA  | 12.8               | 4.1                    | NA                  | 8.7                   |
| 2008    | NA  | 2.02 (5)  | NA  | 2.7                | 0.1                    | NA                  | 2.6                   |
| 2009    | NA  | 1.63      | NA  | 9.2                | 2.7                    | NA                  | 6.5                   |
| 2010    | NA  | 1 92      | NA  | 13.0               | 15                     | NA                  | 11.5                  |
| 2010    | 110                                       | 1.02      | 110                                       | 10.0               | 1.5                    |                     | 11.0                  |
|         |   |           |   | 10                 |                        | 10.5                |                       |
| Average | 2.34                                      | 3.04      | <u>14.9</u> %                             | 13.3 %             | 3.7 %                  | 10.9 %              | 10.9 %                |

Notes: (1) Market-to-Book Ratio equals average of the high and low market price for the year divided by the average book value.

(2) Earnings/Book equals earnings per share for the year divided by the average book value.

(3) On January 2, 2001 Standard & Poor's released Global Industry Classification Standard (GICS) price indexes for all Standard & Poor's U.S. indexes. As a result, all S&P Indexes have been calculated with a common base of 100 at a start date of December 31, 1994. Also, the GICS industrial sector is not comparable to the former S&P Industrial Index and data for the former S&P Industrial Index has been discontinued.

(4) As measured by the Consumer Price Index (CPI).

(5) Ratios for 2006 / 2007 are based upon estimated book values using the actual average price and the estimated book value calculated by adding the 2006 earnings per share to the 2005 / 2006 book value per share and then subtracting the 2006 / 2007 dividends per share as provided by Standard & Poor's Statistical Record -Current Statistics, March 2008, p. 29.

Source of Information: Standard & Poor's Security Price Index Record, 2000 Edition, p. 40 Standard & Poor's Statistical Service, Current Statistics, August 2001, p. 29 Standard & Poor's Statistical Service, Current Statistics, January 2001, p. 36 Standard & Poor's Current Statistics, June 2006, p. 29. Standard & Poor's Current Statistics, August 2007, p. 29.

Standard & Poor's Compustat Services, Inc. PC Plus Research Insight Database Ibbotson SBBI 2011 Valuation Yearbook

### Missouri-American Water Company Indicated Common Equity Cost Rate Through Use of a Risk Premium Model Using an Adjusted Total Market Approach

| Line No. |   | MIEC Witness<br>Gorman's Proxy<br>Group of Eight<br>Water Companies |
|----------|---|---|
|          |   |   |
| 1.       | Prospective Yield on Aaa Rated<br>Corporate Bonds (1)   | 4.37 %  |
| 2.       | Adjustment to Reflect Yield Spread<br>Between Aaa Rated Corporate<br>Bonds and A Rated Public |   |
|          | Utility Bonds   | 0.35 (2)  |
| 3.       | Adjusted Prospective Yield on A Rated<br>Public Utility Bonds                                 | 4.72 %  |
| 4.       | Adjustment to Reflect Bond<br>Rating Difference of Proxy Group                                | 0.18 (3)  |
| 5.       | Adjusted Prospective Bond Yield   | 4.90  |
| 6.       | Equity Risk Premium (4)   | 5.71  |
| 7.       | Risk Premium Derived Common<br>Equity Cost Rate   | <u> </u>  |

Notes: (1) Derived in Note (4) on page 4 of this Schedule.

(2) The average yield spread of A rated public utility bonds over Aaa rated corporate bonds of 0.35% from page 2 of this Schedule.

- (3) Adjustment to reflect the A3 Moody's Bond Rating of the MIEC Witness Gorman's Proxy Group of Eight Water Companies as shown on page 2 of this Schedule. The 18 basis point adjustment is derived by taking 1/3 of the spread between Baa and A Public Utility Bonds (1/3 \* 0.53% = 0.177%, rounded to 0.18%).
- (4) From page 3 of this Schedule.

### Missouri-American Water Company Moody's and Standard & Poor's Bond Ratings for MIEC Witness Gorman's Proxy Group of Eight Water Companies

|   | N     | loody's              | Standa | ard & Poor's               |
|---|-------|----------------------|--------|----------------------------|
|   | Bor   | nd Rating            | Bor    | d Rating                   |
|   | Octo  | ober 2011            | Octo   | ber 2011                   |
|   | Bond  | Numerical            | Bond   | Numerical<br>Weighting (1) |
| MIEC Witness Gorman's Proxy Group<br>of Eight Water Companies | Raung | <u>weighting (1)</u> |        | weighting (1)              |
| American States Water Co. (3)                                 | A2    | 6.0                  | A+     | 5.0                        |
| American Water Works Co., Inc. (4)                            | Baa1  | 8.0                  | A+     | 5.0                        |
| Aqua America, Inc. (5)  | NR    |                      | AA-    | 4.0                        |
| California Water Service Group (6)                            | NR    |                      | AA-    | 4.0                        |
| Connecticut Water Service, Inc. (7)                           | NR    |                      | А      | 6.0                        |
| Middlesex Water Company                                       | NR    |                      | А      | 6.0                        |
| SJW Corporation (8)   | NR    |                      | А      | 6.0                        |
| York Water Company  | NR    |                      | A-     | 7.0                        |
| Average   | A3    | 7.0                  | A+     | 5.4                        |

Notes: (1) From page 3 of Schedule PMA-10.

 Ratings, business risk and financial risk profiles are those of Golden State Water Company.
 Rating, business risk and financial risk profiles are those of Pennsylvania and New Jersey American Water.

(4) Ratings, business risk and financial risk profiles are those of Aqua Pennsylvania, Inc.

(5) Ratings, business risk and financial risk profiles are those of California Water Service Co.

(6) Ratings, business risk and financial risk profiles are those of Connecticut Water Company.

(7) Ratings, business risk and financial risk profiles are those of San Jose Water Co.

(8) Ratings, business risk and financial risk profiles are those of Laclede Gas Company.

Source Information: Moody's Investors Service

Standard & Poor's Global Utilities Rating Service

<u>MoodVs</u> Comparison of Interest Rate Trends for the Three Months Ending September 2011 (1)

| olic Utility Bonds   |                 |                      | Baa over A  |              |           |         | 0.53 %                      |
|----------------------|-----------------|----------------------|-------------|--------------|-----------|---------|-----------------------------|
| Spread - Pub         |                 |                      | A over Aa   |              |           |         | 0.23 %                      |
| tility Bonds         | Baa (Pub.       | Util.) over          | Aaa (Corp.) |              |           |         | 0.88 %                      |
| prporate v. Public U | A (Pub. Util.)  | over Aaa             | (Corp.)     |              |           |         | 0.35 %                      |
| Spread - Corp        | Aa (Pub. Util.) | over Aaa             | (Corp.)     |              |           |         | 0.12 %                      |
|                      | Aa (Pub         |                      | Baa Rated   | 5.11 %       | 5.22      | 5.70    | 5.34 %                      |
|                      |                 | Public Utility Bonds | A Rated     | 4.48 %       | 4.69      | 5.27    | 4.81 %                      |
|                      |                 |                      | Aa Rated    | 4.24 %       | 4.44      | 5.05    | 4.58 %                      |
|                      | Corporate       | Bonds                | Aaa Rated   | 4.09 %       | 4.37      | 4.93    | 4.46 %                      |
|                      |                 |                      | Months      | September-11 | August-11 | July-11 | Average of Last<br>3 Months |

Notes: (1) All yields are distributed yields.

Source of Information: Mergent Bond Record, October 2011, Vol 78, No. 10

### Missouri-American Water Company Judgment of Equity Risk Premium for the MIEC Witness Gorman's Proxy Group of Eight Water Companies <u>d MIEC Witness Gorman's Proxy Group of Ten Natural Gas Distribution Compani</u>

| Line<br>No. |   | Gorman's Proxy<br>Group of Eight<br>Water Companies |
|-------------|---|---|
| 1.          | Calculated equity risk  |   |
|             | premium based on the<br>total market using<br>the beta approach (1)                                     | 7.30  |
| 2.          | Mean equity risk premium<br>based on a study<br>using the holding period<br>returns of public utilities |   |
|             | with A rated bonds (2)  | 4.12  |
| 3.          | Average equity risk premium   | <u> </u>  |

Notes: (1) From page 4 of this Schedule.

(2) From page 6 of this Schedule.

# <u>Missouri-American Water Company</u> Derivation of Equity Risk Premium Based on the Total Market Approach Using the Beta for the MIEC Witness Gorman's Proxy Group of Eight Water Companies and MIEC Witness Gorman's Proxy Group of Ten Natural Gas Distribution Companies

| Line No. |  |  | MIEC Witness<br>Gorman's Proxy<br>Group of Eight<br>Water Companies   |
|----------|--|--|---|
|          |  |  |   |
| 1.       |  | Arithmetic mean total return rate on<br>the Standard & Poor's 500 Composite<br>Index - 1926-2010 (1)   | 11.90 %   |
| 2.       |  | Arithmetic mean yield on<br>Aaa and Aa Corporate Bonds<br>1926-2010 (2)  | (6.10)  |
| 3.       |  | Historical Equity Risk Premium   | <u> </u>  |
| 4.       |  | Forecasted 3-5 year Total Annual<br>Market Return (3)  | 18.29 %   |
| 5.       |  | Prospective Yield an Aaa Rated<br>Corporate Bonds (4)  | (4.37)  |
| 6.       |  | Forecasted Equity Risk Premium   | 13.92_%   |
| 7.       |  | Conclusion of Equity Risk Premium (5)  | 9.86 %  |
| 8.       |  | Adjusted Value Line Beta (6)   | 0.74  |
| 9.       |  | Beta Adjusted Equity Risk Premium  | <u> </u>  |
| Notes:   | <ul> <li>(1)</li> <li>(2)</li> <li>(3)</li> <li>(4)</li> </ul> | Stocks, Bonds, Bills, and Inflation - Market Results for 1926-207<br>Valuation Edition, Morningstar, Inc., 2011 Chicago, IL.<br>From Moody's Industrial Manual and Mergent Bond Record Mon<br>The projected 3-5 year return of the market as calculated by the<br>market appreciation potential plus dividend yield published by V<br>October 21, 2011. The forecasted 3-5 year total annual market<br>(15.99% + 2.30% = 18.29%)<br>Average forecast based upon six quarterly estimates of Aaa rate<br>per the consensus of nearly 50 economists reported in Blue Ch<br>Forecasts dated October 1, 2011 (see page 5 of this Schedule)<br>are detailed below. | 10 Yearbook<br>hthly Update.<br>13 week average<br>(alue Line ended<br>return is 18.29%.<br>ed corporate bonds<br>ip Financial<br>. The estimates |
|          |  | Fourth Quarter 2011<br>First Quarter 2012<br>Second Quarter 2012<br>Third Quarter 2012<br>Fourth Quarter 2012<br>First Quarter 2013  | 4.20 %<br>4.20<br>4.30<br>4.40<br>4.50<br>4.60  |

<u>4.37</u>%

Average 4.60
Average 4.37 %
(5) The average of the historical equity risk premium of 5.80% from Line No. 3 and the forecasted equity risk premium of 13.92% from Line No. 6 ((5.80% + 13.92%) / 2 =
(6) From Line 3 of MPG-16, page 1.

### Consensus Forecasts Of U.S. Interest Rates And Key Assumptions<sup>1</sup>

|                         |                |           |               | Histo         | ry          |             |       |                | Cons        | ensus l     | Foreca      | sts-Qua     | arterly     | Avg.        |
|-------------------------|----------------|-----------|---------------|---------------|-------------|-------------|-------|----------------|-------------|-------------|-------------|-------------|-------------|-------------|
|                         | Av             | erage For | Week End      | ding          | Ave         | rage For M  | Month | Latest $Q^*$   | 4Q          | 1Q          | 2Q          | 3Q          | 4Q          | 1Q          |
| Interest Rates          | <u>Sep. 23</u> | Sep. 16   | <u>Sep. 9</u> | <u>Sep. 2</u> | <u>Aug.</u> | <u>July</u> | June  | <u>3Q 2011</u> | <u>2011</u> | <u>2012</u> | <u>2012</u> | <u>2012</u> | <u>2012</u> | <u>2013</u> |
| Federal Funds Rate      | 0.09           | 0.09      | 0.08          | 0.09          | 0.10        | 0.07        | 0.09  | 0.09           | 0.1         | 0.1         | 0.1         | 0.1         | 0.1         | 0.2         |
| Prime Rate              | 3.25           | 3.25      | 3.25          | 3.25          | 3.25        | 3.25        | 3.25  | 3.25           | 3.3         | 3.3         | 3.3         | 3.3         | 3.3         | 3.3         |
| LIBOR, 3-mo.            | 0.35           | 0.34      | 0.33          | 0.33          | 0.29        | 0.25        | 0.29  | 0.29           | 0.3         | 0.3         | 0.3         | 0.4         | 0.4         | 0.5         |
| Commercial Paper, 1-mo. | 0.09           | 0.09      | 0.09          | 0.09          | 0.11        | 0.09        | 0.11  | 0.10           | 0.1         | 0.2         | 0.2         | 0.2         | 0.3         | 0.3         |
| Treasury bill, 3-mo.    | 0.01           | 0.01      | 0.02          | 0.02          | 0.02        | 0.04        | 0.04  | 0.02           | 0.1         | 0.1         | 0.1         | 0.1         | 0.1         | 0.2         |
| Treasury bill, 6-mo.    | 0.03           | 0.04      | 0.06          | 0.05          | 0.06        | 0.08        | 0.10  | 0.06           | 0.1         | 0.1         | 0.1         | 0.2         | 0.2         | 0.3         |
| Treasury bill, 1 yr.    | 0.09           | 0.10      | 0.12          | 0.10          | 0.11        | 0.19        | 0.18  | 0.13           | 0.1         | 0.2         | 0.2         | 0.3         | 0.4         | 0.5         |
| Treasury note, 2 yr.    | 0.19           | 0.20      | 0.20          | 0.20          | 0.23        | 0.41        | 0.41  | 0.28           | 0.3         | 0.3         | 0.4         | 0.5         | 0.6         | 0.8         |
| Treasury note, 5 yr.    | 0.89           | 0.91      | 0.87          | 0.93          | 1.02        | 1.54        | 1.58  | 1.15           | 1.0         | 1.1         | 1.3         | 1.4         | 1.6         | 1.7         |
| Treasury note, 10 yr.   | 1.99           | 2.03      | 1.99          | 2.17          | 2.30        | 3.00        | 3.00  | 2.43           | 2.1         | 2.3         | 2.4         | 2.6         | 2.7         | 2.8         |
| Treasury note, 30 yr.   | 3.23           | 3.32      | 3.30          | 3.52          | 3.65        | 4.27        | 4.23  | 3.73           | 3.3         | 3.4         | 3.5         | 3.7         | 3.8         | 3.9         |
| Corporate Aaa bond      | 4.10           | 4.14      | 4.11          | 4.34          | 4.37        | 4.93        | 4.99  | 4.47           | 4.2         | 4.2         | 4.3         | 4.4         | 4.5         | 4.6         |
| Corporate Baa bond      | 5.30           | 5.33      | 5.24          | 5.34          | 5.36        | 5.76        | 5.75  | 5.47           | 5.3         | 5.3         | 5.3         | 5.4         | 5.5         | 5.6         |
| State & Local bonds     | 3.85           | 4.07      | 4.05          | 4.14          | 4.02        | 4.52        | 4.51  | 4.18           | 3.9         | 3.9         | 4.0         | 4.1         | 4.2         | 4.2         |
| Home mortgage rate      | 4.09           | 4.09      | 4.12          | 4.22          | 4.27        | 4.55        | 4.51  | 4.31           | 4.1         | 4.1         | 4.2         | 4.3         | 4.5         | 4.6         |
|                         |                |           |               | Histor        | y           |             |       |                | Co          | onsensu     | ıs Fore     | casts-Q     | Juarte      | rly         |
|                         | 4Q             | 1Q        | 2Q            | 3Q            | 4Q          | 1Q          | 2Q    | $3Q^*$         | 4Q          | 1Q          | 2Q          | 3Q          | 4Q          | 1Q          |
| Key Assumptions         | 2009           | 2010      | 2010          | 2010          | 2010        | 2011        | 2011  | 2011           | 2011        | 2012        | 2012        | 2012        | 2012        | 2013        |
| Major Currency Index    | 72.8           | 74.8      | 77.6          | 75.9          | 73.0        | 71.9        | 69.6  | 69.5           | 70.5        | 70.6        | 70.6        | 70.6        | 70.6        | 71.2        |
| Real GDP                | 3.8            | 3.9       | 3.8           | 2.5           | 2.3         | 0.4         | 1.0   | 1.9            | 2.0         | 2.0         | 2.3         | 2.6         | 2.8         | 2.8         |
| GDP Price Index         | 1.1            | 1.5       | 1.5           | 1.4           | 1.9         | 2.5         | 2.4   | 2.0            | 1.8         | 2.0         | 1.9         | 1.9         | 1.9         | 2.1         |
| Consumer Price Index    | 2.7            | 1.3       | -0.5          | 1.4           | 2.6         | 5.2         | 4.1   | 2.7            | 2.1         | 2.1         | 2.1         | 2.3         | 2.2         | 2.3         |

Forecasts for interest rates and the Federal Reserve's Major Currency Index represent averages for the quarter. Forecasts for Real GDP, GDP Price Index and Consumer Price Index are seasonally-adjusted annual rates of change (saar). Individual panel members' forecasts are on pages 4 through 9. Historical data for interest rates except LIBOR is from Federal Reserve Release (FRSR) H.15. LIBOR quotes available from *The Wall Street Journal*. Interest rate definitions are the same as those in FRSR H.15. Treasury yields are reported on a constant maturity basis. Historical data for the Fed's Major Currency Index is from FRSR H.10 and G.5. Historical data for Real GDP and GDP Chained Price Index are from the Bureau of Economic Analysis (BEA). Consumer Price Index (CPI) history is from the Department of Labor's Bureau of Labor Statistics (BLS). *\*Interest rate data for 3Q 2011 Based on historical data through the week ended September 23rd*. *\*Data for 3Q 2011 Major Currency Index also is based on data through week ended September 23rd*. *\*Data for 3Q 2011 Real GDP, GDP Chained Price Index and Consumer Price Index are consensus forecasts based on a special question asked of the panelists this month (see page 14).* 



### U.S. 3-Mo, T-Bills & 10-Yr, T-Note Yield







### <u>Missouri-American Water Company</u> Derivation of Mean Equity Risk Premium Based on a Study <u>Using Holding Period Returns of Public Utilities</u>

| Line No.  |   | Over A Rated<br>Moody's Public Utility<br>Bonds - AUS<br>Consultants Study (1) |
|-----------|---|--|
| 1.        | Arithmetic Mean Holding Period Returns on<br>the Standard & Poor's Utility Index 1926-<br>2010 (2): | 10.69 %  |
| 2.        | Arithmetic Mean Yield on Moody's A Rated Public Utility Yields 1926-2010                            | (6.57)   |
| 3.        | Equity Risk Premium   | 4.12 %   |
| Notes: (1 | ) S&P Public Utility Index and Moody's Public Utility   | Bond Average Annual Yields   |

otes: (1) S&P Public Utility Index and Moody's Public Utility Bond Average Annual Yields 1928-2010, (AUS Consultants - Utility Services, 2011).

(2) Holding period returns are calculated based upon income received (dividends and interest) plus the relative change in the market value of a security over a one-year holding period.

### Missouri-American Water Company

### Correction to MIEC Witness Gorman's CAPM Analysis which Includes Consideration of Forward-Looking Market Returns and the Utilization of the Empirical Capital Asset Pricing Model (ECAPM)

|   | <u>1</u>                           | <u>2</u>                   | <u>3</u>              | <u>4</u>                             | <u>5</u>                  | <u>6</u>                                       |
|---|------------------------------------|----------------------------|-----------------------|--------------------------------------|---------------------------|--|
|   | Value Line<br>Adjusted<br>Beta (1) | Market Risk<br>Premium (2) | Risk-Free<br>Rate (3) | Traditional<br>CAPM Cost<br>Rate (4) | ECAPM<br>Cost Rate<br>(5) | Indicated<br>Common<br>Equity Cost<br>Rate (6) |
| MIEC Witness Gorman's Proxy Group<br>of Eight Water Companies | 0.74                               | 10.55 %                    | 3.90 %                | 11.71 %                              | 12.39 %                   | 12.05 %  |

Notes

(1) From Line 3 of Schedule MPG-16, page 1.

(2) Average of the Ibbotson long-term arithmetic mean risk premium of 6.70% and the projected 3-5 year return of the market as calculated by the 13 week average market appreciation potential published by Value Line ended October 21, 2011 minus MIEC Witness Gorman's projected risk-free rate. The average risk premium is 10.55%. ((6.70% + 14.39%) / 2 = 10.55%)

(3) From Line 1 of MPG-16, page 1.

(4) Calculated as shown on page 2 of Schedule PMA-12, note 3.

(5) Calculated as shown on page 2 of Schedule PMA-12, note 4.

(6) Average of Columns 4 and 5.

### Missouri-American Water Company Brief Summary of MIEC Witness Gorman's Corrected Common Equity Cost Rate

|     |  | MIEC Witness<br>Gorman's Proxy    |
|-----|--|-----------------------------------|
| No. | Principal Methods  | Group of Eight<br>Water Companies |
| 1.  | Discounted Cash Flow Model (DCF) (1)                                   | 11.93 %                           |
| 2.  | Risk Premium Model (RPM) (2)   | 10.61                             |
| 3.  | Capital Asset Pricing Model (CAPM) (3)                                 | 12.05                             |
| 4.  | Indicated Common Equity Cost Rate before Adjustment for Business Risks | 11.53 %                           |
| 5.  | Flotation Cost Adjustment (4)  | 0.16                              |
| 6.  | Financial Risk Adjustment (5)  | (0.21)                            |
| 7   | Business Risk Adjustment (6)   | 0.40                              |
| 8.  | Indicated Common Equity Cost Rate                                      | <u>    11.88 </u> %               |
| 9.  | Recommended Common Equity Cost Rate                                    | <u> </u>                          |

### Notes: (1) From Schedule PMA-29

- (2) From Schedule PMA-33
- (3) From Schedule PMA-34
- (4) From Ms. Ahern's electronic workpapers.
- (5) Financial risk adjustment to reflect the financial risk of the capital structure employed by Missouri-American Water Company for rate making purposes relative to the MIEC Witness Gorman's proxy group as detailed in Ms. Ahern's accompanying rebuttal testimony.
- (6) Business risk adjustment to reflect Missouri-American Water Company's greater business risk due to its small size relative to MIEC Witness Gorman's proxy group as detailed in Ms. Ahern's accompanying rebuttal testimony.

| Missouri-American Water Company | Correction of BJC Witness LaConte's DCF Analysis |
|---------------------------------|--|
|---------------------------------|--|

# Estimated ROE Single Stage DCF with Analyst Growth Rates

|                           | Close            |          |              |        | Analysts' Estim | ated Growth Ra | ites      |       |
|---------------------------|------------------|----------|--------------|--------|-----------------|----------------|-----------|-------|
|                           | 8/2011-11/2011   | 2011     | Div.         | Value  |                 | Yahoo          |           |       |
| <u>Utility</u>            | Avg. Stock Price | Dividend | <u>Yield</u> | Line   | Reuters         | Finance        | Average   | RoE   |
| American States Water     | 34.04            | 1.10     | 3.2%         | 5.50%  | 7.15%           | 7.15%          | 6.60%     | 9.8%  |
| American Water Works      | 29.63            | 0.91     | 3.1%         | 9.50%  | 11.09%          | 8.03%          | 9.54%     | 12.6% |
| Aqua America              | 21.59            | 0.62     | 2.9%         | 10.50% | 7.60%           | 6.67%          | 8.26%     | 11.1% |
| Artesian Resources Corp.  | 17.91            | 0.79     | 4.4%         | 3.60%  | 5.00%           | 4.00%          | 4.20%     | 8.6%  |
| California Water          | 17.68            | 0.62     | 3.5%         | 6.00%  | 7.00%           | 15.00%         | 9.33%     | 12.8% |
| Connecticut Water Service | 22.62            | 0.93     | 4.1%         | 4.00%  | 5.50%           | 3.00%          | 4.17%     | 8.3%  |
| Middlesex Water           | 17.52            | 0.73     | 4.2%         | 6.00%  | -5.00%          | 3.00%          | 4.50% (1) | 8.7%  |
| SJW Corporation           | 26.27            | 0.69     | 2.6%         | 7.50%  | n/a             | 14.00%         | 10.75%    | 13.4% |
| York Water Company        | 16.85            | 0.52     | 3.1%         | 6.00%  | 6.00%           | 6.00%          | 6.00%     | 9.1%  |
| Average                   |                  |          |              |        |                 |                | T         | 10.5% |
| Median                    |                  |          |              |        |                 |                | "         | 9.8%  |

Notes:

(1) The average projected EPS growth rate for Middlesex Water Company does not include the negative 5% reported by Reuters.

Sourceof Information: Schedule BSL-1.

### Missouri-American Water Company

Correction to BJC Witness LaConte's CAPM Analysis which Includes Consideration of Forward-Looking Market Returns and the Utilization of the Empirical Capital Asset Pricing Model (ECAPM)

|  | <u>1</u>                           | <u>2</u>                   | <u>3</u>              | <u>4</u>                             | <u>5</u>               | <u>6</u>                                       |
|--|------------------------------------|----------------------------|-----------------------|--------------------------------------|------------------------|--|
|  | Value Line<br>Adjusted<br>Beta (1) | Market Risk<br>Premium (2) | Risk-Free<br>Rate (3) | Traditional<br>CAPM Cost<br>Rate (4) | ECAPM Cost<br>Rate (5) | Indicated<br>Common<br>Equity Cost<br>Rate (6) |
| BJC Witness LaConte's Proxy Group of<br>Nine Water Companies | 0.72                               | 10.65 %                    | 4.38 %                | 12.05 %                              | 12.79 %                | 12.42 %  |

### Notes

(1) From Line 10 of Schedule BSL-4.

- (2) Average of the Ibbotson long-term arithmetic mean risk premium of 6.70% and the projected 3-5 year return of the market as calculated by the 13 week average market appreciation potential published by Value Line ended November 11, 2011 minus BJC Witness LaConte's projected risk-free rate. The average risk premium is 10.65%. ((6.70% + 14.60%) / 2 = 10.65%)
- (3) From Line 15 of Schedule BSL-4.

(4) Calculated as shown on page 2 of Schedule PMA-12, note 3.

(5) Calculated as shown on page 2 of Schedule PMA-12, note 4.

(6) Average of Columns 4 and 5.

### Missouri-American Water Company Brief Summary of Corrected Common Equity Cost Rate

|     |  | BJC Witness<br>LaConte's Proxy<br>Group of Nine |
|-----|--|---|
| No. | Principal Methods  | Water Companies                                 |
| 1.  | Discounted Cash Flow Model (DCF) (1)                                   | 10.49 %   |
| 3.  | Capital Asset Pricing Model (CAPM) (2)                                 | 12.42   |
| 4.  | Indicated Common Equity Cost Rate before Adjustment for Business Risks | 11.46 %   |
| 5.  | Flotation Cost Adjustment (3)  | 0.16  |
| 6.  | Financial Risk Adjustment (4)  | (0.21)  |
| 7   | Business Risk Adjustment (5)   | 0.40  |
| 8.  | Indicated Common Equity Cost Rate                                      | <u>    11.80  </u> %                            |

- Notes: (1) From Schedule PMA-36.
  - (2) From Schedule PMA-37.
  - (3) From Ms. Ahern's electronic workpapers.
  - (4) Financial risk adjustment to reflect the financial risk of the capital structure employed by Missouri-American Water Company for rate making purposes relative to the proxy group as detailed in Ms. Ahern's direct testimony.
  - (5) Business risk adjustment to reflect Missouri-American Water Company's greater business risk due to its small size relative to the proxy group as detailed in Ms. Ahern's direct testimony.

### <u>Missouri-American Water Company</u> Summary of Cost of Capital and Fair Rate of Return Based upon the Estimated Capital Structure at December 31, 2011

| Type of Capital  | Amounts (1)    | Ratios (1) | Cost Rate  | Weighted<br>Cost Rate |
|------------------|----------------|------------|------------|-----------------------|
| Long-Term Debt   | \$ 423,295,622 | 49.18%     | 6.28% (2)  | 3.09%                 |
| Preferred Equity | \$ 2,223,468   | 0.26%      | 9.35% (2)  | 0.02%                 |
| Common Equity    | \$ 435,252,472 | 50.57%     | 11.85% (3) | 5.99%                 |
| Total            | \$ 860,771,562 | 100.01% *  |            | 9.10%                 |

\* does not add due to rounding

### Notes:

(1) Company-Provided.

- (2) From pages 3 and 4 of this Schedule.
- (2) Based upon informed judgment from the entire study, the principal results of which are summarized on page 2 of this Schedule.

### Missouri-American Water Company Brief Summary of Common Equity Cost Rate

| No. | Principal Methods  | Proxy Group of<br>Nine Water<br>Companies |
|-----|--|---|
| 1.  | Discounted Cash Flow Model (DCF) (1)   | 11.45 %                                   |
| 2.  | Risk Premium Model (RPM) (2)   | 10.34                                     |
| 3.  | Capital Asset Pricing Model (CAPM) (3)   | 11.22                                     |
| 4.  | Market Models Applied to Comparable Risk, Non-Price<br>Regulated Companies (4) | 13.21                                     |
| 5.  | Indicated Common Equity Cost Rate before Adjustment for Business Risks         | 11.50 %                                   |
| 6.  | Financial Risk Adjustment (5)  | (0.21)                                    |
| 7.  | Flotation Cost Adjustment (6)  | 0.16                                      |
| 8.  | Business Risk Adjustment (7)   | 0.40                                      |
| 9.  | Indicated Common Equity Cost Rate  | <u> </u>                                  |
| 10. | Recommended Common Equity Cost Rate  | <u> </u>                                  |

Notes: (1) From page 5 of this Schedule.

- (2) From page 15 of this Schedule.
- (3) From page 21 of this Schedule.
- (4) From page 23 of this Schedule
- (5) Financial risk adjustment to reflect the financial risk of the capital structure employed by Missouri-American Water Company for rate making purposes relative to the proxy group as detailed in Ms. Ahern's direct testimony.
- (6) From page 33 of this Schedule
- (7) Business risk adjustment to reflect Missouri-American Water Company's greater business risk relative to the proxy group as detailed in Ms. Ahern's direct testimony.
Column No.

| 15        | Total   | Cost (4)        | \$ 627,000 | 259,995   | 900,872    | 12,981    | 233,831   | 14,889    | 955,490    | 1,748,325  | 30,061     | 2,694,971  | 6,825,391   | 4,602,554  | 2,091,639  | 1,262,500  | 28,500    | 66,216    | 65,076    | 26,003     | 19,734     | 31,753     | 1,289,961  | 2,007,035  | \$ 25,794,775  |
|-----------|---|-----------------|------------|-----------|------------|-----------|-----------|-----------|------------|------------|------------|------------|-------------|------------|------------|------------|-----------|-----------|-----------|------------|------------|------------|------------|------------|----------------|
| <u>14</u> | Annual<br>Interest                              | Expense (3)     | 623,200    | 257,400   | 892,500    | •         | 222,000   | •         | 913,500    | 1,698,905  |            | 2,644,080  | 6,790,790   | 4,585,000  | 2,058,320  | 1,262,500  |           | •         | •         | •          | •          | •          | 1,256,130  | 1,952,000  | 25,156,325     |
| <u>13</u> | Annual<br>Amortization<br>Debt                  | Expense (2)     | 3,800      | 2,595     | 8,372      | 12,981    | 11,831    | 14,889    | 41,990     | 49,420     | 30,061     | 50,891     | 34,601      | 17,554     | 33,319     |            | 28,500    | 66,216    | 65,076    | 26,003     | 19,734     | 31,753     | 33,831     | 55,035     | \$ 638,451 \$  |
| 12        | Pro Forma<br>Carrying<br>Value                  | at 12/31/10 (1) | 7,941,421  | 2,965,837 | 12,314,413 | (142,787) | 4,249,722 | (215,882) | 17,563,176 | 27,897,206 | 14,166,274 | 56,211,975 | 102,106,139 | 69,799,593 | 24,052,490 | 25,000,000 | (510,625) | (601,462) | (656,183) | (288,204)  | (264,765)  | (471,004)  | 24,083,061 | 38,095,226 | \$ 423,295,622 |
| 티         | Pro Forma<br>Unamortized<br>Issuance<br>Expense | at 12/31/10     | 58,579     | 34,163    | 185,587    | 142,787   | 190,278   | 215,882   | 706,824    | 897,794    | 608,726    | 1,268,025  | 893,861     | 200,407    | 896,843    |            | 510,625   | 601,462   | 656, 183  | 288,204    | 264,765    | 471,004    | 546,939    | 944,774    | 10,583,711     |
| <u>10</u> | Monthly<br>Amortization<br>Debt                 | Expense         | 317        | 216       | 698        | 1,082     | 986       | 1,241     | 3,499      | 4,118      | 2,505      | 4,241      | 2,883       | 1,463      | 2,777      |            | 2,375     | 5,518     | 5,423     | 2,167      | 1,645      | 2,646      | 2,819      | 4,586      | \$ 53,204 \$   |
| ര         | Pro Forma                                       | Adjustments     |            |           |            |           |           |           |            |            |            |            |             |            |            |            |           |           |           |            |            |            |            |            | '<br>\$        |
| 80        | Unamortized<br>Issuance<br>Expense              | at 12/31/10     | 62,379     | 36,757    | 193,959    | 155,768   | 202,109   | 230,771   | 748,814    | 947,213    | 638,786    | 1,318,915  | 928,462     | 217,961    | 930,162    |            | 539,125   | 667,678   | 721,259   | 314,207    | 284,499    | 502,757    | 580,770    | 999,810    | \$ 11,222,162  |
| 7         | Pro Forma<br>Amount<br>Outstanding              | at 12/31/10     | 8,000,000  | 3,000,000 | 12,500,000 |           | 4,440,000 |           | 18,270,000 | 28,795,000 | 14,775,000 | 57,480,000 | 103,000,000 | 70,000,000 | 24,949,333 | 25,000,000 |           |           | •         |            | •          |            | 24,630,000 | 39,040,000 | \$ 433,879,333 |
| 9         | Pro Forma                                       | Adjustments     |            |           |            |           |           |           |            |            |            |            |             |            |            |            |           |           |           |            |            |            |            |            | ,<br>\$        |
| 5         | Amount<br>Outstanding                           | at 12/31/10     | 8,000,000  | 3,000,000 | 12,500,000 | •         | 4,455,000 | •         | 18,405,000 | 28,820,000 | 14,810,000 | 57,480,000 | 103,000,000 | 70,000,000 | 24,951,000 |            | •         |           | •         |            |            |            | 24,660,000 | 39,195,000 | 409,276,000    |
| 4         | Principal                                       | Amount          | 8,000,000  | 3,000,000 | 12,500,000 | 4,950,000 | 4,500,000 | 6,000,000 | 19,000,000 | 29,000,000 | 15,000,000 | 57,480,000 | 103,000,000 | 70,000,000 | 25,000,000 | 25,000,000 | •         |           | •         | 15,000,000 | 12,000,000 | 19,900,000 | 25,000,000 | 40,000,000 | 494,330,000    |
| က၊        | Maturity  | Date            | 06/01/27   | 03/01/25  | 03/01/34   | 01/01/23  | 02/01/28  | 07/01/26  | 11/30/28   | 03/01/30   | 04/01/32   | 12/01/36   | 10/15/37    | 05/31/23   | 12/01/38   | 10/15/37   |           |           |           | 02/01/23   | 06/01/25   | 11/01/26   | 03/01/28   | 03/01/29   | \$             |
| 5         | Issuance  | Date            | 06/01/97   | 04/21/95  | 03/16/94   | 05/18/93  | 02/01/98  | 07/26/96  | 11/01/98   | 03/01/00   | 04/01/02   | 12/20/06   | 10/22/07    | 8/1/08 (5) | 02/04/09   | 11/21/11   |           |           |           | 02/01/93   | 06/01/95   | 11/01/96   | 03/01/98   | 03/01/99   |                |
| ۲I        | Coupon  | Rate            | 7.790%     | 8.580%    | 7.140%     | 5.500%    | 5.000%    | 5.850%    | 5.000%     | 5.900%     | 5.200%     | 4.600%     | 6.593%      | 6.550%     | 8.250%     | 5.050%     |           |           |           | 5.500%     | 5.700%     | 5.500%     | 5.100%     | 5.000%     |                |

6.09% (6)

Notes: (1) Column 7 - Column 11. (2) Column 10 × 12. (3) Column 10 × 12. (4) Column 14 + Column 15. (5) Original issue date was 5/15/08 and held by AWK awaiting Board Approval until 8/1/08. (5) Cost of Long-Term Debt = [Total Cost / Carrying Value].

Source of Information: Company-Provided

|        | <u>12</u>   | Total<br>nnnual<br><u>iost (2)</u>                      |                               | 207,972     | 207,972        | 9.35% (3) |
|--------|-------------|---|-------------------------------|-------------|----------------|-----------|
|        | 1           | Annual <i>A</i><br>Dividends <u>C</u>                   | <del>ب</del><br>ب             | 206,550     | \$ 206,550 \$  |           |
|        | <u>1</u>    | Annual<br>Amortization                                  | ب<br>ج                        | 1,422       | \$ 1,422       |           |
|        | ଚା          | Pro Forma<br>Carrying<br>Value<br>12/31/11 (1)          |                               | 2,223,468   | 2,223,468      |           |
|        | ωI          | o Forma<br>amortized<br>suance<br>xpense<br>12/31/11 at | <del>ن</del>                  | 26,532      | 26,532 \$      |           |
| No.    | 7           | Pr<br>Una<br>Is<br>EE<br>EE                             | <del>ن</del>                  | (1,422)     | (1,422) \$     |           |
| Column | Q           | nortized<br>Jance<br>Sense<br><u>231/11</u> <u>Adiu</u> | \$                            | 27,954      | 27,954 \$      |           |
|        | 10          | Unan<br>Junt Issi<br>anding Ext<br>31/11 at 13          | \$                            | 50,000      | 50,000 \$      |           |
|        |             | Amo<br>Outsta<br><u>eents at 12</u> /                   | <del>ب</del>                  | ),000) 2,2! | ),000) \$ 2,2! |           |
|        | 41          | t<br>ng<br><u>11 Adiustm</u>                            | \$ 000                        | 00 (250     | 00 \$ (250     |           |
|        | ເບ <u>ເ</u> | Amoun<br>Outstandi<br>at 12/31/                         | 36 \$ 96,C                    | 91 2,500,0  | \$ 2,596,0     |           |
|        | -           | lend Date<br><u>Ite</u> <u>Issuec</u>                   | 10/11/6                       | .18% 10/3/9 |                |           |
|        |             | <u>. Par Value</u><br><u>R</u> i                        | tive Preferred<br>100 Par 5.8 | ar Value 9  | referred Stock |           |
|        |             | Type  | Cumula<br>Stock \$<br>Value   | \$100 P     | Total P        |           |

Notes: (1) Column 5 - Column 8. (2) Column 10 + Column 11. (3) Total Cost of Preferred Stock = [Total Annual Cost/Carrying Value].

Source of Information: Company-Provided

Missouri-American Water Company Pro Forma Cost of Preferred Stock at December 31, 2011 Schedule PMA-39 Page 4 of 36

### Missouri-American Water Company Indicated Common Equity Cost Rate Using the Discounted Cash Flow Model for the Proxy Group of Nine Water Companies

|                                     | <u>1</u>                         | <u>2</u>   | <u>3</u>  | <u>4</u>   | <u>5</u>  | <u>6</u>  | <u>7</u>                          | <u>8</u>                                       |
|-------------------------------------|----------------------------------|--|---|--|---|---|-----------------------------------|--|
| Proxy Group of Nine Water Companies | Average<br>Dividend<br>Yield (1) | Value Line<br>Projected<br>Five Year<br>Growth in<br>EPS (2) | Reuters<br>Mean<br>Consensus<br>Projected<br>Five Year<br>Growth Rate<br>in EPS | Zack's Five<br>Year<br>Projected<br>Growth<br>Rate in<br>EPS | Yahoo!<br>Finance<br>Projected<br>Five Year<br>Growth in<br>EPS | Average<br>Projected<br>Five Year<br>Growth in<br>EPS (3) | Adjusted<br>Dividend<br>Yield (4) | Indicated<br>Common<br>Equity Cost<br>Rate (5) |
| American States Water Co.           | 3.22 %                           | 5.50 %   | 7.50 %  | 12.00 %  | 7.57 %  | 8.14 %  | 3.35 %                            | 11.49 %  |
| American Water Works Co., Inc.      | 3.00                             | 9.50   | 11.00   | 7.80   | 8.58  | 9.22  | 3.14                              | 12.36  |
| Aqua America, Inc.                  | 2.86                             | 10.50  | 7.80  | 8.30   | 7.28  | 8.47  | 2.98                              | 11.45  |
| Artesian Resources Corp.            | 4.18                             | NA   | 4.90  | -  | 4.40  | 4.65  | 4.28                              | 8.93   |
| California water Service Group      | 3.38                             | 6.00   | 9.80  | 10.00  | 12.40   | 9.55  | 3.54                              | 13.09  |
| Connecticut Water Service, Inc.     | 3.52                             | NA   | 5.70  | -  | 4.55  | 5.13  | 3.61                              | 8.74   |
| Middlesex Water Company             | 4.02                             | 6.00   | (1.00)  | -  | 2.70  | 2.57  | 4.07                              | 6.64   |
| SJW Corporation                     | 2.92                             | 7.50   | N/A   | -  | 14.00   | 10.75   | 3.08                              | 13.83  |
| York water Company                  | 3.09                             | NA   | 5.60  | -  | 4.90  | 5.25  | 3.17                              | 8.42   |
| Average                             |                                  |  |   |  |   |   |                                   | <u>10.55</u> %                                 |
| Median                              |                                  |  |   |  |   |   |                                   | 11.45 %  |

NA= Not Available

NMF = Not Meaningful Figure

Notes:

- (1) Indicated dividend at 1/3/2011 divided by the average closing price of the last 60 trading days ending 12/30/2011 for each company.
- (2) From pages 6 through 14 of this Schedule.
- (3) Average of columns 2 through 5 excluding negative growth rates.
- (4) This reflects a growth rate component equal to one-half the conclusion of growth rate (from column 6) x column 1 to reflect the periodic payment of dividends (Gordon Model) as opposed to the continuous payment. Thus, for American States Water Co., 3.22% x (1+(1/2 x 8.14%)) = 3.35%.
- (5) Column 6 + column 7.

Source of Information:

Value Line Investment Survey: October 21, 2011 www.reuters.com Downloaded on 01/03/2012 www.zacks.com Downloaded on 01/03/2012 www.yahoo.com Downloaded on 01/03/2012

| AME   | R.  | STAT             | ES                  | WAT                    | ERN                          | YSE-A                   | WR PI                  | ECENT<br>Rice   | 34.49         | P/E<br>RATI    | 16.                   | 0 (Traili<br>Medi | ng: 14.9<br>an: 22.0) | RELATIVE<br>P/E RATIO | 1.1             | 7 DIV'D              | 3.2                | %                     |                    | =                  |                  |
|---|---|------------------|---------------------|------------------------|------------------------------|-------------------------|------------------------|-----------------|---------------|----------------|-----------------------|-------------------|-----------------------|-----------------------|-----------------|----------------------|--------------------|-----------------------|--------------------|--------------------|------------------|
| TIMELIN   | ESS 1   | Raised 9/1       | 6/11                | High:                  | 25.3                         | 26.4                    | 29.0                   | 29.0            | 26.8          | 34.6           | 43.8                  | 46.1              | 42.0                  | 38.8                  | 39.6<br>31.2    | 36.4                 |                    |                       | Target             | Price              | Range            |
| SAFETY  | 3   | B New 2/4/0      | 0                   | LEGEN                  | NDS<br>25 x Divide           | nds o sh                | 10.0                   | 21.0            | 20.0          | 24.0           | 00.0                  | 00.0              | 21.0                  | 10.0                  | 0112            | 00.0                 |                    |                       | 2014               | 2015               | 128              |
| TECHNIC   | CAL 3   | B Lowered 1      | 0/21/11             | dh<br>Re               | vided by In<br>elative Price | terest Rate<br>Strength |                        |                 |               |                |                       |                   |                       |                       |                 |                      |                    |                       |                    |                    | 120              |
| BETA .75  | (1.00 =   | = Market)        | NS                  | 3-for-2 sp<br>Options: | Hit 6/02<br>Yes              | ala racara              | iana                   |                 |               |                |                       |                   |                       |                       | 12.0            |                      |                    |                       |                    |                    | -80<br>-64       |
| P   | rice  | An               | n'i Total<br>Return | Snaded                 | areas muic                   | ale recess              | IONS                   |                 |               |                |                       | 1.11              | (                     |                       |                 |                      |                    |                       |                    |                    | 48               |
| High  | 55 {  | +60%)            | 15%                 |                        |                              |                         |                        |                 |               | .tullut        | HI Hunth              | manter            | Illun III             | 14,000,0              | Tilli           | រិកិតិព្             |                    |                       |                    |                    | 32               |
| Insider   | Decis   | ions             | 170                 | <sub>-11</sub> #       |                              | 1                       | - UHH                  | 111111111       | Tin Internet  | 4,4            |                       |                   | In                    |                       |                 |                      |                    |                       |                    |                    | -24              |
| to Buy  | N D J<br>0 0 0  | F M A<br>0 0 0   | M J J<br>0 0 0      | 141                    | Harthe                       | en li                   |                        |                 |               |                |                       |                   |                       |                       |                 |                      |                    |                       |                    |                    | - 16             |
| Options<br>to Sell  |   | 0 0 0 0 0 0      | 0 0 0 0 0           |                        |                              |                         | ·                      | -               |               |                |                       |                   |                       |                       |                 |                      |                    | % TO                  | T. RETUR           | N 9/11             | - 12             |
| Institut  | ional I   | Decision         | S 202011            |                        |                              |                         |                        |                 | ·····         |                |                       | hind              | 1. 111                |                       | · · · · ·       | li                   |                    |                       | THIS V<br>STOCK    | INDEX              |                  |
| to Buy  | 59  | 51               | 48                  | Shares                 | 8 -                          |                         | 1                      | -               | 1.1.104       | unio           |                       |                   |                       |                       |                 | milli                |                    | 1 yr.<br>3 yr.        | -2.2<br>-3.6       | -4.8<br>25.0       | E                |
| Hid's(000)  | 11086   | 11214            | 11377               | traded                 | 4 -                          |                         |                        |                 |               | 2005           | 2006                  | 2007              | 2008                  | 2000                  | 2010            | 2011                 | 2012               | 5 yr.                 | 2.2                | 16.6               | 1.16             |
| 11 03   | 11 37   | 11 44            | 11 02               | 12.91                  | 12.17                        | 13.06                   | 13.78                  | 13.98           | 13.61         | 14.06          | 15.76                 | 17.49             | 18.42                 | 19.48                 | 21.41           | 22.10                | 22.85              | Revenue               | s per sh           | JD. LLO            | 26.25            |
| 1.75  | 1.75  | 1.85             | 2.04                | 2.26                   | 2.20                         | 2.53                    | 2.54                   | 2.08            | 2.23          | 2.64           | 2.89                  | 3.31              | 3.37                  | 3.40                  | 4.23            | 4.20                 | 4.40               | "Cash F               | low" per s         | sh                 | 4.75             |
| 1.03  | 1.13  | 1.04             | 1.08                | 1.19                   | 1.28                         | 1.35                    | 1.34                   | .78             | 1.05          | 1.32           | 1.33                  | 1.62              | 1.55                  | 1.62                  | 2.22            | 2.10                 | 2.20               | Earnings              | s per sh 4         | h Bm               | 2.50             |
| 2.19  | 2.40  | 2.58             | 3.11                | 4.30                   | 3.03                         | 3.18                    | 2.68                   | 3.76            | 5.03          | 4.24           | 3.91                  | 2.89              | 4.45                  | 4.18                  | 4.24            | 4.00                 | 4.15               | Cap'l Sp              | ending p           | ersh               | 4.75             |
| 10.29   | 11.01   | 11.24            | 11.48               | 11.82                  | 12.74                        | 13.22                   | 14.05                  | 13.97           | 15.01         | 15.72          | 16.64                 | 17.53             | 17.95                 | 19.39                 | 20.26           | 20.55                | 20.80              | Book Va               | lue per sh         | 1<br>1             | 20.00            |
| 11.77   | 11.6         12.6         14.5         15.5         17.1         15.9         16.7         18.3         31.9         23.2         21.9         27.7         24.0         22.6         21.2         15.7         Bold right same         Avg and right same |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       |                 | 19.0                 |                    |                       |                    |                    |                  |
| .78   | 78         79         .84         .81         .97         1.03         .86         1.00         1.82         1.23         1.17         1.50         1.27         1.36         1.41         1.01         Value         Value         Line         Relative P/E Ratio         1.22         2.7%           6.7%         5.8%         5.5%         5.0%         4.2%         3.9%         3.6%         3.5%         3.6%         3.1%         2.5%         2.9%         2.9%         3.0%         estimates         Avg Ann'l Div'd Yield         2.7%           CAPITAL STRUCTURE as of 6/30/11         197.5         209.2         212.7         228.0         236.2         268.6         301.4         318.7         361.0         308.9         420         440         Pavanue (Smith)         52   |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       |                 | 1.25                 |                    |                       |                    |                    |                  |
| b.7%         b.5%         b.5%         b.0%         4.2%         3.9%         3.6%         3.5%         3.6%         3.1%         2.5%         2.9%         2.9%         3.0%         Avg Ann'I Div'd Yield         2.7%           CAPITAL STRUCTURE as of 6/30/11         197.5         209.2         212.7         228.0         236.2         268.6         301.4         318.7         361.0         398.9         420         440         Revenues (\$mill)         52           Total Debt \$352.8 mill.         Due in 5 Yrs \$288.5 mill.         20.4         20.3         11.9         16.5         22.5         23.1         28.0         26.8         29.5         41.4         40.0         43.0         Net Profit (\$mill)         50.   |   |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       | 2.7%            |                      |                    |                       |                    |                    |                  |
| CAPITAL STRUCTURE as of 6/30/11         19/.5         209.2         212.7         228.0         236.2         268.6         301.4         318.7         361.0         398.9         420         440         Revenues (\$mill)         525           Total Debt \$352.8 mill.         Due in 5 Yrs \$288.5 mill.         20.4         20.3         11.9         16.5         22.5         23.1         28.0         26.8         29.5         41.4         40.0         43.0         Net Profit (\$mill)         50.0           LT Debt \$340.4 mill.         LT Interest \$27.6 mill.         38.9%         43.5%         37.4%         47.0%         40.5%         37.8%         38.9%         43.0%         42.0%         Income Tax Rate         60.0%           (LT interest earned: 4.2x: total interest         50.7%         37.4%         47.0%         40.5%         37.8%         38.9%         43.0%         42.0%         Income Tax Rate         60.0% |   |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       | 50.0            |                      |                    |                       |                    |                    |                  |
| LT Debt   | LT Debt \$340.4 mill.         LT Interest \$27.6 mill.         43.0%         38.9%         43.5%         37.4%         47.0%         40.5%         42.6%         37.8%         38.9%         43.2%         Income Tar Rate         40.0%           (LT Interest earned: 4.2x: total interest coverage: 4.0x)         (46% of Cap'l)             12.2%         8.5%         6.9%         3.2%         5.8%         5.0%         AFUDC % to Net Profit         5.0%           54.9%         52.0%         52.0%         47.7%         50.4%         48.6%         46.9%         44.3%         46.0%         46.5%         Long-Term Debt Ratio         47.0%  |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       |                 | 40.0%                |                    |                       |                    |                    |                  |
| Linterest earned: 4.2x: total interest coverage: 4.0x)         (46% of Cap'l)         -         -         -         -         -         -         12.2%         8.5%         6.3%         3.2%         5.8%         5.0%         AFUDC % to Net Profit         5.0%           54.9%         52.0%         52.0%         47.7%         50.4%         48.6%         46.9%         46.2%         45.9%         44.3%         46.0%         46.5%         Long-Term Debt Ratio         47.0%           Leases, Uncapitalized: Annual rentals \$3.3 mill.         44.7%         48.0%         52.3%         49.6%         51.4%         53.1%         53.8%         54.1%         55.7%         54.0%         53.5%         Common Equity Ratio         53.0%  |   |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       | 5.0%            |                      |                    |                       |                    |                    |                  |
| Consistent Annual rentals \$3.3 mill.         54.9%         52.0%         52.0%         47.7%         50.4%         48.6%         46.9%         46.2%         45.9%         44.3%         46.0%         46.5%         Long-Term Debt Ratio         47.0           Leases, Uncapitalized: Annual rentals \$3.3 mill.         44.7%         48.0%         52.3%         49.6%         51.4%         53.1%         55.7%         54.0%         46.5%         Common Equity Ratio         53.0           Penetian Ascele 12/10 \$00.2 mill         447.6         444.4         442.3         480.4         532.5         551.6         569.4         577.0         665.0         677.4         690         705         Total Capital (\$mill)         75  |   |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       | 53.0%           |                      |                    |                       |                    |                    |                  |
| Leases, Uncapitalized: Annual rentals \$3.3 mill.         44.7%         48.0%         52.3%         49.6%         51.4%         53.1%         53.8%         54.1%         55.7%         54.0%         53.5%         Common Equity Ratio         53.1%           Pension Assets-12/10 \$90.2 mill.         447.6         444.4         442.3         480.4         532.5         551.6         569.4         577.0         665.0         677.4         690         705         Total Capital (\$mill)         77           Oblin         511.8         mill         53.8         563.3         602.3         664.2         713.2         750.6         776.4         825.3         866.4         855.0         890         930         Net Plant (\$mill)         16   |   |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       | 750             |                      |                    |                       |                    |                    |                  |
| Pension Assets-12/10 \$90.2 mill.<br>Oblig. \$118.8 mill.         447.6         444.4         442.3         480.4         532.5         551.6         569.4         577.0         665.0         677.4         690         705         Total Capital (\$mill)         77           Oblig. \$118.8 mill.         563.3         602.3         664.2         713.2         750.6         776.4         825.3         866.4         855.0         890         930         Net Plant (\$mill)         10           Pfd Stock None.         6.1%         6.5%         4.6%         5.2%         5.4%         6.0%         6.7%         6.4%         5.9%         7.6%         7.0%         7.5%         Return on Total Cap'l         8.0  |   |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       | 8.0%            |                      |                    |                       |                    |                    |                  |
| Oblig. \$118.8 mill.         6.1%         6.5%         4.6%         5.2%         5.4%         6.0%         6.7%         6.4%         5.9%         7.6%         7.0%         7.5%         Return on Total Cap'l         8.0           Pfd Stock None.         10.1%         9.5%         5.6%         6.6%         8.1%         9.3%         8.6%         8.2%         11.0%         10.5%         11.0%         Return on Shr. Equity         12.0           0.1%         0.5%         5.6%         6.6%         8.5%         8.1%         9.3%         8.6%         8.2%         11.0%         10.5%         11.0%         Return on Shr. Equity         12.0  |   |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       | 12.0%           |                      |                    |                       |                    |                    |                  |
| Common Stock 18,684,812 shs.         10.1%         9.5%         5.6%         6.6%         8.5%         8.1%         9.3%         8.6%         8.2%         11.0%         10.5%         11.0%         Return on Shr. Equity         12.0           common Stock 18,684,812 shs.         10.1%         9.5%         5.6%         6.6%         8.5%         8.1%         9.3%         8.6%         8.2%         11.0%         10.5%         10.0%         Return on Shr. Equity         12.0           as of 8/5/11         3.6%         3.3%         NMF         1.0%         2.8%         2.7%         3.9%         3.1%         3.2%         5.8%         5.0%         6.6%   |   |                  |                     |                        |                              |                         |                        |                 |               |                |                       |                   |                       |                       | 6.0%            |                      |                    |                       |                    |                    |                  |
| MARKE   | T CAP:  | \$650 mill       | ion (Sm             | all Cap)               | 0100144                      | 65%                     | 65%                    | 113%            | 84%           | 67%            | 67%                   | 58%               | 64%                   | 61%                   | 47%             | 52%                  | 52%                | All Div'd             | s to Net P         | Prof               | 51%              |
| (\$ML   | L)  | SITION           | 2009                | 2010                   | 6/30/11                      | BUSIN                   | ESS: An                | nerican         | States Wa     | ater Co        | operate               | s as a            | holding               | ers in t              | he city o       | of Big Be            | ar Lake            | and in a              | areas of           | San Ber            | mardino          |
| Other   | ssets   | 3                | 94.3                | 200.8                  | 160.9                        | Compa                   | ny. Thro<br>iny, it su | oplies wa       | iter to mo    | re than        | 250,000               | custome           | rs in 75              | ployees.              | Officer         | s & dire             | ctors ow           | n 2.9%                | of comm            | ion stoci          | k (4/11          |
| Accts P   | Assets<br>ayable  | 1                | 96.0<br>33.9        | 205.0 36.2             | 172.2 58.0                   | commu                   | inities in             | 10 cou          | nties. Se     | rvice ar       | eas inclu<br>ange Cou | ude the           | greater               | Proxy).<br>Sprowls    | Chairma         | an: Lloye<br>A Addr: | d Ross.<br>630 Eas | Preside<br>t Foothill | nt & Cl<br>Bouleva | EO: Rol<br>urd San | bert J.<br>Dimas |
| Debt Du<br>Other  | Je Je   |                  | 18.1<br>47.7        | 61.4<br>81.2           | 12.4<br>54.2                 | pany a                  | lso provi              | des elect       | ric utility s | services       | to nearly             | 23,250            | custom-               | CA 917                | 73. Tel: 9      | 909-394-3            | 600. Inte          | ernet: ww             | w.aswate           | r.com.             |                  |
| Current   | Liab.   | 3                | 99.7<br>52%         | 178.8                  | 124.6                        | Ame                     | ericar                 | 1 Sta           | tes W         | ater           | does                  | not               | ap-                   | those                 | outs            | ide th               | ne con             | mpany                 | are                | strin              | gent             |
| ANNUA   | L RATE  | S Past           | Pa                  | st Est'o               | 1 '08-'10                    | Wat                     | r to b<br>er Co        | e mis           | far. T        | the (<br>he w  | <b>hapa</b><br>ater   | utility           | far                   | and                   | capı<br>tainir  | ital-in              | tensiv<br>id di    | stribu                | ting               | wate:              | r is             |
| of change<br>Revenue  | e (persh<br>es  | ) 10 Yrs.<br>5.0 | 5Y<br>% 7           | rs. to<br>.5%          | '14-'16<br>4.5%              | surp                    | assed                  | expec           | tation        | s in t         | he Ju                 | ine pe            | riod,                 | high,                 | as ol           | d, dila              | apidat             | ed, sy                | stems              | s, in s            | some             |
| "Cash I<br>Earning  | low"<br>s   | 5.5°<br>4.5      | % 9.<br>% 11        | .5%                    | 4.5%<br>5.5%                 | the tow.                | Indee                  | uarter          | e water       | r util         | ity po                | sted e            | ry in<br>earn-        | cases                 | , requestly.    | uire a<br>and        | will (             | on. T                 | ne inv             | vestm<br>ue to     | ents             |
| Dividen<br>Book V   | ds<br>alue  | 2.0              | % 2<br>% 5          | .5%<br>.0%             | 4.0%<br>2.0%                 | ings                    | of \$0                 | .68 a           | share,        | 45%            | bette                 | r than            | n the                 | away                  | at pr           | ofit m               | argin              | s.                    | and a              | L 43               | <i>C</i>         |
| Cal-  | QUA   | RTERLY RE        | VENUES              | (\$ mill.)             | Full                         | year                    | befor                  | e, on<br>of the | 14%           | reven          | ue gr<br>associ       | owth.<br>ated     | with                  | Time                  | stoc            | k is<br>s. AV        | VR wi              | ed 1<br>ill like      | elv co             | nest)              | e to             |
| endar   | Mar.31  | Jun. 30          | Sep. 30             | Dec. 31                | Year                         | this                    | busin                  | less p          | rovide        | dal            | boost,                | outw              | eigh-                 | do re                 | lative          | ely we               | ll wh              | ile th                | e broa             | ader r             | mar-             |
| 2008  | 68.9<br>79.6  | 80.3<br>93.6     | 85.3                | 84.2                   | 318.7                        | Rate                    | any re                 | eases           | e loss<br>mea | suffe          | red 11                | n the             | sale.                 | ket r                 | eman            | to $12$              | flux a             | as we<br>hs.          | expe               | ct for             | the              |
| 2010  | 88.4  | 95.5             | 111.3               | 103.7                  | 398.9                        | play                    | a role                 | e, as d         | lid bus       | iness          | gene                  | rated             | from                  | That                  | said            | l, it                | loses              | sign                  | ificar             | at lu              | ster             |
| 2012  | 98.0  | 115              | 125                 | 102                    | 440                          | The The                 | milita                 | ry ver          | ated a        | arm            | is he                 | comi              | าศล                   | for                   | n we<br>a be    | look                 | furth              | er ou<br>omic         | t and              | acco               | The              |
| Cal-  | E   | ARNINGS P        | ER SHAR             | E A                    | Full                         | bigg                    | ger pi                 | ece o           | f the         | puzz           | le. Ma                | inager            | nent                  | costs                 | asso            | ciated               | with               | doin                  | g bus              | iness              | will             |
| 2008  | .30   | .53              | .26                 | .43                    | 1.55                         | has                     | been                   | aggre           | essivel       | y tai          | geting                | g mil             | itary                 | proba                 | ably a<br>while | the i                | s han              | g over                | r the              | comp               | any,<br>nice     |
| 2009  | .28   | .64              | .52                 | .18                    | 1.62                         | mak                     | ing ir                 | roads           | in le         | ss sa          | inction               | ned a             | reas.                 | there                 | are             | more-                | appea              | ling o                | livide             | nd-pa              | ying             |
| 2010  | .45   | .68              | .02                 | .35                    | 2.20                         | This                    | busi                   | ness i          | s expe        | cted           | make                  | more              | of a                  | stock                 | s out           | there                | . Clou             | ding i                | matter             | rs slig<br>heet    | Al-              |
| 2012  | .42   | .62              | .76                 | .40                    | 2.20                         | are                     | finaliz                | zed. V          | Ve wou        | ild e          | xpect                 | expar             | sion                  | thoug                 | gh a            | rece                 | nt d               | ebt o                 | offerin            | g he               | lped             |
| Cal-<br>endar   | Mar.31  | Jun.30           | Sep.30              | Dec.31                 | Full<br>Year                 | here                    | to be                  | a cat           | alyst.        | 10-            | rolu                  | norm              | aine                  | reple                 | nish t          | the ca               | sh cot             | ffers a               | i bit,             | additi             | ional            |
| 2007  | .235  | .235             | .235                | .250                   | .96                          | hea                     | vily r                 | egula           | ited, a       | and t          | heref                 | ore la            | acks                  | need                  | ed lo           | oking                | ahea               | id. As                | s a r              | esult,             | we               |
| 2008  | .250  | .250             | .250                | .250                   | 1.00                         | sign                    | ifica                  | at ea           | rning         | s po           | tentia                | al in             | our                   | think                 | that            | the c                | urrent             | t payo                | ut rat             | io ma              | iy be            |
| 2010  | .260  | .260             | .260                | .260                   | 1.04                         | men                     | t is i                 | mprov           | ving, t       | e reg<br>he gi | uidelin               | y env             | et by                 | Andr                  | e J. C          | lostan.              | za                 |                       | Octobe             | er 21,             | 2011             |
| (A) Prim  | ary ear   | nings. Ex        | cludes r            | nonrecurr              | ing add                      | due to ro               | ounding.               |                 |               |                | (C) In m              | illions, ad       | justed fo             | r split.              |                 | Cor                  | npany's            | Financia              | I Streng           | th                 | B++              |
| gains/(los  | ses): (   | 14, 14¢; 0       | 5, 25¢;             | UD, 6¢; 1              | U8, (B)                      | Dividend                | s nistoric             | ally paid       | in early I    | warch,         |                       |                   |                       |                       |                 | Sto                  | UK S Pric          | e stabili             | 19                 |                    | 60               |

gains/(losses): 04, 14¢; 05, 25¢; 05, 05¢; 05, 0¢; 10¢, 10¢) Dividends historically paid in early March, (27¢); 10, (44¢) 11, 20¢. Next earnings report June, September, and December. = Div'd rein-due early November. Quarterly egs. may not vestment plan available. © 2011, Value Line Publishing LLC. All rights reserved. Factual material is obtained from sources believed to be reliable and is provided without warranties of any kind. ThE PUBLISHER IS NOT RESPONSIBLE FOR ANY ERRORS OR OMISSIONS HEREIN. This publication is strictly for subscriber's own, non-commercial, internal use. No pair of it may be reproduced, resold, stored or transmitted in any printed, electronic or other form, or used for generaling or marketing any printed or electronic publication, service or product. To subscribe call 1-800-833-0046.

| Company's Financial Strength | 8++ |
|------------------------------|-----|
| Stock's Price Stability      | 85  |
| Price Growth Persistence     | 60  |
| Earnings Predictability      | 85  |

| AM                     | ERIC  | CAN                      | WAT                  | <b>TER</b>               | VYSE-A               | wĸ                    | R                       | ecent<br>Rice          | 30.0                   | 8 P/E<br>RATIO         | 16.                     | 9 (Traili<br>Media   | ng: 18.3<br>an: NMF) | RELATIV           | 5 <b>1.2</b>           | 3 DIV'D                | 3.1                      | %                       |                           | E                    |                    |
|------------------------|---|--------------------------|----------------------|--------------------------|----------------------|-----------------------|-------------------------|------------------------|------------------------|------------------------|-------------------------|----------------------|----------------------|-------------------|------------------------|------------------------|--------------------------|-------------------------|---------------------------|----------------------|--------------------|
| TIMELIN                | ess 1   | Raised 10                | 0/7/11               | <u> </u>                 |                      |                       |                         |                        |                        |                        |                         | High:                | 23.7                 | 23.0              | 25.8<br>19.4           | 31.0                   |                          |                         | Target                    | Price                | Range              |
| SAFETY                 | 3   | New 7/25                 | /08                  | LEGE                     | NDS<br>elative Price | e Strength            |                         |                        |                        |                        |                         | LOW.                 | 10.0                 | 10.2              | 10.4                   | LUIL                   |                          |                         | 2014                      | 2015                 | 2016               |
| TECHNIC                | CAL 3   | Lowered I                | 8/26/11              | Options:<br>Shaded       | Yes<br>areas indic   | cate recess           | ions                    |                        |                        |                        |                         |                      |                      |                   |                        |                        |                          |                         |                           |                      | 1 80<br>60         |
| 2014                   | 4-16 PR   | OJECTIC                  | NS                   |                          |                      |                       |                         |                        |                        |                        |                         |                      |                      |                   |                        |                        |                          |                         |                           |                      | -50<br>-40         |
| F                      | rice  | Ar<br>Gain               | nn'l Total<br>Return |                          |                      |                       |                         |                        |                        |                        |                         |                      |                      |                   |                        |                        |                          |                         |                           |                      | -30                |
| High<br>Low            | 50 (+<br>35 (+  | 65%)                     | 16%<br>7%            |                          |                      | EV.Bar                |                         |                        |                        |                        |                         |                      |                      | Hunned            | philling,              |                        |                          |                         |                           |                      | 20                 |
| Insider                | Decisi  | ions<br>FMA              | маа                  |                          |                      |                       |                         |                        | -                      |                        |                         |                      | 1.0                  | 1.0               |                        |                        |                          |                         |                           |                      | - 15               |
| to Buy<br>Options      | 0 0 0 0 0 0   | 1 0 0 1 1 0              | 0 0 0 0 0 0          |                          |                      |                       | -                       |                        |                        |                        |                         |                      | •                    | ••••              |                        |                        |                          | -                       |                           | -                    | 10                 |
| to Sell<br>Institut    | 0 0 0<br>lional [   | 0 1 0<br>Decision        | 000                  |                          |                      |                       |                         |                        |                        |                        |                         |                      |                      | ··                |                        |                        |                          | % TO                    | T. RETUR                  | IN 9/11              | -1.5               |
| to Buy                 | 402010<br>145   | 102011<br>168            | 202011<br>138        | Percen                   | t 21 -               |                       |                         |                        | -                      |                        |                         |                      |                      |                   |                        |                        |                          | 1 yr.                   | STOCK<br>34.1             | INDEX<br>-4.8        | E                  |
| to Sell<br>Hid's(000)  | 119<br>145430   | 117<br>145932            | 145<br>145042        | traded                   | 7 -                  |                       |                         |                        |                        |                        |                         |                      |                      | hulli             |                        |                        |                          | 3 yr.<br>5 yr.          | 57.4                      | 16.6                 | F.                 |
| 1995                   | 1996  | 1997                     | 1998                 | 1999                     | 2000                 | 2001                  | 2002                    | 2003                   | 2004                   | 2005                   | 12.09                   | 12.04                | 2008                 | 12 09             | 2010                   | 2011                   | 2012                     | © VAL                   | UE LINE P                 | UB. LLC              | 14-16              |
|                        |   |                          |                      |                          |                      |                       |                         |                        |                        |                        | .65                     | d.47                 | 2.87                 | 2.89              | 3.56                   | 3.70                   | 3.90                     | "Cash F                 | low" per                  | sh                   | 4.20               |
|                        |   |                          |                      |                          |                      |                       |                         |                        |                        |                        | d.97                    | d2.14                | 1.10                 | 1.25              | 1.53                   | 1.75                   | 1.90                     | Earning<br>Div'd De     | s per sh 4                | A<br>hB∎             | 2.25               |
|                        |   |                          |                      |                          |                      |                       |                         |                        |                        |                        | 4.31                    | 4.74                 | 6.31                 | 4.50              | 4.38                   | 4.35                   | 4.50                     | Cap'l Sp                | ending p                  | ersh                 | 4.20               |
|                        |   |                          |                      |                          |                      |                       |                         |                        |                        |                        | 23.86                   | 28.39                | 25.64                | 22.91             | 23.59                  | 24.50                  | 24.05                    | Book Va                 | ilue per sl<br>n Shs Ou   | tst'a C              | 24.05              |
|                        |   |                          |                      |                          |                      |                       |                         |                        |                        |                        |                         |                      | 18.9                 | 15.6              | 14.6                   | Bold fig               | ures are                 | Avg Anr                 | 'I P/E Rat                | lio                  | 19.0               |
|                        |   |                          |                      |                          |                      |                       |                         |                        |                        |                        |                         |                      | 1.14                 | 1.04              | 3.8%                   | estin                  | ates                     | Avg Anr                 | n'l Div'd Y               | ield                 | 1.25               |
| CAPITA                 | L STRU  | CTURE a                  | s of 6/3             | 0/11                     |                      |                       |                         |                        |                        |                        | 2093.1                  | 2214.2               | 2336.9               | 2440.7            | 2710.7                 | 2820                   | 3000                     | Revenue                 | es (\$mill)               |                      | 3500               |
| LT Debt                | ar Dept \$5526.2 6 mill. DUE In 5 Yrs \$252.9 mill d155.8 d342.3 187.2 Debt \$5362.6 mill. LT Interest \$315.0 mill 37.4% tal interest coverage: 2.6x) (56% of Cap'l) 37.4% |                          |                      |                          |                      |                       |                         |                        |                        |                        |                         |                      |                      |                   |                        | 310                    | 340                      | Net Prof                | Tax Rate                  |                      | 415                |
| (Total in              | terest co   | overage: 2               | 2.6x)                | (56% c                   | of Cap'l)            |                       |                         |                        |                        |                        |                         |                      |                      | 12.5%             | 10.0%                  | 10.0%                  | 10.0%                    | AFUDC                   | % to Net I                | Profit               | 15.0%              |
| Leases,<br>Pensior     | Uncapi<br>Assets  | talized: /               | Annual re            | entals \$25              | .7 mill.             |                       |                         |                        |                        |                        | 56.1%                   | 49.1%                | 53.1%                | 56.9%             | 56.8%                  | 56.5%                  | 56.5%                    | Long-Te<br>Commo        | rm Debt F<br>n Equity f   | Ratio<br>Ratio       | 56.5%<br>43.5%     |
| Pfd Sto                | ck \$27.8   | (mill F                  | Oblig. \$1           | 285.5 mil                | Ι.                   |                       |                         |                        |                        |                        | 8692.8                  | 9245.7               | 8750.2               | 9289.0            | 9561.3                 | 9800                   | 10000                    | Total Ca                | ipital (\$mi              | II)                  | 10500              |
| Commo                  | n Stock   | 175 445                  | 661 che              |                          |                      |                       |                         |                        |                        |                        | 8/20.6<br>NMF           | 9318.0<br>NMF        | 9991.8<br>3.7%       | 3.8%              | 4.4%                   | 4.5%                   | 5.0%                     | Return o                | it (\$mill)<br>on Total C | ap'l                 | 13150<br>5.5%      |
| as of 7/2              | 28/11   | 175,445                  | 001 5115             |                          | 8                    |                       |                         |                        |                        |                        | NMF                     | NMF                  | 4.6%                 | 5.2%              | 6.5%                   | 7.5%                   | 8.0%                     | Return o                | on Shr. Eo                | uity                 | 9.0%               |
| MARKE                  | T CAP:  | \$5.3 billi              | on (Larg             | le Cap)                  |                      |                       |                         |                        |                        |                        | NMF                     | NMF                  | 3.0%                 | 1.8%              | 2.8%                   | 3.5%                   | 4.0%                     | Retained                | d to Com                  | Eq                   | 4.5%               |
| CURRE<br>(\$MI         | NT POS<br>上)  | ITION                    | 2009                 | 2010                     | 6/30/11              |                       |                         |                        |                        |                        |                         |                      | 34%                  | 65%               | 56%                    | 52%                    | 51%                      | All Div'd               | is to Net F               | Prof                 | 51%                |
| Cash A<br>Other        | ssets   | 4                        | 22.3                 | 13.1<br>521.2            | 13.5                 | BUSIN                 | ESS: An<br>r-owned      | water ar               | Mater Wo<br>nd wastev  | rks Com<br>vater util  | ipany, In<br>ity in the | U.S., pr             | largest              | employ            | accounti<br>ees. Dep   | ng for or<br>reciation | rate, 2.5                | of reven                | Jues. Has<br>). BlackR    | ock, Inc.            | , owns             |
| Accts P                | ayable  | 1                        | 38.6                 | 534.3<br>199.2           | 1493.2               | service               | s to over<br>ulated b   | 15 millio<br>usiness   | on people<br>assists n | in over :<br>nunicipal | 30 states<br>ities and  | and Can<br>military  | bases                | 6.9% o<br>1%. Pre | f the cor<br>esident & | nmon sto<br>CEO; Je    | ock outst<br>effry Sterk | anding. (<br>ba. Chairr | Off. & dir<br>man; Geo    | . own le<br>orge Mac | ss than<br>kenzie. |
| Other                  | Je  | 1                        | 73.6                 | 44.8<br>530.5            | 458.4<br>710.2       | with th               | e mainte                | enance a               | nd upkee               | p as we                | ell. Regu               | lated ope            | erations<br>biggest  | Address<br>856-34 | s: 1025 L              | aurel Oa               | k Road,                  | Voorhee                 | s, NJ 080                 | )43, Tele            | phone:             |
| Fix. Ch                | g. Cov.   | 2                        | 507.4<br>10%         | 774.5<br>237%            | 1327.6 250%          | Ame                   | ericar                  | 1 Wat                  | ter W                  | orks                   | look                    | s a li               | ittle                | Ame               | ricans                 | ' pock                 | ets, tl                  | ne rec                  | ent su                    | iccess               | has                |
| ANNUA                  | L RATE  | S Past                   | Pa                   | ast Est'o                | d '08-'10            | diff                  | erent                   | these                  | e day                  | s. In                  | line w                  | ith it               | s ag-                | us o              | ptimis                 | stic th                | nat m                    | ore fa                  | vorab                     | le rul               | lings              |
| Revenu<br>"Cash I      | ies<br>Flow"  |                          |                      |                          | 3.5%                 | crea                  | sed it                  | s pres                 | ence i                 | n Mis                  | souri                   | and (                | Dhio,                | for 1             | 8% ea                  | rning                  | s grov                   | with in                 | 2011.                     | : 110W               | TOOK               |
| Earning                | ls<br>ds  |                          |                      |                          | 9.5%<br>8.0%         | whil and              | e selli<br>New          | ing op<br>Mexi         | eratio                 | ns in<br>eanwl         | Texas                   | , Ariz<br>t has      | ona,<br>also         | The               | stock                  | t has<br>espite        | held<br>e the            | firm<br>bro             | since                     | our                  | last               |
| Book V                 | alue  | TCOLVOR                  |                      |                          | Nil                  | anno                  | ounce                   | l tha                  | tit                    | will ]                 | purch                   | ase s                | even                 | sello             | off. A                 | WK is                  | bene                     | fiting                  | not                       | only i               | from               |
| Cal-<br>endar          | Mar.31  | Jun. 30                  | Sep. 30              | (\$ mill.)<br>Dec. 31    | Full<br>Year         | But                   | it's b                  | een k                  | n New                  | ess as                 | s usu                   | al for               | the                  | cepti             | on that                | recent<br>at it is     | s a sat                  | fe hav                  | en du                     | ring t               | imes               |
| 2008                   | 506.8   | 589.4                    | 672.2                | 2 568.5                  | 2336.9               | wat                   | er ut                   | ility.                 | The o                  | compa                  | ny po                   | osted                | 10%                  | of eq             | conom                  | ic ins                 | stabili                  | ty. Th                  | he ma                     | arket                | has                |
| 2010                   | 588.1   | 671.2                    | 786.9                | 9 664.5                  | 2710.7               | a 69                  | 6 top-                  | line a                 | dvanc                  | e. (It                 | should                  | d be r               | oted                 | from              | day                    | to d                   | ay, a                    | nd fe                   | ars o                     | f and                | ther               |
| 2011                   | 645   | 730                      | 865                  | 724.5                    | 3000                 | that the              | the la<br>aforen        | atest l<br>nentio      | ned a                  | of resulterati         | ults ad                 | ccount<br>o the l    | s for                | reces             | to par                 | have i<br>rk the       | many<br>eir m            | on W<br>onev            | until                     | there                | ook-               |
| Cal-                   | E/<br>Mar 31  | ARNINGS F                | ER SHAP              | REA                      | Full                 | ness                  | mode                    | el, but                | the p                  | orior y                | vear's                  | figure               | es do                | signs             | s of s                 | tabili                 | ty. Gi                   | ven t                   | he m                      | urky                 | eco-               |
| 2008                   | .04   | .28                      | .55                  | .23                      | 1.10                 | sult                  | s.) Ar                  | ise w                  | e do<br>roved          | regu                   | latory                  | e past               | iron-                | our 1             | Highe                  | st (1)                 | ranki                    | ng for                  | the c                     | comin                | g six              |
| 2009                   | .19   | .32                      | .52                  | .21                      | 1.25                 | men                   | t was                   | s larg                 | gely r                 | espon                  | sible,                  | as A                 | AWR                  | to 12<br>The      | 2 mont                 | ths.<br>e fad          | es 9                     | bit lo                  | okin                      | o fur                | ther               |
| 2011                   | .24   | .46                      | .78                  | .27                      | 1.75                 | othe                  | r \$10.                 | 7 mill                 | lion in                | annu                   | al rev                  | enues                | , an-                | out,              | how                    | ever.                  | The                      | costs                   | of fi                     | ixing                | and                |
| Cal-                   | QUAR  | TERLY DIV                | /IDENDS              | PAID B.                  | Full                 | for                   | have<br>this            | raise<br>vear          | and r                  | eari                   | to ac                   | estin                | t for                | main<br>on tl     | ntain a<br>he ris      | aging<br>e, and        | water<br>d will          | likely                  | ems w<br>7 eat            | ill rei<br>away      | nain<br>at a       |
| endar                  | Mar.31  | Jun.30                   | Sep.30               | Dec.31                   | Year                 | ong                   | oing                    | mome                   | entun                  | on                     | the re                  | egula                | tory                 | healt             | thy po                 | ortion                 | of the                   | profi                   | ts enj                    | oyed                 | from               |
| 2007                   |   |                          | .20                  | .20                      | .40                  | othe                  | r \$4.8                 | milli                  | on rul                 | ing, a                 | nd ha                   | is an                | addi-                | dend              | l is he                | ealthy                 | , inco                   | me-m                    | inded                     | inves                | tors               |
| 2009 2010              | .20   | .20                      | .21                  | .21                      | .82                  | tion                  | al \$31                 | 15 mi                  | llion o                | do p                   | in ca                   | ses u                | nder                 | have              | bette                  | er alte                | ernativ                  | ves to                  | choos                     | se fro               | m in               |
| 2011                   | .22   | .23                      | .23                  |                          |                      | the                   | money                   | bein                   | g soug                 | ht to                  | make                    | its w                | ay to                | Andr              | re J. C                | Costan                 | za                       |                         | Octobe                    | er 21,               | 2011               |
| (A) Dilut<br>gains (lo | ed earr<br>sses): '(  | nings. Ex<br>08, (\$4.6) | cludes<br>2); '09, ( | nonrecurr<br>(\$2.63). E | Dis- (B)             | nings may<br>Dividend | y not sum<br>s to be pa | due to r<br>aid in Fet | ounding.<br>oruary, Ma | ay, Au-                | (C) In mi<br>(D) Inclu  | llions.<br>Ides inta | ngibles.             | In 2010:          | \$1.251 8              | oil- Sto               | mpany's<br>ock's Pric    | Financia<br>e Stabili   | al Streng<br>ity          | th                   | 8<br>90            |

gains (losses): us, (s4.b2); us, (s2.b3). Dis-continued operations: '06, (44): '11, 94. Next earnings report due early Nov. Quarterly able. © 2011, Value Line Publishing LLC. All rights reserved. Factual material is obtained from sources believed to be reliable and is provided without warranties of any kind. THE PUBLISHER IS NOT RESPONSIBLE FOR ANY ERRORS OR OM/SIGNONS HEREIN. This publication is sticity for subscriber's own, non-commercial, internal use. No part of it may be reproduced, resold, stored or transmitted in any printed, electronic or other form, or used for generating any printed or electronic publication, service or product.

| Earnings Predictability | 15             |
|-------------------------|----------------|
| To subscribe call       | 1-800-833-0046 |

| AQUA AMERICA NYSE-WIR         RECENT PRICE         21.30         P/E RATIO         20.1 (Trailing: 22.2) (Median: 25.0)         RELATIVE         1.47         DIV'D         3.1%           TIMELINESS         3 Lowered 1/21/11         High:         12.0         14.8         15.0         16.8         18.5         29.2         29.8         26.6         22.0         21.5         23.8  |  |                     |                       |                          |                          |                         |                  |                                       |                        |                       |                      |                        |                    | %             | LINE                  | E                       |              |                      |                        |                       |                        |
|---|--|---------------------|-----------------------|--------------------------|--------------------------|-------------------------|------------------|---------------------------------------|------------------------|-----------------------|----------------------|------------------------|--------------------|---------------|-----------------------|-------------------------|--------------|----------------------|------------------------|-----------------------|------------------------|
| TIMELINES   | ss 3   | Lowered 1           | 1/21/11               | High:<br>Low:            | 12.0<br>6.3              | 14.8<br>9.4             | 15.0<br>9.6      | 16.8<br>11.8                          | 18.5<br>14.2           | 29.2<br>17.5          | 29.8<br>20.1         | 26.6<br>18.9           | 22.0<br>12.2       | 21.5<br>15.4  | 23.0<br>16.5          | 23.8<br>19.3            |              |                      | Target                 | Price                 | Range                  |
| SAFETY  | 3  | Lowered 8           | 3/1/03                | LEGEN                    | NDS<br>60 x Divide       | nds p sh                |                  |                                       | 10100                  |                       |                      |                        |                    |               |                       |                         |              |                      | 2014                   | 2010                  | 64                     |
| TECHNICA  | L 3  | Raised 6/           | 10/11                 | div<br>Re                | vided by Inleading Price | erest Rate<br>Strength  |                  |                                       |                        | 4.                    | for-3                |                        |                    |               |                       |                         |              |                      |                        |                       | -48                    |
| 2014-1  | 16 PRO.  | JECTIO              | NS                    | 5-for-4 sp<br>5-for-4 sp | lit 12/00<br>lit 12/01   |                         |                  |                                       |                        | •                     |                      |                        |                    |               |                       |                         |              |                      |                        |                       | 32                     |
| Pric  | ce Ga  | An                  | n'i Total<br>Return   | 5-for-4 sp<br>4-for-3 sp | lit 12/03<br>lit 12/05   |                         |                  |                                       | for-4                  |                       | րոյ <sub>նները</sub> | molty                  |                    |               |                       | ստորւթ                  |              |                      |                        |                       | 24                     |
| High 35<br>Low 25   | +6   | 5%)<br>5%)          | 16%<br>8%             | Shaded                   | areas indic              | ate recess              | ions             | اسس                                   | 41111141               |                       | -                    |                        | արի                | որիսեւ        | աղող                  |                         |              |                      |                        |                       | - 16                   |
| Insider D   | D J F  | MA                  | мјј                   | Challer                  |                          | 10000                   | -471#            | TT.                                   |                        |                       |                      |                        | Stor.              |               |                       |                         | -            |                      |                        |                       | <b>1</b> <sup>12</sup> |
| to Buy 0<br>Options 0   | 0 0 0 1 1 0  | 0010                | 0 0 0 0 0 0           | daman                    | 1                        |                         | ·                | · · · · · · · · · · · · · · · · · · · |                        |                       | ·                    |                        |                    |               |                       |                         |              |                      |                        |                       |                        |
| to Sell 1<br>Institutio   | 1 0 0<br>nal De  | 1 0<br>cislor       | 0 0 0<br>15           | 1                        |                          |                         |                  |                                       |                        |                       |                      |                        | ·····              | 1 .           | ··                    | ·                       |              | % TO                 | T. RETUR               | RN 9/11<br>VL ARITH." |                        |
| to Buy  | 4Q2010   | 102011<br>81        | 202011<br>106         | Percent                  | t 15 -                   |                         |                  |                                       |                        |                       |                      | 1.1111                 |                    |               |                       |                         | -            | 1 yr.                | STOCK<br>8.7           | INDEX<br>-4.8         | E                      |
| to Sell<br>Hid's(000) 5   | 94<br>5463   | 112<br>55308        | 104<br>55457          | traded                   | 5 -                      | anandi                  | ullilu           |                                       | ululuulu               |                       |                      |                        |                    |               |                       |                         |              | 3 yr.<br>5 yr.       | 32.8<br>12.7           | 25.0<br>16.6          | -                      |
| 1995 1  | 996  | 1997                | 1998                  | 1999                     | 2000                     | 2001                    | 2002             | 2003                                  | 2004                   | 2005                  | 2006                 | 2007                   | 2008               | 2009          | 2010                  | 2011                    | 2012         | © VALI               | JE LINE P              | UB. LLC               | 14.16                  |
| 1.84  | 1.86   | 2.02                | 2.09                  | 2.41                     | 2.46                     | 2.70                    | 2.85             | 2.97                                  | 3.48                   | 1.21                  | 4.03                 | 4.52                   | 4.63               | 1.61          | 1.78                  | 1.95                    | 2.05         | "Cash F              | low" per               | sh                    | 2.40                   |
| .29   | .30  | .34                 | .40                   | .42                      | .47                      | .51                     | .54              | .57                                   | .64                    | .71                   | .70                  | .71                    | .73                | .77           | .90                   | 1.05                    | 1.10         | Earnings             | s per sh               | A Ba                  | 1.40                   |
| .22   | .23  | .24                 | .20                   | .27                      | 1.16                     | 1.09                    | 1.20             | 1.32                                  | 1.54                   | 1.84                  | 2.05                 | 1.79                   | 1.98               | 2.08          | 2.37                  | 2.30                    | 2.35         | Cap'l Sp             | ending p               | ersh                  | 2.50                   |
| 2.46  | 2.69   | 2.84                | 3.21                  | 3.42                     | 3.85                     | 4.15                    | 4.36             | 5.34                                  | 5.89                   | 6.30                  | 6.96                 | 7.32                   | 7.82               | 8.12          | 8.51                  | 8.95                    | 9.40         | Book Va              | lue per si             | h<br>Ist'a C          | 11.05                  |
| 12.0  | 15.6   | 17.8                | 22.5                  | 21.2                     | 111.02                   | 23.6                    | 23.6             | 24.5                                  | 25.1                   | 31.8                  | 34.7                 | 32.0                   | 24.9               | 23.1          | 21.1                  | Bold fig                | res are      | Avg Ann              | I P/E Ra               | tio                   | 21.0                   |
| .80   | .98  | 1.03                | 1.17                  | 1.21                     | 1.18                     | 1.21                    | 1.29             | 1.40                                  | 1.33                   | 1.69                  | 1.87                 | 1.70                   | 1.50               | 1.54          | 1.36                  | Value<br>estim          | Line<br>ates | Relative             | P/E Ratio              | 0<br>Gold             | 1.40                   |
| CAPITAL   | 4.9%         3.9%         2.9%         3.0%         3.3%         2.5%         2.5%         2.3%         1.8%         1.8%         2.1%         2.8%         3.1%         3.1%           AL STRUCTURE as of 6/30/11         307.3         322.0         367.2         442.0         496.8         533.5         602.5         627.0         670.5         726.1   |                     |                       |                          |                          |                         |                  |                                       |                        |                       |                      |                        |                    |               | 765                   | 810                     | Revenue      | s (\$mill)           | leid                   | 960                   |                        |
| Total Debt  | AL STRUCTURE as of 6/30/11         307.3         322.0         367.2         442.0         496.8         533.5         602.5         627.0         670.5         726           Jebt \$1559.0 mill. Due in 5 Yrs \$310 mill.         58.5         62.7         67.3         80.0         91.2         92.0         95.0         97.9         104.4         124           Jt \$1468.5 mill.         LT Interest \$66.1 mill.         39.3%         38.5%         39.3%         39.4%         38.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.4%         39.6%         38.9%         39.7%         39.6%         39.7%         39.7%         3 |                     |                       |                          |                          |                         |                  |                                       |                        |                       |                      |                        |                    |               | 124.0                 | 145                     | 155          | Net Prof             | it (\$mili)            |                       | 200                    |
| (LT interes   | t \$1468.5 mill. LT Interest \$66.1 mill.<br>rest earned: 4.5x; total interest coverage:<br>(55% of Cap'l)   |                     |                       |                          |                          |                         |                  |                                       |                        |                       |                      |                        |                    |               | 39.2%                 | 40.0%                   | 40.0%        | Income               | Tax Rate<br>% to Net I | Profit                | 40.0%                  |
| 4.5x)         (55% of Cap'l)           52.2%         54.2%         51.4%         50.0%         52.0%         51.6%           Pension Assets-12/10 \$159.2 mill.         47.7%         45.8%         48.6%         50.0%         48.0%   |  |                     |                       |                          |                          |                         |                  |                                       |                        |                       |                      | 55.4%                  | 54.1%              | 55.6%         | 56.6%                 | 54.0%                   | 52.0%        | Long-Te              | rm Debt F              | Ratio                 | 47.0%                  |
| Pension A   | Assets-1   | 2/10 \$1            | 159.2 mil             | l.<br>blig. \$234        | 4.9 mill.                | 47.7%                   | 45.8%            | 48.6%                                 | 50.0%                  | 48.0%                 | 48.4%                | 44.6%                  | 45.9%              | 44.4%         | 43.4%                 | 46.0%                   | 48.0%        | Common<br>Total Ca   | n Equity I             | Ratio                 | 53.0%<br>2950          |
| Pfd Stock   | None<br>Stock 1  | 38 105              | 123 cha               | 100                      |                          | 1368.1                  | 1490.8           | 1824.3                                | 2069.8                 | 2280.0                | 2506.0               | 2792.8                 | 2997.4             | 3227.3        | 3469.3                | 3630                    | 3795         | Net Plan             | t (\$mill)             | ,                     | 4320                   |
| To Block Nons         Tober Nons         Tobe |  |                     |                       |                          |                          |                         |                  |                                       |                        |                       |                      |                        |                    | 5.9%          | 6.5%                  | 7.0%                    | Return o     | on Total C           | ap'l                   | 8.0%                  |                        |
| CURRENT   | POSIT  | ION                 | 2009                  | 2010                     | 6/30/11                  | 12.3%                   | 12.7%            | 10.2%                                 | 10.7%                  | 11.2%                 | 10.0%                | 9.7%                   | 9.3%               | 9.4%          | 10.6%                 | 11.5%                   | 11.5%        | Return o             | on Com E               | quity                 | 12.5%                  |
| (\$MLL)<br>Cash Ass   | )<br>ets   |                     | 21.9                  | 5.9                      | 6.9                      | 5.1%                    | 5.2%             | 4.2%                                  | 4.6%                   | 4.9%                  | 3.7%                 | 3.2%                   | 2.8%               | 2.7%          | 3.7%                  | 5.0%                    | 4.5%         | Retained             | to Com                 | Eq                    | 5.5%                   |
| Receivabl<br>Inventory  | les<br>(AvgCs  | st)                 | 78.7<br>9.5           | 85.9<br>9.2              | 93.6<br>12.0             | BUSIN                   | ESS: Ag          | ua Amer                               | ca, Inc.               | is the ho             | olding cor           | npany fo               | r water            | others.       | Water su              | ipply reve              | enues '10    | ): residen           | tial, 59.5             | %; com                | mercial,               |
| Other<br>Current A  | ssets  | -1                  | 11.5<br>21.6          | 44.4                     | 178.0                    | and wa                  | stewater         | utilities I                           | hat serve              | e approxi             | imately th           | ree milli              | on resi-           | 14.5%;        | industrial            | & other                 | 26.0%.       | Officers<br>Chairm:  | and dire               | ctors ow              | n 2.0%                 |
| Accts Pay<br>Debt Due   | able   |                     | 57.9<br>87.0          | 45.3<br>28.5             | 42.7<br>90.5             | Jersey                  | , Florida,       | Indiana,                              | and five               | e other s             | states. D            | ivested t              | hree of            | ficer: Ni     | icholas D             | eBenedi                 | tis. Inco    | rporated:            | Pennsyl                | vania. A              | ddress:                |
| Other<br>Current Li   | iab.   | -2                  | 56.1<br>01.0          | 223.7                    | 173.7<br>306.9           | others.                 | Acquired         | business<br>d AquaSo                  | es in '91<br>ource, 7/ | ; telemai<br>03; Cons | weing gi<br>sumers W | oup in 3<br>later, 4/9 | 93; and<br>99; and | ephone        | st Lanca<br>: 610-525 | ster Aver<br>5-1400. Ir | iternet: w   | i Mawr, F<br>ww.aqua | america.               | ania 190<br>com.      | 10. 181-               |
| Fix. Chg.   | Cov.   | 3                   | 46%                   | 290%                     | 340%                     | Aqu                     | a Am             | nerica                                | sho                    | uld e                 | end 2                | 011 c                  | n a                | planı         | ning                  | on fil                  | ing c        | ases                 | in se                  | ven                   | more                   |
| ANNUAL I  | RATES  | Past                | Pa                    | st Est'd                 | '08-'10                  | alon                    | ng r<br>g wit    | 10te.<br>th st                        | Favo<br>ronge:         | rable<br>r-than       | rate                 | ruli<br>cted           | con-               | Jurise        | diction<br>rica's     | track                   | the y        | d. the               | id. Gi<br>ese ru       | lings                 | Aqua<br>will           |
| Revenues  | S  | 8.0                 | % 7                   | .5%                      | 6.5%                     | sum                     | er der           | mand,                                 | are                    | slated                | to b                 | e the                  | key                | likely        | y cont                | ribute                  | e to r       | evenu                | e and                  | l earr                | nings                  |
| Earnings  | 2  | 6.5                 | % 4                   | .5% 1                    | 0.5%                     | The                     | com              | pany                                  | enter                  | ed in                 | to a j               | oint                   | ven-               | Aqua          | a $Am$                | erica                   | is g         | etting               | g out                  | of s                  | ome                    |
| Book Valu   | ue   | 9.0                 | % 7                   | .0%                      | 6.0%                     | ture                    | wit              | h M                                   | LP H                   | Penn                  | Virg                 | inia                   | Re-                | mar           | kets.                 | Man                     | agem         | ent's                | plan                   | to                    | exit                   |
| Cal-<br>endar M   | QUARTI<br>ar.31  | ERLY RE<br>Jun.30   | VENUES<br>Sep.30      | (\$ mill.)<br>Dec.31     | Full<br>Year             | te a                    | fresh            | wate                                  | r pipe                 | eline.                | The p                | roject                 | will               | prog          | ressin                | g smo                   | othly.       | To th                | nis en                 | id, it                | sold                   |
| 2008 1  | 39.3   | 151.0               | 177.1                 | 159.6                    | 627.0                    | be s                    | supply<br>ers in | the I                                 | vater                  | to na                 | atural               | gas<br>PA              | pro-<br>area       | its 1<br>wate | Maine<br>r syst       | oper<br>ems)            | to Co        | s (co                | nsisti<br>icut V       | ng o<br>Water         | for                    |
| 2009 1  | 54.5<br>60.5   | 167.3<br>178.5      | 180.8 207.8           | 167.9<br>179.3           | 670.5                    | of th                   | ne Ma            | rcellu                                | s Sha                  | le. Th                | ie joir              | nt ver                 | ture               | \$53.5        | 5 mill                | ion, ii                 | n the        | secon                | d qua                  | arter.                | The                    |
| 2011 1  | 71.3   | 188.2               | 220                   | 185.5<br>200             | 765                      | has<br>a \$1            | been 1<br>2 mill | namec<br>lion ir                      | itial a                | wate<br>stake         | from                 | nces,<br>each          | part-              | with          | Ame:                  | rican                   | Water        | Wor                  | ano<br>ks (it          | swa                   | pped                   |
| Cal-  | EAR  | NINGS P             | ER SHAR               | E A                      | Full                     | ner.                    | Rang             | e Res                                 | ources                 | has                   | been o               | contra                 | icted              | its M         | lissou                | ri pro                  | pertie       | s in t               | he fir                 | st qu                 | arter                  |
| endar M   | 1ar.31   | Jun.30              | Sep.30                | Dec.31                   | Year                     | anti                    | cipate           | d to b                                | e opei                 | ration                | al by                | the b                  | e is<br>egin-      | Also,         | Aqua                  | a Ame                   | erica        | will b               | e swa                  | appin                 | g its                  |
| 2009  | .14  | .19                 | .20                   | .19                      | .77                      | ning                    | of 20            | 12, th                                | ough                   | no sol                | lid en               | d date                 | e has              | New           | York                  | proper                  | rties t      | o Ame                | Ohio                   | Wate                  | er in<br>ities         |
| 2010 2011   | .16<br>.19   | .22                 | .32                   | .20                      | 1.05                     | one                     | of ma            | iny st                                | eps th                 | ie con                | npany                | is ta                  | king               | Both          | deals                 | s are                   | slate        | d to e               | expan                  | d its                 | cus-                   |
| 2012  | .20  | .25                 | .37                   | .28                      | 1.10                     | to es                   | stablis<br>Marce | h itse                                | If as a                | a majo                | or ben               | eficia<br>a re         | ry of<br>sult      | tome          | r bas                 | e in f                  | ast-gr       | owing<br>a out       | g sect<br>from         | ors, v<br>its u       | vhile<br>nder-         |
| Cal-<br>endar M   | lar.31   | Jun.30              | Sep.30                | Dec.31                   | Full<br>Year             | ther                    | e shou           | ild be                                | a sig                  | nifical               | nt boo               | st to                  | reve-              | perfo         | rming                 | g are                   | as. T        | he de                | eals s                 | should                | d be                   |
| 2007  | .115   | .115                | .125                  | .125                     | .48                      | tome                    | and<br>er bas    | earni<br>e expa                       | ngs a<br>inds.         | s the                 | comp                 | any's                  | cus-               | done          | by th<br>ter.         | le end                  | of th        | is yea               | r or 2                 | :012's                | nrst                   |
| 2008  | .125   | .125                | .125                  | .135<br>.145             | .51                      | Rat                     | e ruli           | ings                                  | are s                  | till o                | n the                | age                    | nda.               | This          | equ                   | ity h                   | as a         | n ab                 | ove                    | indu                  | stry                   |
| 2010  | .145   | .145                | .145                  | .155                     | .59                      | rate                    | rulir            | ngs la                                | st ye                  | ed se<br>ar, ai       | nd is                | curre                  | ently              | Saha          | ina Zi                | itshi                   | 10r 1n       | come (               | Octobe                 | er 21,                | 2011                   |
| (A) Diluted   | egs. Ex  | cl. non             | ec. gain              | s (losses)               | : (B)                    | Dividend                | s historica      | ally paid i                           | n early N              | larch,                |                      |                        |                    |               |                       | Cor                     | npany's      | Financia             | l Streng               | th                    | B+                     |
| 99, (11¢); '<br>Excl. gain f  | 00, 2¢;<br>from dise   | 01, 2¢;<br>c. opera | 02, 5¢;<br>ations: '9 | 03, 4¢.<br>6, 2¢. Ne:    | xt avai                  | a, Sept. 8<br>lable (5% | discoun          | uv d. reir<br>t).                     | ivestmen               | t plan                |                      |                        |                    |               |                       | Prie                    | ck's Pric    | h Persis             | tence                  |                       | 70                     |

Exc. gain from disc. operations: 90, 22, Next available (5% discount), earnings report due late October. (C) In millions, adjusted for stock splits. © 2011, Value Line Publishing LLC, All rights reserved. Factual material is obtained from sources believed to be reliable and is provided without warranties of any kind, THE PUBLISHER IS NOT RESPONSIBLE FOR ANY ERRORS OR OMISSIONS HEFEIN: This publication is strictly for subcerbar's own, non-commercial, internal use. No particulation is used for generating or marketing any printed or electronic publication, service or product.

| subscribe call 1-800-8 | 33-0046. |
|------------------------|----------|
| rnings Predictability  | 100      |
| ock's Price Stability  | 100      |
|                        |          |

| ART                                  | ESI/   | AN R               | ES. (           | COR                | P. 1               | IDQARTN                    | A RE<br>PR           | CENT 17                             | 72 TRAILIN<br>P/E RAT         | 6 <b>19.7</b> | RELATIVE 1.30                     | ) DIV'D 4               | .3% VA             | LUE<br>NE                        |
|--------------------------------------|--|--------------------|-----------------|--------------------|--------------------|----------------------------|----------------------|-------------------------------------|-------------------------------|---------------|-----------------------------------|-------------------------|--------------------|----------------------------------|
|                                      | RA   | KS                 |                 | 19                 | .83                | 20.04                      | 22.62                | 22.33                               | 20.67                         | 19.3          | 1 18.73                           | 19.59                   | 19.99              | High                             |
| PERFOR                               | MANCE  | 2 4                | bove<br>verage  |                    | LEGE               | NDS                        |                      |                                     |                               |               |                                   |                         |                    | 1 (100)                          |
| Technica                             | 1  | 3 4                | verage          | 1 F                | 2 Mos<br>Rel Prin  | e Strength                 | H. H. 1111444        | - <sup>114</sup> 11111              | 1+++·····                     |               | The second                        |                         | · ·····            | 18                               |
| SAFETY                               |  | 2 2                | bove            | 3-for-2<br>3-for-2 | split 7<br>split 7 | /03                        | •                    |                                     |                               |               | • •                               |                         |                    | 13                               |
| BETA F                               | 50   | (1.00 =            | Markot)         | Snaded             | area indi          | cates recession            |                      |                                     |                               |               |                                   |                         |                    | 8                                |
| DEIA .                               |  | (1.00 -            | Markey          |                    |                    |                            |                      | · ·                                 |                               |               |                                   |                         |                    | 5                                |
| Einoneial                            | Cirona   | 4h                 | P.              |                    |                    |                            |                      |                                     |                               |               |                                   | ·····                   |                    | 4                                |
| Deles Cia                            | - Streng   |                    | 400             |                    |                    |                            |                      |                                     |                               |               |                                   |                         |                    | 0                                |
| Price Sta                            | Dility   |                    | 100             |                    |                    |                            |                      |                                     |                               | 1.1.1.1       |                                   |                         |                    | 2                                |
| Price Gro                            | owth Pe  | rsistence          | a 60 .          |                    | _                  |                            |                      |                                     |                               |               |                                   |                         |                    | 400                              |
| Earnings                             | Predic   | tability           | 90              |                    |                    |                            |                      |                                     | huulth                        | hudan         |                                   | 1111111                 |                    | VOL.<br>(thous.)                 |
| © VALUE                              | LINE I   | UBLISH             | ING LLC         | 200                | 3                  | 2004                       | 2005                 | 2006                                | 2007                          | 2008          | 2009                              | 2010                    | 2011               | 2012/2013                        |
| SALES F                              | ER SH  |                    |                 | 6.3                | 20                 | 6.67                       | 7.52                 | 7.77                                | 7.20                          | 7.59          | 8.11                              | 8.48                    |                    |                                  |
| "CASH F                              | LOW"   | PER SH             |                 | 1.                 | 28                 | 1.42                       | 1.56                 | 1.75                                | 1.57                          | 1.65          | 1.84                              | 1.92                    | <br>02 A,B         | 1 10 C/NA                        |
| DIV'DS D                             | ECL'D  | PER SH             |                 |                    | 53                 | .55                        | .58                  | .61                                 | .66                           | .00           | .72                               | .75                     | .52                | 1.10 ///4                        |
| CAP'L SI                             |  | G PER S            | н               | 4.1                | 20                 | 4.82                       | 3.35                 | 5.08                                | 3.66                          | 6.09          | 2.32                              | 2.57                    |                    |                                  |
| COMMON                               | SHS (  | DUTST'G            | (MILL)          | 5.0                | 85                 | 5.93                       | 6.02                 | 6.09                                | 7.30                          | 7.40          | 7.51                              | 7.65                    |                    |                                  |
| AVG ANN                              | L P/E  | RATIO              |                 | 24.1               | 7                  | 25.4                       | 24.2                 | 20.3                                | 21.5                          | 20.1          | 16.4                              | 18.2                    | 19.3               | 16.1/NA                          |
| AVG ANN                              | I'L DIV  | D YIELD            |                 | 3.4                | 4%                 | 3.0%                       | 2.9%                 | 3.1%                                | 3.4%                          | 4.1%          | 4.5%                              | 4.1%                    |                    |                                  |
| SALES (                              | SALES (\$MILL)<br>OPERATING MARGIN<br>DEPRECIATION (\$MILL)      |                    |                 |                    | 3                  | 39.6                       | 45.3                 | 47.3                                | 52.5                          | 56.2          | 60.9                              | 64.9                    |                    | Bold figures                     |
| DEPREC                               | OPERATING MARGIN<br>DEPRECIATION (\$MILL)<br>NET PROFIT (\$MILL) |                    |                 |                    | 6                  | 4.0                        | 4.4                  | 45.6%                               | 45.6%                         | 45.1%         | 6.6                               | 46.5%                   |                    | are consensus<br>earnings        |
| NET PRO                              | NET PROFIT (\$MILL)<br>INCOME TAX RATE                           |                    |                 |                    | 9                  | 4.4                        | 5.0                  | 6.1                                 | 6.3                           | 6.4           | 7.3                               | 7.6                     |                    | estimates                        |
| INCOME TAX RATE<br>NET PROFIT MARGIN |  |                    |                 | 37.9               | 9%<br>8%           | 39.6%<br>11.1%             | 39.9%                | 39.0%                               | 39.8%                         | 40.8%         | 40.1%<br>11.9%                    | 40.0%                   |                    | and, using the<br>recent prices. |
| WORKIN                               | G CAP'   | L (\$MILL          | )               | d10.               | 5                  | d8.7                       | d1.8                 | d8.8                                | 2.5                           | d20.9         | d23.3                             | d27.9                   |                    | P/E ratios.                      |
| LONG-TE                              | ERM DE   | BT (\$MIL<br>MILL) | .L)             | 80.0               | 6                  | 82.4<br>54.9               | 92.4<br>57.8         | 92.1                                | 91.8                          | 107.6         | 106.0                             | 105.1                   | -                  |                                  |
| RETURN                               | ON TO  | TAL CAP            | ۳L              | 4.                 | 5%                 | 5.1%                       | 5.3%                 | 5.8%                                | 5.3%                          | 4.7%          | 5.2%                              | 5.6%                    |                    |                                  |
| RETURN                               | ON SH  | R. EQUIT           | ſY              | 7.4                | 4%                 | 8.0%                       | 8.7%                 | 9.8%                                | 7.4%                          | 7.3%          | 8.0%                              | 8.0%                    |                    |                                  |
| ALL DIV                              | DS TO  | NET PRO            | )F              | 81%                | 470                | 74%                        | 69%                  | 61%                                 | 71%                           | 81%           | 74%                               | 75%                     |                    |                                  |
| A <sub>No.</sub> of a                | nalysts ci   | hanging ea         | arn. est. in l  | ast 8 days         | s: 0 up,           | 0 down, conse              | nsus 5-year ean      | nings growth ne                     | ot available. <sup>B</sup> Ba | sed upon 4 an | alysts' estimates. <sup>C</sup> E | Based upon 4 a          | nalysts' estimates |                                  |
|                                      | . /  | ANNUAL             | RATES           |                    |                    | ASSETS (\$n                | nill.) 2             | 009 2010                            | 6/30/11                       |               | INDU                              | ISTRY: W                | ater Utility       |                                  |
| Sales                                | ge (per s  | ihare)             | 5 Yrs.<br>3.5%  | 1                  | Yr.<br>.5%         | Cash Assets<br>Receivables |                      | .5 .2                               | 2                             | BUSIN         | ESS: Artesia                      | n Resourc               | es Corporati       | on, through its                  |
| "Cash F<br>Earning                   | low"<br>s  |                    | 5.0%<br>5.5%    | 4.                 | .0%<br>.0%         | Inventory                  |                      | 1.2 1.2                             | 1.4                           | subsidia      | ries, provides                    | water, was              | stewater, and      | l other services                 |
| Dividend<br>Book Va                  | ls<br>Iue  |                    | 5.5%            | 4.                 | 5%                 | Current Asse               | ts t                 | 13.2 14.0                           | 11.3                          | on the I      | Delmarva Pen                      | insula. Th              | e company          | distributes and                  |
| Final                                | 0114   | DTEDIV             | SALES /6        | mill \             |                    | Property Pla               | nt                   |                                     |                               | protectic     | on, to residenti                  | ial, comme              | ercial, indust     | rial, municipal,                 |
| Year                                 | 10   | 2Q                 | 3Q              | 4Q                 | Year               | & Equip, a                 | at cost 40           | 03.0 414.6                          |                               | and utili     | ty customers                      | throughou               | t Delaware,        | Maryland, and                    |
| 12/31/09                             | 13.9   | 15.4               | 16.1            | 15.5               | 60.9               | Net Property               | 33                   | 38.1 345.4                          | 352.5                         | Pennsylv      | vania. It also<br>n Delaware a    | provides v              | wastewater s       | ervices to cus-                  |
| 12/31/10                             | 15.0<br>14.8   | 16.0<br>16.5       | 18.0            | 15.9               | 64.9               | Total Assets               | 35                   | <u>7.6</u> <u>12.</u><br>58.9 371.5 | 371.8                         | ments to      | o provide wa                      | astewater               | services in        | Maryland. In                     |
| 12/31/12                             |  |                    |                 |                    |                    |                            | (\$mill)             |                                     | 6 - 1985 A                    | addition      | , Artesian pro                    | vides con               | tract water a      | and wastewater                   |
| Fiscal                               | EA   | RNINGS             | PER SHA         | RE                 | Full               | Accts Payabl               | e                    | 3.7 3.4                             | 2.6                           | operation     | ns, water and                     | sewer service           | rvice line pi      | otection plans,                  |
| 12/31/08                             | .13  | .21                | .35             | .17                | .86                | Other                      |                      | <u>5.1</u> <u>7.9</u>               | 8.0                           | and engi      | ineering servi                    | ces. Artesi             | ian Resource       | es is the parent                 |
| 12/31/09                             | .22  | .27                | .28             | .20                | .97                | Current Liab               | 1                    | 36.5 41.9                           | 38.3                          | holding       | company of A                      | rtesian Wa              | ter Company        | y, Inc., Artesian                |
| 12/31/10                             | .22  | .24                | .38             | .16                | 1.00               |                            |                      |                                     |                               | Artesian      | Wastewater M                      | Inc., Artes<br>Manageme | sian water f       | viaryland, Inc.,                 |
| 12/31/12                             |  |                    |                 |                    |                    | LONG-TERN                  | DEBT AND E           | EQUITY                              |                               | ter Mary      | land, Inc. and                    | three othe              | er entities. H     | as 426 employ-                   |
| Cal-                                 | QUAR<br>10   | TERLY D            | OIVIDENDS<br>30 | S PAID             | Full               | Tatal Data                 | 404.4                | <b>D</b>                            | - F.V NA                      | ees. Cha      | hirman, C.E.C                     | ). & Presi              | dent: Dian         | C. Taylor. Ad-                   |
| 2008                                 | .172   | .178               | .178            | .178               | .71                | LT Debt \$10               | 6.7 mill.            | Due                                 | II O TIS, NA                  | 453-690       | 0. Internet: ht                   | tp://www.               | artesianwate       | r.com.                           |
| 2009                                 | .178   | .178               | .178            | .187               | .72                | Including Ca               | ap. Leases NA        | (5                                  | 3% of Cap'l)                  |               |                                   |                         |                    | J.V.                             |
| 2010 2011                            | .187   | .188<br>.19        | .188            | .189               | .75                | Leases, Unc                | apitalized Anr       | nual rentals N                      | 4                             |               | C                                 | October 21              | , 2011             |                                  |
|                                      | INSTI  | UTIONA             | L DECISIO       | ONS                |                    | Pension Lia                | bility \$.5 mill. ir | n '10 vs. \$.7 mi                   | l. in '09                     | TOTAL         | SHAPEHOLD                         | ER DETI                 | RN                 |                                  |
|                                      | C.C.   | 4Q'10              | 1Q'11           | 20                 | 2'11               | Pfd Stock No               | ne                   | Pfd Div                             | d Paid None                   | IUIAL         | UNANEHOLD                         | Divider                 | nds plus apprecia  | ation as of 9/30/2011            |
| to Buy<br>to Sell                    |  | 23<br>21           | 24<br>19        |                    | 25<br>15           | Common Sto                 | ck 7,675,000 sh      | ares                                | 176/ 06 00                    | 3 Mos.        | 6 Mos.                            | 1 Yr.                   | 3 Yrs              | . 5 Yrs.                         |
| HId's(00                             | 00)  | 2190               | 2308            | 23                 | 47                 |                            |                      | (                                   | 47 % of Cap1)                 | -1.80%        | -8.30%                            | -4.43%                  | 18.87              | % 13.80%                         |

2011 Value Line Publishing LLC. All rights reserved. Factual material is obtained from sources believed to be reliable and is provided without warranties of any kind. THE PUBLISHER IS NOT RESPONSIBLE FOR ANY ERRORS OR OMISSIONS HEREIN. This publication is shirtly for subscriber's own, non-commercial, internal use. No part of it may be reproduced, resold, stored or transmitted in any printed, electronic or other form, or used for generating or marketing any printed or electronic publication, service or product.

| CAL                             | IF0                | RNI  | A WA              | ATER  |  | E-CWT                  | RI                     | ECENT<br>Rice         | 17.8                   | 4 P/E<br>RATI        | 15.                 | 9 (Traili<br>Media  | ng: 19.4<br>an: 22.0) | RELATIVE<br>P/E RATIO | 1.1                    | 6 DIV'D<br>YLD       | 3.5                  | %                        |                            | Ξ                     |                   |
|---------------------------------|--------------------|--|-------------------|---|--|------------------------|------------------------|-----------------------|------------------------|----------------------|---------------------|---------------------|-----------------------|-----------------------|------------------------|----------------------|----------------------|--------------------------|----------------------------|-----------------------|-------------------|
| TIMELIN                         | ESS 3              | Raised 7/                                    | 22/11             | High:<br>Low:   | 15.7<br>10.8                           | 14.3<br>11.4           | 13.4<br>10.2           | 15.7<br>11.8          | 19.0<br>13.0           | 21.1<br>15.6         | 22.9<br>16.4        | 22.7<br>17.1        | 23.3<br>13.8          | 24.1<br>16.7          | 19.8<br>16.9           | 19.4<br>16.7         |                      |                          | Target                     | Price                 | Range             |
| SAFETY                          | 3                  | Lowered                                      | 7/27/07           | LEGEN   | NDS<br>33 x Divide                     | nds p sh               |                        |                       |                        |                      |                     |                     | denira de             |                       |                        |                      |                      |                          | 2014                       | 2010                  | -64               |
| BETA .7                         | CAL 3<br>0 (1.00=  | <ul> <li>Lowered</li> <li>Market)</li> </ul> | 10/7/11           | Re<br>2-for-1 sp  | ided by In<br>Itative Price<br>It 1/98 | erest Hate<br>Strength | -                      |                       |                        | <u>.</u>             | -                   |                     |                       |                       |                        |                      |                      |                          |                            |                       | 48                |
| 2014                            | 4-16 PR            | OJECTIC                                      | NS                | 2-for-1 sp<br>Options:  | lit 6/11<br>Yes<br>areas indir         | ala racass             | ione                   |                       |                        |                      |                     |                     |                       |                       |                        | - 2-for-1            |                      |                          |                            |                       | -32               |
| F<br>High                       | rice<br>35 (+      | Gain<br>⊦95%)                                | Return<br>20%     |   |  |                        |                        | _                     |                        | Lunt (11+),1         | 10 Junt             | hin little          | mull                  | Hipmon                | ակեստ                  | • II,, II •          |                      |                          |                            |                       | 20                |
| Low<br>Insider                  | 20 (+<br>Decis     | lons   | 6%                | 1 <sup>1111</sup> 11,1  | II<br>Internet                         | 1. Thursday            |                        | ոսկետ                 | munat                  |                      |                     |                     |                       |                       |                        |                      |                      | _                        |                            |                       | 12                |
| to Buy                          | N D J<br>0 0 0     | F M A<br>0 19 0                              | M J J<br>0 0 0    | <u></u>   |  |                        |                        |                       |                        |                      |                     |                     |                       |                       |                        |                      |                      |                          |                            |                       | -8                |
| Options<br>to Sell              | 100                | 0 0 0 0 0 0 0                                |                   |   |  | · · · · ·              | ·                      |                       |                        | **********           | *******             |                     |                       | ***                   | ·                      |                      |                      | % TO                     | I<br>T. RETUR              | N 9/11                | -6                |
| Institu                         | 402010             | 102011                                       | ns<br>202011      | Percen  | l<br>t 18 —                            |                        |                        |                       |                        |                      |                     |                     |                       |                       |                        |                      |                      | 1.vr                     | THIS N<br>STOCK            | INDEX                 | L                 |
| to Buy<br>to Sell<br>Hidre(000) | 62<br>48<br>20250  | 56<br>49<br>21158                            | 60<br>48<br>21479 | shares<br>traded  | 12 -<br>6 -                            |                        | hum                    |                       |                        | hundete              | athit               |                     |                       |                       |                        |                      |                      | 3 yr.<br>5 yr.           | 1.3<br>12.4                | 25.0<br>16.6          | -                 |
| 1995                            | 1996               | 1997   | 1998              | 1999  | 2000                                   | 2001                   | 2002                   | 2003                  | 2004                   | 2005                 | 2006                | 2007                | 2008                  | 2009                  | 2010                   | 2011                 | 2012                 | © VAL                    | UE LINE PI                 | UB. LLC               | 4-16              |
| 6.58<br>1.04                    | 7.24               | 7.74   | 7.38              | 7.98  | 8.08                                   | 8.13                   | 8.67                   | 8.18                  | 8.59                   | 8.72                 | 8.10                | 8.88                | 9.90                  | 10.82                 | 11.05<br>1.93          | 11.80                | 12.05                | Revenue<br>"Cash F       | es per sh<br>low" per :    | sh                    | 14.00 2.60        |
| .58                             | .75                | .92  | .73               | .77   | .66                                    | .47                    | .63                    | .61                   | .73                    | .74                  | .67                 | .75                 | .95                   | .98                   | .91                    | 1.10                 | 1.20                 | Earning                  | s per sh                   | A<br>                 | 1.35              |
| 1.09                            | 1.41               | 1.30   | 1.37              | 1.72  | 1.23                                   | 2.04                   | 2.91                   | 2.19                  | 1.87                   | 2.01                 | 2.14                | 1.84                | 2.41                  | 2.66                  | 2.97                   | 2.50                 | 2.75                 | Cap'l Sp                 | ending p                   | ersh                  | 3.15              |
| 5.86                            | 6.11               | 6.50   | 6.69              | 6.71  | 6.45                                   | 6.48                   | 6.56                   | 7.22                  | 7.83                   | 7.90                 | 9.07                | 9.25                | 9.72                  | 10.13                 | 10.45                  | 10.75                | 10.90                | Book Va<br>Commo         | lue per si<br>n Shs Out    | st'a D                | 11.95<br>46.50    |
| 13.7                            | 11.9               | 12.6   | 17.8              | 17.8  | 19.6                                   | 27.1                   | 19.8                   | 22.1                  | 20.1                   | 24.9                 | 29.2                | 26.1                | 19.8                  | 19.7                  | 20.3                   | Bold fig             | vres are             | Avg An                   | I P/E Rat                  | io                    | 20.5              |
| .92<br>6.4%                     | .75<br>5.8%        | 4.6%   | 4.2%              | 4.0%  | 4.3%                                   | 1.39                   | 4.5%                   | 4.2%                  | 3.9%                   | 1.33                 | 1.58                | 3.0%                | 3.1%                  | 3.1%                  | 3.2%                   | estin                | ates                 | Avg An                   | P/E Ratio                  | ield                  | 2.8%              |
| CAPITA                          | L STRU             | CTURE a                                      | as of 6/30        | )/11  | mill                                   | 246.8                  | 263.2                  | 277.1                 | 315.6                  | 320.7                | 334.7               | 367.1               | 410.3                 | 449.4                 | 460.4                  | 505                  | 530                  | Revenu                   | es (\$mill)                | E                     | 650               |
| LT Debt                         | \$478.0            | mill. L                                      | T Interes         | st \$32.0 n   | nill.                                  | 39.4%                  | 39.7%                  | 39.9%                 | 39.6%                  | 42.4%                | 37.4%               | 39.9%               | 39.8                  | 40.6                  | 39.5%                  | 35.0%                | 36.5%                | Income                   | Tax Rate                   |                       | 39.0%             |
| Danalar                         | esteam             | 100: 3.0X;                                   | (5)               | 2% of Ca  | .)<br>p'l)                             | 50.3%                  | 55 3%                  | 10.3%                 | 3.2%                   | 3.3%                 | 10.6%               | 8.3%                | 8.6%                  | 7.6%                  | 4.2%                   | 10.0%                | 10.0%                | AFUDC                    | % to Net I                 | Profit                | 10.0%             |
| Pension                         | Assets             | 5-12/10 \$                                   | Oblig. \$2        | ı.<br>69.9 mill.  |  | 48.8%                  | 44.0%                  | 49.1%                 | 50.8%                  | 51.1%                | 55.9%               | 56.6%               | 58.4%                 | 52.9%                 | 47.6%                  | 48.5%                | 49.0%                | Commo                    | n Equity P                 | Ratio                 | 49.0%             |
| Pid Sto                         | CK None            | 1  |                   |   |  | 402.7<br>624.3         | 453.1                  | 498.4                 | 565.9<br>800.3         | 568.1<br>862.7       | 670.1               | 674.9               | 690.4<br>1112.4       | 794.9                 | 914.7<br>1294.3        | 945<br>1350          | 980                  | Total Ca<br>Net Plan     | pital (\$mi<br>nt (\$mill) | 11)                   | 1125<br>1625      |
| Commo                           | n Stock            | 41,752,0                                     | )32 shs.          |   |  | 5.3%                   | 5.9%                   | 5.6%                  | 6.1%                   | 6.3%                 | 5.2%                | 5.9%                | 7.1%                  | 6.5%                  | 5.5%                   | 6.5%                 | 7.0%                 | Return                   | on Total C                 | ap'l                  | 7.5%              |
| MARKE                           | T CAP:             | \$750 mil                                    | lion (Sm          | all Cap)  |  | 7.2%                   | 9.4%                   | 7.8%                  | 8.9%<br>9.0%           | 9.3%                 | 6.8%                | 8.1%                | 9.9%                  | 9.6%                  | 8.6%                   | 10.0%                | 10.5%                | Return                   | on Shr. Eq                 | quity                 | 11.0%             |
| CURRE<br>(\$MI                  | NT POS             | ITION  | 2009              | 2010  | 6/30/11                                | NMF<br>119%            | 1.0%                   | .7%                   | 2.1%                   | 2.1%                 | 1.0%                | 1.8%                | 3.8%                  | 3.8%                  | 3.0%                   | 4.5%                 | 5.0%                 | Retained<br>All Div'd    | d to Com<br>Is to Net F    | Eq                    | 5.5%<br>52%       |
| Cash A<br>Other                 | ssets              | _  | 9.9<br>82.3       | 42.3<br>83.9  | 32.9<br>98.7                           | BUSIN                  | ESS: Ca                | lifornia W            | ater Serv              | rice Grou            | p provide           | es regula           | ted and               | breakdo               | wn, '10:               | residenti            | al, 72%;             | business                 | , 20%; pu                  | iblic auth            | orities,          |
| Accts P                         | Assets ayable      |  | 92.2<br>43.7      | 126.2<br>39.5   | 131.6<br>51.6                          | nonreg<br>commi        | ulated w<br>unities in | ater sen<br>Californi | vice to r<br>a, Washi  | ngton, N             | 70,200 c<br>ew Mexi | ustomer:<br>co, and | s in 83<br>Hawaii.    | 4%; inc<br>roughly    | lustrial,<br>1,127 ei  | 4%. '10<br>mployees  | reported<br>. Chairm | d depred<br>an: Rob      | iation ra<br>ert W. Fr     | te: 2.3%<br>oy. Presi | dent &            |
| Other                           | ue                 | -  | 25.0<br>41.7      | 26.1  | 35.1<br>44.9                           | Main s<br>Salinas      | ervice and s Valley.   | eas: Sar<br>San Joa   | n Francis<br>aquin Val | co Bay a<br>lev & pa | area, Sad           | ramento<br>os Angel | Valley,<br>les, Ac-   | CEO: P<br>North F     | eter C. N<br>irst Stre | Velson (4<br>et, San | /11 Prox<br>Jose, Ca | y). Inc.: I<br>alifornia | Delaware.<br>95112-45      | Addres<br>98. Tele    | s: 1720<br>phone: |
| Fix. Ch                         | g. Cov.            | 4  | 110.4             | 390%  | 300%                                   | quired                 | Rio Gra                | nde Cor               | p; West                | Hawaii               | Utilities (         | 9/08). R            | evenue                | 408-367               | -8200. lr              | nternet: w           | ww.calw              | atergroup                | o.com.                     |                       |                   |
| ANNUA<br>of change              | L RATE<br>(per sh) | S Past<br>10 Yrs                             | Pa<br>5 Y         | st Est'c<br>rs. to  | 1 '08-'10<br>'14-'16                   | We<br>Gro              | look<br>up to          | for (<br>gain         | alifo<br>1 fur         | rnia<br>ther         | Water               | · Ser               | vice<br>n in          | ket, a<br>relati      | and C<br>ive st        | WT is<br>tabilit     | y sin                | tterei                   | nt as s<br>r Jul           | seen b<br>y rev       | iew.              |
| "Cash                           | ies<br>Flow"       | 3.0<br>4.0                                   | % 4.<br>% 6.      | .5%<br>.5%  | 5.0%<br>5.5%                           | the                    | seco                   | nd ha                 | alf of                 | the                  | year                | Rate                | e in-                 | The o                 | currer                 | tock                 | d is a               | nothe                    | r selli                    | ng poi                | int.              |
| Dividen                         | js<br>ids          | 3.0  | % 6.<br>% 1       | .0%   | 6.0%<br>3.0%                           | quai                   | ter, e                 | enabli                | ng th                  | e wa                 | ter p               | ovide               | r to                  | ing                   | furth                  | er ou                | t. CW                | T, an                    | d mos                      | st util               | ities             |
| Cal-                            | QUAR               | TERLY RE                                     | VENUES (          | \$ mill.)E  | Full                                   | post<br>terir          | bette<br>n, sug        | r-than<br>gestir      | ng tha                 | t addi               | tional              | in th               | e in-                 | avera                 | nat m<br>ages v        | vhen                 | typic                | ally t<br>are g          | cood, a                    | and w                 | e do              |
| endar<br>2008                   | Mar.31             | Jun.30                                       | Sep.30            | Dec.31  | Year                                   | may                    | be in                  | the<br>estin          | pipelin                | ne. As<br>for t      | a rea<br>he ba      | sult, v<br>ck ha    | we've                 | expea                 | t the                  | mark                 | et to 1<br>e cos     | recove                   | r by 2<br>runi             | 2014-2<br>ning        | 016.<br>and       |
| 2009                            | 86.6               | 116.7  | 139.2             | 106.9   | 449.4                                  | the                    | year,                  | and                   | look                   | for he               | ealthy              | top-                | and                   | main                  | tainir                 | ng a w               | ater                 | utility                  | servi                      | ces p                 | lant,             |
| 2010                            | 90.3               | 131.4  | 140.5             | 105.5<br>115  | 460.4<br>505                           | The                    | re co                  | uld b                 | e son                  | ne m                 | ore g               | ood r               | iews                  | with                  | it, is                 | a ve                 | ry ex                | pensiv                   | ve un                      | dertal                | ting.             |
| 2012<br>Cal-                    | 103<br>E/          | 135<br>ARNINGS I                             | 170<br>PER SHAR   | 122<br>E A  | 530<br>Full                            | its c                  | the h                  | orizo<br>capita       | n, too                 | o. CW<br>licatio     | T rec<br>n in a     | ently<br>in att     | filed                 | Fede                  | ral a:<br>elv st       | nd st<br>ringer      | ate rent. an         | equire<br>d sys          | ement:<br>tems             | s are<br>are g        | ex-<br>row-       |
| endar                           | Mar.31             | Jun.30                                       | Sep.30            | Dec.31  | Year                                   | to in                  | icreas                 | e its i               | return                 | on e                 | quity               | a full              | per-                  | ing o                 | older                  | by th                | e day                | . Ma                     | ny re                      | quire                 | sig-              |
| 2008                            | .01                | .24  | .53               | .16   | .95                                    | proc                   | ess is                 | unp                   | redict                 | able,                | but t               | he re               | ecent                 | plete                 | overl                  | hauls.               | Thes                 | e cost                   | s are                      | not li                | kely              |
| 2010                            | .05                | .25  | .49<br>.59        | .49 .12 .91 climate appears to have warmed for utili-<br>.59 .17 1.10 ties, particularly in the Golden State. If a  |  |                        |                        |                       |                        |                      |                     |                     |                       |                       |                        | for C                | WT                   | soon,<br>on th           | creat<br>ae cos            | t sid                 | e of              |
| 2012                            | .07<br>OUAR        | .32  | .62<br>/IDENDS F  | .62 .19 1.20 favorable decision is handed down by year-<br>ledger. Indeed, these<br>any necessary capital<br>MS PAID <sup>8</sup> • Full end, as expected, this would likely force us any necessary capital   |  |                        |                        |                       |                        |                      |                     |                     |                       |                       |                        |                      | exper<br>l reg       | ises, a<br>uirem         | long<br>ents.              | with<br>will          |                   |
| endar                           | Mar.31             | Jun.30                                       | Sep.30            | <b>NOSFAID</b> Full end, as expected, this would likely force us any necessary capital requirements, with <u>p.30 Dec.31 Year</u> to bump up our current 2012 estimates. likely temper earnings advances out the divergence of the divergence |  |                        |                        |                       |                        |                      |                     |                     |                       |                       |                        |                      | t to                 |                          |                            |                       |                   |
| 2007 2008                       | .145<br>.147       | .145<br>.147                                 | .145<br>.147      | 145 .145 .58 Now may be a good time for many mid-decade and thereafter. While the div<br>147 .147 .59 seeking to avoid getting caught up in dend is certainly a plus, CWT still lac   |  |                        |                        |                       |                        |                      |                     |                     |                       |                       |                        |                      | acks                 |                          |                            |                       |                   |
| 2009 2010                       | .148               | .148<br>.149                                 | .148<br>.149      | 8 148 .59 the recent market volatility to consid- relative total-return potential,<br>9 149 .60 er initiating a position here. Water are better income vehicles on t  |  |                        |                        |                       |                        |                      |                     |                     |                       |                       |                        | tial, a<br>on th     | and t<br>e ma        | here<br>rket,            |                            |                       |                   |
| 2011                            | .154               | .154   | .154              |   |  | utili                  | ty sto                 | cks ar                | e gen                  | erally               | less s              | uscep               | tible                 | espec                 | cially                 | in the               | Elect                | ric U                    | tility i                   | ndust                 | ry.<br>2011       |
| (A) Basi                        | EPS. E             | Excl. nonr                                   | ecurring          | gain (loss  | s); (B)                                | Dividend               | s historic             | ally paid             | in early F             | eb.,                 | (C) Incl.           | deferred            | charges.              | In '10: \$2           | 2.2 mill.,             | Co                   | mpany's              | Financia                 | al Streng                  | th                    | B+                |

| Company's Financial Strength | 0.1 |
|------------------------------|-----|
| Stock's Price Stability      | 90  |
| Price Growth Persistence     | 60  |
| Earnings Predictability      | 85  |
|                              |     |

(A) Basic EPS. Excl. nonrecurring gain (loss):
 (B) Dividends historically paid in early Feb., May, Aug., and Nov. ■ Div'd reinvestment plan available.
 (C) Incl. deferred charges. In '10: \$2.2 mill., \$0.05/sh.
 (D) In millions, adjusted for splits.
 (D) In millions, adjusted for splits.
 (D) In millions, adjusted for splits.
 (E) Excludes non-reg. rev.
 © 2011, Value Line Publishing LLC. All rights reserved. Factual material is obtained from sources believed to be reliable and is provided without warranties of any kind.
 (D) In millions, adjusted for splits.
 (E) Excludes non-reg. rev.
 To subscribe call 1-800-833-0046.

| CONN, WATE                    | ER SI   | ERVIC   | ES NDQ-                       | CTWS PR                                | 14 TRAILING<br>P/E RATI | 0 19.4 RE      | LATIVE 1.28              | B DIV'D 3.         | .5% VAI                              | LUE<br>NE                                      |                           |  |  |  |  |
|-------------------------------|---|---|-------------------------------|--|-------------------------|----------------|--------------------------|--------------------|--------------------------------------|--|---------------------------|--|--|--|--|
| RANKS                         |   | 30.4  | 29.76                         | 28.17                                  | 27.71                   | 25.61          | 28.95                    | 26.44              | 27.90                                | 28.27  | High                      |  |  |  |  |
| PERFORMANCE 3                 | Average   | 24.00   | GENDS                         | 21.51                                  | 20.23                   | 22.40          | 10.20                    | 17.01              | 20.00                                | 20.27  | 45                        |  |  |  |  |
| Technical 3                   | Average   |   | Nos Mov Avg<br>Price Strength |  |                         |                |                          |                    |                                      |  | 30                        |  |  |  |  |
|                               | Average<br>Above  | Shaded area   | indicales recession           | h+++++++++++++++++++++++++++++++++++++ | ++++1111                |                | ,                        | The second         | المليم وللم                          | <u>  1</u> 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | 22.5                      |  |  |  |  |
| SAFETY Z                      | Average   |   |                               |  | 1                       |                | 1                        | data.              | 11.                                  |  |                           |  |  |  |  |
| BETA .80 (1.00 =              | = Market)   |   |                               |  |                         |                |                          |                    |                                      |  |                           |  |  |  |  |
|                               |   |   |                               | ···.·                                  |                         |                |                          |                    |                                      |  | 9                         |  |  |  |  |
| Financial Strength            | B+  |   |                               |  | ·····.                  |                |                          |                    |                                      |  | 6                         |  |  |  |  |
| Price Stability               | 95  |   |                               |  |                         |                |                          |                    | ·······                              |  | 4                         |  |  |  |  |
| Price Growth Persistence      | e 25  |   | -                             |  |                         |                |                          |                    |                                      |  | 3                         |  |  |  |  |
| Earnings Predictability       | 80  | -   |                               |  |                         |                |                          | ht.h. III          |                                      |  | 550<br>                   |  |  |  |  |
|                               |   | 11111111  | վաղկու                        | d d d d d d d d d d d d d d d d d d d  |                         | hullin         | 4 millitt                |                    |                                      |  | (thous.)                  |  |  |  |  |
| © VALUE LINE PUBLISH          | HING LLC  | 2003  | 2004                          | 2005                                   | 2006                    | 2007           | 2008                     | 2009               | 2010                                 | 2011   | 2012/2013                 |  |  |  |  |
| SALES PER SH                  |   | 5.91  | 6.04                          | 5.81                                   | 5.68                    | 7.05           | 7.24                     | 6.93               | 7.65                                 |  |                           |  |  |  |  |
| EARNINGS PER SH               |   | 1.09  | 1.16                          | .88                                    | .81                     | 1.05           | 1.11                     | 1.19               | 1.13                                 | 1.28 <sup>A,B</sup>                            | 1.32 <sup>C</sup> /NA     |  |  |  |  |
| DIV'DS DECL'D PER SH          |   | .83   | .84                           | .85                                    | .86                     | .87            | .88                      | .90                | .92                                  |  |                           |  |  |  |  |
| BOOK VALUE PER SH             | SH  | 1.49  | 1.58                          | 1.96                                   | 1.96                    | 2.24           | 12.23                    | 3.28               | 13.05                                |  |                           |  |  |  |  |
| COMMON SHS OUTST'G            | (MILL)  | 7.97  | 8.04                          | 8.17                                   | 8.27                    | 8.38           | 8.46                     | 8.57               | 8.68                                 |  |                           |  |  |  |  |
| AVG ANN'L P/E RATIO           |   | 23.5  | 22.9                          | 28.6                                   | 29.0                    | 23.0           | 22.2                     | 18.4               | 20.7                                 | 21.2   | 20.6/NA                   |  |  |  |  |
| AVG ANN'L DIV'D YIELD         |   | 3.0%  | 3.1%                          | 3.4%                                   | 3.6%                    | 3.6%           | 3.6%                     | 4.1%               | 3.9%                                 | -  |                           |  |  |  |  |
| SALES (\$MILL)                | 100   | 47.1  | 48.5                          | 47.5                                   | 46.9                    | 59.0           | 61.3                     | 59.4               | 66.4                                 |  | Bold figures              |  |  |  |  |
| DEPRECIATION (\$MILL)         | 1   | 52.1%   | 6.0                           | 48.3%                                  | 43.7%                   | 40.8%          | 49.0%                    | 6.4                | 40.7%                                |  | are consensus<br>earnings |  |  |  |  |
| NET PROFIT (\$MILL)           |   | 9.2   | 9.4                           | 7.2                                    | 6.7                     | 8.8            | 9.4                      | 10.2               | 9.8                                  |  | estimates                 |  |  |  |  |
| INCOME TAX RATE               |   | 17.9%   | 22.9%                         |  | 23.5%                   | 32.4%          | 27.2%                    | 19.5%              | 35.2%                                |  | and, using the            |  |  |  |  |
| WORKING CAP'L (\$MILL         | _)  | d3.9  | d.7                           | 13.0                                   | 14.3%                   | 8.1            | d3.3                     | d13.1              | d14.070                              |  | P/E ratios.               |  |  |  |  |
| LONG-TERM DEBT (\$MI          | ĹL)   | 64.8  | 66.4                          | 77.4                                   | 77.3                    | 92.3           | 92.2                     | 112.0              | 111.7                                |  |                           |  |  |  |  |
| SHR. EQUITY (\$MILL)          | ויס   | 84.2  | 88.7                          | 94.9                                   | 96.7                    | 100.9          | 104.2                    | 109.3              | 114.0                                |  |                           |  |  |  |  |
| RETURN ON SHR. EQUI           | TY  | 10.9%   | 10.6%                         | 7.5%                                   | 6.9%                    | 8.7%           | 9.0%                     | 9.3%               | 8.6%                                 |  |                           |  |  |  |  |
| RETAINED TO COM EQ            | 05  | 3.2%  | 3.1%                          | .3%                                    | NMF                     | 1.6%           | 1.9%                     | 2.3%               | 1.6%                                 |  |                           |  |  |  |  |
| ALL DIV DS TO NET PRO         | or<br>am est in   | ast 8 days: 0   | up. 0 down. cons              | 95%<br>ensus 5-vear ear                | ninas arowth not        | available, BBa | 1976<br>sed upon 6 analy | sts' estimates. CE | 0176<br>Based upon 6 ai              | <br>nalysts' estimates                         |                           |  |  |  |  |
| ANNUAL                        | RATES   |   | 1                             |  |                         |                |                          | INDU               | STRY: Wa                             | ater Utility                                   |                           |  |  |  |  |
| of change (per share)         | 5 Yrs.  | 1 Yr.   | Cash Asset                    | mill.) 2<br>S                          | 5.4 1.0                 | .8             |                          |                    | ecticut Water Service Inc. primarily |  |                           |  |  |  |  |
| Sales<br>"Cash Flow"          | 4.0%  | 10.5%<br>5.5%   | Receivables                   | s<br>wa cost)                          | 6.5 10.1                | 16.4           | BUSINES                  | SS: Conne          | cticut Wat                           | ter Service, Inc. primarily                    |                           |  |  |  |  |
| Earnings                      | 1.5%  | -5.0%   | Other                         |  | 7.0 7.6                 | 2.0            | through th               | s a water ut       | s: Water A                           | ctivities. Re                                  | al Estate Trans-          |  |  |  |  |
| Book Value                    | 3.0%  | 3.0%  | Current Ass                   | ets :                                  | 20.0 20.4               | 20.4           | actions, a               | nd Services        | and Ren                              | tals. The W                                    | /ater Activities          |  |  |  |  |
| Fiscal QUARTERLY              | SALES (   | mill.) F  | III Property, Pl              | ant                                    |                         |                | segment s                | upplies publ       | lic drinking                         | g water to its                                 | customers. Its            |  |  |  |  |
| Year 1Q 2Q                    | 3Q  | 4Q Ye   | ar Accum Dep                  | at cost 44<br>reciation 12             | 48.2 471.6 23.0 127.4   |                | Real Estat               | e Transactio       | ons segmen                           | oldings The                                    | d in the sale of          |  |  |  |  |
| 12/31/09 13.4 15.2            | 16.6  | 14.2 5  | 0.4 Net Propert               | y 3                                    | 25.2 344.2              | 355.1          | Rentals se               | gment prov         | ides contra                          | acted service                                  | es to water and           |  |  |  |  |
| 12/31/11 17.5 17.4            | 21.0  | 13.7  | Total Asset                   | s 4                                    | 15.3 425.2              | 430.7          | wastewate                | r utilities a      | and other                            | clients, as                                    | well as leases            |  |  |  |  |
| 12/31/12                      |   |   |                               | S (Smill.)                             |                         |                | certain pro              | operties to 1      | third partie                         | s. This segu                                   | nent's services           |  |  |  |  |
| Fiscal EARNINGS               | 30 PER SHA  | ARE F   | Accts Paya                    | ble                                    | 6.5 6.6                 | 6.8            | ties: Line               | backer, its s      | ervice line                          | protection                                     | plan for public           |  |  |  |  |
| 12/31/08 20 35                | 34  | 22 1  | 11 Other                      |  | 1.6 <u>20.3</u>         | 1.5            | drinking v               | vater custon       | ners; and p                          | provision of                                   | bulk deliveries           |  |  |  |  |
| 12/31/09 .13 .27              | .67   | .12 1   | 19 Current Lial               | b :                                    | 33.1 35.1               | 34.7           | of emerge                | ncy drinkin        | g water to                           | businesses                                     | and residences            |  |  |  |  |
| 12/31/10 .12 .27              | .54   | .20 1.  | 13                            |  |                         |                | via tanke                | r truck. As        | s of Augu                            | ist 9, 2011<br>wimately 90                     | 000 customers             |  |  |  |  |
| 12/31/12 .21                  |   | .78 Drovided drinking water to approximately 90 or 300,000 people in 55 towns. Has 2  |                               |  |                         |                |                          |                    |                                      |  | 04 employees.             |  |  |  |  |
| Cal- QUARTERLY                | DIVIDEND  | IDS PAID Full as of 6/30/11 Chairman, C.E.O. & President: Eric V  |                               |  |                         |                |                          |                    |                                      |  | aburg. Inc.: CT.          |  |  |  |  |
| endar 1Q 2Q                   | 30  | Q 4Q Year Total Debt \$137.8 mill. Due in 5 Yrs. NA Address: 93 West Main Street, Clinton, (860) 669-8636. Internet: http://www.ctw |                               |  |                         |                |                          |                    |                                      |  | 1 06413. Tel.:            |  |  |  |  |
| 2008 .218 .218 .218 .218 .218 | .222  | 22 .222 .88 Li Deut (11.4 mill.<br>28 .228 .90 (860) 669-8636. Internet: h  |                               |  |                         |                |                          |                    |                                      | www.ctwate                                     | J.V.                      |  |  |  |  |
| 2010 .228 .228                | .228         .233         .233         .92         Leases, Uncapitalized Annual rentals NA         October 21, 20 |   |                               |  |                         |                |                          |                    | . 2011                               |  |                           |  |  |  |  |
| 2011 .233 .233                | .200  |   | Pension Li                    | ability \$16.7 mil                     | l. in '10 vs. \$14.9    | ) mill. in '09 |                          |                    |                                      |  |                           |  |  |  |  |
| INSTITUTIONA                  | AL DECISIO  | UNS 2014  | Pfd Stock \$                  | .8 mill.                               | Pfd Div                 | d Paid NMF     | TOTAL S                  | HAREHOLD           | ER RETUR<br>Dividen                  | URN<br>lends plus appreciation as of 9/30/2011 |                           |  |  |  |  |
| to Buy 27                     | 25  | 21  | Common St                     | ock 8 722 000 st                       | nares                   |                | 3 Mos.                   | 6 Mos.             | 1 Yr.                                | 3 Yrs  | . 5 Yrs.                  |  |  |  |  |
| to Sell 19<br>Hid's(000) 2764 | 19<br>2769  | 27  |                               |  | (5                      | 1% of Cap'l)   | -1.33%                   | -3 32%             | 8 33%                                | -2 749   | 36.93%                    |  |  |  |  |
|                               | 2100  | 2120  |                               |  |                         |                | 1.0070                   | 0.06.70            | 0.0070                               |  |                           |  |  |  |  |

©2011 Value Line Publishing LLC. All rights reserved. Factual material is obtained from sources believed to be reliable and is provided without warranties of any kind. THE PUBLISHER IS NOT RESPONSIBLE FOR ANY ERRORS OR OMISSIONS HEREIN. This publication is strictly for subscriber's own, non-commercial, internal use. No part of it may be reproduced, resold, stored or transmitted in any printed, electronic or other form, or used for generating or marketing any printed or electronic publication, service or product.

| MID                     | DLE                        | SEX           | WA                  | TER NDQ-MSEX RECENT 17.94 PIE RATIO 19.1 (Trailing: 19.1) RELATIVE 1.39 PIE RATIO 1.39 VLD 4.1% VALUE |   |  |                       |                        |                        |  |                   |                         |                     |                   |                        |                        |                      |                      |                         |                      |                     |
|-------------------------|----------------------------|---------------|---------------------|---|---|--|-----------------------|------------------------|------------------------|--|-------------------|-------------------------|---------------------|-------------------|------------------------|------------------------|----------------------|----------------------|-------------------------|----------------------|---------------------|
| TIMELIN                 | ESS 3                      | New 10/2      | 1/11                | High:<br>Low:   | 17.0<br>12.5  | 18.7<br>14.7   | 20.0<br>13.7          | 21.2<br>15.8           | 21.8<br>16.7           | 23.5<br>17.1                           | 20.5<br>16.5      | 20.2<br>16.9            | 19.8<br>12.0        | 17.9<br>11.6      | 19.3<br>14.7           | 19.3<br>16.5           |                      |                      | Target<br>2014          | Price                | Range<br>2016       |
| SAFETY                  | 2                          | New 10/2      | 1/11                | LEGEN   | NDS<br>20 x Divide  | nds p sh<br>lerest Bate  |                       |                        |                        |  |                   |                         |                     | 0.594             |                        |                        |                      |                      |                         |                      | -64                 |
| BETA .7                 | 5 (1.00 =                  | Market)       | WI1                 | 3-for-2 sp  | lative Price  | Strength   | _                     |                        |                        |  |                   |                         |                     |                   |                        |                        |                      |                      |                         |                      | 48                  |
| 2014                    | 4-16 PR                    | DJECTIC<br>Ar | NS<br>n'l Total     | Options: 1<br>Shaded  | No<br>areas indic   | ate recess   | ions                  |                        |                        |  |                   |                         | 12.00               |                   |                        |                        |                      |                      |                         |                      | -32                 |
| P<br>High               | rice (+                    | Gain<br>40%)  | Return<br>12%       |   |   | n ilmir  |                       | in the                 | այր <sup>եր</sup> ներ  | 11111111111111111111111111111111111111 | որդիրո            | 44LUART                 | no halli            | 1                 | ակում                  | ուսուն                 |                      |                      |                         |                      | 20                  |
| Insider                 | Decisi                     | ons           | 3%                  | 1111  | Harming   |  | 1                     |                        |                        |  |                   |                         | <u> </u>            | dillin.           |                        |                        |                      |                      |                         |                      | - 12                |
| to Buy                  | N D J                      | F M A         | 0 0 0<br>0 0 0      |   |   | ·····  |                       | · ·.·.                 | ·····                  |  |                   |                         |                     | <i>.</i>          |                        |                        |                      |                      |                         |                      | -8                  |
| to Sell                 |                            | 0 1 0         | 0 0 0               |   |   | 12 au  |                       |                        |                        |  | ·····             | ······                  | *******             |                   | ·                      |                        |                      | % то                 | T. RETUR                | N 9/11               | -0                  |
| to Buy                  | 402010<br>39               | 102011<br>25  | 2Q2011<br>31        | Percent   | 1 12 <del>-</del>   |  |                       |                        |                        | _                                      |                   |                         |                     | -11               |                        | ****                   |                      | 1 yr.                | STOCK<br>5.5            | INDEX<br>-4.8        | E                   |
| to Sell<br>Hid's(000)   | 21<br>6031                 | 25<br>6200    | 24<br>6377          | traded  | 4 -   | millin   | ահորու                | ավան                   | uullinu                | ullillul                               | illiluli          | ահենու                  |                     |                   | puttitu                | hilld                  |                      | 3 yr.<br>5 yr.       | 7.8                     | 16.6                 | -                   |
| 1995<br>4.56            | 4.52                       | 4.72          | <b>1998</b><br>4.39 | <b>1999</b><br>5.35   | 2000<br>5.39  | 2001<br>5.87   | 2002<br>5.98          | 2003<br>6.12           | 2004<br>6.25           | 2005<br>6.44                           | 6.16              | 6.50                    | 6.79                | 6.75              | 6.60                   | 6.75                   | 6.90                 | Revenue              | es per sh               | UB. LLC              | 8.25                |
| 1.01                    | .94                        | 1.02          | 1.02                | 1.19  | .99   | 1.18   | 1.20                  | 1.15                   | 1.28                   | 1.33                                   | 1.33              | 1.49                    | 1.53                | 1.40              | 1.55                   | 1.60                   | 1.70                 | "Cash F              | low" per                | sh                   | 2.00                |
| .00                     | .60                        | .07           | .58                 | .76   | .61   | .60  | .63                   | .65                    | .73                    | .67                                    | .62               | .67                     | .89                 | .71               | .50                    | .33                    | .74                  | Div'd De             | cl'd per s              | sh <sup>B</sup> ∎    | .80                 |
| 1.08                    | .73<br>5.85                | 1.20 6.00     | 2.68<br>6.80        | 2.33  | 1.32  | 1.25   | 1.59                  | 1.87                   | 2.54<br>8.02           | 2.18<br>8.26                           | 2.31<br>9.52      | 1.66                    | 2.12                | 1.49              | 1.90                   | 1.50                   | 1.55<br>11.25        | Cap'l Sp<br>Book Va  | ending p<br>lue per sl  | ersh<br>h            | 1.75<br>11.75       |
| 8.30                    | 8.41                       | 8.54          | 9.82                | 10.00   | 10.11   | 10.17  | 10.36                 | 10.48                  | 11.36                  | 11.58                                  | 13.17             | 13.25                   | 13.40               | 13.52             | 15.57                  | 15.70                  | 16.00                | Commo                | n Shs Ou                | tst'g C              | 17.00               |
| .82                     | .90                        | .77           | .79                 | 1.00  | 1.87  | 1.26   | 1.28                  | 1.71                   | 1.39                   | 1.46                                   | 1.23              | 1.15                    | 1.19                | 1.40              | 1.13                   | Value                  | Line<br>Line         | Relative             | P/E Ratio               |                      | 1.15                |
| 6.5%                    | 6.4%                       | 6.3%          | 5.4%                | 4.4%  | 4.2%  | 3.8%   | 3.7%                  | 3.5%                   | 3.4%                   | 3.5%                                   | 3.7%              | 3.7%                    | 4.0%                | 91.2              | 4.2%                   | 106                    | 110                  | Avg Anr              | es (\$mill)             | ield                 | 3.6%                |
| Total De                | bt \$139<br>\$134.7        | .1 mill. D    | ue in 5             | Yrs \$25.0<br>st \$6.0 mi   | mill.   | 7.0  | 7.8                   | 6.6                    | 8.4                    | 8.5                                    | 10.0              | 11.8                    | 12.2                | 10.0              | 14.3                   | 15.0                   | 16.0                 | Net Pro              | lit (\$mill)            |                      | 21.0                |
| (LT inter               | est cove                   | rage: 4.5     | x)                  | (43% 0  | f Cap'l)  | 34.8%<br>7.4%  | 7.2%                  | 32.8%                  | 7.3%                   | 7.5%                                   | 33.4%<br>7.5%     | 7.5%                    | 7.7%                | 7.7%              | 7.5%                   | 7.5%                   | 7.5%                 | AFUDC                | % to Net I              | Profit               | 7.0%                |
| Pensior                 | Assets                     | -12/10 \$     | 30.0 mill           | (1070 0   | , cop.i   | 53.6%<br>43.9%   | 52.1%                 | 53.8%                  | 53.8%<br>42.5%         | 55.3%<br>41.3%                         | 49.5%             | 49.0%                   | 45.6%               | 46.6%             | 43.1%                  | 43.0%                  | 43.0%                | Long-Te              | rm Debt F<br>n Equity F | Ratio<br>Ratio       | 42.0%<br>56.0%      |
| Pfd Sto                 | ck 3.4 m                   | ill Pfd D     | Oblig. \$4          | 2.1 mill.<br>till   |   | 164.5  | 168.0                 | 181.1                  | 214.5                  | 231.7                                  | 264.0             | 268.8                   | 259.4               | 267.9             | 310.5                  | 315                    | 325                  | Total Ca             | ipital (\$mi            | ill)                 | 375                 |
| Commo                   | n Stock                    | 15.618.3      | 17 shs              |   |   | 199.1<br>5.6%  | 6.0%                  | 230.9<br>5.0%          | 5.1%                   | 288.0<br>5.0%                          | 5.1%              | 5.6%                    | 366.3<br>5.8%       | 5.0%              | 405.9                  | 425<br>5.0%            | 5.0%                 | Return of            | on Total C              | ap'l                 | 5.5%                |
| as of 8/                | 2/11                       |               |                     |   |   | 9.1%<br>9.3%   | 9.6%                  | 7.9%                   | 8.5%<br>9.0%           | 8.2%<br>8.6%                           | 7.5%              | 8.6%<br>8.7%            | 8.6%<br>8.9%        | 7.0%              | 8.1%                   | 8.5%<br>8.5%           | 8.5%<br>8.5%         | Return of Return of  | on Shr. Ec<br>on Com E  | quity<br>quity       | 10.5%<br>10.5%      |
| MARKE                   | T CAP:                     | \$275 mil     | lion (Sm            | all Cap)  | 6/20/11   | .5%  | 1.3%                  | NMF                    | .9%                    | .6%                                    | 1.3%              | 1.8%                    | 2.0%                | .1%               | 2.1%                   | 2.0%                   | 2.0%                 | Retaine              | d to Com                | Eq                   | 3.5%                |
| (\$MI<br>Cash A         | L)<br>ssets                | mon           | 4.3                 | 2.5   | 4.3   | BUSIN  | ESS: Mi               | ddlesex \              | Vater Co               | npany e                                | ngages i          | n the ow                | nership             | 2010, th          | ne Middle              | esex Sys               | tem acco             | ounted fo            | r 64% of                | total re             | venues.             |
| Other<br>Current        | Assets                     | -             | 17.7                | 20.3  | 4.3 BUSINESS: Middlesex Water Company engages in the ownership 2010, the Middlesex System accounted for 64% of total and operation of regulated water utility systems in New Jersey, Del-<br>25.5 ware, and Pennsvitania It also norrates water and wastewater President CEO and Chairmap: Dennis W. Doll Officer |  |                       |                        |                        |  |                   |                         |                     |                   |                        | corporat               | ed: NJ.<br>irectors  |                      |                         |                      |                     |
| Accts P<br>Debt De      | ayable<br>Je               |               | 4.3                 | 6.4   | 5.7   | system<br>NJ and   | IS under              | contract (<br>Middlese | on behalf<br>s System  | of munic                               | ipal and water se | private c<br>ervices to | lients in<br>60.000 | own 3.3<br>Group. | 9% of th               | e commo                | on stock;<br>Addres  | BlackRo<br>s: 1500   | ck, 7.0%<br>Ronson      | ; The Va<br>Road, Is | inguard<br>elin, NJ |
| Current                 | Liab.                      | _             | 60.7                | 40.7  | 43.7  | retail   | customer              | s, primar              | ily in Mi              | dlesex                                 | County,           | New Jer                 | rsey, in            | 08830.            | Tel.: 732              | -634-150               | 0. Interne           | et: www.r            | niddlesex               | water.co             | m.                  |
| ANNUA                   | L RATE                     | S Past        | 25%<br>Pa           | st Est'd  | 415%  | We<br>ny   | to Th                 | e Val                  | ue Li                  | ne In                                  | Water<br>westn    | ent                     | ipa-<br>Sur-        | tion.<br>flowi    | Howe<br>ng, I          | ever, 1<br>Middle      | n ord                | er to<br>will l      | have                    | the w                | vest                |
| Revenu                  | e (per sh)<br>ies<br>Flow" | 10 Yrs<br>3.0 | 5Y                  | rs. to  | '14'16<br>3.5%  | vey.<br>1897   | The<br>, and          | comp                   | any<br>s regu          | was<br>ilated                          | incorp            | orate<br>r ser          | d in<br>vices       | heav              | ily in<br>ructui       | repai<br>re, wł        | ring a<br>nich v     | and ir<br>vill h     | nprovi<br>ampei         | ing it<br>the        | s in-<br>bot-       |
| Earning                 | ls<br>ds                   | 2.5           | % 4<br>% 1          | .5%   | 6.0%<br>2.0%  | to r   | esider                | ntial a                | and co                 | omme                                   | rcial             | custo                   | ners                | tom               | line.                  | All to                 | ld, we               | e proj               | ect th                  | at an                | nual<br>mid-        |
| Book V                  | alue                       | 4.5           | % 5                 | .5%   | 2.0%  | nia.   | It al                 | so ow                  | ns an                  | d op                                   | erates            | noni                    | egu-                | singl             | e-digi                 | t rate                 | to 20                | 14-20                | 16.                     | ube u                | haalt               |
| Cal-<br>endar           | Mar.31                     | Jun. 30       | Sep. 30             | (\$ mm.)<br>Dec. 31   | Full<br>Year  | reve   | nues                  | comes                  | from                   | tems.                                  | Midd              | lesex                   | Sys-                | may               | app                    | beal                   | to s                 | ome                  | cons                    | ervat                | tive,               |
| 2008                    | 20.8<br>20.6               | 23.0<br>23.1  | 25.7<br>25.5        | 21.5<br>22.0  | 91.0  | abou   | , whi<br>it 60,0      | ch pi<br>000 cu        | ovides<br>stome        | s wat                                  | ter se<br>New J   | ervice                  | s to                | sister            | me-on<br>ntcy          | of its                 | d in busi            | ness                 | ors.<br>allow           | The<br>s for         | con-<br>the         |
| 2010 2011               | 21.6<br>24.0               | 26.5<br>26.1  | 29.6<br>30.0        | 25.0<br>25.9  | 102.7   | The  | com                   | pany<br>s are          | 's ne                  | ar- a                                  | ind l             | ong-t                   | erm                 | stock             | to la                  | argely<br>certai       | avoio                | d sha                | rp pri<br>times         | ce sw<br>. Its       | rings<br>Beta       |
| 2012                    | 25.0                       | 27.0          | 31.0                | 27.0  | 110   | num  | ber c                 | f rate                 | case                   | s that                                 | at are            | awa                     | iting               | is 0.             | 75, a                  | nd th                  | e equ                | ity c                | arries                  | a S                  | afety               |
| endar                   | Mar.31                     | Jun. 30       | Sep. 30             | Dec. 31   | Year  | reco   | up e                  | xpens                  | es tl                  | nat                                    | stemn             | red                     | from                | comp              | ared                   | to the                 | e othe               | er wa                | ter ut                  | ilities              | un-                 |
| 2008                    | .15                        | .26<br>.21    | .35<br>.29          | .13<br>.12  | .89   | elev<br>mitt   | ated<br>ed a          | main                   | tenano<br>st to        | the I                                  | itlays,<br>Delawa | it<br>are P             | sub-<br>ublic       | der h             | <i>Value</i><br>nighes | <i>Line</i><br>st divi | dend                 | age, A<br>yield      | (recen                  | sex o<br>tly 4.      | ffers<br>1%),       |
| 2010 2011               | .11                        | .31<br>.23    | .37<br>.35          | .17<br>.18  | .96   | Serv   | vice (                | commi                  | ssion                  | for<br>\$6.9                           | an ir<br>millio   | creas                   | e in<br>veral       | and t             | the pa                 | yout a                 | appea<br>aid a       | rs sec<br>divid      | ure. I<br>end e         | n fact               | , the<br>vear       |
| 2012                    | .18                        | .25           | .37                 | .20   | 1.00  | rate   | case                  | rulin                  | igs ar                 | e exp                                  | bected            | over                    | the                 | since             | 191                    | 2. Ho                  | wever                | , inve               | estors                  | that                 | are                 |
| endar                   | Mar.31                     | Jun.30        | Sep.30              | Dec.31  | Year  | adva   | ance r                | evenu                  | es and                 | d sha                                  | re net            | . Fur                   | ther-               | not               | apply                  | here.                  | Midd                 | llesex               | stock                   | typi                 | cally               |
| 2007 2008               | .173<br>.175               | .173<br>.175  | .173<br>.175        | .175<br>.178  | .69   | .69 more, and most important, water is one of, trades in a tight band, and already<br>.70 if not the most, essential part of life. priced at the low end of our projected Ta |                       |                        |                        |  |                   |                         |                     |                   |                        | Tar-                   |                      |                      |                         |                      |                     |
| 2009 2010               | .178                       | .178<br>.180  | .178<br>.180        | .180<br>.183  | .71   | Wat  | er pro                | nd de                  | s, the                 | refore<br>for                          | , are<br>water    | almos                   | st as<br>nt to      | get I             | Price<br>ranke         | Range<br>d 3 (A        | e for 2<br>verag     | 2014–<br>e) for      | 2016.<br>Timel          | It is<br>iness       | also                |
| 2011                    | .183                       | .183          | .183                |   |   | cont   | inue                  | to gro                 | ow alc                 | ng w                                   | ith th            | ie poj                  | pula-               | Ian (             | Gendle                 | er                     | 0                    |                      | Octobe                  | er 21,               | 2011                |
| (A) Dilute<br>early Not | ed earnir<br>vember.       | ngs. Next     | earning             | s report d  | lue (B)<br>May  | Divident<br>, Aug., a  | ds histor<br>nd Nover | ically pa<br>nber.= Di | id in mi<br>v'd reinve | d-Fed.,<br>stment                      | (C) In mi         | llions, ad              | justed for          | r splits.         |                        | Sto                    | npany's<br>ck's Pric | Financia<br>e Stabil | al Streng<br>ity        | th                   | 8+<br>95            |
|                         |                            |               |                     |   | plan  | availabl   | e.                    |                        |                        |  |                   |                         |                     |                   |                        | Ear                    | nings Pi             | redictabl            | lity                    |                      | 30<br>85            |

© 2011, Value Line Publishing LLC. All rights reserved. Factual material is obtained from sources believed to be reliable and is provided without warranties of any kind. THE PUBLISHER IS NOT RESPONSIBLE FOR ANY ERRORS OR OMISSIONS HEREIN. This publication is strictly for subscriber's own, non-commercial, internal use. No part of it may be reproduced, tesold, stored or transmitted in any printed, electronic or other form, or used for generating or marketing any printed or electronic publication, service or product.

| SJV                   | VCC                   | RP.              | NYSE-                    | -S.W RECENT 23.38 P/E RATIO 22.9 (Trailing: 26.9) RELATIVE 1.67 VID 3.0% |                            |                         |                      |               |                      |           |               |                         |           |                    |                         |                 | E                    |                       |                         |               |  |
|-----------------------|-----------------------|------------------|--------------------------|--|----------------------------|-------------------------|----------------------|---------------|----------------------|-----------|---------------|-------------------------|-----------|--------------------|-------------------------|-----------------|----------------------|-----------------------|-------------------------|---------------|--|
| TIMELIN               | ESS 3                 | Raised 8         | 12/11                    | High:  | 20.3                       | 17.8                    | 15.1                 | 15.0          | 19.6<br>14.6         | 27.8      | 45.3          | 43.0                    | 35.1      | 30.4               | 28.2                    | 26.8<br>20.9    |                      |                       | Target                  | Price         | Range                                    |
| SAFETY                | 3                     | New 4/22         | /11                      | LEGEN  | NDS                        | nds o sh                |                      | 12.0          | 14.0                 | 10.1      |               |                         | 20.0      | 10.2               | 2.1.0                   |                 |                      |                       | 2014                    | 2015          | 2010                                     |
| TECHNI                | cal 3                 | Lowered          | 9/30/11                  | div<br>Re  | ided by In<br>lative Price | terest Rate<br>Strength | -                    |               |                      |           |               |                         |           |                    |                         |                 |                      |                       |                         |               | $\begin{bmatrix} 80 \\ 60 \end{bmatrix}$ |
| BETA .9               | 0 (1.00 =             | Market)          | NS                       | 3-for-1 sp<br>2-for-1 sp   | lit 3/04<br>lit 3/06       |                         |                      |               |                      |           |               |                         |           |                    |                         |                 |                      |                       |                         |               | <b>50</b>                                |
| 201                   | de lo Filo            | A                | nn'l Total               | Shaded   | No<br>areas indic          | ate recess              | ions                 |               |                      |           | الر           | hlulul                  | hun       |                    |                         |                 | _                    |                       |                         |               | L <sub>30</sub>                          |
| High                  | 40 (1                 | 70%)             | 17%                      |  |                            |                         |                      |               |                      |           | 14111         | 14                      |           | Tillion            | , լ <sup>լլ</sup> կյյու | • التعن         |                      |                       |                         |               | -25                                      |
| Insider               | Decisi                | ions             | 970                      |  |                            | 1                       | -                    |               | 1"ppnt               | 1111      |               |                         |           | 1.0                |                         |                 |                      |                       |                         |               | 15                                       |
| toBuy                 | N D J<br>0 1 0        | F M A<br>0 1 0   | M J J<br>1 0 0           |  | "mante                     | 1 miles                 | 6.0 <sub>000</sub> . | n'tutte.      |                      |           |               | .                       |           |                    |                         |                 |                      |                       |                         |               | 10                                       |
| Options<br>to Sell    | 010                   | 0 0 0 0 0 0      | 0 0 0                    |  |                            |                         |                      | ·             |                      |           | ·····         |                         | ·*···     |                    |                         |                 | -                    | % TO                  | I<br>T DETIIC           | <br> N 0/11   | -7.5                                     |
| Institu               | tional D              | Decision         | 15                       |  |                            |                         |                      |               |                      |           |               |                         |           |                    |                         |                 |                      | /* 10                 | THIS STOCK              | L ARITH.      |  |
| to Buy                | 34                    | 26               | 31                       | Percent<br>shares  | t 21 -<br>14 -             |                         |                      |               |                      |           |               |                         | d dd a    |                    |                         |                 |                      | 1 yr.<br>3 yr.        | -9.0                    | -4.8<br>25.0  | FI                                       |
| to Sell<br>Hkd's(000) | 8640                  | 8648             | 8839                     | traded   | 7 -                        | lunum                   | unun                 | untin         | uuuuu                | milition  | uulluu        |                         |           | hillion            | mullu                   | huult           |                      | 5 yr.                 | -17.5                   | 16.6          | F  |
| 1995                  | 1996                  | 1997             | 1998                     | 1999   | 2000                       | 2001                    | 2002                 | 2003          | 2004                 | 2005      | 10.25         | 2007                    | 12 12     | 11.69              | 2010                    | 12 65           | 12 50                | Rovonue               | UE LINE P               | UB. LLC       | 12 25                                    |
| 4.99                  | 1.43                  | 1.27             | 1.26                     | 1.43   | 1.23                       | 1.49                    | 1.55                 | 1.75          | 1.89                 | 2.21      | 2.38          | 2.30                    | 2.44      | 2.21               | 2.37                    | 2.60            | 2.70                 | "Cash F               | low" per                | sh            | 3.00                                     |
| .59                   | .96                   | .80              | .76                      | .87  | .58                        | .77                     | .78                  | .91           | .87                  | 1.12      | 1.19          | 1.04                    | 1.08      | .81                | .84                     | 1.00            | 1.10                 | Earning               | s per sh                | A             | 1.40                                     |
| .35                   | 1.06                  | .38              | 1.81                     | 1.77   | .41                        | 2.63                    | 2.06                 | .49           | 2.31                 | 2.83      | 3.87          | 6.62                    | 3.79      | 3.17               | .68                     | 3.80            | 3.75                 | Cap'l Sp              | endina p                | ersh          | 4.45                                     |
| 5.58                  | 6.31                  | 7.02             | 7.53                     | 7.88   | 7.90                       | 8.17                    | 8.40                 | 9.11          | 10.11                | 10.72     | 12.48         | 12.90                   | 13.99     | 13.66              | 13.75                   | 13.70           | 14.25                | Book Va               | lue per s               | h             | 16.20                                    |
| 19.50                 | 19.02                 | 19.02            | 19.01                    | 18.27  | 18.27                      | 18.27                   | 18.27                | 18.27         | 18.27                | 18.27     | 18.28         | 18.36                   | 18.18     | 18.50              | 18.55                   | 18.60           | 20.00                | Common                | n Shs Ou                | tst'g C       | 22.50                                    |
| .66                   | .43                   | .65              | .68                      | .88  | 2.15                       | .95                     | .94                  | .88           | 1.04                 | 1.05      | 1.27          | 1.77                    | 1.58      | 1.91               | 1.89                    | Value           | Line                 | Relative              | P/E Ratio               |               | 1.65                                     |
| 6.0%                  | 5.7%                  | 4.3%             | 3.9%                     | 3.0%   | 2.1%                       | 3.0%                    | 3.4%                 | 3.5%          | 3.0%                 | 2.4%      | 2.0%          | 1.7%                    | 2.3%      | 2.8%               | 2.8%                    | estim           | ates                 | Avg Ann               | n'l Div'd Y             | īeld          | 2.5%                                     |
| CAPITA<br>Total De    | L STRU                | CTURE a          | as of 6/30               | )/11<br>Vrs 564 3  | mill                       | 136.1                   | 145.7                | 149.7         | 166.9                | 180.1     | 189.2         | 206.6                   | 220.3     | 216.1              | 215.6                   | 235             | 250                  | Revenue               | es (\$mill)             |               | 300                                      |
| LT Debt               | \$344.8               | mill.            | T Interes                | st \$17.0 n  | nill.                      | 34.5%                   | 40.4%                | 36.2%         | 42.1%                | 41.6%     | 40.8%         | 39.4%                   | 39.5%     | 40.4%              | 39.7%                   | 40.0%           | 40.0%                | Income                | Tax Rate                |               | 39.0%                                    |
| (L1 inter<br>coverag  | rest earn<br>e: 3.0x) | ed: 3.2x:        | total inter              | rest<br>(57% o   | f Cap'l)                   | 4.4%                    | 4.2%                 | 1.6%          | 2.1%                 | 1.6%      | 2.1%          | 2.7%                    | 2.3%      | 2.0%               | 3.6%                    | 5.0%            | 5.0%                 | AFUDC                 | % to Net I              | Profit        | 5.0%                                     |
| 1 03505               | Uncani                | talizod: /       | Annual re                | ntale \$4.2  | mill                       | 42.4%                   | 41.7%                | 45.6%         | 43.7%                | 42.6%     | 41.8%         | 47.7%                   | 46.0%     | 49.4%              | 46.3%                   | 57.5%           | 55.0%                | Long-Te               | rm Debt F<br>n Fauity F | Ratio         | 48.5%                                    |
| Leudea                | oncupi                | talled. 7        |                          | 111013 94.2  |                            | 259.4                   | 263.5                | 306.0         | 328.3                | 341.2     | 391.8         | 453.2                   | 470.9     | 499.6              | 550.7                   | 600             | 635                  | Total Ca              | pital (\$mi             | 11)           | 755                                      |
| Pension               | n Assets              | -12/10 \$        | 10.8 mill.<br>Oblig. \$5 | 8.8 mill.  |                            | 367.8                   | 390.8                | 428.5         | 456.8                | 484.8     | 541.7         | 645.5                   | 684.2     | 718.5              | 785.5                   | 815             | 860                  | Net Plan              | t (\$mill)              |               | 1000                                     |
| Pfd Sto               | ck None               |                  | -                        |  |                            | 9.4%                    | 9.3%                 | 10.0%         | 8.7%                 | 10.6%     | 9.7%          | 8.2%                    | 5.8%      | 6.0%               | 6.1%                    | 4.5%            | 7.5%                 | Return o              | on Total C              | api           | 5.5%<br>8.0%                             |
| Commo                 | n Stock               | 18,577,6         | 30 shs.                  |  |                            | 9.4%                    | 9.3%                 | 10.0%         | 8.7%                 | 10.6%     | 9.7%          | 8.2%                    | 8.0%      | 6.0%               | 6.1%                    | 7.5%            | 7.5%                 | Return o              | on Com E                | quity         | 8.0%                                     |
| MARKE                 | T CAP:                | \$425 mil        | lion (Sm                 | all Cap)   |                            | 4.1%                    | 3.8%                 | 4.7%          | 3.6%                 | 5.6%      | 5.2%          | 3.5%                    | 3.3%      | 1.2%               | 1.2%                    | 2.0%            | 2.5%                 | Retained<br>All Div'd | to Com                  | Eq            | 3.0%<br>62%                              |
| CURRE                 | NT POS                | ITION            | 2009                     | 2010   | 6/30/11                    | BUSIN                   | ESS: S.              | W Corp        | oration e            | angages   | in the        | productio               | n. pur-   | Austin.            | Texas.                  | The con         | npany of             | ffers nor             | regulate                | d water-      | related                                  |
| Cash A                | ssets                 |                  | 1.4                      | 1.7  | 45.4                       | chase,                  | storage,             | purificati    | on, distril          | oution, a | nd retail     | sale of w               | ater. It- | service            | s, includi              | ng water        | system o             | peration              | s, cash r               | emittanc      | es, and                                  |
| Current               | Assets                |                  | 28.0                     | 38.0   | 83.8                       | serve a                 | a populat            | ion of ap     | o approx<br>proximat | ely one   | million p     | connectio<br>cople in t | he San    | mainter            | real est                | ate inves       | vices. 53<br>Iments. | Has 375               | i employ                | ees. Ch       | airman:                                  |
| Debt D                | 'ayable<br>ue         |                  | 6.6<br>6.9               | 5.5<br>5.1   | 9.8<br>7.9                 | Jose a                  | rea and              | 8,700 co      | nnections            | that set  | ve appro      | ximately<br>San Anto    | 36,000    | Charles<br>San Jos | J. Toen                 | iskoetter.      | Inc.: CA             | . Addres              | is: 110 V               | I. Taylor     | Street,                                  |
| Other                 | Liab.                 |                  | 18.5                     | 18.6   | 21.7                       | Rate                    | e inci               | ease          | a are                | reall     | v hel         | ning                    | S.IW      | The                | stoc                    | k has           | hee                  | n do                  | ing                     | relati        | velv                                     |
| Fix. Ch               | g. Cov.               |                  | 352%                     | 400%   | 250%                       | Cor                     | p                    | Inde          | ed, t                | he w      | ater          | utility                 | got       | well               | latel                   | y. It l         | nas he               | eld its               | grou                    | nd fo         | r the                                    |
| ANNUA<br>of chang     | L RATE<br>(persh)     | S Past<br>10 Yrs | Pa<br>5 Y                | st Est'd<br>rs. to   | '14'16                     | earn                    | ings g               | growth        | 1 back               | c on t    | rack          | in the                  | sec-      | most               | part                    | since           | our                  | July                  | revie                   | w, de         | spite                                    |
| Revenu<br>"Cash       | les<br>Flow"          | 6.5<br>6.0       | % 5.<br>% 3.             | .5%<br>.5%   | 2.0%<br>4.0%               | digit                   | top-l                | ine ga        | in.                  | in Br     | .19 10        | u uo                    | ubic      | many               | y outs                  | ide th          | e wat                | er util               | lity in                 | dustr         | y.                                       |
| Earning               | js<br>ids             | 2.0<br>5.0       | 0% -1.<br>0% 5.          | .5%<br>.5%   | 7.5%<br>3.5%               | <br>mal                 | and                  | are           | e lil<br>esh d       | kely      | to            | cont                    | inue      | But                | it sti                  | ill do          | es no                | ot sta                | nd of                   | ut in         | any                                      |
| Book V                | alue                  | 6.0              | 6.                       | .5%  | 2.5%                       | We'v                    | e inc                | reased        | lour                 | secon     | d-half        | and                     | 2012      | wate               | r util                  | lity s          | pace                 | is ap                 | pealir                  | ng at         | this                                     |
| Cal-<br>endar         | Mar.31                | Jun. 30          | Sep. 30                  | () mill.)<br>Dec. 31   | Full<br>Year               | estir                   | nates                | to acc        | atory                | for th    | e add         | ed ber                  | nefits    | time               | , inve                  | estors          | have<br>g veh        | bett                  | to ch                   | owth          | and                                      |
| 2008                  | 41.3                  | 60.0             | 69.5                     | 49.5   | 220.3                      | may                     | well                 | prove         | light                | if fa     | vorab         | le rul                  | lings,    | It is              | an a                    | verage          | selec                | ction i               | in bot                  | h reg         | ards,                                    |
| 2009 2010             | 40.0                  | 58.2<br>54.1     | 69.3<br>70.3             | 48.6<br>50.8   | 216.1                      | which                   | h we                 | are n         | ot ant               | icipat    | ing at        | t this                  | time,     | and                | also                    | lacks           | 3- to                | 5-ye                  | ar ap                   | opreci        | ation                                    |
| 2011                  | 43.7                  | 59.0             | 77.3                     | 55.0   | 235                        | How                     | vever                | onnig         | eratir               | ng co     | osts          | are                     | also      | that               | it is u                 | inder           | and th               | he cos                | ts of a                 | loing         | busi-                                    |
| 2012<br>Cal.          | 47.0<br>FI            | ARNINGS I        | PER SHAR                 | EA   | Eull                       | like                    | ly to                | conti         | nue i                | to mo     | ount.         | Water                   | dis-      | ness               | that                    | are li          | ikely                | to con                | ntinue                  | e to s        | well.                                    |
| endar                 | Mar.31                | Jun. 30          | Sep. 30                  | Dec. 31  | Year                       | and                     | fede                 | ral st        | tanda                | rds.      | Mean          | while,                  | the       | the d              | compa                   | ny fro          | m goi                | ing ou                | it and                  | mak           | ing a                                    |
| 2008                  | .15                   | .34              | .44                      | .15  | 1.08                       | majo                    | ority d              | of pip        | elines               | and       | waste         | water                   | sys-      | splas              | sh in                   | the a           | cquisi               | tion 1                | marke                   | t. Th         | e in-                                    |
| 2010                  | .05                   | .24              | .44                      | .11  | .84                        | As a                    | resul                | t, ope        | rating               | g costs   | are o         | expect                  | ed to     | ists               | great                   | oppor           | tunity               | to fi                 | urther                  | build         | d out                                    |
| 2011 2012             | .03                   | .29              | .54                      | .14  | 1.00                       | rem                     | ain o                | n an          | upw                  | ard       | trajec        | tory,                   | thus      | the h              | ousine                  | ess mo          | del vi               | a exp                 | ansio                   | n into        | new                                      |
| Cal-                  | QUAR                  | TERLY DI         | VIDENDS F                | PAID B.  | Full                       | case                    | ing a                | ny o<br>prove | t the ments          | afore     | ement<br>SJW. | ioned                   | the       | shee               | t and                   | a dea           | urth o               | f cash                | on h                    | a ba          | how-                                     |
| endar                 | Mar.31                | Jun.30           | Sep.30                   | Dec.31   | Year                       | mea                     | ntime                | , is n        | ot exa               | actly     | flush         | with                    | cash,     | ever,              | mak                     | e suc           | h an                 | und                   | ertaki                  | ng h          | ighly                                    |
| 2007                  | .15                   | .15              | .15                      | .15  | .60                        | desp                    | nte a<br>simi        | recen         | t debi<br>nare       | and/or    | ing.<br>deb   | we su<br>t offe         | spect     | cerns              | cely, a                 | and, v<br>r the | vorse<br>susta       | yet, inabil           | raise                   | some<br>f the | divi-                                    |
| 2009                  | .165                  | .165             | .165                     | .165   | .66                        | will                    | be re                | quire         | d in o               | order     | to fo         | ot the                  | bill,     | dend               | if sor                  | nethir          | ng doe               | sn't g                | ive.                    |               | 0011                                     |
| 2011                  | 173                   | .173             | .173                     | .11  | .00                        | ther                    | eby fu               | rther         | dıluti               | ng fut    | ure g         | ains.                   |           | Andr               | е J. С                  | ostan           | za                   | (                     | Jctobe                  | r 21,         | 2011                                     |
| (A) Dilu              | leg ear               | ings. E)         | ciudes r                 | nonrecurri   | ing   due                  | to round                | ing.                 |               |                      |           | (c) in m      | mons.                   |           |                    |                         | Cor             | npany's              | rmancia               | n otreng                | wit .         | DT                                       |

(A) Linuted earnings. Excludes nonrecurring oue to rounding. Losses : 03, S1.97; '04, S3.78; '05, S1.09; '06, '(B) Dividends historically paid in early March, S16.36; '08, S1.22; '10, 46¢. Next earnings June, September, and December. = Div/d rein-report due late Oct. Quarterly egs. may not add | vestment plan available. © 2011, Value Line Publishing LLC. All rights reserved. Factual material is obtained from sources believed to be reliable and is provided without warranties of any kind. THE PUBLISHER IS NOT RESPONSIBLE FOR ANY ERRORS OR OMISSIONS HEREIN. This publication is strictly for subscriber's own, non-commercial, internal use. No part of it may be reproduced, resold, stored or transmitted in any printed, electronic or other form, or used for generating or marketing any printed or electronic publication, service or product.

| Stock's Price Stability 70<br>Price Growth Persistence 75<br>Earnings Predictability 85 | Company's Financial Strongth | D.+ |
|---|------------------------------|-----|
| Price Growth Persistence 75<br>Earnings Predictability 85                               | Stock's Price Stability      | 70  |
| Earnings Predictability 85  | Price Growth Persistence     | 75  |
|   | Earnings Predictability      | 85  |

| YORK WATER CO                                | ) NDQYO           | RW   | RE<br>PR  | ICE 16.4          | 40 TRAILING                                 | <sup>3</sup> 22.2 | RELATIVE 1.46         | DIV'D 3                           | .2% VA                      | LUE<br>NE                      |  |  |  |  |
|--|-------------------|--|---|-------------------|---|-------------------|-----------------------|-----------------------------------|-----------------------------|--------------------------------|--|--|--|--|
| RANKS  | 13.49             | 14.03  | 17.87   | 20.99             | 18.55                                       | 16.5              | 0 17.95               | 18.00                             | 18.14                       | High                           |  |  |  |  |
| PERFORMANCE 3 Average                        | LEGE              | NDS  | 11.07   | 10.00             | 15.45                                       | 0.2               | .0 5.14               | 12.00                             | 10.01                       | 1 2011                         |  |  |  |  |
| Technical 3 Average                          | 12 Mo             | s Mov Avg  | 1000  | mul Hulling       |   | 1                 |                       | 111                               | • 14111                     |                                |  |  |  |  |
| Above  | 2-for-1 split     | 5/02   | Hill -  |                   |   | 1                 |                       | 111111                            |                             |                                |  |  |  |  |
| SAFEIY Z Average                             | Shaded area ind   | icates recession   |   | · · · · ·         |   |                   | ······                |                                   | Contraction of the          |                                |  |  |  |  |
| BETA .70 (1.00 = Market)                     |                   | ···  | h   |                   | [····.'·,·                                  |                   |                       |                                   |                             | 5                              |  |  |  |  |
|  |                   |  |   |                   |   |                   |                       |                                   |                             | 4                              |  |  |  |  |
| Financial Strength B++                       |                   |  |   |                   |   |                   |                       |                                   |                             | 3                              |  |  |  |  |
| Price Stability 95                           |                   |  |   |                   |   |                   |                       |                                   |                             | 2                              |  |  |  |  |
| Price Growth Persistence 70                  |                   |  |   |                   |   |                   |                       |                                   | 1 1 1                       |                                |  |  |  |  |
| Earnings Predictability 100                  |                   |  |   |                   |   |                   |                       | uttellin.                         |                             | 700<br>VOL                     |  |  |  |  |
|  | سللس              | սավակ  | սևևևսս  | لللسسا            | ահրոր                                       | dudhut            |                       |                                   |                             | (thous.)                       |  |  |  |  |
| © VALUE LINE PUBLISHING LLC                  | 2003              | 2004   | 2005  | 2006              | 2007  | 2008              | 2009                  | 2010                              | 2011                        | 2012/2013                      |  |  |  |  |
| REVENUES PER SH                              | 2.17              | 2.18   | 2.58  | 2.56              | 2.79  | 2.89              | 2.95                  | 3.07                              |                             |                                |  |  |  |  |
| EARNINGS PER SH                              | .47               | .49  | .56   | .58               | .57   | .57               | .64                   | .71                               | .75 <sup>A,B</sup>          | .80 <sup>C</sup> /NA           |  |  |  |  |
| DIV'D DECL'D PER SH                          | .37               | .39  | .42   | .45               | .48   | .49               | .51                   | .52                               | -                           |                                |  |  |  |  |
| BOOK VALUE PER SH                            | 4.06              | 4.65   | 4.85  | 5.84              | 5.97  | 6.14              | 6.92                  | 7.19                              | -                           |                                |  |  |  |  |
| COMMON SHS OUTST'G (MILL)                    | 9.63              | 10.33  | 10.40   | 11.20             | 11.27                                       | 11.37             | 12.56                 | 12.69                             |                             |                                |  |  |  |  |
| AVG ANN'L P/E RATIO                          | 24.5              | 25.7   | 26.3  | 31.2              | 30.3  | 24.6              | 21.9                  | 20.7                              | 21.9                        | 20.5/NA                        |  |  |  |  |
| AVG ANN'L DIV'D YIELD                        | 3.2%              | 3.1%   | 2.9%  | 2.5%              | 2.8%  | 3.5%              | 6 3.6%                | 3.5%                              |                             |                                |  |  |  |  |
| REVENUES (\$MILL)                            | 20.9              | 22.5   | 26.8  | 28.7              | 31.4  | 32.8              | 37.0                  | 39.0                              |                             | Bold figures                   |  |  |  |  |
| INCOME TAX RATE                              | 4.4               | 4.8  | 36.7%   | 34.4%             | 36.5%                                       | 36.19             | 6 37.9%               | 38.5%                             |                             | earnings                       |  |  |  |  |
| AFUDC % TO NET PROFIT                        |                   |  |   | 7.2%              | 3.6%  | 10.19             | 6                     | 1.2%                              |                             | estimates                      |  |  |  |  |
| LONG-TERM DEBT RATIO                         | 43.4%             | 42.5%<br>57.5%   | 44.1%   | 48.3%             | 6 45.7%<br>6 54.3%                          | 48.3%             | -                     | and, using the                    |                             |                                |  |  |  |  |
| TOTAL CAPITAL (\$MILL)                       | 69.0              | 83.6   | 90.3  | 126.5             | 125.7                                       | 153.4             | 160.1                 | 176.4                             |                             | P/E ratios.                    |  |  |  |  |
| NET PLANT (\$MILL)                           | 116.5             | 140.0  | 155.3   | 174.4             | 191.6                                       | 211.4             | 222.0                 | 228.4                             |                             |                                |  |  |  |  |
| RETURN ON SHR. EQUITY                        | 8.5%              | 10.0%  | 8.4%  | 9.3%              | 9.5%  | 9.2%              | 6 8.6%                | 9.8%                              | 2                           |                                |  |  |  |  |
| RETURN ON COM EQUITY                         | 11.4%             | 10.0%  | 11.6%   | 9.3%              | 9.5%  | 9.29              | 6 8.6%                | 9.8%                              |                             |                                |  |  |  |  |
| RETAINED TO COM EQ                           | 2.6%              | 2.1%   | 3.0%  | 2.2%              | 1.7%  | 1.49              | 6 1.9%<br>78%         | 2.7%                              | -                           |                                |  |  |  |  |
| ANo. of analysts changing earn. est. in      | last 8 days: 0 up | , 0 down, conse  | nsus 5-year ean                                     | nings growth not  | available. BBa                              | sed upon 4 an     | alysts' estimates. CB | Based upon 4 a                    | nalysts' estimates          |                                |  |  |  |  |
| ANNUAL RATES                                 |                   | ASSETS /S  | nill ) 2  | 000 2010          | 6/30/11                                     |                   | INDU                  | STRY: Wa                          | ater Utility                |                                |  |  |  |  |
| of change (per share) 5 Yrs.                 | 1 Yr.             | Cash Assets  |   | .0 1.3            | 3.1   |                   |                       |                                   | 9                           | • •                            |  |  |  |  |
| "Cash Flow" 7.0%                             | 12.0%             | Receivables<br>Inventory (Av                                     | va cost)  | 5.4 6.3<br>.7 .6  | 6.1<br>.7                                   | f water in York   |                       |                                   |                             |                                |  |  |  |  |
| Earnings 5.0%<br>Dividends 5.0%              | 11.0%<br>2.0%     | Other Other  | -   | 1.0 .6            | 1.5   | County            | and Adams C           | County, Pennsylvania. The company |                             |                                |  |  |  |  |
| Book Value 8.5%                              | 4.0%              | Current Asse   | 315   | 7.1 0.0           | 11.4  | supplies          | water for res         | idential, c                       | commercial,                 | industrial, and                |  |  |  |  |
| Fiscal QUARTERLY SALES (                     | imill.) Full      | Property, Pla  | ant<br>at cost 24                                   | 50.4 270.8        |   | other cu          | s 700 feet lo         | has two r                         | eservoirs, 1<br>e feet high | and creates a                  |  |  |  |  |
| Year 1Q 2Q 3Q                                | 4Q Year           | Accum Depr   | eciation  | 38.4 42.4         |   | reservoi          | r covering a          | pproximat                         | ely 165 ac                  | res containing                 |  |  |  |  |
| 12/31/10 9.0 9.7 10.5                        | 9.2 37.0 9.8 39.0 | Other  |   | 19.7 22.7         | 230.2                                       | about 8           | 70 million ga         | illons of                         | water; and                  | Lake Redman,                   |  |  |  |  |
| 12/31/11 9.6 10.5                            |                   | Total Assets   | 24  | 48.8 259.9        | 264.8                                       | which i           | s 1,000 feet          | long and                          | 52 feet hi                  | gh, creating a hat holds about |  |  |  |  |
| Eisent FARNINGS PER SHA                      | RE Eul            | LIABILITIES  | (\$mill.)   | 1919 - 1912 (     |   | 1.3 bill          | ion gallons of        | f water. I                        | n addition,                 | it possesses a                 |  |  |  |  |
| Year 1Q 2Q 3Q                                | 4Q Year           | Accts Payab<br>Debt Due  | le  | 1.4 1.2<br>9.3 .0 | 1.7   | 15-mile           | pipeline fror         | n the Su                          | squehanna                   | River to Lake                  |  |  |  |  |
| 12/31/08 .11 .13 .15                         | .18 .57           | Other  | -   | 3.9 4.1           | 4.0   | Redman            | that provides         | s access t                        | o an additi                 | onal supply of                 |  |  |  |  |
| 12/31/09 .13 .17 .18<br>12/31/10 .15 .18 .21 | .16 .64           | Current Liab   |   | 14.0 5.3          | 5.6   | entered           | into an agreen        | nent to pro                       | vide water s                | ervice to Cross                |  |  |  |  |
| 12/31/11 .17 .19 .21                         | .18               |  |   |                   |   | Keys Vi           | llage in Adam         | s County,                         | PA. Cross K                 | eys Village is a               |  |  |  |  |
| 12/31/12 .17                                 |                   | LONG-TERM<br>as of 6/30  | M DEBT AND E<br>0/11                                | QUITY             |   | continui          | ng-care retire        | ment con                          | nmunity cu                  | rrently serving                |  |  |  |  |
| endar 1Q 2Q 3Q                               | 4Q Year           | Total Debt \$  | e on a gro  | lent: Jeffrev     | R. Hines. Inc.:                             |                   |                       |                                   |                             |                                |  |  |  |  |
| 2008 .121 .121 .121                          | .121 .48          | LT Debt \$85   | Market S  | treet, York,      | PA 17401. Tel.:                             |                   |                       |                                   |                             |                                |  |  |  |  |
| 2009 .126 .126 .126                          | .126 .50          |  | ap, Leases NA                                       | 45-3601. Intern   | net: http://                                | www.yorkw         | ater.com. J.V.        |                                   |                             |                                |  |  |  |  |
| 2010 .120 .120 .128                          | .120 .51          | Leases, Und  | capitalized Anr                                     | nual rentals NA   |   | October 21, 2011  |                       |                                   |                             |                                |  |  |  |  |
| INSTITUTIONAL DECISI                         | DNS               | Pension Lia  | Liability \$9.8 mill. in '10 vs. \$8.8 mill. in '09 |                   |   |                   |                       |                                   | RN                          |                                |  |  |  |  |
| 4Q'10 1Q'11                                  | 2Q'11             | Pfd Stock No   | ne  | Pfd Div'd         | 'd Paid None Dividends plus appreciation as |                   |                       |                                   |                             | ation as of 9/30/2011          |  |  |  |  |
| to Buy 25 20<br>to Sell 16 21                | 27                | Common Stock 12,743,000 shares (59% of Cool) 3 Mos. 6 Mos. 1 Yr. |   |                   |   |                   | 3 Yr                  | s. 5 Yrs.                         |                             |                                |  |  |  |  |
| Hid's(000) 3107 3080                         | 3163              |  | (52% of Cap'l) -1.44% -5.57%                        |                   |   |                   |                       |                                   | 44.94                       | % 0.10%                        |  |  |  |  |

©2011 Value Line Publishing LLC. All rights reserved. Factual material is obtained from sources believed to be reliable and is provided without warranties of any kind. THE PUBLISHER IS NOT RESPONSIBLE FOR ANY ERRORS OR OMISSIONS HEREIN. This publication is strictly for subscriber's own, non-commercial, internal use. No part of it may be reproduced, resold, stored or transmitted in any printed, electronic or other form, or used for generating or marketing any printed or electronic publication, service or product.

### Missouri-American Water Company Indicated Common Equity Cost Rate Through Use of a Risk Premium Model Using an Adjusted Total Market Approach

| Line No. |   | Proxy Group of<br>Nine Water<br>Companies |
|----------|---|---|
|          |   |   |
| 1.       | Prospective Yield on Aaa Rated<br>Corporate Bonds (1)   | 4.23 %                                    |
| 2.       | Adjustment to Reflect Yield Spread<br>Between Aaa Rated Corporate<br>Bonds and A Rated Public |   |
|          | Utility Bonds   | 0.44 (2)                                  |
| 3.       | Adjusted Prospective Yield on A Rated<br>Public Utility Bonds                                 | 4.67 %                                    |
| 4.       | Adjustment to Reflect Bond<br>Rating Difference of Proxy Group                                | 0.22 (3)                                  |
| 5.       | Adjusted Prospective Bond Yield   | 4.89                                      |
| 6.       | Equity Risk Premium (4)   | 5.45                                      |
| 7.       | Risk Premium Derived Common<br>Equity Cost Rate   | <u> </u>                                  |

Notes: (1) Derived in Note (4) on page 19 of this Schedule.

(2) The average yield spread of A rated public utility bonds over Aaa rated corporate bonds of 0.44% from page 18 of this Schedule.

- (3) Adjustment to reflect the A3 Moody's bond rating of the proxy group of nine water companies as shown on page 16 of this Schedule. The 22 basis point adjustment is derived by taking 1/3 of the spread between Baa2 and A2 Public Utility Bond yields. (1/3 \* 0.67% = 0.22%)
- (4) From page 18 of this Schedule.

# Missouri-American Water Company Comparison of Bond Ratings, Business Risk and Financial Risk Profiles for the Proxy Group of Nine Water Companies

Standard & Poor's

Moody's

|                                     |         | Bo             | nd Rating                  |                | Bo                         | nd Rating        |                            |                              |                            |                               |                            |
|-------------------------------------|---------|----------------|----------------------------|----------------|----------------------------|------------------|----------------------------|------------------------------|----------------------------|-------------------------------|----------------------------|
|                                     |         | Jan            | uary 2012                  |                | Jan                        | uary 2012        |                            | 1 1                          |                            |                               |                            |
|                                     |         | Bond<br>Rating | Numerical<br>Weiahtina (1) | Bond<br>Rating | Numerical<br>Weiahtina (1) | Credit<br>Rating | Numerical<br>Weidhting (1) | Business Risk<br>Profile (2) | Numerical<br>Weiahtina (1) | Financial Risk<br>Profile (2) | Numerical<br>Weiahtina (1) |
|                                     |         | 0              |                            |                |                            | 2                |                            |                              |                            |                               |                            |
| American States Water Co. (3)       |         | A2             | 6.0                        | 4+             | 5.0                        | 4+               | 5.0                        | Excellent                    | 1.0                        | Intermediate                  | 3.0                        |
| American Water Works Co., Inc. (4)  |         | Baa1           | 8.0                        | A+             | 5.0                        | BBB+             | 8.0                        | Excellent                    | 1.0                        | Aggressive                    | 5.0                        |
| Aqua America, Inc. (5)              |         | NR             | :                          | -AA-           | 4.0                        | A+               | 5.0                        | Excellent                    | 1.0                        | Intermediate                  | 3.0                        |
| Artesian Resources Corp.            |         | NR             | :                          | NR             | :                          | NR               | :                          | NR                           | :                          | NR                            | :                          |
| California Water Service Group (6)  |         | NR             | :                          | -AA-           | 4.0                        | A+               | 5.0                        | Excellent                    | 1.0                        | Intermediate                  | 3.0                        |
| Connecticut Water Service, Inc. (7) |         | NR             | :                          | A              | 6.0                        | A                | 6.0                        | Excellent                    | 1.0                        | Intermediate                  | 3.0                        |
| Middlesex Water Company             |         | NR             | :                          | A              | 6.0                        | -A               | 7.0                        | Excellent                    | 1.0                        | Intermediate                  | 3.0                        |
| SJW Corporation (8)                 |         | NR             | :                          | A              | 6.0                        | A                |                            | Excellent                    | 1.0                        | Intermediate                  | 3.0                        |
| York Water Company                  |         | NR             | :                          | A-             | 7.0                        | -A               | 7.0                        | Excellent                    | 1.0                        | Intermediate                  | 3.0                        |
|                                     | Average | A3             | 7.0                        | A+             | 5.4                        | A                | 6.1                        | Excellent                    | 1.0                        | Intermediate                  | 3.3                        |
|                                     |         |                |                            |                |                            |                  |                            |                              |                            |                               |                            |
|                                     |         |                |                            |                |                            |                  |                            |                              |                            |                               |                            |

Notes: (1) (2) (3) (4) (5) (6) (6) (7) (8)

From page 3 of Schedule PMA-10. From page 3 of Schedule PMA-10. From Standard & Por's Issuer Ranking: U.S. Investor-Owned Water Utilities, Strongest to Weakest, October 7, 2011. Ratings, business risk and financial risk profiles are those of Golden State Water Company. Rating, business risk and financial risk profiles are those of Pennsylvania and New Jersey American Water. Ratings, business risk and financial risk profiles are those of Apua Pennsylvania. Inc. Ratings, business risk and financial risk profiles are those of Apua Pennsylvania. Inc. Ratings, business risk and financial risk profiles are those of Comedicut Water Service Co. Ratings, business risk and financial risk profiles are those of Comedicut Water Company. Ratings, business risk and financial risk profiles are those of San Jose Water Co.

Source Information:

Moody's Investors Service Standard & Poor's Global Utilities Rating Service

<u>MoodVs</u> Comparison of Interest Rate Trends for the Three Months Ending November 2011 (1).

| olic Utility Bonds   |                 |                      | Baa over A  |             |            |              | 0.67 %                      |
|----------------------|-----------------|----------------------|-------------|-------------|------------|--------------|-----------------------------|
| Spread - Pub         |                 |                      | A over Aa   |             |            |              | 0.30 %                      |
| Itility Bonds        | Baa (Pub.       | Util.) over          | Aaa (Corp.) |             |            |              | 1.11 %                      |
| prporate v. Public U | A (Pub. Util.)  | over Aaa             | (Corp.)     |             |            |              | 0.44 %                      |
| Spread - Co          | Aa (Pub. Util.) | over Aaa             | (Corp.)     |             |            |              | 0.14 %                      |
|                      |                 |                      | Baa Rated   | 4.93 %      | 5.24       | 5.11         | 5.09 %                      |
|                      |                 | Public Utility Bonds | A Rated     | 4.25 %      | 4.52       | 4.48         | 4.42 %                      |
|                      |                 |                      | Aa Rated    | 3.92 %      | 4.21       | 4.24         | 4.12 %                      |
|                      | Corporate       | Bonds                | Aaa Rated   | 3.87 %      | 3.98       | 4.09         | 3.98 %                      |
|                      |                 |                      | Months      | November-11 | October-11 | September-11 | Average of Last<br>3 Months |

Notes: (1) All yields are distributed yields.

Source of Information: Mergent Bond Record, December 2011, Vol 78, No. 12.

### Missouri-American Water Company Judgment of Equity Risk Premium for the Proxy Group of Nine Water Companies

| Line<br>No. |   | Proxy Group of Nine<br>Water Companies |
|-------------|---|--|
|             |   |  |
| 1.          | Calculated equity risk<br>premium based on the<br>total market using                                    |  |
|             | the beta approach (1)   | 6.78                                   |
| 2.          | Mean equity risk premium<br>based on a study<br>using the holding period<br>returns of public utilities |  |
|             | with A rated bonds (2)  | 4.12                                   |
| 3.          | Average equity risk premium   | <u> </u>                               |

- Notes: (1) From page 19 of this Schedule.
  - (2) From page 8 of Schedule PMA-10.

### <u>Missouri-American Water Company</u> Derivation of Equity Risk Premium Based on the Total Market Approach Using the Beta for <u>the Proxy Group of Nine Water Companies</u>

| Line No | ) <u>.</u> |   | Proxy Group of<br>Nine Water<br>Companies             |
|---------|------------|---|---|
|         |            |   |   |
| 1.      |            | Arithmetic mean total return rate on<br>the Standard & Poor's 500 Composite<br>Index - 1926-2010 (1)  | 11.90 %   |
| 2.      |            | Arithmetic mean yield on  |   |
|         |            | 1926-2010 (2)   | (6.10)  |
| 3.      |            | Historical Equity Risk Premium  | <u> </u>  |
| 4.      |            | Forecasted 3-5 year Total Annual<br>Market Return (3)   | 17.80 %   |
| 5.      |            | Prospective Yield an Aaa Rated<br>Corporate Bonds (4)   | (4.23)  |
| 6.      |            | Forecasted Equity Risk Premium  | <u>    13.57 </u> %                                   |
| 7.      |            | Conclusion of Equity Risk Premium (5)   | 9.69 %  |
| 8.      |            | Adjusted Value Line Beta (6)  | 0.70  |
| 9.      |            | Beta Adjusted Equity Risk Premium   | <u> </u>  |
| Notes:  | (1)        | Stocks, Bonds, Bills, and Inflation - Market Results for 1926-201 Valuation Edition, Morningstar, Inc., 2011 Chicago, IL.   | 0 Yearbook  |
|         | (2)        | From Moody's Industrial Manual and Mergent Bond Record Mor  | nthly Update.   |
|         | (3)        | From page 22 of this Schedule.  |   |
|         | (4)        | Average forecast based upon six quarterly estimates of Aaa rate<br>per the consensus of nearly 50 economists reported in Blue Chi<br>Forecasts dated January 1, 2012 (see page 20 of this Schedule<br>are detailed below. | ed corporate bonds<br>p Financial<br>). The estimates |
|         |            | First Quarter 2012<br>Second Quarter 2012<br>Third Quarter 2012<br>Fourth Quarter 2012<br>First Quarter 2013<br>Second Quarter 2013   | 4.00 %<br>4.00<br>4.20<br>4.30<br>4.40<br>4.50        |
|         |            | Average   | 4.23 %  |
|         | (5)        | The average of the historical equity risk premium of 5.80% from   | Line No. 3 and the                                    |

(5) The average of the historical equity risk premium of 5.80% from Line No. 3 and the forecasted equity risk premium of 13.57% from Line No. 6 ((5.80% + 13.57%) / 2 = 9.69%.

(6) Median beta derived from page 21 of this Schedule.

### Consensus Forecasts Of U.S. Interest Rates And Key Assumptions<sup>1</sup>

|                         |        |           |         | Histo  | ry   |            |       | Cons           | ensus 1 | Foreca      | sts-Qu      | arterly     | Avg.        |             |
|-------------------------|--------|-----------|---------|--------|------|------------|-------|----------------|---------|-------------|-------------|-------------|-------------|-------------|
|                         | Av     | erage For | Week En | ding   | Ave  | rage For M | Month | Latest Q*      | 1Q      | 2Q          | 3Q          | 4Q          | 1Q          | 2Q          |
| Interest Rates          | Dec.23 | Dec.16    | Dec.9   | Dec.2  | Nov. | Oct.       | Sep.  | <u>4Q 2011</u> | 2012    | <u>2012</u> | <u>2012</u> | <u>2012</u> | <u>2013</u> | <u>2013</u> |
| Federal Funds Rate      | 0.07   | 0.07      | 0.08    | 0.08   | 0.08 | 0.07       | 0.08  | 0.07           | 0.1     | 0.1         | 0.1         | 0.1         | 0.1         | 0.2         |
| Prime Rate              | 3.25   | 3.25      | 3.25    | 3.25   | 3.25 | 3.25       | 3.25  | 3.25           | 3.3     | 3.3         | 3.3         | 3.3         | 3.3         | 3.3         |
| LIBOR, 3-mo.            | 0.57   | 0.55      | 0.54    | 0.52   | 0.48 | 0.41       | 0.35  | 0.32           | 0.5     | 0.5         | 0.5         | 0.5         | 0.5         | 0.5         |
| Commercial Paper, 1-mo. | 0.10   | 0.12      | 0.09    | 0.11   | 0.10 | 0.09       | 0.08  | 0.10           | 0.1     | 0.2         | 0.2         | 0.2         | 0.2         | 0.3         |
| Treasury bill, 3-mo.    | 0.01   | 0.01      | 0.01    | 0.02   | 0.01 | 0.02       | 0.01  | 0.01           | 0.0     | 0.1         | 0.1         | 0.1         | 0.1         | 0.2         |
| Treasury bill, 6-mo.    | 0.04   | 0.05      | 0.05    | 0.06   | 0.05 | 0.05       | 0.04  | 0.05           | 0.1     | 0.1         | 0.1         | 0.1         | 0.2         | 0.3         |
| Treasury bill, 1 yr.    | 0.12   | 0.11      | 0.11    | 0.13   | 0.11 | 0.11       | 0.10  | 0.11           | 0.1     | 0.2         | 0.2         | 0.3         | 0.3         | 0.5         |
| Treasury note, 2 yr.    | 0.26   | 0.25      | 0.24    | 0.26   | 0.25 | 0.28       | 0.21  | 0.26           | 0.3     | 0.3         | 0.4         | 0.5         | 0.6         | 0.8         |
| Treasury note, 5 yr.    | 0.86   | 0.85      | 0.90    | 0.94   | 0.91 | 1.06       | 0.90  | 0.95           | 1.0     | 1.1         | 1.2         | 1.4         | 1.5         | 1.7         |
| Treasury note, 10 yr.   | 1.90   | 1.94      | 2.04    | 2.04   | 2.01 | 2.15       | 1.98  | 2.04           | 2.0     | 2.2         | 2.3         | 2.5         | 2.6         | 2.8         |
| Treasury note, 30 yr.   | 2.90   | 2.95      | 3.05    | 3.02   | 3.02 | 3.13       | 3.18  | 3.04           | 3.1     | 3.2         | 3.4         | 3.5         | 3.7         | 3.8         |
| Corporate Aaa bond      | 3.85   | 3.92      | 4.02    | 4.00   | 3.87 | 3.98       | 4.09  | 3.93           | 4.0     | 4.0         | 4.2         | 4.3         | 4.4         | 4.5         |
| Corporate Baa bond      | 5.18   | 5.20      | 5.30    | 5.28   | 5.14 | 5.37       | 5.27  | 5.25           | 5.2     | 5.3         | 5.4         | 5.5         | 5.6         | 5.7         |
| State & Local bonds     | 3.92   | 3.92      | 3.93    | 4.12   | 4.05 | 4.13       | 4.01  | 4.03           | 4.0     | 4.1         | 4.2         | 4.2         | 4.3         | 4.4         |
| Home mortgage rate      | 3.91   | 3.94      | 3.99    | 4.00   | 3.99 | 4.07       | 4.11  | 4.00           | 4.0     | 4.0         | 4.2         | 4.3         | 4.4         | 4.6         |
|                         |        |           |         | Histor | y    |            |       |                | Co      | onsensu     | is Fore     | casts-Q     | Juarte      | rly         |
|                         | 1Q     | 2Q        | 3Q      | 4Q     | 1Q   | 2Q         | 3Q    | 4Q*            | 1Q      | 2Q          | 3Q          | 4Q          | 1Q          | 2Q          |
| Key Assumptions         | 2010   | 2010      | 2010    | 2010   | 2011 | 2011       | 2011  | 2011           | 2012    | 2012        | 2012        | 2012        | 2013        | 2013        |
| Major Currency Index    | 74.8   | 77.6      | 75.9    | 73.0   | 71.9 | 69.6       | 69.9  | 72.3           | 73.6    | 73.9        | 74.0        | 74.0        | 74.0        | 74.2        |
| Real GDP                | 3.9    | 3.8       | 2.5     | 2.3    | 0.4  | 1.3        | 1.8   | 3.1            | 2.0     | 2.1         | 2.4         | 2.6         | 2.6         | 2.8         |
| GDP Price Index         | 1.5    | 1.5       | 1.4     | 1.9    | 2.5  | 2.5        | 2.6   | 1.7            | 1.8     | 1.8         | 1.9         | 1.9         | 2.0         | 2.0         |
| Consumer Price Index    | 1.3    | -0.5      | 1.4     | 2.6    | 5.2  | 4.1        | 3.1   | 1.6            | 1.9     | 2.1         | 2.3         | 2.2         | 2.2         | 2.2         |

Forecasts for interest rates and the Federal Reserve's Major Currency Index represent averages for the quarter. Forecasts for Real GDP, GDP Price Index and Consumer Price Index are seasonally-adjusted annual rates of change (saar). Individual panel members' forecasts are on pages 4 through 9. Historical data for interest rates except LIBOR is from Federal Reserve Release (FRSR) H.15. LIBOR quotes available from *The Wall Street Journal*. Interest rate definitions are the same as those in FRSR H.15. Treasury yields are reported on a constant maturity basis. Historical data for the Fed's Major Currency Index is from FRSR H.10 and G.5. Historical data for Real GDP and GDP Chained Price Index are from the Bureau of Economic Analysis (BEA). Consumer Price Index (CPI) history is from the Department of Labor's Bureau of Labor Statistics (BLS). *\*Interest rate data for 4Q 2011 Major Currency Index also is based on data through the week ended December 23rd. \*Data for 4Q 2011 Major Currency Index also is based on a special question asked of the panelists this month (see page 14).* 









### Missouri-American Water Company Indicated Common Equity Cost Rate Through Use of the Traditional Capital Asset Pricing Model (CAPM) and Empirical Capital Asset Pricing Model (ECAPM)

|                                     | <u>1</u>    | 2           | <u>3</u>   | <u>4</u>            | <u>5</u>           | <u>6</u>            |
|-------------------------------------|-------------|-------------|------------|---------------------|--------------------|---------------------|
|                                     | Mahaa Lia a |             |            | The stitle shell    | 504544             | Indicated           |
|                                     | Value Line  | Market Diak | Dials Free |                     | ECAPM<br>Cost Data | Common              |
| Proxy Group of Nine Water Companies | Beta        | Premium (1) | Rate (2)   | Rate (3)            | (4)                | Rate (5)            |
| American States Water Co.           | 0.75        | 10.53 %     | 3.45 %     | 11.35 %             | 12.01 %            |                     |
| American Water Works Co., Inc.      | 0.65        | 10.53       | 3.45       | 10.29               | 11.22              |                     |
| Aqua America, Inc.                  | 0.65        | 10.53       | 3.45       | 10.29               | 11.22              |                     |
| Artesian Resources Corp.            | 0.60        | 10.53       | 3.45       | 9.77                | 10.82              |                     |
| California Water Service Group      | 0.70        | 10.53       | 3.45       | 10.82               | 11.61              |                     |
| Connecticut Water Service, Inc.     | 0.80        | 10.53       | 3.45       | 11.87               | 12.40              |                     |
| Middlesex Water Company             | 0.75        | 10.53       | 3.45       | 11.35               | 12.01              |                     |
| SJW Corporation                     | 0.90        | 10.53       | 3.45       | 12.93               | 13.19              |                     |
| York Water Company                  | 0.70        | 10.53       | 3.45       | 10.82               | 11.61              |                     |
| Average                             |             |             |            | <u>11.06</u> %      | <u>11.79</u> %     | <u>11.43</u> %      |
| Median                              |             |             |            | <u>    10.82 </u> % | 11.61              | <u>    11.22 </u> % |

See page 22 for notes.

### Missouri-American Water Company Development of the Market-Required Rate of Return on Common Equity Using the Capital Asset Pricing Model for the Proxy Group of Nine AUS Utility Reports Water Companies Adjusted to Reflect a Forecasted Risk-Free Rate and Market Return

### Notes:

(1) For reasons explained in Ms. Ahern's accompanying direct testimony, from the thirteen weeks ending January 6, 2012, <u>Value Line Summary & Index</u>, a forecasted 3-5 year total annual market return of 17.80% can be derived by averaging the thirteen weeks ended January 6, 2012 forecasted total 3-5 year total appreciation, converting it into an annual market appreciation and adding the <u>Value Line</u> average forecasted annual dividend yield.

The 3-5 year average total market appreciation of 78% produces a four-year average annual return of 15.46% (( $1.78^{.25}$ ) - 1). When the average annual forecasted dividend yield of 2.34% is added, a total average market return of 17.80% (2.34% + 15.46%) is derived.

The thirteen week forecasted total market return of 17.80% minus the forecasted risk-free rate of 3.45% (developed in Note 2) is 14.35% (17.80% - 3.45%). The Morningstar, Inc. (Ibbotson Associates) calculated market premium of 6.70% for the period 1926-2010 results from a total market return of 11.90% less the average income return on long-term U.S. Government Securities of 5.20% (11.90% - 5.20% = 6.70%). This is then averaged with the 14.35% <u>Value Line</u> market premium resulting in an 10.53% market premium. The 10.53% market premium is then multiplied by the beta in column 1 of page 21 of this Schedule.

(2) The average forecast based upon six quarterly estimates of 30-year Treasury Note yields per the consensus of nearly 50 economists reported in the <u>Blue Chip Financial Forecasts</u> dated January 1, 2012 (see page 20 of this Schedule). The estimates are detailed below:

|                     | <u>30-Year</u>      |
|---------------------|---------------------|
|                     | Treasury Note Yield |
| First Quarter 2012  | 3.10                |
| Second Quarter 2012 | 3.20                |
| Third Quarter 2012  | 3.40                |
| Fourth Quarter 2012 | 3.50                |
| First Quarter 2013  | 3.70                |
| Second Quarter 2013 | <u>3.80</u>         |
|                     |                     |
| Average             | <u>3.45%</u>        |
|                     |                     |

(3) The traditional Capital Asset Pricing Model (CAPM) is applied using the following formula:

 $R_{\rm S} = R_{\rm F} + \beta \left( R_{\rm M} - R_{\rm F} \right)$ 

Where  $R_S$  = Return rate of common stock  $R_F$  = Risk Free Rate  $\beta$  = Value Line Adjusted Beta  $R_M$  = Return on the market as a whole

(4) The empirical CAPM is applied using the following formula:

 $R_{S} = R_{F} + .25 (R_{M} - R_{F}) + .75 \beta (R_{M} - R_{F})$ 

 $\begin{array}{ll} \mbox{Where} & R_S = \mbox{Return rate of common stock} \\ R_F = \mbox{Risk-Free Rate} \\ \beta & = \mbox{Value Line Adjusted Beta} \\ R_M = \mbox{Return on the market as a whole} \end{array}$ 

 Source of Information:
 Value Line Summary & Index

 Blue Chip Financial Forecasts, January 1, 2012

 Value Line Investment Survey, October 21, 2011

 Standard Edition and Small and Mid-Cap Edition

 Ibbotson<sup>®</sup> SBBI<sup>®</sup> 2011 Valuation Yearbook – Market Results for

 Stocks, Bonds, Bills, and Inflation – 1926 – 2010, Morningstar, Inc., 2011 Chicago, IL

### <u>Missouri-American Water Company</u> Summary of Cost of Equity Models Applied to the Proxy Group of Non Price Regulated Companies Comparable in Total Risk to the <u>Proxy Group of Nine Water Companies</u>

| Principal Methods                             |         | Forty-Two Non<br>Price Regulated<br>Companies | _  |
|---|---------|---|----|
| Projected Return on Book<br>Common Equity (1) |         | 14.00   | %  |
| Average of Market-Based<br>Models (2)         |         | 12.41   | _% |
|   | Average | 13.21   | %  |

### Notes:

- (1) From page 27 of this Schedule.
- (2) Average of the results of the DCF (12.84%), RPM (12.72%), and CAPM / ECAPM
  - (11.68%) analyses as shown on pages 28,

29, and 32 of this Schedule, respectively.

### <u>Missouri-American Water Company</u> Basis of Selection of Comparable Risk <u>Domestic Non-Price Regulated Companies</u>

| Proxy Group of Nine Water<br>Companies  | Value Line<br>Adjusted<br>Beta   | Unadjusted<br>Beta   | Residual<br>Standard Error<br>of the<br>Regression   |
|---|--|--|--|
| American States Water Co.<br>American Water Works Co., Inc.<br>Aqua America, Inc.<br>Artesian Resources Corp.<br>California Water Service Group<br>Connecticut Water Service, Inc.<br>Middlesex Water Company<br>SJW Corporation<br>York Water Company<br>Average | 0.80<br>0.65<br>0.65<br>0.60<br>0.70<br>0.80<br>0.75<br>0.90<br>0.70<br>0.73 | $\begin{array}{c} 0.62 \\ 0.43 \\ 0.41 \\ 0.33 \\ 0.53 \\ 0.64 \\ 0.56 \\ 0.84 \\ 0.48 \\ \hline 0.54 \end{array}$ | 3.6318<br>3.7667<br>2.8589<br>2.5296<br>3.5690<br>2.8819<br>2.7573<br>4.3983<br>3.3729<br>3.3074 |
| Beta Range (+/- 2 std. Devs. of Beta)<br>2 std. Devs. of Beta   | 0.40<br>0.14   | 0.68   |  |
| Residual Std. Err. Range (+/- 2 std.<br>Devs. of the Residual Std. Err.)  | 3.0168   | 3.5980   |  |
| Std. dev. of the Res. Std. Err.   | 0.1453   |  |  |
| 2 std. devs. of the Res. Std. Err.  | 0.2906   |  |  |

### <u>Missouri-American Water Company</u> Proxy Group of Non-Price Regulated Companies Comparable in Total Risk to the <u>Proxy Group of Nine Water Companies</u>

|                                    |             |            | Residual         |
|------------------------------------|-------------|------------|------------------|
|                                    |             |            | Standard         |
| Proxy Group of Forty-Two Non Price | VL Adjusted | Unadjusted | Error of the     |
| Regulated Companies                | Beta        | Beta       | Regression       |
| Gallagher (Arthur J.)              | 0.70        | 0.54       | 3.0668           |
| Amaen                              | 0.65        | 0.43       | 3.5824           |
| AutoZone Inc.                      | 0.70        | 0.53       | 3,3352           |
| Bristol-Myers Sauibb               | 0.75        | 0.57       | 3.1171           |
| Brown & Brown                      | 0.70        | 0.49       | 3.1582           |
| Capitol Fed Finl                   | 0.65        | 0.43       | 3 2572           |
| CVS Caremark Corp                  | 0.80        | 0.66       | 3 0354           |
| Forest Labs                        | 0.80        | 0.63       | 3 3743           |
| Gen-Probe                          | 0.80        | 0.68       | 3 3384           |
| Hashro Inc                         | 0.00        | 0.60       | 3 3919           |
| IAC/InterActiveCorp                | 0.65        | 0.00       | 3 2905           |
| Investors Bancorn                  | 0.00        | 0.55       | 3 4028           |
| J&J Snack Foods                    | 0.70        | 0.00       | 3 3652           |
| Lancaster Colony                   | 0.75        | 0.40       | 3 3268           |
| Lincare Holdings                   | 0.65        | 0.00       | 3 5487           |
| McKesson Corp                      | 0.00        | 0.40       | 3 3801           |
| Medtronic Inc                      | 0.70        | 0.66       | 3 5135           |
| Medro Health Solutions             | 0.00        | 0.00       | 3 5446           |
| Marsh & McLennan                   | 0.70        | 0.50       | 3 0499           |
| MAXIMUS Inc                        | 0.75        | 0.50       | 3 4659           |
| Microsoft Corp                     | 0.75        | 0.68       | 3 0865           |
| Northwest Bancshares               | 0.00        | 0.00       | 3 3107           |
| Owens & Minor                      | 0.75        | 0.02       | 3 3015           |
|                                    | 0.70        | 0.47       | 3.5915           |
| Peoples United Finl                | 0.00        | 0.03       | 2 0079           |
| Peoples United Fillin              | 0.05        | 0.40       | 3.0970           |
| Romins, inc.                       | 0.00        | 0.00       | 3.0494           |
| Shorwin-Williams                   | 0.00        | 0.07       | 3.3940           |
| Smerwin-Williams                   | 0.70        | 0.01       | 3.4209<br>2.0447 |
| Sara Loo Corp                      | 0.70        | 0.40       | 3 0463           |
| Stariovelo Inc                     | 0.00        | 0.00       | 3.0403           |
| Stendycle Inc.                     | 0.05        | 0.40       | 2.0677           |
| Struker Corp                       | 0.70        | 0.52       | 2.0077           |
| Toloflox Inc                       | 0.00        | 0.07       | 3.2403           |
| Telellex IIIC.                     | 0.80        | 0.00       | 3.2493           |
| Walaroon Co                        | 0.80        | 0.07       | 3.0200           |
|                                    | 0.75        | 0.00       | 2 4090           |
| Wois Markets                       | 0.75        | 0.30       | 2 05 4 0         |
| Wetson Dharman                     | 0.05        | 0.45       | 3.0349           |
| Parklay (M/ D)                     | 0.75        | 0.50       | 3.1400           |
| World Wrootling Ent                | 0.70        | 0.51       | J.ZZIZ           |
|                                    | 0.00        | 0.05       | 3.4001           |
| Allegnany Corp.                    | 0.80        | 0.67       | 3.2459           |
| Average                            | 0.74        | 0.57       | 3.2800           |
| Proxy Group of Nine Water          |             |            |                  |
| Companies                          | 0.73        | 0.54       | 3.3074           |

### <u>Missouri-American Water Company</u> Basis of Selection of Group of Domestic, Non-Price Regulated Companies <u>Comparable in Total Risk to the Proxy Group of Nine Water Companies</u>

- (1) The criteria for selection of the proxy group of forty-two non-utility companies was that the non-utility companies be domestic and have a meaningful projected rate of return on book common equity, shareholder's equity, net worth or partner's capital for the years 2014-2016, as reported in Value Line Investment Survey (Standard Edition). The proxy group of forty-two non-utility companies was selected based upon the proxy group of nine water companies unadjusted beta range of 0.40 0.68 and standard error of the regression range of 3.0168 3.5980. These ranges are based upon plus or minus two standard deviations of the unadjusted beta and standard error of the regression as detailed in Ms. Ahern's direct testimony. Plus or minus two standard deviations captures 95.50% of the distribution of unadjusted betas and standard errors of the regression.
- (2) The standard deviation of group of nine water companies' standard error of the regression is 0.1392. The standard deviation of the standard error of the regression is calculated as follows:

Standard Deviation of the Std. Err. of the Regr. = <u>Standard Error of the Regression</u>  $\sqrt{2N}$ 

where: N = number of observations. Since Value Line betas are derived from weekly price change observations over a period of five years, N = 259

Thus, 0.1453 =  $\frac{3.3074}{\sqrt{518}}$  =  $\frac{3.3074}{22.7596}$ 

| Source of Information: | Value Line, Inc., December 15, 2011             |
|------------------------|---|
|                        | Value Line Investment Survey (Standard Edition) |

## Missouri-American Water Company Comparable Earnings Analysis for the Proxy Group of Non Price Regulated Companies Comparable to the <u>Proxy Group of Nine Water Companies(1)</u>

|                              |          |             |                               | Rate of Return on<br>Equity, Net Wort<br>Capi | Book Common<br>h, or Partner's<br>tal |              |
|------------------------------|----------|-------------|-------------------------------|---|---------------------------------------|--------------|
|                              |          |             |                               |   | 5-Year Pro                            | ected (2)    |
|                              | VL       | the distant | Residual<br>Standard<br>Error | Standard                                      | <b>5</b> V                            | Quadratile T |
| Proxy Group of Forty-Two Non | Adjusted | Unadjusted  | of the                        | Deviation of                                  | 5 Year                                | Student's I  |
| Price Regulated Companies    | Dela     | Dela        | Regression                    | Dela  | Projection                            | Statistic    |
| Gallagher (Arthur J.)        | 0.70     | 0.54        | 3.0668                        | 0.0637  | 14.00 %                               | (0.3)        |
| Amgen                        | 0.65     | 0.43        | 3.5824                        | 0.0744  | 15.50                                 | (0.2)        |
| AutoZone Inc.                | 0.70     | 0.53        | 3.3352                        | 0.0693  | NMF                                   | (1.8)        |
| Bristol-Myers Squibb         | 0.75     | 0.57        | 3.1171                        | 0.0647  | 23.00                                 | 0.6          |
| Brown & Brown                | 0.70     | 0.49        | 3,1582                        | 0.0656  | 11.50                                 | (0.6)        |
| Capitol Fed, Finl            | 0.65     | 0.43        | 3.2572                        | 0.0676  | 3.50                                  | (1.5)        |
| CVS Caremark Corp.           | 0.80     | 0.66        | 3.0354                        | 0.0630  | 11.00                                 | (0,7)        |
| Forest Labs.                 | 0.80     | 0.63        | 3.3743                        | 0.0701  | 9.50                                  | (0.8)        |
| Gen-Probe                    | 0.80     | 0.68        | 3,3384                        | 0.0693  | 12.50                                 | (0.5)        |
| Hasbro, Inc.                 | 0.75     | 0.60        | 3.3919                        | 0.0704  | 30.00                                 | 1.4          |
| IAC/InterActiveCorp          | 0.65     | 0.47        | 3.2905                        | 0.0779  | 7.50                                  | (1.0)        |
| Investors Bancorp            | 0.75     | 0.55        | 3.4028                        | 0.0707  | 9.50                                  | (0.8)        |
| J&J Snack Foods              | 0.70     | 0.48        | 3.3652                        | 0.0699  | 13.00                                 | (0.5)        |
| Lancaster Colony             | 0.75     | 0.56        | 3.3268                        | 0.0691  | 18.00                                 |              |
| Lincare Holdings             | 0.65     | 0.45        | 3.5487                        | 0.0737  | 25.00                                 | 0.8          |
| McKesson Corp.               | 0.75     | 0.56        | 3.3801                        | 0.0702  | 16.50                                 | (0.1)        |
| Medtronic, Inc.              | 0.80     | 0.66        | 3.5135                        | 0.0730  | 16.00                                 | (0.1)        |
| Medco Health Solutions       | 0.70     | 0.50        | 3.5446                        | 0.0736  | 22.50                                 | 0.6          |
| Marsh & McLennan             | 0.75     | 0.58        | 3.0499                        | 0.0633  | 19.00                                 | 0.2          |
| MAXIMUS Inc.                 | 0.75     | 0.61        | 3.4659                        | 0.0720  | 28.50                                 | 1.2          |
| Microsoft Corp.              | 0.80     | 0.68        | 3.0865                        | 0.0641  | 36.00 (3)                             | 2.0          |
| Northwest Bancshares         | 0.75     | 0.62        | 3.3107                        | 0.0687  | 7.50                                  | (1.0)        |
| Owens & Minor                | 0.70     | 0.47        | 3.3915                        | 0.0704  | 13.00                                 | (0.5)        |
| OReilly Automotive           | 0.80     | 0.63        | 3.5477                        | 0.0737  | 13.00                                 | (0.5)        |
| Peoples United Finl          | 0.65     | 0.40        | 3.0978                        | 0.0643  | 6.50                                  | (1.1)        |
| Rollins, Inc.                | 0.80     | 0.66        | 3.0494                        | 0.0633  | 32.00                                 | 1.6          |
| Ross Stores                  | 0.80     | 0.67        | 3,5940                        | 0.0746  | 41.50 (3)                             | 2.6          |
| Sherwin-Williams             | 0.70     | 0.51        | 3,4289                        | 0.0712  | 26.00                                 | 0.9          |
| Smucker (J.M.)               | 0.70     | 0.48        | 3.0447                        | 0.0632  | 11.00                                 | (0,7)        |
| Sara Lee Corp.               | 0.80     | 0.66        | 3.0463                        | 0.0633  | NMF                                   | (1.8)        |
| Stericycle Inc.              | 0.65     | 0.46        | 3.2191                        | 0.0668  | 15.50                                 | (0.2)        |
| Safeway Inc.                 | 0.70     | 0.52        | 3.0677                        | 0.0637  | 16.50                                 | (0,1)        |
| Stryker Corp.                | 0.80     | 0.67        | 3,2465                        | 0.0674  | 14.00                                 | (0.3)        |
| Teleflex Inc.                | 0.80     | 0.68        | 3,2493                        | 0.0675  | 9.50                                  | (0.8)        |
| TJX Companies                | 0.80     | 0.67        | 3.0258                        | 0.0628  | 44.00 (3)                             | 2.8          |
| Walgreen Co.                 | 0.75     | 0.60        | 3.2564                        | 0.0676  | 20.50                                 | 0.3          |
| WD-40 Co.                    | 0.75     | 0.56        | 3.4989                        | 0.0727  | 17.50                                 | 0.0          |
| Weis Markets                 | 0.65     | 0.45        | 3.0549                        | 0.0634  | 9.50                                  | (0.8)        |
| Watson Pharmac.              | 0.75     | 0.56        | 3.1485                        | 0.0654  | 14.00                                 | (0.3)        |
| Berkley (W.R.)               | 0.70     | 0.51        | 3.2272                        | 0.0670  | 13.50                                 | (0.4)        |
| World Wrestling Ent.         | 0.80     | 0.65        | 3.4061                        | 0.0707  | 17.00                                 | (0.0)        |
| Alleghany Corp.              | 0.80     | 0.67        | 3.2459                        | 0.0674  | 5.50                                  | (1.2)        |
| Average                      | 0.74     | 0.57        | 3.2807                        | 0.0684  |                                       |              |
|                              |          |             |                               |   |                                       |              |

| Average for the Proxy Group of<br>Nine Water Companies | 0.73 | 0.54 | 3.3074 ( | 1) 0.0697 |        |
|--|------|------|----------|-----------|--------|
| Median (4)   |      |      |          |           | 14.75% |
| Conservative Median (5)                                |      |      |          |           | 14.00% |

Notes:

(1) From page 26 of this Schedule.
(2) From Value Line Investment Survey, various issues for the years 2014 - 2016.
(3) The student's T statistic associated with these returns exceeds 1.96 at the 95% level of confidence. Therefore, they have been excluded, as outliers, to arrive at proper projected returns as fully explained in Ms. Ahern's direct
(4) Modian five year projected rate of return on book common equity, shareholders' equity, net worth, or partners' capital

have been excluded, as outliers, to arrive at proper projected returns as runy explained in twis. Amenus onect
(4) Median five year projected rate of return on book common equity, shareholders' equity, net worth, or partners' capital including returns identified as outliers as outlined in note (3) above.
(5) Median five year projected rate of return on book common equity, shareholders' equity, net worth, or partners' capital excluding returns identified as outliers as outlined in note (3) above.

### Missouri-American Water Company DCF Results for the Proxy Group of Non-Utility Companies Comparable in Total Risk to the Proxy Group of Nine Water Companies

| Proxy Group of Forty-<br>Two Non Price Regulated<br>Companies | Average<br>Dividend<br>Yield | Value Line<br>Projected<br>Five Year<br>Growth in<br>EPS | Reuters Mean<br>Consensus<br>Projected Five<br>Year Growth<br>Rate in EPS | Zack's<br>Five Year<br>Projected<br>Growth<br>Rate in<br>EPS | Yahoo!<br>Finance<br>Projected<br>Five Year<br>Growth in<br>EPS | Average<br>Projected<br>Five Year<br>Growth Rate<br>in EPS | Adjusted<br>Dividend<br>Yield | Indicated<br>Common<br>Equity<br>Cost Rate |
|---|------------------------------|--|---|--|---|--|-------------------------------|--|
| Gallagher (Arthur J.  | 4.33 %                       | 9.00 %   | 9.80 %  | 9.60 %   | 9.77 %  | 9.54 %   | 4.54 %                        | 14.08 %                                    |
| Amgen   | 0.00                         | 7.00   | 7.00  | 9.00   | 7.55  | 7.64   | -                             | N/A  |
| AutoZone Inc.   | 0.00                         | 13.50  | 15.00   | 15.40  | 14.07   | 14.49  | -                             | N/A  |
| Bristol-Myers Squibb  | 4.05                         | 8.00   | 0.00  | 1.50   | -0.65   | 3.17   | 4.11                          | 7.28                                       |
| Brown & Brown   | 1.52                         | 7.00   | 11.00   | 13.30  | 13.00   | 11.08  | 1.60                          | 12.68                                      |
| Capitol Fed. Fini   | 2.70                         | 8.00   | 3.00  | 3.00   | 3.00  | 4.25   | 2.76                          | 7.01                                       |
| CVS Caremark Corp.  | 1.34                         | 0.00   | 11.00   | 11.60  | 10.45   | 10.44  | 1.40                          | 11.04                                      |
| Con Broke   | 0.00                         | 11.00  | 0.00  | -2.60  | -5.05   | 0.00   | -                             | IN/A                                       |
| Hashro Inc  | 0.00                         | 10.00  | 12.00   | 14.50  | 13.14   | 12.00  | - 3.58                        | 1/A<br>1/ 81                               |
| IAC/InterActiveCorp   | 0.00                         | 34.00  | 36.00   | 35.40  | 36.03   | 35.36  | 5.50                          | N/A  |
| Investors Bancorn In  | 0.00                         | NMF  | 15.00   | 15.00  | 15.00   | 15.00  |                               | N/A  |
| 1& I Spack Foods  | 0.00                         | 10.50  | N/A   | 13.00  | 0.00  | 5 25   | 96 0                          | 6.21                                       |
| Lancaster Colony  | 1 97                         | 6.00   | 10.00   |  | 10.00   | 8.67   | 2.06                          | 10.73                                      |
| Lincare Holdings  | 3 41                         | 12.00  | 15.00   | 17.50  | 10.00   | 14.83  | 3.66                          | 18 49                                      |
| McKesson Corp.  | 1.02                         | 12.00  | 12.00   | 12.20  | 14.02   | 12.56  | 1.09                          | 13.65                                      |
| Medtronic, Inc.   | 2.79                         | 5.50   | 7.00  | 7.50   | 6.78  | 6.70   | 2.88                          | 9.58                                       |
| Medco Health Solutio  | 0.00                         | 11.00  | 14.00   | 12.80  | 14.27   | 13.02  | -                             | N/A  |
| Marsh & McLennan  | 2.94                         | 28.50  | 10.00   | 10.70  | 11.39   | 15.15  | 3.16                          | 18.31                                      |
| MAXIMUS Inc.  | 0.90                         | 18.50  | 7.00  | 4.00   | 7.00  | 9.13   | 0.94                          | 10.07                                      |
| Microsoft Corp.   | 2.46                         | 12.00  | 10.00   | 11.10  | 9.69  | 10.70  | 2.59                          | 13.29                                      |
| Northwest Bancshares  | 3.60                         | 15.50  | 5.00  | 5.00   | 5.00  | 7.63   | 3.74                          | 11.37                                      |
| Owens & Minor   | 2.74                         | 10.00  | 9.50  | 13.00  | 9.53  | 10.51  | 2.88                          | 13.39                                      |
| OReilly Automotive  | 0.00                         | 13.50  | 16.00   | 17.20  | 17.09   | 15.95  | -                             | N/A  |
| Peoples United Fin  | 5.13                         | 21.00  | 21.00   | 20.00  | 21.68   | 20.92  | 5.66                          | 26.58                                      |
| Rollins, Inc.   | 1.32                         | 13.50  | N/A   | -  | 10.00   | 11.75  | 1.40                          | 13.15                                      |
| Ross Stores   | 0.99                         | 18.50  | 11.00   | 12.60  | 10.77   | 13.22  | 1.05                          | 14.27                                      |
| Sherwin-Williams  | 1.74                         | 11.00  | 11.00   | 10.90  | 11.13   | 11.01  | 1.83                          | 12.84                                      |
| Smucker (J.M.)  | 2.53                         | 9.50   | 7.60  | 8.00   | 7.63  | 8.18   | 2.63                          | 10.81                                      |
| Sara Lee Corp.  | 2.54                         | 10.50  | 8.40  | 6.00   | 9.13  | 8.51   | 2.65                          | 11.16                                      |
| Stericycle Inc.   | 0.00                         | 13.00  | 17.00   | 17.50  | 18.00   | 16.38  | -                             | N/A  |
| Sareway Inc.  | 2.98                         | 6.50   | 8.40  | 10.40  | 8.51  | 8.45   | 3.11                          | 11.56                                      |
| Stryker Corp.   | 1.50                         | 8.00   | 11.00   | 10.80  | 10.65   | 10.11  | 1.57                          | 11.08                                      |
| Telefiex Inc.   | 2.31                         | 9.00   | 13.00   | 10.00  | 14.90   | 11.73  | 2.40                          | 14.10                                      |
| Walgreen Co   | 2.71                         | 13.00  | 12.00   | 12.40  | 0.54  | 12.74  | 2.86                          | 14.07                                      |
| WD-40 Co  | 2.71                         | 9.50   | 12.00   | 12.40  | 12.00   | 11.24  | 2.00                          | 14.10                                      |
| Weis Markets  | 2.00                         | 6.50   | N/A   | 12.00  | 0.00  | 3.25   | 3.04                          | 6.29                                       |
| Watson Pharmac  | 0.00                         | 11.50  | 12 00   | 12 20  | 12 21   | 11.98  | -                             | N/A  |
| Berkley (W.R.)  | 0.97                         | 11.50  | 11.00   | 11.30  | 9.50  | 10.83  | 1.02                          | 11.85                                      |
| World Wrestling Ent.  | 4.90                         | 5.00   | 8.50  | 7.50   | 10.00   | 7.75   | 5.09                          | 12.84                                      |
| Alleghany Corp.   | 0.00                         | 10.00  | N/A   | -  | 0.00  | 5.00   | -                             | N/A  |
| Average   |                              |  |   |  |   |  |                               | 12.65                                      |
| Median  |                              |  |   |  |   |  |                               | 12.84                                      |

NA= Not Available

NMF= Not Meaningful Figure

(1) Ms. Ahern's application of the DCF model to the domestic, non-price regluated comparable risk companies is identical to the application of the DCF to her proxy group of water companies. She uses the 60 day average price and the spot indicated dividend as of January 2, 2012 for her dividend yield and then adjusts that yield for 1/2 the average projected growth rate in EPS, which is calculated by averaging the 5 year projected growth in EPS provided by Value Line, www.reuters.com, www.zacks.com, and www.yahoo.com (excluding any negative growth rates) and then adding that growth rate to the adjusted dividend yield.

Source of Information: Value Line Investment Survey: www.reuters.com Downloaded on 01/03/2012 www.zacks.com Downloaded on 01/03/2012 www.yahoo.com Downloaded on 01/03/2012

### Missouri-American Water Company Indicated Common Equity Cost Rate Through Use of a Risk Premium Model Using an Adjusted Total Market Approach

| Line No. |   | Proxy Group of<br>Forty-Two Non<br>Price Regulated<br>Companies |
|----------|---|---|
| 1.       | Prospective Yield on Baa Rated<br>Corporate Bonds (1) | 5.45 %  |
| 2.       | Equity Risk Premium (2)                               | 7.27  |
| 3.       | Risk Premium Derived Common<br>Equity Cost Rate       | <u> </u>  |

Notes: (1) Average forecast based upon six quarterly estimates of Baa rated corporate bonds per the consensus of nearly 50 economists reported in Blue Chip Financial Forecasts dated January 1, 2012 (see page 20 of this Schedule). The estimates are detailed below.

| First Quarter 2012  | 5.20 | %  |
|---------------------|------|----|
| Second Quarter 2012 | 5.30 |    |
| Third Quarter 2012  | 5.40 |    |
| Fourth Quarter 2012 | 5.50 |    |
| First Quarter 2013  | 5.60 |    |
| Second Quarter 2013 | 5.70 | _  |
|                     |      |    |
| Average             | 5.45 | _% |

(2) From page 31 of this Schedule.

### Comparison of Bond Ratings for the Proxy Group of Non Price Regulated Companies Comparable in Total Risk to the <u>Proxy Group of Nine Water Companies</u>

|   | Moody's<br>Bond Rating<br>December 2011 |                            | Stand<br>Bo<br>Dece | ard & Poor's<br>nd Rating<br>ember 2011 |
|---|---|----------------------------|---------------------|---|
| Proxy Group of Forty-Two Non<br>Price Regulated Companies | Bond<br>Rating                          | Numerical<br>Weighting (1) | Bond<br>Rating      | Numerical<br>Weighting (1)              |
| Gallagher (Arthur J.)                                     | NR                                      |                            | NR                  |   |
| Amgen   | Baa1                                    | 8.0                        | A+                  | 5.0                                     |
| AutoZone Inc.   | Baa2                                    | 9.0                        | BBB                 | 9.0                                     |
| Bristol-Myers Squibb                                      | A2                                      | 6.0                        | A+                  | 5.0                                     |
| Brown & Brown   | NR                                      |                            | NR                  |   |
| Capitol Fed. Finl   | NR                                      |                            | NR                  |   |
| CVS Caremark Corp.  | Baa2                                    | 9.0                        | BBB+                | 8.0                                     |
| Forest Labs.  | Baa2                                    | 9.0                        | NR                  |   |
| Gen-Probe   | NR                                      |                            | NR                  |   |
| Hasbro, Inc.  | NR                                      |                            | BBB+                | 8.0                                     |
| IAC/InterActiveCorp                                       | Ba2                                     | 12.0                       | NR                  |   |
| Investors Bancorp   | NR                                      |                            | NR                  |   |
| J&J Snack Foods   | NR                                      |                            | NR                  |   |
| Lancaster Colony  | NR                                      |                            | NR                  |   |
| Lincare Holdings  | NR                                      |                            | NR                  |   |
| McKesson Corp.  | Baa2                                    | 9.0                        | A-                  | 7.0                                     |
| Medtronic. Inc.   | A1                                      | 5.0                        | AA-                 | 4.0                                     |
| Medco Health Solutions                                    | Baa3                                    | 10.0                       | BBB+                | 8.0                                     |
| Marsh & McLennan  | Baa2                                    | 9.0                        | BBB-                | 10.0                                    |
| MAXIMUS Inc.  | NR                                      |                            | NR                  |   |
| Microsoft Corp.   | Aaa                                     | 1.0                        | AAA                 | 1.0                                     |
| Northwest Bancshares                                      | NR                                      |                            | NR                  |   |
| Owens & Minor   | Ba2                                     | 12.0                       | BBB-                | 10.0                                    |
|   | Baa3                                    | 10.0                       | NR                  |   |
| Peoples United Finl                                       | A2                                      | 60                         | NR                  |   |
| Rollins Inc   | NR                                      |                            | NR                  |   |
| Ross Stores   | NR                                      |                            | NR                  |   |
| Sherwin-Williams  | A3                                      | 7.0                        | Δ                   | 6.0                                     |
| Smucker (IM)  | A3                                      | 7.0                        | NR                  |   |
| Sara Lee Corp   | Baa1                                    | 8.0                        | BBB                 | 9.0                                     |
| Stericycle Inc  | NR                                      | 0.0                        | NR                  | 0.0                                     |
| Safeway Inc   | Baa2                                    | 9.0                        | BBB                 | 9.0                                     |
| Stryker Corp  | A3                                      | 7.0                        | A+                  | 5.0                                     |
| Teleflex Inc  | Ba3                                     | 13.0                       | NR                  | 5.0                                     |
| T.IX Companies  | A3                                      | 7.0                        | NR                  |   |
| Walgreen Co   | A2                                      | 6.0                        | Δ                   | 6.0                                     |
| WD-40 Co  | NR                                      |                            | NR                  |   |
| Weis Markets  | NR                                      |                            | NR                  |   |
| Watson Pharmac  | Baa3                                    | 10.0                       | NR                  |   |
| Berkley (W R )  | Baa2                                    | 9.0                        | BRR+                | 8.0                                     |
| World Wrestling Ent                                       | NR                                      |                            | NR                  |   |
| Alleghany Corp.   | Baa2                                    | 9.0                        | NR                  |   |
| Average   | Baa1                                    | 8.3                        | A-                  | 6.9                                     |

Notes:

(1) From page 3 of Schedule PMA-10.

Source of Information: Standard & Poor's Bond Guide December 2011 www.moodys.com; downloaded 1/3/2012

### <u>Missouri-American Water Company</u> Derivation of Equity Risk Premium Based on the Total Market Approach Using the Beta for the Proxy Group of Non Price Regulated Companies Comparable in Total Risk to the <u>Proxy Group of Nine Water Companies</u>

| Line No. |     |  | Proxy Group of<br>Forty-Two Non<br>Price Regulated<br>Companies |
|----------|-----|--|---|
| 1.       |     | Arithmetic mean total return rate on<br>the Standard & Poor's 500 Composite<br>Index - 1926-2010 (1) | 11.90 %   |
| 2.       |     | Arithmetic mean yield on<br>Aaa and Aa Corporate Bonds<br>1926-2010 (2)                              | (6.10)  |
| 3.       |     | Historical Equity Risk Premium   | <u> </u>  |
| 4.       |     | Forecasted 3-5 year Total Annual<br>Market Return (3)  | 17.80 %   |
| 5.       |     | Prospective Yield an Aaa Rated<br>Corporate Bonds (4)  | (4.23)  |
| 6.       |     | Forecasted Equity Risk Premium   | <u>    13.57  </u> %  |
| 7.       |     | Conclusion of Equity Risk Premium (5)  | 9.69 %  |
| 8.       |     | Adjusted Value Line Beta (6)   | 0.75  |
| 9.       |     | Beta Adjusted Equity Risk Premium  | <u> </u>  |
| Notes:   | (1) | Ibbotson Associates 2011 Valuation Yearbook - Market Results<br>Morningstar, Inc., 2011 Chicago, IL. | for 1926-2010,  |

(2) From Moody's Industrial Manual and Mergent Bond Record Monthly Update.

(3) From page 22 of this Schedule.

(4) Average forecast based upon six quarterly estimates of Aaa rated corporate bonds per the consensus of nearly 50 economists reported in Blue Chip Financial Forecasts dated January 1, 2012 (see page 20 of this Schedule). The estimates are detailed below.

| First Quarter 2012  | 4.00 % |
|---------------------|--------|
| Second Quarter 2012 | 4.00   |
| Third Quarter 2012  | 4.20   |
| Fourth Quarter 2012 | 4.30   |
| First Quarter 2013  | 4.40   |
| Second Quarter 2013 | 4.50   |
| Average             | 4.23 % |

- (5) The average of the historical equity risk premium of 5.80% from Line No. 3 and the forecasted equity risk premium of 13.57% from Line No. 6 ((5.80% + 13.57%) / 2 = 9.69%.
- (6) Median beta derived from page 21 of this Schedule.

### Missouri-American Water Company

Traditional CAPM and ECAPM Results for the Proxy Group of Non Price Regulated Companies Comparable in Total Risk to the Proxy Group of Nine Water Companies

| Proxy Group of Forty-Two<br>Non Price Regulated<br>Companies | Value Line<br>Adjusted<br>Beta | Market Risk<br>Premium (1) | Risk-Free<br>Rate (2) | Traditional<br>CAPM Cost<br>Rate (3) | ECAPM Cost<br>Rate (4) | Indicated<br>Common Equity<br>Cost Rate (5) |
|--|--------------------------------|----------------------------|-----------------------|--------------------------------------|------------------------|---|
| Gallagher (Arthur, I.)                                       | 0.70                           | 10 53                      | 3 45                  | 10.82                                | 11 61                  |   |
| Amgen  | 0.70                           | 10.53                      | 3.45                  | 10.02                                | 11.01                  |   |
| AutoZone Inc   | 0.00                           | 10.53                      | 3 45                  | 10.23                                | 11.61                  |   |
| Bristol-Myers Squibb   | 0.75                           | 10.53                      | 3 45                  | 11.35                                | 12.01                  |   |
| Brown & Brown  | 0.70                           | 10.53                      | 3 45                  | 10.82                                | 11.61                  |   |
| Capitol Fed. Finl  | 0.65                           | 10.53                      | 3 45                  | 10.29                                | 11.22                  |   |
| CVS Caremark Corp.   | 0.80                           | 10.53                      | 3 45                  | 11.87                                | 12.40                  |   |
| Forest Labs  | 0.80                           | 10.53                      | 3 45                  | 11.87                                | 12.40                  |   |
| Gen-Probe  | 0.80                           | 10.53                      | 3 45                  | 11.87                                | 12.10                  |   |
| Hasbro Inc.  | 0.75                           | 10.53                      | 3 45                  | 11.35                                | 12.01                  |   |
| IAC/InterActiveCorp  | 0.65                           | 10.53                      | 3.45                  | 10.29                                | 11.22                  |   |
| Investors Bancorp  | 0.75                           | 10.53                      | 3.45                  | 11.35                                | 12.01                  |   |
| J&J Snack Foods  | 0.70                           | 10.53                      | 3.45                  | 10.82                                | 11.61                  |   |
| Lancaster Colony   | 0.75                           | 10.53                      | 3.45                  | 11.35                                | 12.01                  |   |
| Lincare Holdings   | 0.65                           | 10.53                      | 3.45                  | 10.29                                | 11.22                  |   |
| McKesson Corp.   | 0.75                           | 10.53                      | 3.45                  | 11.35                                | 12.01                  |   |
| Medtronic, Inc.  | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Medco Health Solutions                                       | 0.70                           | 10.53                      | 3.45                  | 10.82                                | 11.61                  |   |
| Marsh & McLennan   | 0.75                           | 10.53                      | 3.45                  | 11.35                                | 12.01                  |   |
| MAXIMUS Inc.   | 0.75                           | 10.53                      | 3.45                  | 11.35                                | 12.01                  |   |
| Microsoft Corp.  | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Northwest Bancshares   | 0.75                           | 10.53                      | 3.45                  | 11.35                                | 12.01                  |   |
| Owens & Minor  | 0.70                           | 10.53                      | 3.45                  | 10.82                                | 11.61                  |   |
| OReilly Automotive   | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Peoples United Finl  | 0.65                           | 10.53                      | 3.45                  | 10.29                                | 11.22                  |   |
| Rollins, Inc.  | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Ross Stores  | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Sherwin-Williams   | 0.70                           | 10.53                      | 3.45                  | 10.82                                | 11.61                  |   |
| Smucker (J.M.)   | 0.70                           | 10.53                      | 3.45                  | 10.82                                | 11.61                  |   |
| Sara Lee Corp.   | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Stericycle Inc.  | 0.65                           | 10.53                      | 3.45                  | 10.29                                | 11.22                  |   |
| Safeway Inc.   | 0.70                           | 10.53                      | 3.45                  | 10.82                                | 11.61                  |   |
| Stryker Corp.  | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Teleflex Inc.  | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| TJX Companies  | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Walgreen Co.   | 0.75                           | 10.53                      | 3.45                  | 11.35                                | 12.01                  |   |
| WD-40 Co.  | 0.75                           | 10.53                      | 3.45                  | 11.35                                | 12.01                  |   |
| Weis Markets   | 0.65                           | 10.53                      | 3.45                  | 10.29                                | 11.22                  |   |
| Watson Pharmac.  | 0.75                           | 10.53                      | 3.45                  | 11.35                                | 12.01                  |   |
| Berkley (W.R.)   | 0.70                           | 10.53                      | 3.45                  | 10.82                                | 11.61                  |   |
| World Wrestling Ent.   | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Alleghany Corp.  | 0.80                           | 10.53                      | 3.45                  | 11.87                                | 12.40                  |   |
| Average  |                                |                            |                       | 11.22                                | 11.91                  | 11.57                                       |
| Median   |                                |                            |                       | 11.35                                | 12.01                  | 11.68                                       |

Notes:

(1) From note 1 on page 22 of this Schedule.

(2) From note 2 on page 22 of this Schedule.

(3) Derived from the model shown on page 22 of this Schedule, note 3.

(4) Derived from the model shown on page 22 of this Schedule, note 4.
(5) Average of CAPM and ECAPM cost rates.

Missouri-American Water Company Derivation of the Floatation Cost Adjustment to the Cost of Common Equity.

## Equity Issuances and Floatation Costs of the Parent Since 2009

| olumn 10]  | tation Cost<br>centage (7)          | 4.33%                   | 4.33%       |               |  |
|------------|-------------------------------------|-------------------------|-------------|---------------|--|
| <u>Ö</u>   | s Flo                               | -                       |             |               |  |
| Column 9]  | Floatation Cost:<br>(6)             | 8,717,000               | 8,717,000   |               |  |
| _          | Total F                             | \$                      | ω           |               |  |
| [Column 8] | Vet Proceeds (5)                    | 192,418,000             | 192,418,000 |               |  |
|            | Total                               | \$                      | ω           |               |  |
| [Column 7] | oss Equity Issue<br>efore Costs (4) | 201,135,000             | 201,135,000 |               |  |
|            | P 9                                 | θ                       | φ           |               |  |
| [Column 6] | Net Proceeds<br>per Share (3)       | \$ 16.7320              |             |               |  |
| Column 5]  | derwriting<br>Discount              | 0.5180                  |             | ment          |  |
| <u>0</u>   | 5                                   | φ                       |             | it Adjust     |  |
| olumn 4]   | tet Pressure<br>(2)                 | 0.2400                  |             | Flotation Cos |  |
| <u>D</u>   | Mark                                | ŝ                       |             |               |  |
| [Column 3] | Offering Price<br>per Share         | \$ 17.2500              |             |               |  |
| [Column 2] | Market Price<br>per Share           | \$ 17.4900              |             |               |  |
| [Column 1] | Shares Issued                       | 11,500,000              |             |               |  |
|            | Transaction (1)                     | Primary Equity Offering |             |               |  |
|            | Date                                | 06/10/09                |             |               |  |

 Provession
 Average DCF

 Average DCF
 Average DCF

 Average DV
 Cost Rate

 Average Dvidend Yield
 Projected EPS

 Average Dvidend Yield
 Projected EPS

 Provy Group of Nine
 3.35

 Mater Companies
 3.47

0.16 %

10.71 %

Flotation Cost Adjustment (10)

DCF Cost Rate Adjusted for Flotation (9)

See page 34 of this Schedule for notes.

### Missouri-American Water Company Notes to Accompany the Derivation of the Flotation Cost Adjustment to the Cost of Common Equity

- (1) Company-provided.
- (2) Column 2 Column 3.
- (3) Column 2 the sum of columns 4 and 5.
- (4) Column 1 \* Column 2.
- (5) Column1 \* Column 6.
- (6) Column1 \* (the sum of columns 4 and 5).
- (7) (Column 7 Column 8) divided by Column 7.
- (8) Using the average growth rate from Schedule 7.
- (9) Adjustment for flotation costs based on adjusting the average DCF constant growth cost rate in accordance with the following:

$$K = \frac{D(1+0.5g)}{P(1-F)} + g ,$$

where g is the growth factor and F is the percentage of flotation costs.

(10) Flotation cost adjustment of 0.16% equals the difference between the flotation adjusted average DCF cost rate of 10.71% and the unadjusted average DCF cost rate of 10.55% of the proxy group of nine water companies.

Source of Information:

Company provided information

|   |    |  |   |                                     |     |  |   |   | Page 35 of 36   |
|---|----|--|---|-------------------------------------|-----|--|---|---|---|
|   | 41 | Spread from<br>Applicable Size<br>Premium for (4     |   | 0.42%                               | (E) |  | -0.38%<br>0.81%<br>1.01%  | 1.81%<br>1.82%<br>2.65%<br>2.94%<br>6.36%<br>arbook   | corresponds to the<br>Column 3 etc For  |
|   | က၊ | Applicable Size<br>Premium (3)                       | 2.27%   | 1.85%                               | (D) | Recent Average<br>Market<br>Capitalization<br>( millions ) | <ul> <li>\$ 51,109.438</li> <li>\$ 10,350.159</li> <li>\$ 5,468.472</li> <li>\$ 3.214.606</li> </ul>    | <ul> <li>\$ 2,264.447</li> <li>\$ 1,565.828</li> <li>\$ 1,062.538</li> <li>\$ 662.101</li> <li>\$ 370.152</li> <li>\$ 103.121</li> <li>Yer</li> </ul> | <ul> <li>decile (Column (A))</li> <li>bottom of this page.</li> <li>a - Line No. 3 of (</li> </ul>  |
| ed upon<br>e NYSE/AMEX/NASDAQ   | N  | Applicable Decile of<br>the NYSE/AMEX/<br>NASDAQ (2) | 7-8   | 6-7                                 | (C) | Recent Total Market<br>Capitalization<br>( millions )      | <ul> <li>\$ 8,586,385.656</li> <li>1,873,378.709</li> <li>1,022,604.243</li> <li>594.702.185</li> </ul> | 482,327.242<br>360,140.550<br>304,948.414<br>239,018.595<br>181,744.805<br>136,119.075 *F∣  | of this page. The appropriate<br>hich is found in Column 1.<br>is provided on Column (E) on th<br>umn 3 and Line No. 1b, Colum<br>2 is derived as follows 0.42% |
| ritcan Water Company<br>nt Risk Adjustment Bas<br>e Decile Portfolios of th |    | on on January 2,<br>(1)<br>(times larger)            |   | 1.3                                 | (B) | Number of<br>Companies<br>( millions )                     | 168<br>181<br>187   | 213<br>230<br>361<br>491<br>1320  | is Schedule.<br>Irmn (D) on the bottom<br>on of the proxy group, w<br>premium to the decile<br>in 3 – Line No. 2 Colu<br>6 in Column 4, Line No                 |
| <u>Missouri-Ame</u><br>erivation of Investmer<br>es' Size Premia for th     | -1 | Market Capitalizati<br>2012<br>( millions )          | \$ 776.975  | \$ 1,039.534                        | (A) | Decile   | ← 0 0 4   | n o n o O   | From page 36 of th<br>Gleaned from Colu<br>market capitalizatio<br>Corresponding risk<br>Line No. 1a Colum<br>example, the 0.42%                                |
| D.<br>Ibbotson Associate  |    |  | Missouri-American Water Company<br>Based Upon the Proxy Group of Nine Water Companies | Proxy Group of Nine Water Companies |     |  | Largest   | Smallest  | Notes:<br>(1)<br>(3)<br>(4)   |
|   |    | Line No.   |   | N                                   |     |  |   |   |   |

Schedule PMA-39

|   |   |   |   | 2  |   | ମ<br>ମ  |                                   | 41                              | ام   |                                    | JO  |
|---|---|---|---|--|---|---|-----------------------------------|---------------------------------|--|------------------------------------|---|
| Company Excha   | ange  | Common Stock Shares<br>Outstanding at Fiscal<br>Year End 2010<br>( millions )   | Book <sup>v</sup><br>Share<br>Year Er   | Value per<br>at Fiscal<br>nd 2010 (1)  | Total Com<br>Fiscal Ye<br>( mi                              | mon Equity at<br>ar End 2010<br>Illions)  | Closir<br>Market<br>January       | g Stock<br>Price on<br>02, 2012 | Market-to-Book<br>Ratio on January<br>02, 2012 (2) | Capi<br>Janu                       | Market<br>talization on<br>ary 02, 2012<br>(3)<br>millions) |
| Missouri-American Water Company                             |   | NA  |   | NA   | \$  | 415.717 (4)   |                                   | NA                              |  |                                    |   |
| Based Upon the Proxy Group of Nine Water<br>Companies       |   |   |   |  |   |   |                                   |                                 | 186.9 % (  | 5) \$                              | 776.975 (6)   |
| Proxy Group of Nine Water Companies                         |   |   | e   |  | e   |   | ¢                                 |                                 |  | e                                  |   |
| American states water Co.<br>American Water Works Co., Inc. |   | 18.631  | ቀ ዓ   | 20.204<br>23.614   | <i>ө</i> ө  | 377.541<br>4,132.272  | <i>ө</i> ө                        | 34.900<br>31.860                | 1/2.2 %  | <del>ه</del> ه                     | 5,575.373   |
| Aqua America, Inc.  |   | 138.449   | \$  | 8.481  | \$  | 1,174.254   | \$                                | 22.050                          | 260.0  |                                    |   |
| Artesian Resources Corp.                                    |   | 7.517   | ഴ   | 12.657   | s   | 95.146  | ŝ                                 | 18.830                          | 148.8  | ф                                  | 141.547   |
| California Water Service Group                              |   | 41.666  | Ь   | 10.453   | s   | 435.526   | ŝ                                 | 18.260                          | 174.7  | Ф                                  | 760.821   |
| Connecticut Water Service, Inc.                             |   | 8.677   | в   | 13.134   | ŝ   | 113.963   | ¢                                 | 27.130                          | 206.6  | Ф                                  | 235.403   |
| Middlesex Water Company                                     |   | 15.566  | в   | 11.132   | ¢   | 173.279   | ŝ                                 | 18.660                          | 167.6  | θ                                  | 290.462   |
| SJW Corporation   |   | 18.552  | Ь   | 13.747   | в   | 255.032   | ŝ                                 | 23.640                          | 172.0  | в                                  | 438.558   |
| York Water Company  |   | 12.692  | θ   | 7.190  | в   | 91.257  | ф                                 | 17.640                          | 245.3  | ω                                  | 223.888   |
| Average   |   | 48.527  | ф   | 13.408   | \$  | 760.919   | ÷                                 | 23.663                          | 186.9 %  | φ                                  | 1,039.534   |
| NA= Not Avai  | ilable  |   |   |  |   |   |                                   |                                 |  |                                    |   |
| _   | Notes: (1)<br>(2)<br>(3)<br>(4)<br>(5)<br>(6) | Column 3 / Column 1.<br>Column 4 / Column 2.<br>Column 5 * / Column 3.<br>From Financial Statements of<br>The market-to-book ratio of N<br>of Nine Water Companies at<br>Missouri-American Water Co<br>02, 2012 of the Proxy Group<br>02, 2012 of the Proxy Group | f Missouri- <i>I</i><br>lissouri-Am<br>January 02<br>mpany's co<br>of Nine Wa | American Water<br>lerican Water C<br>; 2012.<br>mmon stock, if<br>ter Companies,<br>llion. | Company for<br>ompany on Ja<br>traded, would<br>186.9%, and | Fiscal Year End 20<br>nuary 02, 2012 is :<br>trade at a market-t<br>Missouri-American | 010.<br>assumed tr<br>o-book rati | be equal to t<br>o equal to the | he market-to-book rati<br>average market-to-bo     | o of the I<br>ok ratio<br>ok ratio | Proxy Group<br>at January<br>2012                           |

Missouri-American Water Company Market Capitalization of Missouri-American Water Company and the Proxy Group of Nine Water Companies

Source of Information: 2010 Annual Forms 10K yahoo.finance.com